PLANT 1 WATERFRONT PROBING SUMMARY REPORT SITE: FORMER BP HARBOR ISLAND TERMINAL CLEANUP SITE ID: 4426

SEAPORT MIDSTREAM PARTNERS LLC. SEATTLE TERMINAL 1652 SW LANDER STREET SEATTLE, WASHINGTON

CONSENT DECREE NO. 00-2-05714-8SEA

FEBRUARY 2020

Submitted to Washington State Department of Ecology Northwest Regional Office 3190 160th Avenue SE Bellevue, Washington 98008-5452

> Prepared for TLP Management Services, LLC 1670 Broadway Suite 3100 Denver, CO 80202

Prepared by TechSolve Environmental, Inc. 7518 NE 169th Street Kenmore, Washington 98028 www.techsolveinc.com

PLANT 1 WATERFRONT PROBING SUMMARY REPORT SITE: FORMER BP HARBOR ISLAND TERMINAL CLEANUP SITE ID: 4426

SEAPORT MIDSTREAM PARTNERS, LLC. SEATTLE TERMINAL 1652 SW LANDER STREET SEATTLE, WASHINGTON

CONSENT DECREE NO. 00-2-05714-8SEA

FEBRUARY 2020

Scott K. Larsen, снмм Project Scientist/Project Manager TechSolve Environmental, Inc.

Jany & Roberts

Larry E. Roberts, LG, LHG WA # 1148 Principal, WA Licensed Geologist/Hydrogeologist TechSolve Environmental, Inc.



TABLE OF CONTENTS

List of Abbreviations and Acronyms	••••
Executive Summary	
1. Introduction	1
1.1. Waterfront Probing Objective	1
2. Site Description and History	1
2.1. Site Description	1
2.2 Site History	2
2.2.1 Cleanup Criteria	3
2.2.2 Waterfront Remedial Actions	4
3. Waterfront Probing Investigation	5
3.1. Probing Locations	6
3.2. Probing Field Activities	6
3.2.1. Site Preparation and Utility Location	6
3.2.2. Probing and Well Installation	6
3.2.3. Soil Logging, Field Screening, and Soil Sampling	7
3.2.4. Surveying	8
3.2.5. Decontamination Procedures	8
3.2.6. Waste Handling and Disposal	9
3.3 Groundwater Sampling	9
3.4 Sample Handling and Analytical Methodology	10
3.4.1. Quality Assurance and Quality Control	10
4. Results	10
4.1 Field Observations	10
4.1.1. Soil Observations	10
4.1.2 Groundwater Observations	11
4.1.3 IHS Field Screening	11
4.2 Soil Analytical Results	12
4.3 Groundwater Sampling and Analytical Results	13
5. Summary and Findings	13
5.1. IHSs Remaining in Soil	14
5.1.1 Lateral and Vertical Soil TPH Contours	.14
5.1.2 Residual LNAPL Saturation in Soil	14
5.2. IHSs Remaining in Groundwater	15
5.3. Compliance with Performance and Confirmational Criteria	15
5.3.1. LNAPL Removal Performance Criterion	15
5.3.2. Waterway Sheen Criterion	16
5.3.3. Dissolved Phase IHS Criterion	17
5.4. Recommendations for Ongoing Remedial Actions	. 17
5.5. Data Gaps	18
6. Conclusions	19
7. References	20

Tables

Table 1 – Groundwater Monitoring Analytical Results for TPH and Benzene

- Table 2 Waterfront Groundwater Recovery Wells Petroleum Hydrocarbon History
- Table 3 Waterfront Probing Field and Laboratory Groundwater Sampling Results
- Table 4 Waterfront Probing Field and Laboratory Soil Sampling Results
- Table 5 Containment Boom Sheen Monitoring

Figures

- Figure 1 Site Location Map
- Figure 2 Areas of Remediation Plant 1
- Figure 3 Cumulative Waterfront LNAPL Recovery Through September 2019
- Figure 4 Plant 1 Probing Investigation, Boring Locations
- Figure 5 Plant 1 Waterfront Hydrograph, 2012 through 2019
- Figure 6 Plant 1 Waterfront, Existing Well Location Map
- Figure 7 Plant 1 Probing Investigation, A-A' Plant 1 Waterfront Cross Section Map
- Figure 8 Plant 1 Probing Investigation, Groundwater Monitoring Analytical Results
- Figure 9 Plant 1 Probing Investigation, Soil Maximum Total TPH Contour Map
- Figure 10 Plant 1 Probing Investigation, A-A' TPH Contour Cross Section Map
- Figure 11 Plant 1 Probing Investigation, B-B' TPH Contour Cross Section Map
- Figure 12 Plant 1 Probing Investigation, C-C' TPH Contour Cross Section Map

Appendices

- Appenxic A Groundwater Monitoring and Recovery Well Hydrocarbon Graphs
- Appendix B Soil Boring Logs
- Appendix C Sampling and Analysis Plan, November 1999, Exhibit F of Consent Decree No. 00-2-05714-8SEA
- Appendic D Data Validation Summary Sheets for Soil and Groundwater

List of Abbreviations and Acronyms

BGSBelow Ground SurfaceCAPCleanup Action PlanCOCChain of CustodycPAHsCarcinogenic Polycyclic Aromatic HydrocarbonsCSMConceptual Site ModelCULCleanup LevelEDREngineering Design ReportGWCMCPGroundwater Compliance Monitoring and Contingency ProgramIDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-04Total Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	APS	Applied Professional Services, Inc.
CAPCleanup Action PlanCOCChain of CustodycPAHsCarcinogenic Polycyclic Aromatic HydrocarbonsCSMConceptual Site ModelCULCleanup LevelEDREngineering Design ReportGWCMCPGroundwater Compliance Monitoring and Contingency ProgramIDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per Litermg/kgMilligrams per kilogramNLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	BGS	Below Ground Surface
COCChain of CustodycPAHsCarcinogenic Polycyclic Aromatic HydrocarbonsCSMConceptual Site ModelCULCleanup LevelEDREngineering Design ReportGWCMCPGroundwater Compliance Monitoring and Contingency ProgramIDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-0Total Petroleum Hydrocarbons – Oil RangeTPH-0Total Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	CAP	Cleanup Action Plan
cPAHsCarcinogenic Polycyclic Aromatic HydrocarbonsCSMConceptual Site ModelCULCleanup LevelEDREngineering Design ReportGWCMCPGroundwater Compliance Monitoring and Contingency ProgramIDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per Litermg/kgMilligrams per kilogramNLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-0Total Petroleum Hydrocarbons – Oil RangeTPH-0Total Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	COC	Chain of Custody
CSMConceptual Site ModelCULCleanup LevelEDREngineering Design ReportGWCMCPGroundwater Compliance Monitoring and Contingency ProgramIDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSoil Vapor ExtractionTPHTotal Petroleum Hydrocarbons – Diesel RangeTPH-QTotal Petroleum Hydrocarbons – Oil RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	cPAHs	Carcinogenic Polycyclic Aromatic Hydrocarbons
CULCleanup LevelEDREngineering Design ReportGWCMCPGroundwater Compliance Monitoring and Contingency ProgramIDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSoil Vapor ExtractionTPHTotal Petroleum Hydrocarbons – Diesel RangeTPH-QTotal Petroleum Hydrocarbons – Oil RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	CSM	Conceptual Site Model
EDREngineering Design ReportGWCMCPGroundwater Compliance Monitoring and Contingency ProgramIDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSoil Vapor ExtractionTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Oil RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	CUL	Cleanup Level
GWCMCPGroundwater Compliance Monitoring and Contingency ProgramIDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/IMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-QTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	EDR	Engineering Design Report
IDInner DiameterIHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-OTotal Petroleum Hydrocarbons – Diesel RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	GWCMCP	Groundwater Compliance Monitoring and Contingency Program
IHSIndicator Hazardous SubstanceLNAPLLight Non-Aqueous Phase Liquidµg/lMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSoil Vapor ExtractionTPHTotal Petroleum Hydrocarbons – Diesel RangeTPH-QTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	ID	Inner Diameter
LNAPLLight Non-Aqueous Phase Liquidμg/lMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	IHS	Indicator Hazardous Substance
µg/lMicrograms per Litermg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	LNAPL	Light Non-Aqueous Phase Liquid
mg/kgMilligrams per kilogramMLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	μg/l	Micrograms per Liter
MLLWMean Low Low WaterNAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	mg/kg	Milligrams per kilogram
NAD-83North American Datum of 1983NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	MLLW	Mean Low Low Water
NAVD-88North American Vertical Datum of 1988NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	NAD-83	North American Datum of 1983
NGVD-29National Geodetic Vertical Datum of 1929PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeVOCsVolatile Organic Compounds	NAVD-88	North American Vertical Datum of 1988
PIDPhotoionization DetectorPPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeVOCsVolatile Organic Compounds	NGVD-29	National Geodetic Vertical Datum of 1929
PPMParts Per MillionQAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	PID	Photoionization Detector
QAQuality AssuranceQCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	PPM	Parts Per Million
QCQuality ControlRI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	QA	Quality Assurance
RI/FSRemedial Investigation / Feasibility StudySAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	QC	Quality Control
SAPSampling and Analysis PlanSVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	RI/FS	Remedial Investigation / Feasibility Study
SVESoil Vapor ExtractionTPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	SAP	Sampling and Analysis Plan
TPHTotal Petroleum HydrocarbonsTPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	SVE	Soil Vapor Extraction
TPH-DTotal Petroleum Hydrocarbons – Diesel RangeTPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	ТРН	Total Petroleum Hydrocarbons
TPH-GTotal Petroleum Hydrocarbons – Gasoline RangeTPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	TPH-D	Total Petroleum Hydrocarbons – Diesel Range
TPH-OTotal Petroleum Hydrocarbons – Oil RangeVOCsVolatile Organic Compounds	TPH-G	Total Petroleum Hydrocarbons – Gasoline Range
VOCs Volatile Organic Compounds	TPH-O	Total Petroleum Hydrocarbons – Oil Range
	VOCs	Volatile Organic Compounds

Executive Summary

This report presents the results of a soil and groundwater investigation along the waterfront at Plant 1 of the former BP Harbor Island Terminal (the Site). The investigation was conducted to evaluate the effectiveness of ongoing and completed remedial actions in obtaining cleanup objectives and to evaluate the remaining extent of known indicator hazardous substances (IHSs) (benzene, gasoline, diesel, and oil) and separate phase hydrocarbons (i.e. free product or Light Non-Aqueous Phase Liquid [LNAPL]) in subsurface soils and groundwater. Remedial actions have been ongoing since the early 1990s in this area of the Site to address historical environmental impacts. In 2000, remedial action requirements for the Site were formalized in Consent Decree No. 00-2-05714-8SEA. The Consent Decree's Groundwater Compliance Monitoring and Contingency Program (GWCMCP) identified performance and confirmational criteria to be used in determining when ongoing remedial actions have completed their remedial objectives. This investigation was intended to aid in determining if these criteria have been met. The results will also be utilized in an upcoming evaluation of how a new seawall installed along the waterfront affected site hydrology.

This investigation involved the completion of 41 direct push borings from ground surface, through the vadose zone, and approximately 10 feet into the saturated zone. Soils were field screened for IHS and LNAPL impacts, laboratory analyses were performed on the most impacted soils observed in each boring, and a total of 11 of the borings were converted to temporary piezometer wells. These wells were screened across the vadose zone and into shallow groundwater and were subsequently developed and sampled for the presence of LNAPL and dissolved IHSs in groundwater.

Field screening and laboratory analyses showed much of the soil to be below associated cleanup levels (CULs), with a few remaining areas containing residual hydrocarbon impacts along a narrow vertical profile, at and just below the soil/groundwater interface (i.e. the smear zone). Most soils in the unsaturated zone showed little indication of the presence of IHSs or residual LNAPL. IHS concentrations detected from analyzing soil samples collected from the most impacted soil observed in each boring (typically 2 to 3 feet below the groundwater interface) were mainly at or below expected residual LNAPL saturation concentrations, indicating that mobile LNAPL has been removed throughout the area of investigation to the extent practicable.

Groundwater sampling and analyses showed no free LNAPL to be present in any of the 11 temporary wells. No sheen on groundwater was observed in four wells, a slight sheen was observed in six wells, and a moderate sheen was observed in one well. Dissolved IHSs detected in shallow groundwater were below the CULs (established for the more deeply screened conditional point of compliance wells) in seven of the temporary wells. Gasoline was above the CUL in two of the temporary wells and diesel was above the CUL in two other temporary wells. The groundwater monitoring results appear similar to results from recent monitoring in nearby compliance and recovery wells. The groundwater monitoring results from this investigation and past sampling events indicate that remedial actions performed to date have removed LNAPL to the extent practicable.

The results of this investigation show that remedial actions along the Plant 1 waterfront have reduced or removed most of the preexisting soil impacts in the unsaturated zone and that no free LNAPL was

detected on groundwater. The results indicate that the performance criterion from the GWCMCP requiring the lack of measurable product (LNAPL) thickness to discontinue recovery system operations has been met. It appears that the current remedial system has recovered LNAPL to the extent practicable and further operation of the existing groundwater pump and treatment system is unlikely to provide additional environmental benefit.

1. Introduction

This report presents the results of a soil and groundwater investigation along the waterfront at Plant 1 of the former BP Harbor Island Terminal (the Site) (Figure 1). The soil and groundwater investigation was conducted in accordance with the requirements of Consent Decree No. 00-2-05714-8SEA (Ecology, 2000), the Sampling and Analysis Plan (TechSolve, 1999b), the Plant 1 Waterfront Probing Work Plan submitted to Ecology on January 7, 2019 (TechSolve, 2019a), and incorporates Ecology's contingencies to the work plan's approval (Ecology, 2019).

1.1. Waterfront Probing Objective

This investigation was conducted to delineate the extent of remaining IHSs and LNAPL in a historically known source area along the Plant 1 waterfront (Figure 2). This probing investigation evaluates the effectiveness of remedial actions in achieving Site cleanup criteria, including if remaining IHSs and LNAPL have been recovered to the extent practicable. These evaluations are focused on determining if ongoing remedial actions should continue, if additional contingency actions are needed to meet cleanup objectives, or if remedial actions have met associated performance criteria. The primary objectives of this waterfront probing investigation are to:

- Evaluate the spatial (vertical and horizontal) distribution of IHSs remaining in the known historical subsurface plume located beneath the warehouse and loading rack area.
- Evaluate the extent of remaining mobile LNAPL available for capture.
- Identify any remaining IHS source areas with elevated impacts along the waterfront.
- Identify the potential for IHS source areas to create sheen in the adjacent waterway.
- Identify former IHS source areas that have met cleanup standards.
- Evaluate continuing or discontinuing remedial actions.
- Aid in the reevaluation of waterfront hydrologic and geologic conditions.

This investigation is also intended to support a planned hydraulic evaluation of potential changes to groundwater and surface water interactions due to the installation of a new seawall along the northern half of the waterfront at Plant 1 (i.e. evaluate changes to the historic Conceptual Site Model [CSM]), as requested by Ecology (Ecology, 2015). A hydraulic evaluation involving a background water level assessment has been proposed to Ecology under a separate work plan (TechSolve, 2019c) to reevaluate site hydrogeology. The new seawall was installed in 2017 and 2018 to improve the long-term seismic stability of the Site, as previously detailed to Ecology (TechSolve, 2018b). The results of the background water level assessment will be used in conjunction with the results of this probing investigation to meet the objectives listed above.

2. Site Description and History

2.1. Site Description

The Site is located on Harbor Island, a 455-acre man-made island, which lies between the East and West Waterways of the Duwamish River (Figure 1). Shallow soils, the upper 6 to 16 feet or so, were deposited hydraulically from dredged sediments of the Duwamish River. Deeper soils are native deposits that have been shown to be unconsolidated silty, fine-to coarse-grained sand with thin

interbeds of silt and clay (Weston 1993). The native sediments were deposited by the fluvial /deltaic environment of the Duwamish River delta.

Currently, about 95 percent of the island is covered by impervious surfaces, following island-wide redevelopment that occurred in the early 2000s. The island's pervious surfaces, which are areas of groundwater recharge, consist primarily of land adjacent to aboveground storage tanks and by railroad tracks.

Plant 1 occupies about 12 acres on the western portion of the island, along the West Waterway of the Duwamish River. Plant 1 was constructed in the 1930s and has operated as a bulk fuel storage and transfer facility under several owners since that time. Seaport Midstream Partners, LLC is the current property owner and TLP Management Services, LLC manages ongoing operations at the Site.

Remedial Investigation/Feasibility Studies (RI/FS) (Geraghty & Miller, 1994, 1996, and 1997) showed IHSs present in groundwater and soil at the Site, including highly weathered total petroleum hydrocarbons (TPH) as diesel (TPH-D), with lesser amounts of weathered gasoline (TPH-G) and heavier oil (TPH-O), likely from historic spills at the Site. The RI/FS showed that a primary area of impact at the Site was a petroleum based LNAPL plume located beneath the warehouse and loading rack area adjacent to the Duwamish Waterway (Figure 2). The vertical distribution of this plume was contained within an approximate 5-foot smear zone extending from the vadose zone, through the capillary fringe, and into the saturated zone. The vertical width of the smear zone along the waterfront has been attributed to seasonal groundwater fluctuation of around 2 feet and an additional 1 or 2 feet of daily groundwater fluctuation from diurnal tidal changes, which alter surface water elevations by up to 16 feet in the adjacent waterway. The RI attributed the reduced tidal response in shallow groundwater directly inland of the waterway to the following:

- A lack of continuous horizontal stratification in the shallower hydraulic fill, which does not maintain pathways as transmissive to surface waters as those in the deeper native deposits.
- That several barriers to shallow groundwater flow exist along Plant 1 including the warehouse foundation installed to approximately 22.5 feet beneath the warehouse floor (elevation of approximately -7 feet NAVD 88) and the historical wooden island bulkhead that is estimated to have been completed to an elevation of about -5 feet NAVD88.

In addition, it is likely that the low conductivity of the shallow sediments, combined with the high, unconfined storage coefficient, contribute significantly to the attenuation of the tidal effects on inland groundwater levels.

The new seawall installed in 2017 and 2018 along the northern portion of Plant 1 has created an additional barrier that may further attenuate the tidal response on inland groundwater, further isolating groundwater from surface water in nearby upland areas. The seawall consists primarily of an interlocking steel sheet pile wall, with anchored tiebacks. The sheet pile wall was completed to an elevation of approximately -68 feet NAVD88.

2.2 Site History

Consent Decree No. 00-2-05714-8SEA was entered into with Ecology in 2000, which established cleanup criteria and identified remedial actions to be implemented at the Site. The Consent Decree identified remedial actions for the Plant 1 Waterfront area and for inland Plant 1 soils in the Cleanup

Action Plan (CAP) (Ecology, 1999) and in the Engineering Design Report (EDR) (TechSolv and AG&M, 2000).

Remedial actions implemented at Plant 1, include:

- Pumping and treatment for an LNAPL plume and dissolved hydrocarbon recovery along the waterfront of Plant 1 (interim actions were implemented in the early 1990s and final actions were implemented in the early 2000s following implementation of the Consent Decree).
- Excavation of accessible TPH impacted soil "hot spots" in the inland portions of Plant 1.
- Natural attenuation of inaccessible soils.
- Air Sparging and Soil Vapor Extraction (SVE) for accelerated mass removal of residual hydrocarbons in inaccessible soils at Plant 1.
- Remedial action performance and groundwater compliance monitoring.
- Deed restrictions.
- Institutional controls.

2.2.1 Cleanup Criteria

IHSs and Site cleanup levels (CULs) for the Site were identified and defined in the CAP and are summarized below.

The TPH cleanup action level for subsurface soil at the primary area of concern (Plant 1) was established to meet remedial objectives for protecting surface water at property boundaries and shorelines of the Duwamish Waterway. The soil Total TPH (TPH-G+TPH-D+TPH-O) CUL is also protective for other chemical constituents in petroleum product (i.e., benzene, toluene, ethylbenzene, xylenes [BTEX]) and is:

Total TPH 10,000 milligrams per kilogram (mg/kg)

Site groundwater CULs established by Ecology were based on surface water standards, to be protective of aquatic organisms in the Duwamish River. These standards were based on adopted ambient water quality criteria (Washington Administrative Code 173-201A and Section 304 of the Federal Clean Water Act). Surface water standards were not established for TPH when the CAP was approved; therefore, groundwater CULs for TPH-G, TPH-D, and TPH-O were selected by Ecology as protective cleanup goals. Site groundwater CULs are:

Product (LNAPL)	No sheen
Benzene	71 micrograms per liter (μg/L)
Carcinogenic Polycyclic Aromatic Hydrocarbons (cPAHs)	0.031 μg/L
TPH-G	1,000 μg/L
TPH-D	10,000 µg/L
ТРН-О	10,000 μg/L

The achievement of the CULs listed above is to be met at various conditional points of compliance, which were established in the Groundwater Compliance Monitoring and Contingency Program (GWCMCP) (TechSolv, 1999a), Exhibit F of the Consent Decree (Ecology, 2000), and are not applicable at all areas of the Site.

Compliance criteria were established in the GWCMCP for both performance monitoring and for confirmational monitoring. Performance monitoring is intended to document that the cleanup action is performing as anticipated. Confirmational monitoring is intended to monitor the long-term effectiveness of the cleanup action once cleanup and performance and compliance criteria have been met.

Many of the confirmational and performance monitoring criteria have been met at the Site, as previously documented to Ecology (TechSolve, 2019b) and monitoring is ongoing to evaluate if remaining criteria have been met. This report is intended to directly evaluate if the separate phase hydrocarbon performance criteria for Plant 1 (Section 2.2.3 of the GWCMCP) have been met. The performance criterion for separate phase hydrocarbons (LNAPL) is the removal to the maximum extent practicable or a lack of measurable product thickness in compliance monitoring wells, product recovery systems and until a persistent sheen is no longer observed on the waterway. A measurable thickness of free product is defined as greater than or equal to 0.01 feet thick.

This report is also intended to aid in the indirect evaluation of whether criteria have been met, including the following:

- Plant 1 performance criteria for dissolved TPH constituents in groundwater, which is obtainment of the CULs listed above for Benzene, TPH-G, TPH-D, and TPH-O in confirmational monitoring wells (i.e. monitoring wells at the property boundary and downgradient of plume areas).
- Plant 1 confirmational criteria for separate phase hydrocarbons, which is the lack of sheen in associated compliance monitoring wells for a period of a year.
- Plant 1 confirmational criteria for groundwater, which is obtainment of the CULs listed above in confirmational monitoring wells for four quarters or the concentrations of analytes have stabilized and reached equilibrium.

2.2.2 Waterfront Remedial Actions

Groundwater remedial actions have been conducted along the waterfront at Plant 1 since 1992. An interim groundwater/LNAPL recovery system operated from 1992 through 2002, and an interim soil vapor extraction (SVE) system operated from 1996 through 2002. Final remediation systems were installed in 2002, as described in the EDR (TechSolv and AG&M, 2000a), and included SVE, groundwater/LNAPL recovery, and air sparging.

Operation of the waterfront SVE system was discontinued in 2008, as the system no longer recovered measurable concentrations of hydrocarbons and no longer enhanced biodegradation in inaccessible hot spot soils. Air sparging along the waterfront was also discontinued in 2008, as SVE air monitoring data indicated air-sparging operations no longer volatilized measurable quantities of hydrocarbons and likely increased biofouling in LNAPL/groundwater recovery wells, as documented in past reports (e.g. TechSolve, 2018a).

While many of the remedial actions have been completed at Plant 1, the groundwater/LNAPL recovery system continues to operate along the waterfront to depress groundwater and capture LNAPL and shallow groundwater containing dissolved hydrocarbons. Additionally, containment booms are monitored and maintained on the West Duwamish Waterway adjacent to Plant 1 to contain and capture occasional sheens that have appeared on surface water.

Monitoring is conducted at Plant 1 in accordance with the GWCMCP and SAP to confirm that remedial actions have attained cleanup standards and/or have met defined performance standards. Monitoring results show that ongoing and completed remedial actions have improved groundwater, soil, and surface water conditions at Plant 1, including reductions in free LNAPL, reduced dissolved hydrocarbon concentrations in groundwater, and decreased frequency of hydrocarbon sheens in the Duwamish Waterway (TechSolve, 2018a).

Reduced LNAPL and dissolved hydrocarbon capture by ongoing remedial actions has been attributed to most recoverable IHSs having been captured or reduced. Performance monitoring data along the waterfront at Plant 1 indicate that the groundwater/LNAPL recovery system has likely recovered LNAPL to the extent technically practicable, as required by the Consent Decree. Several lines of evidence support this hypothesis, including both system recovery data and monitoring data from recovery and monitoring wells. Asymptotic recovery curves (Figure 3) indicate limited mobile LNAPL likely remains along the waterfront at Plant 1, as most of the available LNAPL has been recovered during the 26 years that interim (1992-2002) and final (2002-2018) groundwater LNAPL recovery systems have operated at the Site. Monitoring for concentrations of dissolved phase IHSs in groundwater over time has shown decreasing or stable concentrations in confirmation compliance wells (Table 1 and Appendix A) and recovery wells (Table 2 and Appendix A) along the Plant 1 waterfront by the warehouse and loading rack areas. Much of this area now appears to no longer contain recoverable LNAPL and dissolved IHSs are mainly below CULs.

System performance and monitoring data collected prior to this investigation indicated that LNAPL has been recovered to the extent technically practicable and this investigation was intended to provide additional data to support or disprove this conclusion. While reductions in LNAPL capture over time were believed to be due to a lack of remaining LNAPL, it was possible that reduced LNAPL capture was the result of decreased system efficiencies (e.g. biofouling of wells decreasing well yields), detailed in the Waterfront Probing Work Plan (TechSolve, 2019a).

3. Waterfront Probing Investigation

The following tasks were completed as part of the waterfront probing investigation:

- Selection of specific probing locations and well locations.
- Location of subsurface utilities in the work area.
- Direct push probing of soils at 41 boring locations to 15 to 20 feet below ground surface (BGS) by licensed drillers.
- Field screen soil cores for IHSs and LNAPL.
- Backfilling and sealing of select boring locations.

- Installation and development of piezometer wells in 11 completed borings and development of the wells for groundwater sampling and for potential pumping studies, to be evaluated as part of the Site hydraulic evaluation.
- Laboratory analysis of soil cores and groundwater samples for concentrations of IHSs and LNAPL impacts.
- Reevaluation of the CSM with updated data from the investigation.
- Preparation of this summary report to detail the investigation results and provide recommendations for the continuation, discontinuation or implementation of contingency remedial actions to meet the cleanup criteria of the Consent Decree.

3.1. Probing Locations

Probing locations were focused on the area of the Site along the Plant 1 waterfront with known historical IHS impacts (Figure 2). The probing locations (Figure 4) were selected to collect representative soil and groundwater samples on close enough spacing to accurately evaluate conditions along the waterfront. These locations were selected to cover accessible locations at the Site, along the Plant 1 waterfront where active remedial actions are ongoing. Probing locations were selected utilizing a grid pattern with row spacing set at 20 feet on center. Site restrictions, subsurface utility locations, subsurface and overhead obstructions, and drill rig limitations required the abandonment and relocation of a limited number of borings as field conditions dictated.

3.2. Probing Field Activities

3.2.1. Site Preparation and Utility Location

Prior to initiating field activities, potential probing areas were delineated and marked in the field. These areas were then checked for subsurface utilities by placing a public utility request. A ground penetrating radar survey was then conducted on all boring locations by Applied Professional Services (APS). A privately conducted utility location was next conducted by APS to further confirm and evaluate the findings from the public locate and the ground penetrating radar survey. Probing locations (Figure 4) were adjusted as necessary based upon the location of identified utilities.

3.2.2. Probing and Well Installation

Direct-push soil borings were advanced by Washington State licensed drillers from Cascade Drilling, July 22nd through July 31st, 2019. Construction of all soil borings and subsequent decommissioning was conducted in accordance with WAC 173-160. All probing was overseen by TechSolve geologists registered in the state of Washington, or scientists working under the supervision of a registered geologist. All onsite staff utilized throughout the project were 40-hour HAZWOPER trained and had undergone site specific training on the Site, project, and potential hazards.

Cascade Drilling used a Geoprobe[®] Model 7822DT direct push track mounted probe rig for the completion of all borings. Soils from all borings were collected from ground surface, continuously above, through, and below the historically known smear zone in 5-foot intervals using new acrylic liners for each sample interval. Soil borings were completed to a depth of 15 feet BGS outside of the warehouse, and to a depth of 20 feet BGS inside the warehouse. The deeper boring depths were selected as the warehouse sits on an elevated foundation that is on average 3.5 feet higher than surrounding ground surface. All borings were completed to depths below where LNAPL and IHS impacts were detected with field screening, indicating sampling was completed across the entire historically known smear zone in all borings.

A total of 11 of the completed borings were converted to temporary piezometer wells (Figure 4). Wells were installed in these borings to:

- Evaluate remaining LNAPL thickness on groundwater.
- Evaluate for LNAPL and sheen presence on groundwater.
- Collect groundwater samples for the analysis of dissolved phase IHSs.
- Potentially use in additional studies that may be conducted as part of the Site hydraulic evaluation.

All piezometer wells were installed with 1-inch inner diameter (ID) PVC riser-pipe in the unsaturated zone. The piezometer wells were screened with 1-inch ID Geoprobe prepackaged well screen, which are 5-feet long 0.01-inch slotted PVC well screen surrounded with stainless steel mesh and filled with 20/40 sand. The annulus around the prepackaged well screens was filled with 20/40 washed silica sand from 1 foot below to 2 feet above the screened interval. Bentonite chipping, grouting, and installation of flush mount monuments were then conducted in accordance with WAC 173-360 to complete well construction.

All temporary piezometer wells were screened through the vadose zone and into the saturated zone, with all screened intervals extending above the highest historically measured groundwater elevations in this area of the Site. Historical groundwater elevations along the Plant 1 waterfront are detailed in the hydrograph included as Figure 5. The locations of the wells included on the hydrograph are shown on Figure 6.

Wells completed in the elevated warehouse were screened from 7 feet BGS to depth, and wells outside the warehouse were screened from 3 feet BGS to depth. Seven wells were installed with 10 feet of well screen and the remaining four wells (MW-02-PI, MW-07-PI, MW-10-PI, MW-11-PI) were installed with 15 feet of well screen. The additional 5 feet of screen were added to these four wells for additional drawdown capacity if additional studies (e.g. pumping tests) are needed near these wells as part of the upcoming hydraulic evaluation. Locations for these more deeply screened wells were selected to be near current recovery wells (Wells RW-4, RW-2, and RW-9 [Figure 6]) with the highest remaining IHS impacts detected (Table 2 and Appendix A) where, if needed, additional pumping tests are most likely to be conducted.

All wells were developed the week of August 19, following completion of well drilling. All wells were gauged prior to development and no free LNAPL was observed in any well. Each well was then developed for several hours utilizing a combination of pumping and surging with Waterra Hydrolift-2 pumps with dedicated tubing and foot valves. Over five well volumes were removed from each well and well development was continued until turbidity decreased and stabilized.

3.2.3. Soil Logging, Field Screening, and Soil Sampling

Soil cores were logged (Appendix B) for lithologic description and were described using the Unified Soil Classification System, in accordance with Section 3: Field Procedures, of the SAP (Appendix C). All major soil layers encountered were similar to classification results from previous studies (e.g. the RI/FS). Soil cores were retained from all 11 of the borings converted to piezometer wells and these soils can be analyzed to confirm grain size distribution and textural classifications if needed as part of future studies.

Soil cores were visually screened for staining indicative of petroleum impacts and for hydrocarbon odors. Field screening of soils for LNAPL and sheen was performed at one-foot intervals in all borings with a sheen test, where about 1 teaspoon of soil is placed in a gold pan containing water, contents are mixed and the water surface is visually observed for the presence of a petroleum sheen or LNAPL. Sheen and LNAPL observed were classified as:

- NS (no sheen) No visible sheen on the water surface.
- SS (slight sheen) Light, colorless, dull sheen; spread is irregular, not rapid.
- MS (moderate sheen) Pronounced sheen over limited area; some color/iridescence; spread is irregular, sometimes rapid; sheen does not spread over entire water surface.
- HS (heavy sheen) Heavy sheen with pronounced color/iridescence; spread is rapid; the entire water surface is covered with sheen.
- HS w/LNAPL (heavy sheen with LNAPL) Heavy sheen with free LNAPL observed on water.

A headspace vapor analysis was also performed at one-foot intervals in all borings by placing soil in an enclosed bag and measuring the presence of organic vapors with a photoionization detector (PID). The PID utilized was a MiniRae model 3000 with a 10.6 eV lamp, which was calibrated daily with 100 parts per million (ppm) isobutylene gas.

Field screening results were used to direct further sampling of undisturbed soils from each boring for laboratory analyses as detailed in Sections 4.1.1 and 4.1.3. Sample collection during the probing investigation was conducted in accordance with the requirements of the SAP (Appendix C). The preparation of soil samples for the analysis of volatile organic compounds (VOCs) was conducted in accordance with EPA Method 5035A.

Detailed soil boring logs and field notes were recorded to document environmental conditions encountered and probing activities, respectively. The soil boring logs are included in Appendix B.

3.2.4. Surveying

Boring surface completion points, piezometer wells top of casing points, and piezometer well lid points were surveyed by licensed surveyors from Pacific Geomatic Services, Inc. for both location and elevation. All points were surveyed vertically for elevation to within 0.01 foot and for horizontal location to within 0.1 foot. Points were surveyed for location to the Horizontal Datum NAD-83 and for elevation to the Vertical Datum NAVD-88. All elevations listed in this report are in the NAVD-88 datum unless otherwise noted. Note that elevations provided in historical reports referenced in this report (e.g. the RI/FS) were typically to the no longer used NGVD-29 Datum, and elevations provided by others (e.g. for the new seawall) are often to Mean Low Low Water (MLLW). Comparison of points at the Site that have been surveyed in multiple datums show NGVD-29 elevations are on average 3.4 feet lower than NAVD-88, and MLLW elevations are on average 2.4 feet higher than NAVD-88.

3.2.5. Decontamination Procedures

All down-hole and reusable equipment was decontaminated prior to use at new locations to prevent cross-contamination of borings, wells, samples, as well as to prevent the potential transport of contaminants offsite. Decontamination procedures were conducted in accordance with Section 3.1.4 of the SAP (Appendix C). Decontamination waste materials were managed according to the procedures outlined below.

3.2.6. Waste Handling and Disposal

All investigation derived wastes were managed in accordance with Dangerous Waste regulations under 173-303 WAC, as well as Solid Waste Handling Standards under 173-350 WAC, and in accordance with Section 3.1 of the SAP (Appendix C). Generated wastes and their disposal method included the following:

- Two 55-gallon drums containing 1,400 pounds of excavated soils from probing cores. Soils were sampled, profiled, and disposed of as non-hazardous waste by Clean Harbors under a non-hazardous manifest.
- One 55-gallon drum containing 300 pounds of asphalt from coring, disposed of as nonhazardous waste by Clean Harbors under a non-hazardous manifest.
- Decontamination rinse water and fluids, which were processed through the permitted remediation water treatment system onsite and disposed of to the sanitary sewer.
- Disposable personal protection equipment and sampling equipment, which were added to the Site's solid waste stream.

3.3 Groundwater Sampling

Groundwater was sampled from the temporary piezometer wells on September 26th and September 27th, 2019. Sampling was conducted during the lowest low tide periods for the month of September, when groundwater was discharging into surface waters, as requested by Ecology (Ecology, 2019). Sampling was conducted from 0828 to 1225 on September 26 and from 0817 to 1045 on September 27. NOAA tidal predictions for the Lockheed Harbor Island Shipyard Station 9447110 during the September 26th sampling period was from -0.65 to 3.0 feet MLLW (-3.0 to 0.65 feet NAVD88) and from -0.55 to 2.0 feet MLLW (-2.9 to -0.35 feet NAVD88) for the September 27th sampling period. Groundwater elevations measured in the wells ranged from 7.71 to 8.9 feet MLLW (5.36 to 6.55 feet NAVD88) during these sampling periods (Table 3). These data show that during the sampling periods, groundwater elevations were higher than adjacent surface water elevations and groundwater was discharging to surface water throughout the sampling of all wells.

Groundwater was sampled in accordance with the requirements of the SAP (Appendix C). Prior to sampling, all wells were gauged for LNAPL presence/thickness, depth to groundwater, and total depth with an oil/water interface metering probe. Calculations were made to determine the volume of water in each well. The presence of LNAPL and sheen on groundwater were further evaluated by collecting groundwater at the groundwater surface with a new polyethylene bailer, pouring the water into a beaker, and visually observing the water surface for the presence of a petroleum sheen or LNAPL. LNAPL and sheen observations were classified utilizing the same scale listed in Section 3.2.3. If no LNAPL was observed in any well groundwater surface with a polyethylene bailer. Recovered groundwater was visually monitored for parameters throughout sampling and purging, including sheen presence and strength. A minimum of three well volumes of water were purged from each well prior to sampling. Groundwater samples were collected for laboratory analysis once measured field parameters, including; pH, temperature, and conductivity, had stabilized. All field measurements and sampling information was documented and retained on water sampling logs.

3.4 Sample Handling and Analytical Methodology

Sample handling and analyses were conducted in accordance with the standards and procedures identified in the SAP (Appendix C).

Soil samples were selected from each boring for laboratory analysis of IHS concentrations based upon the results of field screening. Soil samples selected for analyses were collected from boring depths with the highest impacts detected with field screening methodologies. Soils were analyzed for concentration of IHSs including TPH-G, TPH-D, and TPH-O utilizing NWTPH-Gx and NWTPH-Dx analytical methods. TPH-G samples were collected in accordance with EPA method 5035A protocol for preparing samples for the analysis of VOCs.

Groundwater samples were collected with polyethylene bailers from shallow groundwater (i.e. at the groundwater surface where the highest concentration of IHSs would be expected). Groundwater samples were analyzed for concentrations of IHSs including TPH-G, TPH-D, TPH-O, and benzene, using NWTPH-Gx, NWTPH-Dx, and EPA-8260 analytical methods, respectively.

All soil and groundwater samples were placed in laboratory provided sample containers, chilled coolers, and delivered to the ALS Environmental - Everett Washington Laboratory under standard chain of custody (COC) protocol. All analyses were performed by ALS Environmental, an Ecology accredited laboratory (Accreditation #: C601-19). Full data package reports were provided by ALS Environmental documenting all analyses performed, which are available upon request and were not included as appendices to this report due to size.

3.4.1. Quality Assurance and Quality Control

Quality assurance (QA) objectives and procedures for field sampling, chain-of-custody, and laboratory analyses are identified in the SAP (Appendix C). The procedures identified in the SAP were used to ensure data are usable, defensible, and can be used for decision making. Specific procedures for sampling, chain-of-custody, calibration, laboratory analysis, reporting of data, internal quality control (QC), audits, preventive maintenance, and corrective actions that were identified in the SAP were utilized during this probing investigation.

Data validation summaries are provided in Appendix D and detail the QA/QC criteria for soil and groundwater field sampling, chain-of-custody, and laboratory analyses. The data validation summaries detail the review of analytical data reports, if data were validated, and whether the QA/QC criteria were met. All laboratory analytical results were reported with associated QA/QC data, and limited data (two results total) were qualified with a J qualifier (the associated values are approximated). All laboratory reports are retained at the TechSolve office.

4. Results

4.1 Field Observations

4.1.1. Soil Observations

Similar soil conditions were encountered in all 41 borings (Figure 4), with black to grey fine silty sands encountered from ground surface (elevations ranging from about 11 to 15 feet) to an elevation of

approximately -3 feet. Deeper soils transitioned to sands, silt, and clay. The shallower and deeper soils are representative of soils encountered during the RI, discussed in Section 2.1. The approximate soil boundaries and interfaces are shown on Figure 7. Soil boring logs detailing soil lithology are included in Appendix B.

4.1.2 Groundwater Observations

Groundwater was encountered at an elevation of between 5 to 7 feet in all borings and during groundwater monitoring of the temporary piezometer wells (Figure 8 and Table 3). The approximate 2 feet of variation in recorded groundwater elevations along the waterfront likely reflect the effects of diurnal tidal exchange, which are discussed in Section 2.1.

4.1.3 IHS Field Screening

All soil cores were field screened from all borings (Figure 4) using the methodology detailed in Section 3.2.3. Visual and odor screening in general showed hydrocarbon odor and staining to be limited to soils at and a few feet below the groundwater interface. Shallow unsaturated soils were mostly free of hydrocarbon odor and staining. Hydrocarbon odor and staining decreased with depth in the saturated zone, with little to no detections at boring termination depths. Odor and visual staining screening results were consistent with field screening and laboratory analytical results (i.e. the highest IHS impacts were consistently detected across all screening and laboratory analysis methods).

The results from field screening soils for LNAPL and sheen are provided in Table 4. Field screening for LNAPL and sheen was conducted at one-foot intervals in all borings, utilizing the sheen test detailed in Section 3.2.3. Screening results showed a heavy sheen or heavy sheen with LNAPL to be present in 27 of the 41 borings, with limited free LNAPL detected in 7 borings. Heavy sheens were typically limited to a 1 to 2-foot vertical band in each boring, at an elevation of around 3 feet. Wider vertical bands of heavy sheen, from 4 feet to 6 feet thick, were only observed in four borings (B-02, B-03, B-20, and B-43). Moderate to slight sheens were typically observed at shallower and deeper depths than where heavy sheens were observed. Slight to moderate sheens were typically observed in borings over a vertical band ranging from 1 to 12 feet in width. The shallowest sheen observed was in Boring B-43 at an elevation of 8 feet and the deepest sheen observed was in Boring B-06 at an elevation of -4.5 feet.

The results from headspace vapor analysis of soils are provided in Table 4. Headspace vapor analysis was conducted at one-foot intervals in all borings with a PID utilizing the test procedure detailed in Section 3.2.3. Soils tested with a PID showing elevated headspace vapor results typically also had greater LNAPL and sheen detected with the field screening results detailed in the previous paragraph. The highest PID readings (i.e. greater than 100 ppm) were typically found over a 1 to 2-foot vertical band, at an elevation of around 3 feet. Lower PID readings (i.e. less than 100 ppm) were typically observed at shallower and deeper depths than where the highest PID readings were observed. Lower PID readings were typically observed in borings over a vertical band ranging from 1 to 12 feet in width. The shallowest PID readings detected above 100 PPM were from boring B-43, at an elevation of 8 feet, and the deepest PID readings detected above 100 PPM was in Boring B-17 at an elevation of -1.5 feet.

Field screening results (PID and LNAPL/sheen screening) indicate that the unsaturated soils in all borings, except for Boring B-43, showed minimal to no signs of IHS impacts. Field screening results from the most impacted borings showed the highest soil impacts to be located within a 2 to 4-foot vertical band. IHS impacts were not detected in most borings below an average elevation of -3.1 feet

MSL. The highest concentrations of IHSs detected in soil were, on average, 2.7 feet below the elevation of static groundwater. Groundwater was encountered at an average elevation of 5.4 feet during probing and the average static groundwater elevation measured in the temporary piezometer wells was 6.0 feet. The highest soil impacts were detected at an average elevation of 3.3 feet.

The combined results from the visual screening, odor screening, LNAPL and sheen screening, and headspace vapor analyses were utilized to identify the most impacted soils from each boring. The most impacted soils identified in each boring were then collected for laboratory analyses.

4.2 Soil Analytical Results

Soil samples were selected from each boring for laboratory analysis of IHS concentrations based upon the results of field screening. Soil samples selected for analyses were collected from boring depths with the highest impacts detected with field screening methodologies. These soil samples were collected on average at an elevation of 3.3 feet, with the highest sampling elevation at 5.5 feet in Boring B-09 and the lowest sampling elevation at -1.5 feet in Boring B-17.

Soils were analyzed for concentration of IHSs including TPH-G, TPH-D, and TPH-O utilizing NWTPH-Gx and NWTPH-Dx analytical methods. The results from these analyses are presented in Table 4.

A comparison of the analytical results from soils analyses to the established soil CUL of 10,000 mg/kg for Total TPH (TPH-G+TPH-D+TPH-O) showed the following:

- Twenty-eight of the 41 borings were below the CUL.
- Seven borings (B-32, B-15, B-06, B-24, B-21, B-13, and B-43) exceeded the CUL by less than 3,000 mg/kg.
- Three borings (B-14, B-43, and B-01) exceeded the CUL by less than CUL by less than 8,000 mg/kg.
- Two borings (B-20 and B-19) were more than double the CUL.
- One boring (B-02) was more than triple the CUL.

An evaluation of the TPH fractions detected in the soil analyses shows that on average, Total TPH was comprised of 13% TPH-G, 78% TPH-D, and 9% TPH-O. Borings with TPH-G fractions greater than 20% (as high as 52% TPH-G in Boring B-41) were all below the associated Total TPH CUL. All but one of the borings that exceeded the Total TPH CUL had a TPH-D fraction greater than 80%. The analysis of the TPH fractions indicates that the TPH present is primarily diesel range with lesser amounts of gasoline and lube oil.

An evaluation of TPH fractions and chromatograms indicated that remaining TPH is weathered to highly weathered. Individual chromatograms from the TPH-D/O and TPH-G soil analyses were provided by the laboratory. The response signals plotted on the chromatograms, primarily from the TPH-D/O analyses, in general, lack clearly defined peaks and appear as unresolved humps, indicating that the detected TPH is highly weathered. A laboratory chemist reviewed all chromatograms and provided notations on the chromatograms estimating product type. These notations indicated that the TPH product type to be primarily weathered to highly weathered diesel. The full data package containing the chromatograms was not included as an appendix due to size and can be provided upon request.

4.3 Groundwater Sampling and Analytical Results

The 11 temporary piezometer wells (Figure 8) were gauged for the presence of LNAPL and sampled for laboratory analyses on September 26th and 27th, 2019, following well development and during a low tide period, as detailed in Section 3.3. Groundwater was collected at the groundwater surface (i.e. where the highest concentration of IHSs would be expected) for visual sheen and LNAPL observations and for laboratory analyses. Groundwater samples were analyzed for concentrations of IHSs including TPH-G, TPH-D, TPH-O, and benzene, using NWTPH-Gx, NWTPH-Dx, and EPA-8260 analytical methods, respectively. The results from the gauging and analyses are presented in Table 3.

Well gauging with an oil/water interface metering probe showed no indication of measurable or detectable free LNAPL to be present in any of the 11 temporary piezometer wells. The presence of LNAPL and sheen on groundwater was then visually evaluated from groundwater collected at the groundwater surface, which showed the following:

- No sheen was observed on groundwater in four temporary piezometer wells (MW-04-PI, MW-05-PI, MW-06-PI, and MW-08-PI)
- A slight sheen was observed on groundwater in six temporary piezometer wells (MW-01-PI, MW-02-PI, MW-03-PI, MW-07-PI, MW-09-PI, and MW-10-PI)
- A moderate sheen was observed on groundwater in one temporary piezometer well (MW-11-PI)
- A heavy sheen was not observed on groundwater in any temporary piezometer well.
- No free LNAPL was observed on groundwater in any temporary piezometer well.

A comparison of results from the groundwater analyses for dissolved phase IHSs in the temporary piezometer wells to the dissolved phase groundwater CULs established for the conditional points of compliance showed the following:

- Seven of 11 wells were below all dissolved phase groundwater CULs.
- All 11 wells were below the benzene CUL of 71 μ g/L.
- Two wells (MW-10-PI at 2,200 µg/L and MW-11-PI at 6,300 µg/L) exceeded the TPH-G CUL of 1,000 µg/L.
- Two wells (MW-02-PI and MW-03-PI both at 13,000 $\mu g/L)$ exceeded the TPH-D CUL of 10,000 $\mu g/L.$

Note that the dissolved phase CULs listed above are applicable to confirmational monitoring wells, which monitor groundwater downgradient of the plume and where groundwater is discharging to surface waters. These dissolved phase CULs are not applicable to the temporary piezometer wells, which represent groundwater conditions within a historically identified source area.

5. Summary and Findings

This investigation evaluates the extent of remaining IHSs in soils and groundwater along the Plant 1 waterfront. This area of the Site (Figure 2) was previously identified in the RI as containing TPH and LNAPL in excess of applicable CULs. This area of the Site contains an active fuel loading rack, offices, a warehouse, and other infrastructure that make it inaccessible to remedial actions such as excavation. In-situ remedial actions have been ongoing in this area of the Site to improve site conditions since the late 1990s. While many of these actions have been completed, a groundwater/LNAPL recovery system

continues to operate along the waterfront to depress groundwater and capture LNAPL and shallow groundwater.

5.1. IHSs Remaining in Soil

Field screening results and laboratory analyses of soils from completed borings (Table 4) indicate that much of the soil in the investigation area is at or below the associated CUL for Total TPH. Soils exceeding the CUL appear to be confined within a 2 to 4-foot vertical band extending downward from the groundwater surface. The highest concentrations of IHSs were encountered, on average, 2.7 feet below the elevation of static groundwater. There were little to no detections of IHSs in soil in shallow unsaturated soils or in soils at depths below the warehouse foundation, island bulkhead, and new seawall, which function as barriers to groundwater and surface water exchange.

5.1.1 Lateral and Vertical Soil TPH Contours

Two-dimensional contour maps were developed to estimate the lateral and vertical extents of Total TPH in soil in the study area. Figure 9 provides a lateral estimation of TPH using the laboratory analytical results collected from soils exhibiting the highest TPH impacts in each boring. Figure 9 represents the highest concentrations of total TPH expected throughout the study area. Figure 10, Figure 11, and Figure 12 provide vertical estimations of the extent of total TPH. The locations of these north/south cross-sectional figures are highlighted on Figure 9. The TPH concentration contours plotted on Figures 10-12 utilize PID and sheen test screening results to extrapolate a total TPH value for each one-foot interval in each boring. Estimated total TPH values for each one-foot interval were calculated by multiplying the boring's laboratory derived total TPH concentration by a ratio of the PID and sheen test screening results were then gridded utilizing Kriging in Surfer and contour maps were developed and manually adjusted to reduce bias (e.g. the overestimation of TPH in the unsaturated zone and the underestimation of TPH in the upper several feet of the saturated zone).

The lateral estimations of the highest concentrations of total TPH, shown in Figure 9, indicate that soils exceeding the CUL are located by the south end of the warehouse (by B-02 and nearby borings), near the center of the warehouse (by borings B-19 and B-20), and by the loading rack in the northern part of the study area. Figure 9 also indicates that the highest total TPH concentrations expected are below the soil total TPH CUL in portions of the warehouse and directly north of the warehouse.

The vertical estimations of total TPH shown on Figures 10-12 indicate that the bulk of total TPH exceeding the soil total TPH CUL is limited to the upper 5 feet of saturated soil in a few isolated areas, also identified in Figure 9. The vertical estimations indicate that a few small "hot spots" remain in soil and that most soils in the investigation area are below the soil total TPH CUL.

5.1.2 Residual LNAPL Saturation in Soil

While some soils in the investigation area exceed the total TPH CUL, the distribution and concentrations appear insufficient to suggest the presence of mobile and recoverable LNAPL. Numerous documents (e.g. MassDEP, 2016) have been published identifying residual LNAPL concentration values, below which LNAPL is theoretically immobile in subsurface soils, and, by extension is:

• the maximum level of LNAPL that can exist in soil and not be mobile; and

• the minimum level of LNAPL that will remain in soil after the completion of conventional remedial recovery efforts.

Residual LNAPL concentration values are dependent on specific soil and LNAPL physical properties, so actual site-specific values will vary. However, various researchers have published values for these metrics since the early 1960s and a collection of these values appears in API's Soil and Groundwater Bulletin No. 9 (Brost et al., 2000). These values can be used as a line of evidence to determine if LNAPL is mobile onsite. Published values for conditions most closely matching Site conditions (i.e. weathered diesel range LNAPL in fine sands to silty sands) range from 13,333 mg/kg to 98,100 mg/kg, with several values in the 30,000 to 34,000 mg/kg range.

The actual residual LNAPL concentration values for soils in the investigation are likely to be on the higher end of the published values, given that:

- LNAPL detected in the investigation was primarily weathered to highly weathered diesel, which indicates an older release, typically less mobile than a new, or fresh, release.
- Elevated TPH soil concentrations were only detected over narrow vertical profiles less than 4 feet in thickness, indicating a diminished LNAPL source area.
- The highest TPH soil concentrations were typically encountered 2 feet below static groundwater. If LNAPL were mobile, highest TPH concentrations would be expected at the groundwater interface, where LNAPL would accumulate.

5.2. IHSs Remaining in Groundwater

Groundwater sampling and analytical results (Figure 8 and Table 3) showed no free LNAPL to be present throughout the investigation area and dissolved phase IHSs were mainly below CULs applicable to the more deeply screened (i.e. where groundwater and surface water exchange is occurring) point of compliance wells.

Groundwater gauging results showed no measurable free LNAPL to be present on groundwater in the 11 temporary piezometer wells. Visual inspections of groundwater from these wells showed no signs of LNAPL presence or that of a heavy sheen in the 11 wells. One well (MW-11-PI) exhibited a moderate sheen, six wells (MW-01-PI, MW-02-PI, MW-03-PI, MW-07-PI, MW-09-PI, MW-10-PI) exhibited a slight sheen, and four wells (MW-04-PI, MW-05-PI, MW-06-PI, MW-08-PI) exhibited no sheen on groundwater.

Groundwater sampling and laboratory analysis results showed dissolved phase IHSs to be below point of compliance CULs in 7 of the 11 temporary piezometer wells. Two wells (MW-10-PI and MW-11-PI) exceeded the TPH-G CUL and two wells (MW-02-PI and MW-03-PI both at 13,000) exceeded the TPH-D CUL. All 11 wells were below the benzene CUL.

5.3. Compliance with Performance and Confirmational Criteria

5.3.1. LNAPL Removal Performance Criterion

The performance criterion for the removal of separate phase hydrocarbons (LNAPL) to the maximum extent practicable or a lack of measurable product (LNAPL) thickness in compliance monitoring wells appears to have been met in the investigation area. As detailed in previous sections, no measurable

LNAPL was detected on groundwater in any of the temporary piezometer wells installed during this investigation. Detected TPH in soil appears to be below residual saturation limits, indicating that any remaining LNAPL is not mobile or recoverable by the active groundwater/LNAPL recovery system.

5.3.2. Waterway Sheen Criterion

The performance criterion for a persistent sheen to no longer be observed on the waterway was also evaluated as part of this investigation. Specifically, this investigation was intended to evaluate if there was a pathway for LNAPL migration from inland soils and groundwater to surface water in locations where a sheen has historically been observed on surface water. Oil sorbent booms have been maintained in the waterway to contain sheens that have appeared on surface water. Two booms are currently located alongside the warehouse (Figure 6) to best contain occasional sheens that have been observed in the waterway. Booms and the waterway are monitored weekly, at a minimum, for the presence of oil sheens and the monitoring results are included in Table 5. Sheen events over the last decade (2010 through 2019) have been infrequent, with 75 light sheens observed at various times along the warehouse in the 1,428 inspections conducted. A light sheen was last observed in the waterway in August of 2019.

It has been theorized in previous reports that the observed sheens originated from small cracks and discontinuities in the concrete warehouse foundation and underlying sheet piling, or island bulkhead. The foundation and bulkhead act as a "hanging" wall, trapping LNAPL while allowing groundwater to flow beneath the base of the foundation and bulkhead.

This investigation indicates that sheens detected in the waterway are not originating from an ongoing source of LNAPL migrating through discontinuities in the concrete warehouse foundation and underlying sheet piling, or island bulkhead. This is supported by data from this investigation showing a lack of LNAPL in the temporary piezometer wells, and data showing that detected TPH in soil appears to be below residual LNAPL saturation limits. The lack of an ongoing inland LNAPL source causing surface water sheen is also supported by the ongoing sheen monitoring data (Table 5), which show that no sheens have been observed over the last decade at low tide periods when groundwater to surface water gradients are highest. All observed sheens have occurred at medium to high tidal stages.

Sheen observances have occurred in what has been historically referred to as "targeting" where small bubbles create a sheen in the waterway, several feet away from the shoreline. The mechanism for sheen occurrence most closely resembles what has been defined by others (ITRC, 2018) as ebullition, where:

"Biodegradation of naturally-occurring organic compounds or of the petroleum itself that is affecting sediments, can generate gases that migrate upward through sediments due to buoyancy. NAPLs are hydrophobic and preferentially attach to the surface of a gas bubble passing through, which subsequently transports the NAPL to the air-water interface. NAPL-coated bubbles reaching the surface of a water body yield a sheen because the surface tension of water is much higher than petroleum."

Determining the source of the observed sheens in the waterway was beyond the scope of this investigation. Sources could be from the adjacent island bulkhead, which was created with creosote timbers, or from a historic release that migrated or occurred waterward of the existing concrete warehouse foundation, underlying sheet piling, and island bulkhead. The source is

assumed to be limited, as sheens have only been observed in 5% of the monitoring conducted over the last decade, the observed sheens have been light to very light in strength, are not widespread, and have occurred in limited and defined locations that have been contained and captured with booms.

5.3.3. Dissolved Phase IHS Criterion

The Plant 1 performance criterion for dissolved TPH constituents in groundwater is obtainment of the CULs listed in Section 2.2.1 for Benzene, TPH-G, TPH-D, and TPH-O in confirmational monitoring wells (i.e. monitoring wells at the property boundary and downgradient of plume areas). The Plant 1 confirmational criteria for groundwater is obtainment of these CULs in confirmational monitoring wells for four quarters or until analytes have stabilized and reached equilibrium.

While the obtainment of the dissolved phase CULs is not a criterion for shallow groundwater in the area that was investigated in this report, shallow groundwater was evaluated in the study area to determine if remaining dissolved phase IHSs posed a potential risk to the confirmational monitoring wells, which are screened in deeper groundwater, where groundwater and surface water exchange is occurring.

The potential for dissolved phase IHS migration from shallow groundwater in the investigation area to deeper groundwater, representing the point of compliance, in excess of CULs appears unlikely due to the following:

- Shallow groundwater in the former source area, which is contained by bulkheads, seawalls and foundations, is predominantly below the dissolved phase CULs applicable to deeper uncontained groundwater that discharges to surface water. Analytical results showed dissolved phase IHSs to be below the CULs in 7 of the 11 temporary piezometer wells, indicating shallow groundwater meets the deeper groundwater's CULs near these wells.
- TPH-G concentrations in the confirmational monitoring well (AMW-04) nearest the two wells that exceeded the TPH-G CUL (MW-10-PI and MW-11-PI) has been below the CUL in every quarter since monitoring was initiated in 2000 (Table 1 and Appendix A).
- TPH-D concentrations in the confirmational monitoring well (AMW-02) nearest the two wells that exceeded the TPH-D CUL (MW-02-PI and MW-03-PI) has been below the CUL in every quarter since monitoring was initiated in 2000 (Table 1 and Appendix A).

Additional evaluations relating to the obtainment of the dissolved phase IHS criteria for performance and confirmational monitoring have been proposed as part of a planned site hydraulic evaluation, which is currently being formalized with input from Ecology.

5.4. Recommendations for Ongoing Remedial Actions

As detailed in Section 2.2.2, a groundwater/LNAPL recovery system continues to operate along the waterfront to depress groundwater and capture LNAPL and shallow groundwater containing dissolved hydrocarbons. The groundwater/LNAPL recovery system pumps shallow groundwater from 9 recovery wells (RW-1, RW-2, RW-4, RW-5, RW-6, RW-7, RW-8, RW-9, RW-10) located along the waterfront at Plant 1 (Figure 6). System performance data, collected prior to this investigation (TechSolve 2018a) indicate that the system has likely recovered LNAPL to the extent technically practicable, as required by the Consent Decree. As detailed, several lines of evidence support this hypothesis. However, these lines of evidence were likely insufficient to fully determine if LNAPL has

been recovered to the extent technically practicable, as it was possible that reduced LNAPL capture was the result of decreased system efficiencies, detailed in the Waterfront Probing Work Plan (TechSolve, 2019a) and previous Annual Site Reports (e.g. TechSolve, 2018a).

This investigation provides additional supporting evidence, as detailed in the previous sections, that recoverable LNAPL is no longer present throughout the investigation area, which represents the expected capture limits of the system. This investigation indicates that continued operation of the groundwater/LNAPL recovery system is unlikely to recover additional LNAPL, and that LNAPL has been recovered throughout the investigation area to the extent technically practicable. Continued operation of the groundwater/LNAPL recovery system is unlikely to reduce remaining IHSs detected during this investigation and addressing these remaining IHSs is likely unnecessary, as the cleanup criteria set forth in the Consent Decree appear to have been met.

The groundwater/LNAPL recovery system was designed (TechSolv and AG&M, 2000a) to provide hydraulic control and groundwater capture throughout the investigation area. While the system locally depresses groundwater around individual recovery wells, the system was not intended to, and is incapable of, depressing groundwater throughout the investigation area to depths where the highest soil impacts were observed in this investigation. The highest soil impacts detected in this investigation appeared to be residual, immobile, and were encountered on average 2.7 feet below static groundwater. As such, another remedial strategy would be required to address remaining soils with residual and immobile impacts that exceed the total TPH CUL. However, as the cleanup criteria appear to have been met in this area of the Site, continued compliance with the provisions of the existing Restrictive Covenant entered into under the Consent Decree is likely sufficient to meet Site cleanup objectives. The provisions of the Restrictive Covenant prevent the alteration, modification, or removal of existing structures (i.e. the warehouse, loading rack, and paved drive areas) in any manner that may result in the release or exposure to the environment of the remaining contaminated soil or create a new exposure pathway without prior written approval from Ecology.

5.5. Data Gaps

Boring B-43 soil results indicate that a shallow source area remains beneath the loading rack, in the area directly north of the investigation area. While additional investigative activities by the loading rack may be beneficial, the access to this area is limited. The loading rack is an active fueling rack in continual operation and access in the load rack area is limited due to substantial aboveground and belowground infrastructure. Ground surface at the load rack is covered with concrete and further covered with a roof and gutter system that both function as effective caps in this area. Measurable LNAPL in no longer detected in shallow groundwater adjacent to the load rack in Recovery Wells RW-2 and RW-4. Dissolved phase IHSs in nearby point of compliance monitoring walls (AMW-05 to the north of the loading rack and AMW-04 to the south of the loading rack) have been below CULs in every quarter since monitoring was initiated in 2000 (Table 1 and Appendix A).

Borings B-13 and B-01 collected at the southern property boundary indicate that limited soil impacts may extend to the south of the property. While additional investigative activities along the southwestern property boundary may be beneficial, the access to this area is also limited. Extensive belowground stormwater infrastructure both for the Site and a separate City of Seattle stormwater line exist directly south of these borings, limiting access for any additional investigation. Impacted soils in this area do not appear to be impacting groundwater, as all dissolved phase IHSs were below CULs and LNAPL was not detectable in temporary piezometer well MW-01-PI.

6. Conclusions

This investigation was conducted to evaluate the effectiveness of ongoing and completed cleanup actions in obtaining cleanup criteria and to evaluate the remaining extent of historical known IHSs and separate phase hydrocarbons (LNAPL) in subsurface soils and groundwater along the Plant 1 waterfront.

The results of this investigation along the Plant 1 waterfront show that:

- Soil impacts in the unsaturated zone are mainly below the soil total TPH CUL, which is attributed to historical remedial actions, such as SVE, that removed impacts in the unsaturated zone.
- Maximum soil impacts detected in 68 percent of the borings (28 out of 41) were below the TPH CUL.
- Soils in excess of the total TPH CUL in the saturated zone do not appear to extend beneath the warehouse foundation, historical island bulkhead, or the new seawall installed along the northern half of Plant, which act as barriers to groundwater/surface water exchange.
- Soils in excess of the total TPH CUL are in a narrow vertical band, typically starting at the groundwater interface and are, on average, less than 4 feet in thickness. The highest soil impacts detected were, on average, 2.7 feet below the elevation of static groundwater.
- The remaining LNAPL contained in the soils is primarily heavily weathered diesel, which is residual in nature, is not mobile, and is not partitioning into the dissolved phase in groundwater to any appreciable concentration.
- Groundwater monitoring of the 11 temporary piezometer wells showed no measurable or detectable free LNAPL on groundwater in the investigation area.
- Remaining soil impacts identified in the investigation area do not appear to be sufficient to create a sheen in the adjacent waterway. Data from this investigation and from historical sheen observations in the waterway could not identify the source of the historic sheens, but the source is assumed to be minimal and not recoverable by upland remedial actions, such as the existing groundwater/LNAPL recovery system.
- The performance criterion from the GWCMCP requiring the lack of measurable product (LNAPL) thickness in the product recovery system appears to have been met.
- The operating groundwater/LNAPL recovery system appears to have recovered LNAPL to the extent practicable.
- Further operation of the existing groundwater/LNAPL recovery system is unlikely to provide additional remedial benefit.

To conservatively test that LNAPL has been recovered to the extent practicable it may be appropriate to discontinue groundwater/LNAPL recovery system pumping and groundwater recovery on a trial basis, where the system is maintained in an operable state and operation can be resumed promptly if measurable and recoverable LNAPL is detected. Increased monitoring of shallow groundwater throughout the investigation area can be conducted following trial system shutdown to aid in evaluating if performance criteria (e.g. no measurable product in the recovery system) have been met. The trial discontinuation of remedial system operations has been previously conducted elsewhere at the Site (e.g. discontinuation of the Inland SVE system) with Ecology's approval.

Further evaluations of the obtainment of performance and confirmational criteria will be conducted as part of the planned hydraulic evaluation (TechSolve 2019c). The primary objectives of the hydraulic investigation are to:

- Evaluate changes to natural gradients and surface water/groundwater relationships due to the new seawall by evaluating current conditions against historical models.
- Evaluate the relationship between surface water and shallow and deep groundwater levels, potential exchanges, and the current natural gradient across the Site.
- Evaluate if the existing compliance monitoring well network, located along the waterfront and behind the new seawall, continues to adequately monitor the zone of groundwater and surface water exchange.
- Aide in future evaluations of whether modifications to ongoing remedial actions are necessary to attain cleanup standards and achieve Site closure.

The results of this soil and groundwater investigation indicate that performance criteria for ongoing remedial actions appear to have been met at the established points of compliance. The hydraulic evaluation will aid in determining if the existing points of compliance remain appropriate for the Site, or if changes in site hydrology due to the new seawall require new or modified points of compliance. Once the hydraulic investigation is completed, further evaluations can be completed to determine if cleanup actions are complete, if performance and compliance criteria have been met, if points of compliance should be modified, and if additional actions are warranted to meet specific compliance criteria.

7. References

Brost et al.; *Non-Aqueous Phase Liquid (NAPL) Mobility Limits in Soil*; API Soil & Groundwater Research Bulletin No. 9; June 2000.

Geraghty & Miller, Inc. (Geraghty & Miller). 1994. Remedial Investigation, ARCO Harbor Island Terminal 21T, Seattle, Washington. July 6, 1994.

______. 1996. Technical Memorandum Summarizing Supplemental Remedial Investigation Activities and Results, ARCO Harbor Island Terminal 21T, Seattle, Washington. June 28, 1996.

______. 1997. Final Focused Feasibility Study, ARCO Harbor Island Terminal 21T, Seattle, Washington. April 17, 1997.

Interstate Technology Regulatory Council (ITRC). 2018. LNAPL-3: LNAPL Site Management: LCSM Evolution, Decision Process, and Remedial Technologies. Appendix E–LNAPL Sheens Appendix. March 2018.

Massachusetts Department of Environmental Protection (MassDEP). 2016. LIGHT NONAQUEOUS PHASE LIQUIDS (LNAPL) AND THE MCP: GUIDANCE for SITE ASSESSMENT AND CLOSURE, Policy #WSC-16-450. February 19, 2016.

TechSolv Consulting Group, Inc. (TechSolv) (now TechSolve Environmental Inc. [TechSolve]). 1999a. Groundwater Compliance Monitoring and Contingency Program, ARCO Harbor Island Terminal 21T, Seattle, Washington. November 19, 1999.

______. 1999b. Sampling and Analysis Plan (SAP), ARCO Harbor Island Terminal 21T, Seattle, Washington. November 1999.

______. 2018a. 2017 Annual Site Report, BP West Coast Products Terminal, 1652 SW Lander Street, Seattle, Washington. April 2018.

______. 2018b. Water Quality Monitoring Summary Report, North Bulkhead Replacement Project, Site: BP Harbor Island Terminal, Cleanup Site ID:4426. June 27, 2018

______. 2019a. Plant 1 Waterfront Probing Work Plan, Site: BP Harbor Island Terminal, Cleanup Site ID:4426. Dated: December 2018. Submitted: January 3, 2019.

______. 2019b. Email from Scott Larsen to Jerome Cruz. RE: BP Harbor Island table of questions and comments on site activities vs. consent decree and CAP requirements. April 5, 2019 1:40PM.

______. 2019c. Plant 1 Hydraulic Evaluation Work Plan, Site: BP Harbor Island Terminal, Cleanup Site ID:4426. Dated: December 2018. Submitted: April 24, 2019.

TechSolv Consulting Group, Inc. and ARCADIS Geraghty & Miller. (TechSolv and AG&M). 2000s. Engineering Design Report, ARCO Harbor Island Terminal 21T, Seattle, Washington. September 2000.

Washington State Department of Ecology. (Ecology). 1999. Cleanup Action Plan (CAP), ARCO Terminal 21T, Seattle, Washington. November 19, 1999.

______. 2000. Consent Decree No. 00-2-05714-8SEA. March 24, 2000.

______. 2015. Letter from Maura S. O'Brien to Paul Supple. Subject: Ecology Comments for Proposed Bulkhead Replacement at BP West Coast Products Terminal Harbor Island, former ARCO Site at 1652 SW Lander Street, Seattle, WA Consent Decree No. 00-2-05714-8SEA and Cleanup Id No. 4426. September 3, 2015.

______. 2019. Email from Jerome Cruz to Scott Larsen. RE: Hydraulic Evaluation Work Plan for Former BP Harbor Island Terminal Site. May 28, 2019 4:53PM.

Weston, Roy F., Inc. (Weston). 1993 Remedial Investigation and Feasibility Study, Harbor Island. Prepared for U.S. Environmental Protection Agency, Contract No. 68-W9-0046, February 1993.

Tables

Table 1 – Groundwater Monitoring Analytical Resultsfor TPH and Benzene

Table 2 – Waterfront Groundwater Recovery Wells Petroleum Hydrocarbon History

Table 3 – Waterfront Probing Field and Laboratory Groundwater Sampling Results

Table 4 – Waterfront Probing Field and Laboratory Soil Sampling Results

Table 5 – Containment Boom Sheen Monitoring

	elle: I elfilei Bl II				
Well	Date	TPH-G WTPH-G (μg/L)	TPH-D WTPH-DX (μg/L)	TPH-O WTPH-DX (μg/L)	Benzene EPA 8021 & 8260 (µg/L)
Plant 1					
	12/21/2000	ND	1 310		14.0
	3/28/2001	50.3	2,600		14.0 60.6
	6/12/2001	105 11	2,000		470
	10/4/2001		944		470
	10/4/2001				152
	12/12/2001	ND 152	1700 J		1,260
AMVV-01	3/7/2002	153	1,410	ND	1,410
AMVV-01	6/12/2002	143 J	2,100		1,680
AMVV-01	9/19/2002	139 J	571 J	ND UJ	1,180
AMVV-01	12/17/2002	196	2,190	ND	/4.6
AMW-01	3/26/2003	101	2,100	ND	933
AMW-01	6/27/2003	ND	2,090	ND	1,260
AMW-01	9/18/2003	55	2,140	ND	48.5
AMW-01	12/22/2003	136	1750 J	ND	571
AMW-01	3/8/2004	ND UJ	ND	ND	961
AMW-01	6/16/2004	138	386	ND	1,540
AMW-01	9/28/2004	83	ND	ND	292
AMW-01	12/6/2004	103	ND	ND	411
AMW-01	3/10/2005	113	ND	ND	812
AMW-01	6/21/2005	129	ND	ND	1,130
AMW-01	9/27/2005	77	ND UJ	ND	181 J
AMW-01	12/13/2005	ND UJ	342	ND	132
AMW-01	3/21/2006	88	ND	ND	363
AMW-01	7/6/2006	ND UJ	ND	ND	912
AMW-01	9/18/2006	91.7	ND	ND	7.38
AMW-01	12/12/2006	1,650 J	ND UJ	ND UJ	539 J
AMW-01	3/21/2007	89.9	ND	ND	457
AMW-01	6/6/2007	61	ND	ND	486
AMW-01	9/12/2007	65	ND	ND	157
AMW-01	12/18/2007	ND	ND	ND	10.6 J
AMW-01	3/25/2008	ND	ND	ND	76
AMW-01	6/25/2008	64.9	ND	ND	370
AMW-01	9/17/2008	55.0	ND	ND	162
AMW-01	12/16/2008	ND	ND	ND	330
AMW-01	3/11/2009	ND	ND	ND	374
AMW-01	6/10/2009	ND	R	R	240 J
AMW-01	9/16/2009	ND	ND	ND	7.4
AMW-01	12/16/2009	ND	ND	ND	280
AMW-01	3/30/2010	ND	ND	ND	310
AMW-01	6/9/2010	ND	720	ND	280
AMW-01	9/14/2010	ND	ND	ND	69.7
AMW-01	12/14/2010	ND	ND	ND	282
AMW-01	3/22/2011	ND	ND	ND	247
AMW-01	6/22/2011	ND	300 J	ND	39.6
AMW-01	9/27/2011	ND	ND	ND	22.2
AMW-01	12/20/2011	ND	ND	ND UJ	151
AMW-01	3/20/2012	ND	ND	ND	178
Cleanup Lev	el	1,000	10,000	10,000	71
Method Repo	ortina Limit	50	250	750	0.5

					Ponzono
\M/oll	Dete				
weii	Dale	(ug/L)			
		(μg/L)	(µg/L)	(µg/Ľ)	(µg/L)
Plant 1, con	tinued				
AMW-01	6/21/2012	ND	ND	ND	77
AMW-01	9/10/2012	ND	ND	ND	38.7 J
AMW-01	12/19/2012	ND	ND	ND	61.2
AMW-01	3/19/2013	ND	ND	ND	110
AMW-01	6/25/2013	ND	ND	ND	12
AMW-01	9/10/2013	ND	ND	ND	17
AMW-01	12/10/2013	ND	ND	ND	17
AMW-01	3/11/2014	ND	990 J	ND	77
AMW-01	6/10/2014	ND UJ	1,100	ND	7.3
AMW-01	9/9/2014	ND	440 J	ND UJ	8.4
AMW-01	12/9/2014	ND	1,500	ND	20
AMW-01	3/10/2015	ND U	1,200 J	ND	68
AMW-01	6/9/2015	ND	450	ND	50
AMW-01	9/22/2015	ND	250	ND	12
AMW-01	12/15/2015	ND	430 J	ND UJ	38 J
AMW-01	3/8/2016	ND	320 J	ND UJ	24
AMW-01	6/8/2016	ND	1,200 J	ND UJ	4.1
AMW-01	9/8/2016	ND	1,300	ND	5.1
AMW-01	12/6/2016	ND U	800 J	ND	7.3
AMW-01	3/7/2017	230 J	1,300 J	1,100 J	1.0
AMW-01	6/7/2017	ND	ND UJ	ND	1.9
AMW-01	9/12/2017	ND	ND	ND	2.4
AMW-01	12/5/2017	ND	ND	ND	1.0
AMW-01	3/20/2018	240	ND	ND	ND
AMW-01	6/19/2018	ND UJ	480	710	ND
AMW-01	9/11/2018	ND	ND UJ	ND UJ	ND
AMW-01	12/11/2018	ND	610	ND	ND
AMW-01	3/12/2019	ND	ND	ND	ND
AMW-01	6/18/2019	ND	270	ND	ND
AMW-01	9/24/2019	ND	350 J	ND UJ	ND
AMW-01	12/17/2019	ND	ND	ND	ND
	12/21/2000	ND	803	ND	3 14
ΔΜ///_02	3/28/2001	Not acce	ssible due to earth	nauske damade to	warehouse
	6/13/2001	ND			3 88 11
ΔΜ\//_02	10/4/2001		1 200	ND	10.00 0
	12/12/2001	ND	1,200		5 47
ΔΜ\//_02	3/7/2002	Not accessible du	e to repair of earth	nauske damage tr	warehouse
AMW-02	6/12/2002		2 420	ND	1 49
	9/19/2002		2,420 105 I		1.40
ΔΜ\//_02	12/17/2002		1 800		4.08
AMW-02	3/26/2003	ND	2 200	ND	00 5.23
AMW-02	6/27/2003	ND	1 680		1 11
AMW-02	9/18/2003	ND	2 430	790	2 01
AMW-02	12/22/2003	ND	1.880.1	ND	ND
AMW-02	3/8/2004	ND	ND	ND	ND
Cleanup Lev	el	1,000	10,000	10,000	71
Method Reporting Limit		50	250	750	0.5

Table 1.Groundwater Monitoring Analytical Results for TPH and Benzene
Plant 1 Waterfront Confirmation Wells
Site: Former BP Harbor Island Terminal

Well	Date	TPH-G WTPH-G (µg/L)	TPH-D WTPH-DX (μg/L)	TPH-O WTPH-DX (μg/L)	Benzene EPA 8021 & 8260 (μg/L)
Plant 1, cont	tinued				
AMW-02	6/16/2004	ND	ND	ND	2.40
AMW-02	9/28/2004	ND	ND	ND	0.85
AMW-02	12/8/2004	ND	ND	ND	23.2
AMW-02	3/10/2005	ND	ND	ND	38.4
AMW-02	6/21/2005	ND	ND	ND	16.1
AMW-02	9/27/2005	ND	ND	ND	9.04
AMW-02	12/13/2005	ND	366	ND	7.26
AMW-02	3/21/2006	ND	ND	ND	2.16
AMW-02	7/6/2006	ND	ND	ND	41.1
AMW-02	9/18/2006	ND	ND	ND	3.18
AMW-02	12/12/2006	84.5 UJ	ND UJ	ND UJ	25.8 J
AMW-02	3/21/2007	ND	ND	ND	92.2
AMW-02	6/6/2007	ND	ND	ND	442
AMW-02	9/12/2007	ND	ND	ND	4.03 J
AMW-02	12/18/2007	ND	ND	ND	66.2
AMW-02	3/25/2008	75.9	ND	ND	343
AMW-02	6/25/2008	ND	ND	ND	125
AMW-02	9/17/2008	ND	ND	ND	30.7
AMW-02	12/16/2008	ND	ND	ND	189
AMW-02	3/11/2009	ND	ND	ND	421
AMW-02	6/10/2009	ND	R	R	100
AMW-02	9/16/2009	ND	ND	ND	12
AMW-02	12/16/2009	ND	ND	ND	110
AMVV-02	3/30/2010	ND	1,000	ND	210
AMVV-02	6/9/2010	ND	1,000	260	130
AMVV-02	9/14/2010	ND	ND	ND	22.6
AMVV-02	12/14/2010			ND	96.2 J
	3/22/2011				149
	0/22/2011				20.0
	9/27/2011				0.0
	2/20/2011				12.2
	3/20/2012 6/21/2012				31.0 92 5
	0/21/2012				02.3 10.7 I
	9/10/2012				12.7 J 12.7
	3/10/2012				0.3
	6/25/2013				9.0 13.0
$\Delta M M_{-02}$	0/20/2013				8.1
$\Delta M M_{-02}$	12/10/2013				5.7
	3/11/2014				19.0
$\Delta M M_{-02}$	6/10/2014		320		12.0
	0/10/2014 0/0/2014		270		29.0
ΔΜ\\/_02	12/9/2014		530		23.0 15.0
AMW-02	3/10/2015		370	ND	ND
AMW-02	6/9/2015	ND	ND	ND	3.1
AMW-02	9/22/2015	ND	ND	ND	2.0
		1.000	10.000	10,000	74
Mothed Dec	el Inting Limit	1,000 E0	10,000	750	<u> </u>
иченной керс	nung Linnt	50	200	150	0.5

Table 1.Groundwater Monitoring Analytical Results for TPH and Benzene
Plant 1 Waterfront Confirmation Wells
Site: Former BP Harbor Island Terminal

		TPH-G	TPH-D	TPH-O	Benzene
Well	Date	WTPH-G	WTPH-DX	WTPH-DX	EPA 8021 & 8260
		(μg/L)	(μg/L)	(μg/L)	(μg/L)
Plant 1, cont	tinued				
AMW-02	12/15/2015	ND	ND	ND	4.4
AMW-02	3/8/2016	ND	290	ND	1.9
AMW-02	6/8/2016	ND	840	ND	3.0
AMW-02	9/8/2016	ND	810	ND	15.0
AMW-02	12/6/2016	ND	510	ND	4.4
AMW-02	3/7/2017	ND	850	740	ND
AMW-02	6/6/2017	ND	ND	ND	2.7
AMW-02	9/12/2017	ND	ND	ND	1.1
AMW-02	12/5/2017	ND	ND	ND UJ	0.96
AMW-02	3/20/2018	53.0	ND	ND	2.30
AMW-02	6/19/2018	ND	ND	ND	0.92
AMW-02	9/11/2018	ND	300	ND	1.20
AMW-02	12/11/2018	ND	560	ND	1.50
AMW-02	3/12/2019	ND	ND	ND	ND
AMW-02	6/18/2019	ND	630	ND	2.40
AMW-02	9/24/2019	ND	260	ND	12.0
					87 , 67 J, 61 J, 64 J
AMW-02	12/17/2019	ND	ND	ND	(69.75 Average)**
AMW-03	12/21/2000	127	1,420	ND	ND
AMW-03	3/28/2001	Not accessible du	ie to earthquake da	amage to wareho	use.
AMW-03	6/13/2001	ND	745	ND	ND
AMW-03	10/4/2001	ND	1,210	ND	ND
AMW-03	12/12/2001	ND	1,080 J	ND UJ	ND
AMW-03	3/7/2002	Not accessible du	ie to earthquake da	amage to wareho	use.
AMW-03	6/12/2002	ND	1,070	ND	ND
AMW-03	9/19/2002	ND UJ	643 J	ND UJ	ND UJ
AMW-03	12/17/2002	ND	1,160	ND	ND
AMW-03	3/26/2003	ND	1,240	ND	ND
AMW-03	6/27/2003	ND	713	ND	ND
AMW-03	9/18/2003	ND	1,050	ND	ND
AMW-03	12/22/2003	ND	374 J	ND	ND
AMW-03	3/8/2004	ND	ND	ND	ND
AMW-03	6/16/2004	ND	ND	ND	1.02
AMW-03	9/28/2004	ND	ND	ND	ND
AMW-03	12/8/2004	ND	ND UJ	ND UJ	ND
AMW-03	3/10/2005	ND	ND	ND	1.56
AMW-03	6/21/2005	ND	ND	ND	0.99
AMW-03	9/27/2005	ND	ND UJ	ND	0.997
AMW-03	12/13/2005	ND	ND	ND	0.828
AMW-03	3/21/2006	ND	ND	ND	2.770
AMW-03	7/6/2006	ND	ND	ND	2.28
AMW-03	9/18/2006	ND	ND	ND	ND
AMW-03	12/12/2006	ND UJ	ND UJ	ND UJ	0.974 J
AMW-03	3/21/2007	ND	ND	ND	ND
AMW-03	6/6/2007	ND	ND	ND	ND
Cleanup Leve	el	1,000	10,000	10,000	71
Method Repo	ortina Limit	50	250	750	0.5

		TPH-G	TPH-D	TPH-O	Benzene
Well	Date	WTPH-G	WTPH-DX	WTPH-DX	EPA 8021 & 8260
		(µg/L)	(µg/L)	(µg/L)	(µg/L)
Plant 1, cont	inued				
AMW-03	9/12/2007	ND	ND	ND	ND UJ
AMW-03	12/18/2007	ND	ND	ND	ND
AMW-03	3/25/2008	ND	ND	ND	ND
AMW-03	6/25/2008	ND	ND	ND	ND
AMW-03	9/17/2008	ND	ND	ND	ND
AMW-03	12/16/2008	ND	ND	ND	ND
AMW-03	3/11/2009	ND	ND	ND	ND
AMW-03	6/10/2009	ND	R	R	ND
AMW-03	9/16/2009	ND	ND	ND	ND
AMW-03	12/16/2009	ND	ND	ND	ND
AMW-03	3/30/2010	ND	400	ND	ND
AMW-03	6/9/2010	ND	230	ND	ND
AMW-03	9/14/2010	ND	ND	ND	ND
AMW-03	12/14/2010	ND	ND	ND	ND
AMW-03	3/22/2011	ND	ND	ND	0.54
AMW-03	6/22/2011	ND	ND	ND	ND
AMW-03	9/27/2011	ND	ND	ND	ND
AMW-03	12/20/2011	ND	ND	ND	ND
AMW-03	3/20/2012	ND	ND	ND	0.52
AMW-03	6/21/2012	ND	ND	ND	ND
AMW-03	9/10/2012	ND	ND	ND	ND
AMW-03	12/19/2012	ND	ND	ND	ND
AMW-03	3/19/2013	ND	ND	ND	ND
AMW-03	6/25/2013	ND	ND	ND	ND
AMW-03	9/10/2013	ND	ND	ND	ND
AMW-03	12/10/2013	ND	ND	ND	ND
AMW-03	3/11/2014	ND	320 J	ND	ND
AMW-03	6/10/2014	ND UJ	430	ND	ND
AMW-03	9/9/2014	ND	360	ND	ND
AMW-03	12/9/2014	ND	570	ND	ND
AMW-03	3/10/2015	ND U	650	ND	ND
AMW-03	6/9/2015	ND	410	ND	ND
AMW-03	9/22/2015	ND	ND	ND	ND
AMW-03	12/15/2015	ND	ND	ND	ND
AMW-03	3/8/2016	ND	250	ND U	ND
AMW-03	6/8/2016	ND	840	ND	ND
AMW-03	9/7/2016	ND	330	ND	ND
AMW-03	12/6/2016	ND	820	ND	ND U
AMW-03	3/7/2017	ND	890	510	ND
AMW-03	6/6/2017	ND	ND	ND	ND
AMW-03	9/12/2017	ND	ND	ND	ND
AMW-03	12/5/2017	ND	ND	ND	ND
AMW-03	3/20/2018	ND	ND	390	ND
AMW-03	6/19/2018	ND	ND	ND	ND
AMW-03	9/11/2018	ND	ND	ND	ND
AMW-03	12/11/2018	ND	370	ND	ND
Cleanup Leve	el	1,000	10,000	10,000	71
Method Repo	ortina Limit	50	250	750	0.5

Well	Date	TPH-G WTPH-G (μg/L)	TPH-D WTPH-DX (µg/L)	TPH-O WTPH-DX (μg/L)	Benzene EPA 8021 & 8260 (μg/L)	
Plant 1 con	tinued					
	3/12/2010					
	6/18/2010					
	0/10/2019					
AIVIV-03	9/24/2019	ND	ND	ND	ND	
AIVIVV-03	12/17/2019	ND	ND	ND	ND	
AMW-04	12/21/2000	ND	1,570	ND	0.66	
AMW-04	3/28/2001	ND	1,660	ND	0.766	
AMW-04	6/13/2001	ND	987	ND	ND	
AMW-04	10/4/2001	ND	379	ND	ND	
AMW-04	12/12/2001	ND	930 J	ND UJ	ND	
AMW-04	3/7/2002	ND	519	ND	2.94	
AMW-04	6/12/2002	ND	1.200	ND	0.63	
AMW-04	9/19/2002	ND UJ	760 J	ND UJ	1.45 J	
AMW-04	12/17/2002	ND	1 070	ND	ND	
AMW-04	3/26/2003	ND	1 240	ND	0.84	
AMW-04	6/27/2003	ND	875	ND	ND	
AMW-04	9/18/2003	ND	1 660	ND	ND	
AMW-04	12/22/2003	ND	686.1	ND	1 73	
AMW-04	3/8/2004	ND		ND	ND	
	6/16/2004					
	0/27/2004					
	12/6/2004					
	2/10/2004					
	3/10/2005					
	9/27/2005					
	12/12/2005					
	12/13/2005					
	3/21/2000					
AIVIV-04	7/0/2000	ND UJ	ND	ND	ND UJ	
AIVIV-04	9/18/2006					
AMVV-04	12/12/2006	ND UJ	ND UJ	ND UJ	ND UJ	
AMVV-04	3/21/2007	ND	ND	ND	0.64	
AMW-04	6/6/2007	ND	ND	ND	ND	
AMW-04	9/12/2007	ND	ND	ND	ND UJ	
AMW-04	12/18/2007	ND	ND	ND	ND	
AMW-04	3/26/2008	ND	ND	ND	ND	
AMW-04	6/25/2008	ND	ND	ND	ND	
AMW-04	9/17/2008	ND	ND	ND	ND	
AMW-04	12/16/2008	ND	ND	ND	0.63	
AMW-04	3/11/2009	ND	ND	ND	ND	
AMW-04	6/10/2009	ND	R	R	ND	
AMW-04	9/16/2009	ND	ND	ND	ND	
AMW-04	12/16/2009	ND UJ	ND	ND	ND	
AMW-04	3/30/2010	ND	610	ND	0.57	
AMW-04	6/9/2010	ND	430	ND	ND	
AMW-04	9/14/2010	ND	ND	ND	ND	
AMW-04	12/14/2010	ND	ND	ND	ND	
Cleanup Lev	el	1,000	10,000	10,000	71	
Method Reporting Limit		50	250	750	0.5	

		TPH-G	TPH-D	TPH-O	Benzene
Well	Date	WTPH-G	WTPH-DX	WTPH-DX	EPA 8021 & 8260
		(μg/L)	(µg/L)	(μg/L)	(μg/L)
Plant 1. con	tinued				
AMW-04	3/22/2011	ND	ND	ND	ND
AMW-04	6/22/2011	ND	ND	ND	ND
AMW-04	9/27/2011	ND	ND	ND	ND
AMW-04	12/27/2011	ND	ND	ND	ND
AMW-04	3/20/2012	ND	ND	ND	ND
AMW-04	6/21/2012	ND	ND	ND	ND
AMW-04	9/10/2012	ND	ND	ND	ND
AMW-04	12/19/2012	ND	ND	ND	ND
AMW-04	3/19/2013	ND	ND	ND	ND
AMW-04	6/25/2013	ND	ND	ND	ND
AMW-04	9/10/2013	ND	ND	ND	ND
AMW-04	12/10/2013	ND	ND	ND	ND
AMW-04	3/11/2014	ND	780 J	ND	ND
AMW-04	6/10/2014	ND UJ	400	ND	ND
AMW-04	9/9/2014	ND	480	ND	ND
AMW-04	12/9/2014	ND	630	ND	ND
AMW-04	3/10/2015	ND U	590	ND	ND
AMW-04	6/9/2015	ND	420	ND	ND
AMW-04	9/22/2015	ND	ND	ND	ND
AMW-04	12/15/2015	ND	ND	ND	ND
AMW-04	3/8/2016	ND	390	ND U	ND
AMW-04	6/8/2016	ND	860	ND	ND
AMW-04	9/8/2016	ND	800	ND	ND
AMW-04	12/6/2016	ND	830	ND	ND U
AMW-04	3/7/2017	ND	830	640	ND
AMW-04	6/6/2017	ND	ND	ND	ND
AMW-04	9/12/2017	ND	ND	ND	ND
AMW-04	12/5/2017	ND	ND	ND	ND
AMW-04	3/20/2018	74 J	ND	ND	ND
AMW-04	6/19/2018	ND	300	ND	ND
AMW-04	9/11/2018	ND	ND	ND	ND
AMW-04	12/11/2018	ND	500	ND	ND
AMW-04	3/12/2019	59	ND	ND	ND
AMW-04	6/18/2019	ND	ND	ND	ND
AMW-04	9/24/2019	ND	ND	ND	ND
AMW-04	12/17/2019	ND	ND	ND	ND
AM\\/_05	12/21/2000	ND	1 450	ND	ND
AMW-05	3/28/2001	ND	1,400		ND
AMW-05	6/13/2001		440		
AMW-05	10/4/2001	71411	318	ND	ND
AMW-05	12/12/2001	ND	940.1		ND
AMW-05	3/7/2002	ND	1 100	ND	2 12
AMW-05	6/12/2002	78	1 180	ND	0 701
AMW-05	9/19/2002	NDUJ	760 J	NDUJ	1.45 J
AMW-05	12/17/2002	ND	1,820	ND	ND
			·,*		
Cleanup Lev	el	1,000	10,000	10,000	71
Method Reporting Limit		50	250	750	0.5

Table 1.Groundwater Monitoring Analytical Results for TPH and Benzene
Plant 1 Waterfront Confirmation Wells
Site: Former BP Harbor Island Terminal
Mall	Data	TPH-G			Benzene
Weil	Date	(ug/L)			(ug/L)
		(µg/Ľ)	(µg/Ľ)	(μg/Ε)	(µg/Ľ)
Plant 1, cont	tinued				
AMW-05	3/26/2003	ND	1,900	ND	0.577
AMW-05	3/27/2003	ND	381 J	ND UJ	ND
AMW-05	9/19/2003	ND	2,150	ND	ND
AMW-05	12/22/2003	ND	1,420 J	ND	0.833
AMW-05	3/8/2004	ND	ND	ND	ND
AMW-05	6/16/2004	ND	ND	ND	ND
AMW-05	9/27/2004	ND	ND	ND	ND
AMW-05	12/6/2004	ND	ND	ND	ND
AMW-05	3/10/2005	ND	ND	ND	ND
AMW-05	6/21/2005	ND	ND	ND	ND
AMW-05	9/27/2005	ND	ND UJ	ND	ND
AMW-05	12/13/2005	ND	ND	ND	0.727
AMW-05	3/21/2006	ND	ND	ND	0.692
AMW-05	7/6/2006	ND	ND	ND	ND
AMW-05	9/18/2006	ND	ND	ND	ND
AMW-05	12/12/2006	ND UJ	ND UJ	ND UJ	0.565 J
AMW-05	3/21/2007	ND	ND	ND	1.11
AMW-05	6/6/2007	ND	ND	ND	ND
AMW-05	9/12/2007	ND	ND	ND	ND UJ
AMW-05	12/18/2007	ND	ND	ND	ND
AMW-05	3/26/2008	ND	ND	ND	ND
AMW-05	6/25/2008	ND	ND	ND UJ	ND
AMW-05	9/17/2008	ND	ND	ND UJ	
AMW-05	12/16/2008	ND	ND	ND	0.768
AMW-05	3/11/2009	ND	ND	ND	0.885
AMVV-05	6/10/2009	ND 54	R	R	
AMVV-05	9/16/2009	54	ND	ND	ND
	12/16/2009		ND 800		
	3/30/2010		890		1.3 ND
	0/9/2010		040 ND		
	9/14/2010 12/11/2010				
	2/22/2011				
AMW-05	6/22/2011				
AMW-05	0/22/2011				
AMW-05	12/20/2011				
ΔΜ\Λ/_05	3/20/2012	ND	ND	ND	
AMW-05	6/21/2012	ND	ND	ND	ND
AMW-05	9/10/2012	ND	ND	ND	ND
AMW-05	12/19/2012	ND	ND	ND	ND
AMW-05	3/19/2013	ND	ND	ND	ND
AMW-05	6/25/2013	ND	ND	ND	ND
AMW-05	9/10/2013	ND	ND	ND	ND
AMW-05	12/10/2013	ND	ND	ND	ND
AMW-05	3/11/2014	ND	ND	ND	ND
AMW-05	6/10/2014	ND UJ	560	ND	ND
Cleanup Lev	el	1,000	10,000	10,000	71
Method Repo	orting Limit	50	250	750	0.5

Table 1.Groundwater Monitoring Analytical Results for TPH and Benzene
Plant 1 Waterfront Confirmation Wells
Site: Former BP Harbor Island Terminal

	Site: Former BP H	arbor Island Tern	ninal		
Well	Date	TPH-G WTPH-G (μg/L)	TPH-D WTPH-DX (μg/L)	TPH-O WTPH-DX (μg/L)	Benzene EPA 8021 & 8260 (μg/L)
Plant 1, cont	tinued				
AMW-05	9/9/2014	ND	300	ND	ND
AMW-05	12/9/2014	ND	460	ND	ND
AMW-05	3/10/2015	ND	480	ND	ND
AMW-05	6/9/2015	ND	300	ND	ND
AMW-05	9/22/2015	ND	ND	ND	ND
AMW-05	12/15/2015	ND	ND	ND	ND
AMW-05	3/8/2016	ND	ND	ND U	ND
AMW-05	6/8/2016	ND	850	ND	ND
AMW-05	9/8/2016	ND	1,300	ND	2.0
AMW-05	12/6/2016	ND	420	ND	ND U
AMW-05	3/7/2017	ND	910	1,000	ND
AMW-05	6/6/2017	ND	ND	ND	ND
AMW-05	9/12/2017	ND	ND	ND	ND
AMW-05	12/5/2017	ND	ND	ND	ND
AMW-05	3/20/2018	ND	ND	340	ND
AMW-05	6/19/2018	ND	ND	ND	ND
AMW-05	9/11/2018	ND	ND	ND	ND
AMW-05	12/11/2018	ND	320	ND	ND
AMW-05	3/12/2019	51	ND	ND	ND
AMW-05	6/18/2019	ND	ND	ND	ND
AMW-05	9/24/2019	ND	ND	ND	ND
AMW-05	12/17/2019	ND	ND	ND	ND
Cleanup Leve	el	1,000	10,000	10,000	71
Method Reporting Limit		50	250	750	0.5

Table 1. Groundwater Monitoring Analytical Results for TPH and Benzene Plant 1 Waterfront Confirmation Wells

Note: Values in **bold** exceed the cleanup level.

J	Estimated value.
µg/L	Micrograms per liter.
NA	Not analyzed.
ND	Constituent not detected above reporting limit.
NS	Not sampled.
ТРН	Total petroleum hydrocarbons.
TPH-D	Total petroleum hydrocarbons as diesel.
TPH-G	Total petroleum hydrocarbons as gasoline.
TPH-O	Total petroleum hydrocarbons as oil.
U	Undetected.
WTPH-DX	Washington State Method for Analysis of Diesel and Oil in Water - Extended.
WTPH-G	Washington State Method for Analysis of Gasoline in Water.

EPA 8021 or EPA 9260 - EPA Methods for Analysis of Benzene in Water.

** AMW-02 benzene result from 12/17/2019 of 69.75 is the average of 4 analyses performed from the sample.

Well ID	Date	Gasoline mg/l	Diesel mg/l	Oil mg/l	Benzene ug/l	Toluene ug/l	Ethylbenzene ug/l	Xylenes (total) ug/l
RW-10	Nov-03	0.625			1.2	0.892	2.42	3.07
RW-10	Aug-04	0.661	36.2	3.46	0.5	0.5	0.653	1.99
RW-10	Feb-05	0.473	1.21	0.75	0.5	0.5	0.5	1.41
RW-10	Nov-05	0.420	13.3	1.63	0.5	0.5	0.5	1
RW-10	Mar-06	0.066	4.14	0.75	0.5	0.5	0.5	1
RW-10 RW-10	Nov-06 May-07	0.930	0.40 0.255	0.5	0.5	0.5	0.5	1
RW-10	Nov-07	0.246	4 65	0.841	0.5	0.5	0.5	1
RW-10	Apr-08	0.235	1.91	0.515	0.5	0.5	0.5	1
RW-10	Nov-08	0.347	8.21	0.946	0.5	0.5	0.5	1
RW-10	Apr-09	0.448	5.95	0.804	0.5	0.5	0.5	1.36
RW-10	Nov-09	0.320	5.2	0.78	0.5	1	1	2
RW-10	Apr-10	0.460	2.3	0.49	0.5	1	1	2
RW-10	Nov-10	0.251	2.4	0.65	0.5	1	1	3
RW-10	Apr-11 Nov-11	0.6	1.5	0.08	0.5	1	1	3
RW-10	Apr-12	0.366	0.22	0.33	0.5	1	1	3
RW-10	Nov-12	0.1	0.11	0.11	0.5	0.5	0.5	1.5
RW-10	Apr-13	0.2	0.36	0.49	0.5	0.5	0.5	0.5
RW-10	Nov-13	0.13	0.25	0.25	0.5	0.5	0.5	1
RW-10	Apr-14	0.16	1.6	0.73	0.14	0.16	0.13	0.13
RW-10	Nov-14	0.11	0.78	0.36	1.0	1.0	1.0	3.0
KVV-10	Apr-15	0.091	0.97	0.8	2.0	2.0	3.0	3.0
RW-10	NOV-15 Apr-16	0.67	1.5 1.0	0.28	4.3	2.0	3.0	0.73
RW-10	Nov-16	0.069	0.77	0.32	0.2	0.2	0.2	0.5
RW-10	Apr-17	0.5	0.11	0.25	2.0	2.0	3.0	3.0
RW-10	Nov-17	0.069	0.36	0.25	0.2	0.2	0.2	0.5
RW-10	Apr-18	0.12	0.33	0.26	0.2	0.2	0.2	0.5
RW-10	Nov-18	0.12	3.4	2.8	0.2	0.2	0.2	0.5
RW-10	Apr-19	0.073	1.6	0.96	0.2	0.2	0.2	0.5
RW-10	Average	0.053	1.7	0.84	0.5	2.0	2.0	4.0
T(W-10	, it of age	0.0	0.0	0.0	-	0.0	1.0	1.7
RW-9	Nov-03	13.1	04.0	2.10	5	43.2	146	1180
RW-9 RW-9	Aug-04 Eeb-05	1.24	94.9 22.1	2.19 <15	0.5	0.5	1.23	1.04
RW-9	Nov-05	0.568	4 31	0 708	0.5	0.5	0.968	1 45
RW-9	Mar-06	0.166	1.68	0.75	0.5	0.5	0.5	1
RW-9	Nov-06	0.359	5.98	1.17	0.5	0.5	0.647	1.09
RW-9	May-07	0.402	2.08	0.5	5.43	0.5	1.4	1.49
RW-9	Nov-07	0.184	70.1	11.6	0.5	0.5	0.5	1
RW-9	Apr-08	0.170	18.2	2.94	3.21	0.5	0.5	1
RVV-9	Nov-08	0.130	49.5	8.21	0.5	0.5	0.5	1
RW-9	Apr-09	0.260	40.1	6.8	0.5	0.5	0.5	2
RW-9	Apr-10	60	110	0.0 24	0.5	1	1	2
RW-9	Nov-10	0.207	2.0	0.53	0.5	1	1	3
RW-9	Apr-11	1.12	276	45.9	0.5	1	1	3
RW-9	Nov-11	0.289	2.3	0.39	0.5	1	1	3
RW-9	Apr-12	0.113	33.2	5.3	0.72	1	1	3
RW-9	Nov-12	0.1	8.2	8.4	0.5	0.5	0.5	1.5
RVV-9	Apr-13 Nov 12	U.1 0.062	44.0	8.5 2.6	0.5	0.5	0.5 0.5	U.5
RW-9	Apr-1/	0.00Z	56.0	2.0 16	0.5	0.5	0.5	1 0 1 2
RW-9	Nov-14	0.14	7.1	2.7	1.0	1.0	1.0	3.0
RW-9	Apr-15	0.18	14.0	4.9	2.0	2.0	3.0	3.0
RW-9	Nov-15	0.32	7.6	3.0	2.0	2.0	3.0	3.0
RW-9	Apr-16	1.5	180.0	38.0	2.0	2.0	3.0	3.0
RW-9	Nov-16	0.17	12.0	3.8	0.2	0.2	0.2	0.5
RW-9	Apr-17	0.5	64.0	17.0	2.0	2.0	3.0	3.0
DN/ 0	NOV-17	0.14	14.0	4.4 2.2	0.2	0.2	0.2	U.5 0 F
RW-9	Api-10 Nov-19	0.000 0.000	17.0	3.3 79	0.2	0.2	0.2 0.2	0.5
RW-9	Apr-19	0.05	87	2.8	0.2	0.2	0.2	0.5
RW-9	Nov-19	0.054	7.5	2.4	0.2	2.0	2.0	4.0
RW-9	Average	0.9	39.8 8.1 1.1 2.1 5.6		38.6			
Groundwater	Cleanup Level	1.0	10.0	10.0	71			
Reporting L	imits/Units	0.05 mg/l	0.25 mg/l	.750 mg/l	0.5 ug/l	Varies	Varies	Varies

Wall ID	Data	Gasoline	Diesel	Oil mg/l	Benzene	Toluene ug/l	Ethylbenzene	Xylenes (total) ug/l
	Nov 02	0.267	iiig/i	ing/i	0.5	0.5	0.797	0.00
	NUV-03	0.307	10.9	2.10	0.5	0.5	0.787	2.23
	Aug-04 Ecb. 05	0.101	19.0	2.19	0.5	0.5	0.55	2.13
	Feb-05	0.210	2.30	0.75	0.5	0.5	0.564	3.04
	Nov-05	0.099	0.575	0.721	0.5	0.5	0.5	1
	Nat-06	0.050	1.44	0.75	0.5	0.5	0.5	1
	NOV-00	0.050	3.30	0.762	0.5	0.5	0.5	1
	Nay-07	0.000	0.273	0.5	0.5	0.5	0.5	1
	NOV-07	0.065	0.29	0.543	0.5	0.5	0.5	1
	Api-00	0.007	2.279	0.029	0.5	0.5	0.5	1
	NOV-00	0.000	0.05	0.492	0.5	0.5	0.5	1
DW/ 8	Api-09	0.091	0.200	0.470	0.5	0.5	0.5	1
	Apr 10	0.140	1.5	0.47	0.5	1	1	2
	Api-10 Nov 10	0.100	1.1	0.49	0.5	1	1	2
	Apr 11	0.105	1.0	0.39	0.5	1	1	3
	Api-11 Nov 11	0.0995	2.0	0.39	0.5	1	1	3
	Apr 12	0.185	1.7	0.39	0.5	1	1	3
	Apr-12	0.05	1.3	0.39	0.5	0.5	0.5	15
	Apr 12	0.165	4.0	0.52	0.5	0.5	0.5	1.5
	Api-13 Nov 12	0.002	2.7	0.52	0.5	0.5	0.5	0.5
	NOV-13	0.1	0.02	0.25	0.5	0.5	0.5	0.52
	Api-14 Nov 14	0.13	3.40	0.91	1.0	1.0	1.0	2.0
	NOV-14	0.14	5.0	3.2	7.0	1.0	1.0	3.0
	Apr-15	0.13	5.Z	2.0	2.0	2.0	3.0	3.0
	Apr 16	0.39	19.0	1.5	0.91	2.0	3.0	3.0
	Api-10 Nov-16	0.20	7.6	0.64	0.64	2.0	0.2	0.5
	Apr 17	0.25	2.0	0.07	2.00	2.0	3.0	3.0
	Api-17 Nov 17	0.5	2.0	1.30	2.00	2.0	3.0	3.0
	Apr 18	0.12	J.0 4.0	1.30	0.20	0.2	0.2	0.5
	Api-10 Nov 18	0.11	4.0	1.20	0.20	0.2	0.2	2.0
	$\Delta pr_{-}10$	0.10	0.95	0.26	0.20	0.2	0.44	2.0
DW/ 8	Nov 19	0.091	0.35	0.20	0.20	2.0	2.00	4.0
	Average	0.001	3.7	12	0.50	0.8	0.9	1.8
		0.1.10	0.1	1.2	0.0	0.0	0.0	0.07
RW-7	Nov-03	0.148		4.0	0.5	0.5	0.518	2.87
RW-7	Aug-04	0.050	7.6	1.2	0.5	0.5	0.5	1.09
	Feb-05	0.050	1.21	0.70	0.5	0.5	0.5	1
	NOV-05	0.050	0.35	0.720	0.5	0.5	0.5	1
	Nov 06	0.050	0.25	0.75	0.5	0.5	0.5	1
	Mov-00	0.003	3.10	0.54	0.5	0.5	0.5	
		0 4 1 4	0.40		0.5	0.5	05	1
	Nov 07	0.414	0.49	0.515	0.5	0.5	0.5	1
	Nov-07	0.414 0.187 0.062	0.49 0.25 0.25	0.515	0.5 0.5	0.5 0.5	0.5 0.5	1
	Nov-07 Apr-08	0.414 0.187 0.063 0.071	0.49 0.25 0.25 0.236	0.575 0.5 0.5	0.5 0.5 0.5	0.5 0.5 0.5	0.5 0.5 0.5	1 1 1
RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr 09	0.414 0.187 0.063 0.071 0.123	0.49 0.25 0.25 0.236 0.238	0.5 0.5 0.5 0.472	0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5	1 1 1 1
RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09	0.414 0.187 0.063 0.071 0.123 0.075	0.49 0.25 0.25 0.236 0.238 0.69	0.515 0.5 0.472 0.476 0.47	0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 0.5	1 1 1 1 1 2
RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr 10	0.414 0.187 0.063 0.071 0.123 0.075 0.140	0.49 0.25 0.25 0.236 0.238 0.69 0.85	0.515 0.5 0.472 0.476 0.47 0.49	0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1	0.5 0.5 0.5 0.5 0.5 1	1 1 1 1 2 2
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46	0.575 0.5 0.472 0.476 0.47 0.49 0.4	0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1	0.5 0.5 0.5 0.5 0.5 1 1	1 1 1 1 2 2 3
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1	0.575 0.5 0.472 0.476 0.47 0.49 0.4 0.4	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1	0.5 0.5 0.5 0.5 1 1 1	1 1 1 1 2 2 3 3
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.4	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1	0.5 0.5 0.5 0.5 1 1 1 1	1 1 1 2 2 3 3 3
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.41 0.42	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1	0.5 0.5 0.5 0.5 1 1 1 1 1	1 1 1 2 2 3 3 3 3 3
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.1	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.4 0.41 0.4 0.42 0.37	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 5	0.5 0.5 0.5 1 1 1 1 1 5	1 1 1 1 2 2 3 3 3 3 3 5
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.1 0.081	0.49 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.4 0.41 0.4 0.42 0.37 0.5	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 1 1 1 1 1 0.5 0.5	0.5 0.5 0.5 1 1 1 1 0.5 0.5	1 1 1 2 2 3 3 3 3 3 1.5 0 5
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.1 0.081 0.05	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.4 0.4 0.41 0.4 0.42 0.37 0.5 0.24	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 1 1 1 1 1 0.5 0.5	0.5 0.5 0.5 1 1 1 1 0.5 0.5 0.5	1 1 1 2 2 3 3 3 1.5 0.5 1
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.1 0.081 0.05 0.07	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.4 0.41 0.42 0.37 0.5 0.24 0.6	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 1 1 1 1 1 0.5 0.5 0.5 0.16	0.5 0.5 0.5 1 1 1 1 1 0.5 0.5 0.5 0.17	1 1 1 2 2 3 3 3 1.5 0.5 1 0.23
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.05 0.05 0.05 0.05 0.07 0.064	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 0.5 0.5 0.16 1.0	0.5 0.5 0.5 0.5 1 1 1 1 1 0.5 0.5 0.17 1.0	1 1 1 2 2 3 3 3 3 1.5 0.5 1 0.23 30
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Apr-15	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.05 0.05 0.05 0.07 0.064 0.073	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 20	0.5 0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.17 1.0 3.0	1 1 1 2 2 3 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.05 0.05 0.07 0.064 0.073 0.11	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2 0.41	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.88	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 2.0 2.0	0.5 0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.5 0.17 1.0 3.0 3.0	1 1 1 2 2 3 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Nov-15 Nov-15 Apr-16	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.05 0.05 0.07 0.064 0.073 0.11 0.26	0.49 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2 0.41 7.9	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.41 0.4 0.41 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.88 2.5	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 2.0 2.0 2.0	0.5 0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.5 0.17 1.0 3.0 3.0 3.0	1 1 1 2 2 3 3 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 3.0
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15 Apr-16 Nov-16	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.07 0.081 0.05 0.07 0.064 0.073 0.11 0.26 0.11	0.49 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.63 0.45 2.4 0.92 5.2 0.41 7.9 0.89	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.88 2.5 0.25	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 2.0 2.0 0.2	0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.5 0.17 1.0 3.0 3.0 0.2	1 1 1 2 2 3 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 0.5
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15 Apr-16 Nov-16 Apr-17	$\begin{array}{c} 0.414\\ 0.187\\ 0.063\\ 0.071\\ 0.123\\ 0.075\\ 0.140\\ 0.11\\ 0.207\\ 0.05\\ 0.07\\ 0.061\\ 0.081\\ 0.05\\ 0.07\\ 0.064\\ 0.073\\ 0.11\\ 0.26\\ 0.11\\ 0.5\\ \end{array}$	0.49 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.63 0.45 2.4 0.92 5.2 0.41 7.9 0.89 0.75	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.88 2.5 0.25 0.27	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 2.0 2.0 2.0 2.0 2.0	0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.5 0.17 1.0 3.0 3.0 0.2 3.0	1 1 1 2 2 3 3 3 3 3 3 5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15 Apr-16 Nov-16 Apr-17 Nov-17	$\begin{array}{c} 0.414\\ 0.187\\ 0.063\\ 0.071\\ 0.123\\ 0.075\\ 0.140\\ 0.11\\ 0.207\\ 0.05\\ 0.05\\ 0.01\\ 0.081\\ 0.05\\ 0.07\\ 0.064\\ 0.073\\ 0.11\\ 0.26\\ 0.11\\ 0.26\\ 0.11\\ 0.5\\ 0.05\\ 0.05\\ \end{array}$	0.49 0.25 0.236 0.236 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.45 2.4 0.92 5.2 0.41 7.9 0.89 0.75 0.21	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.88 2.5 0.25 0.25 0.25 0.25	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 2.0 2.0 2.0 0.2 0.2	0.5 0.5 0.5 1 1 1 1 1 1 0.5 0.5 0.17 1.0 3.0 3.0 3.0 3.0 2.2 0.2	1 1 1 2 2 3 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 0.5 3.0 0.5 3.0 0.5
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15 Apr-16 Nov-16 Apr-17 Nov-17 Apr-18	$\begin{array}{c} 0.414\\ 0.187\\ 0.063\\ 0.071\\ 0.123\\ 0.075\\ 0.140\\ 0.11\\ 0.207\\ 0.05\\ 0.05\\ 0.05\\ 0.07\\ 0.081\\ 0.05\\ 0.07\\ 0.064\\ 0.073\\ 0.11\\ 0.26\\ 0.11\\ 0.26\\ 0.11\\ 0.5\\ 0.05\\ 0.061\\ \end{array}$	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2 0.41 7.9 0.89 0.75 0.21 1.2	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.25 1.6 0.88 2.5 0.25 0.25 0.27 0.26 0.26	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 2.0 2.0 2.0 0.2 0.2 0.2 0.2	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 0.5 0.5 0.5 0.17 1.0 3.0 3.0 3.0 3.0 0.2 0.2	1 1 1 2 2 3 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 0.5 3.0 0.5 0.5
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15 Apr-16 Nov-16 Apr-17 Nov-17 Apr-18 Nov-18	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.07 0.081 0.05 0.07 0.064 0.073 0.11 0.26 0.11 0.26 0.11 0.5 0.05 0.061 0.065	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2 0.41 7.9 0.89 0.75 0.21 1.2 0.48	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.88 2.5 0.25 0.27 0.26 0.26 0.26	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 1 0.5 0.5 0.5 0.5 0.16 1.0 2.0 2.0 2.0 0.2 2.0 0.2 0.2 0.2 0.2 0	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 1 0.5 0.5 0.5 0.5 0.17 1.0 3.0 3.0 3.0 3.0 0.2 0.2 0.2 0.2	1 1 1 2 2 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 0.5 3.0 0.5 0.5 0.5
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15 Apr-16 Nov-16 Apr-17 Nov-17 Apr-18 Nov-18 Apr-19	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.01 0.081 0.05 0.07 0.064 0.073 0.11 0.26 0.11 0.26 0.11 0.26 0.11 0.26 0.11 0.5 0.05 0.05 0.07 0.064 0.073 0.05 0.05 0.05 0.05 0.07 0.064 0.073 0.05 0.05 0.05 0.05 0.05 0.05 0.07 0.064 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.07 0.064 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.07 0.064 0.05 0.0	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2 0.41 7.9 0.89 0.75 0.21 1.2 0.48 0.25	0.515 0.5 0.47 0.476 0.47 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.88 2.5 0.25 0.27 0.26 0.26 0.26 0.26	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 2.0 2.0 2.0 0.2 0.2 0.2 0.2 0.2 0.2 0	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 1 0.5 0.5 0.5 0.17 1.0 3.0 3.0 3.0 0.2 3.0 0.2 0.2 0.2 0.2	1 1 1 1 2 2 3 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 0.5 3.0 0.5 0.5 0.5 0.5
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15 Apr-16 Nov-15 Apr-16 Nov-16 Apr-17 Nov-17 Apr-18 Nov-18 Apr-19 Nov-19	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.01 0.081 0.05 0.07 0.064 0.073 0.11 0.26 0.11 0.26 0.11 0.26 0.11 0.26 0.11 0.5 0.05 0.	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2 0.41 7.9 0.89 0.75 0.21 1.2 0.48 0.25 0.25	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.4 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.25 1.6 0.25 0.25 0.27 0.26 0.26 0.26 0.25	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 0.5 0.5 0.5 0.16 1.0 2.0 2.0 2.0 0.2 2.0 0.2 0.2 0.2 0.2 0	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 5 0.5 0.5 0.17 1.0 3.0 3.0 3.0 0.2 3.0 0.2 0.2 0.2 0.2 2.0	1 1 1 1 2 2 3 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Apr-15 Nov-15 Apr-16 Nov-15 Apr-16 Nov-16 Apr-17 Nov-17 Apr-18 Nov-18 Apr-19 Nov-19 Nov-19 Average	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.05 0.07 0.064 0.073 0.11 0.26 0.11 0.26 0.11 0.26 0.11 0.5 0.05 0.061 0.065 0.07 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.05 0.05 0.07 0.05 0.05 0.07 0.05 0	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2 0.41 7.9 0.89 0.75 0.21 1.2 0.48 0.25 0.25 1.3	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.41 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.25 1.6 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.25 0.27 0.26 0.25 0.26	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 0.5 0.5 0.16 1.0 2.0 2.0 0.2 0.2 0.2 0.2 0.2 0	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 1 1 0.5 0.5 0.17 1.0 3.0 3.0 3.0 0.2 0.2 0.2 0.2 0.2 0.9 0.9	1 1 1 1 2 2 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.
RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7 RW-7	Nov-07 Apr-08 Nov-08 Apr-09 Nov-09 Apr-10 Nov-10 Apr-11 Nov-11 Apr-12 Nov-12 Apr-13 Nov-13 Apr-14 Nov-14 Apr-15 Nov-15 Apr-16 Nov-15 Apr-16 Nov-16 Apr-17 Nov-17 Apr-18 Nov-18 Apr-19 Nov-19 Average r Cleanup Level	0.414 0.187 0.063 0.071 0.123 0.075 0.140 0.11 0.207 0.05 0.05 0.05 0.07 0.064 0.073 0.11 0.26 0.11 0.26 0.11 0.26 0.11 0.5 0.05 0.061 0.065 0.07 0.05 0.07 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.07 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.07 0.05 0.07 0.05 0.07 0.05 0.07 0.05 0.07 0.064 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.07 0.05 0.07 0.064 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.07 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.07 0.05 0.07 0.05 0.05 0.07 0.05 0.05 0.05 0.05 0.07 0.05	0.49 0.25 0.25 0.236 0.238 0.69 0.85 0.46 1.1 0.13 0.21 0.32 0.63 0.45 2.4 0.92 5.2 0.41 7.9 0.89 0.75 0.21 1.2 0.48 0.25 0.25 1.3 10.0	0.515 0.5 0.472 0.476 0.47 0.49 0.4 0.41 0.41 0.42 0.37 0.5 0.24 0.6 0.25 1.6 0.25 1.6 0.25 0.25 0.25 0.27 0.26 0.26 0.26 0.75 0.6 10.0	0.5 0.5 0.5 0.5 0.5 0.5 0.5 0.5	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 0.5 0.5 0.16 1.0 2.0 2.0 0.2 0.2 0.2 0.2 0.2 0	0.5 0.5 0.5 0.5 1 1 1 1 1 1 1 1 1 1 0.5 0.5 0.5 0.17 1.0 3.0 3.0 3.0 0.2 0.2 0.2 0.2 0.2 0.9 0.9	1 1 1 1 2 2 3 3 3 1.5 0.5 1 0.23 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.

Wall ID	Dato	Gasoline mg/l	Diesel ma/l	Oil ma/l	Benzene ug/l	Toluene ug/l	Ethylbenzene	Xylenes (total) ug/l
	Nov 02	0.959	0.72	1.24	1.02	0.759	2.71	2 20
	NUV-03	0.000	0.73 21 C	2.09	1.03	0.750	2.71	3.39
	Feb 05	1.00	18 0	2.00	10.5	4.66	2.1	4.10
RW-1	Nov-05	0.547	2 19	0.75	0.5	4.00	4.00	1 67
RW-1	Mar-06	0.047	4 78	0.802	0.5	0.5	0.5	1
RW-1	Nov-06	0.173	3 28	0.002	0.5	0.5	0.5	1
RW-1	May-07	0.081	0.972	0.526	0.5	0.5	0.5	1
RW-1	Nov-07	0.056	0.596	0.505	0.5	0.5	0.5	1
RW-1	Apr-08	0.068	0.25	0.5	0.5	0.5	0.5	1
RW-1	Nov-08	0.050	0.274	0.472	0.5	0.5	0.5	1
RW-1	Apr-09	0.074	0.332	0.481	0.5	0.5	0.5	1
RW-1	Nov-09	0.073	0.44	0.47	0.5	1	1	2
RW-1	Apr-10	0.071	0.31	0.49	0.5	1	1	2
RW-1	Nov-10	0.143	0.32	0.39	0.5	1	1	3
RW-1	Apr-11	0.0991	0.95	0.39	0.5	1	1	3
RW-1	Nov-11	0.14	6.9	1.6	0.5	1	1	3
RW-1	Apr-12	0.131	0.86	0.4	0.53	1	1	3
RW-1	Nov-12	0.1	0.23	0.35	0.5	0.5	0.5	1.5
RW-1	Apr-13	0.15	0.47	0.5	0.5	0.5	0.5	0.5
RW-1	Nov-13	0.12	0.4	0.25	0.5	0.5	0.5	1
RW-1	Apr-14	0.17	0.9	0.34	0.3	0.16	0.35	0.44
RW-1	NOV-14	0.19	0.72	0.25	1.0	1.0	1.0	3.0
RW-1	Apr-15	0.18	5.0	1.2	2.0	2.0	3.0	3.0
	NOV-15 Apr 16	0.52	0.90	0.10	2.0	2.0	3.0	3.0
	Nov-16	0.24	0.63	0.09	0.22	2.0	0.25	0.5
RW-1	Apr-17	0.5	0.00	0.26	2 00	2.0	3.00	3.0
RW-1	Nov-17	0.086	0.85	0.078	0.26	0.2	0.00	0.5
RW-1	Apr-18	0.2	0.69	0.26	0.23	0.2	0.31	0.5
RW-1	Nov-18	0.16	1.5	0.36	0.20	0.2	0.20	0.5
RW-1	Apr-19	0.11	0.73	0.25	0.20	0.2	0.25	0.5
RW-1	Nov-19	0.11	0.25	0.75	0.50	2.0	2.00	4.0
RW-1	Average	0.2	3.1	0.6	1.0	0.9	1.2	2.4
RW-6	Nov-03	1.81			569	23.1	10	116
RW-6	Aug-04	0.067	0.25	0.75	0.5	0.5	0.5	1
RW-6	Feb-05	0.101	0.25	0.75	0.5	0.5	0.788	1.3
RW-6	Nov-05	8.19	115	14.7	7.62	2.56	53.6	524
RW-6	Mar-06	31.80	560	300	12.7	9.15	96.7	568
RW-6	Nov-06	1.14	26.8	1.05	0.591	0.5	0.636	10
RW-6	May-07	1.02	38.9	5.05	34	1.44	16.6	15.2
RW-6	Nov-07	0.05	1.9	5.32	0.5	0.5	0.5	1
RW-6	Apr-08	0.33	5.56	0.542	10.2	1.22	9.56	6.9
RW-6	Nov-08	0.05	0.734	0.472	0.5	0.5	0.5	1
RW-6	Apr-09	0.175	1.14	0.476	6.93	0.5	3.08	3.32
RW-6	NOV-09	0.050	0.73	0.47	0.5	1	9	2
RVV-0	Apr-10	1.10	3.2	0.49	53	2	9.4	0.7
	Δρr-11	0.200	2.5	0.39	15.1	1	95	67
RW-6	Nov-11	0.595	0.37	0.38	0.5	1	9.0 1	3
RW-6	Apr-12	0.00	0.98	0.00	1 1	1	1	.3
RW-6	Nov-12	0.1	0.11	0.11	0.5	0.5	0.5	1.5
RW-6	Apr-13	0.18	1.1	0.49	0.82	0.5	0.5	0.55
RW-6	Nov-13	0.052	0.29	0.25	0.5	0.5	0.5	1
RW-6	Apr-14	0.19	1.4	0.36	2.1	0.34	1.3	0.64
RW-6	Nov-14	0.068	0.46	0.25	1.0	1.0	1.0	3.0
RW-6	Apr-15	0.13	0.46	0.26	2.0	2.0	3.0	3.0
RW-6	Nov-15	0.097	0.6	0.14	2.0	2.0	3.0	3.0
RW-6	Apr-16	0.21	6.3	2.4	2.0	2.0	3.0	3.0
RW-6	Nov-16	0.18	1.3	0.32	0.2	0.2	0.2	0.5
RW-6	Apr-17	0.5	0.66	0.51	2.0	2.0	3.0	3.0
RW-6	Nov-17	0.05	0.22	0.27	0.2	0.2	0.2	0.5
KW-6	Apr-18	U.11	0.54	0.25	0.2	0.2	0.2	0.5
KW-6	NOV-18	0.086	0.58	0.25	0.2	0.2	0.2	0.5
	Api-19 Nov-10	0.053	0.14	0.25	0.2	0.2 2.0	0.2 2.0	0.5 4 0
RW-0	Average	15	24 9	10 9	22 R	1 9	73	40.5
Groundwater	r Cleanun Level	1.0	10.0	10.0	71	1.0	1.5	40.0
Reporting L	imits/Units	0.05 mg/l	0.25 mg/l	.750 mg/l	0.5 ug/l	Varies	Varies	Varies

		Gasoline	Diesel	Oil	Benzene		Ethylbenzene	Xylenes
Well ID	Date	mg/l	mg/l	mg/l	ug/l	Toluene ug/l	ug/l	(total) ug/l
RW-5	Nov-03	2.10	4.13	0.75	5.21	0.657	83.5	186
RW-5	Aug-04	7.60	14.5	1.55	1.93	1.67	324	630
RW-5	Feb-05	3.18	17.4	15	37.8	40	38.5	287
RVV-5	Nov-05 Mar 06	19.60	1240	361	43.2 1.06	42	66.2 8.02	879
	Nov 06	0.741	0	7.5 1.67	1.06	24.2	0.03	129
	May-07	2 920	13 9	2.01	22.1	0.5	16.7	4.23
RW-5	Nov-07	1 430	2 16	0.639	1.08	0.5	1.87	2 07
RW-5	Apr-08	0 240	7 71	2 17	5.64	0.5	1 19	1 48
RW-5	Nov-08	1.520	0.916	0.472	6.32	0.5	2.85	3.55
RW-5	Apr-09	0.873	11.7	2.45	93.3	2.42	8.74	16.5
RW-5	Nov-09	0.066	0.4	0.47	0.5	1	1	2
RW-5	Apr-10	0.570	1.4	0.49	7.3	1	15	29
RW-5	Nov-10	0.785	0.9	0.39	30.5	1	2	5.3
RW-5	Apr-11	0.801	1.3	0.41	10.3	1	3.5	7
RW-5	Nov-11	0.18	1.2	0.39	9.2	1	5.6	3.9
RW-5	Apr-12	0.746	0.35	0.41	14.1	1	6.8	26
RW-5	NOV-12	0.1	0.38	0.41	1.6	0.5	0.5	1.5
	Apr-13 Nov 13	0.10	20 0.25	2.Z 0.25	0.57	0.5	0.5	0.5
RW-5	Apr-14	0.22	2.8	0.23	5.2	0.5	19	4 1
RW-5	Nov-14	0.40	17	0.56	10	10	1.0	3.0
RW-5	Apr-15	0.45	2.4	0.89	3.2	2.0	3.0	3.0
RW-5	Nov-15	0.39	2.2	0.36	2.0	2.0	3.0	3.0
RW-5	Apr-16	0.63	2.4	0.82	2.0	2.0	3.0	3.3
RW-5	Nov-16	0.72	4.4	1.00	0.59	0.2	0.40	0.41
RW-5	Apr-17	0.5	0.51	0.26	2.00	2.0	3.00	3.00
RW-5	Nov-17	0.32	1	0.26	1.10	0.3	2.60	0.74
RW-5	Apr-18	0.45	0.56	0.28	1.30	0.3	1.30	1.20
RW-5	Nov-18	0.25	4.2	2.10	1.40	0.2	0.76	4.50
RW-5	Apr-19	0.4	0.62	0.26	1.40	0.33	1.10	0.73
RW-5	Average	1.05	0.55 A3 A	12.8	0.50	2.0	2.0	<u>4.0</u> 72.1
<u> </u>	N 00		-10.1		0.0	11.2		
RW-4	Nov-03	4.89	CO4	450	36.1	44.3	337	281
	Aug-04	182.0	2 610	150	017 247	2820	2750	15,200
RW-4	Nov-05	45.4	3 650	1820	347	6940	1100	8 010
RW-4	Mar-06	26.1	440	150	30.2	654	346	3 340
RW-4	Nov-06	7.23	139	5.26	65.2	157	47	1,090
RW-4	May-07	0.82	8.08	0.543	3.97	0.547	3.89	77.5
RW-4	Nov-07	1.29	0.553	0.543	1.97	0.536	3.5	106
RW-4	Apr-08	0.07	2.91	0.532	0.5	0.5	0.5	4.57
RW-4	Nov-08	0.73	6.43	0.472	6.86	0.5	3.6	28.2
RW-4	Apr-09	0.565	7.93	0.481	8.17	0.5	1.43	18.3
RW-4	Nov-09	5.5	25	1.2	22	1.9	30	310
RW-4	Apr-10	4.2	10	0.49	46	1.6	24	155
	$\Delta pr_{-}11$	2.01	20 20 5	0.00	59.9 67 0	1.0		47.9 159
RW-4	Αρι-ΤΙ Νον-11	J.75 4 51	56.2	1.Z	18 5	1.2	44.0	08 3
RW-4	Apr-12	6.24	38.1	1.4	56.8	1.0	45.3	106
RW-4	Nov-12	0.771	10.7	9.2	7.5	0.5	3.9	10.1
RW-4	Apr-13	1.1	7.1	0.5	16	0.5	5.4	2.32
RW-4	Nov-13	0.77	0.63	0.25	12	0.5	6.2	12
RW-4	Apr-14	3.7	50	2.7	14	0.49	14	22
RW-4	Nov-14	1.9	8.7	0.57	15	1.0	16	23
RW-4	Apr-15	3.0	4.1	0.35	13	2.0	18	18
RW-4	Nov-15	2.3	18	0.95	13	0.45	5.3	7.6
KW-4	Apr-16	3.1 0.86	22	1.4	12	2.0	/	3.0 0 5
	NOV-10	0.00	00	2.9 0.06	1.9 0 /	0.2	U.41 1	0.0
	Apr-17	2.2	22	0.90 1.2	ö.4 22	2.U 1.2	4	3.U 1 2
RW-4	Apr-18	1.3	20	1.2	43	0.26	2.1	0.6
RW-4	Nov-18	1.2	33	1.2	12	0.34	0.45	11
RW-4	Apr-19	0.77	28	1.1	5.5	0.20	2.9	0.74
RW-4	Nov-19	0.48	35	3.8	10	2.0	2.0 2.0 2.0	
<u>RW-4</u>	Average 12.6 260 94 59.6 574.7		574.7	178.8	1135.9			
Groundwater	Cleanup Level	1.0	10.0	10.0	71			
Reporting L	imits/Units	0.05 mg/l	0.25 mg/l	.750 mg/l	0.5 ug/l	Varies	Varies	Varies

Well ID	Dato	Gasoline mg/l	Diesel ma/l	Oil ma/l	Benzene	Toluene ug/l	Ethylbenzene	Xylenes (total) ug/l
	Nov 02	2.07	iiig/i	ing/i	020	260	24.5	124
RW-2	Aug-04	2.07	46	1 / 1	2 270	382	34.5	1 1 1 8 0
RW-2	Feb-05	4.65	1.02	0.75	1.690	450	296	752
RW-2	Nov-05	2.82	0.76	0.708	1,540	299	159	353
RW-2	Mar-06	2.39	6.84	3.75	1,120	112	138	224
RW-2	Nov-06	13.10	14.3	1.05	1,830	516	410	1,810
RW-2	May-07	8.25	6.35	0.505	254	33.1	237	1,150
RW-2	Nov-07	3.55	3.32	0.538	895	5	79.4	172
RW-2	Apr-08	2.06	10.0	0.515	245	5	58	190
RW-2	Δpr-00	1.42 0.497	0.864	0.401	360 /0	4.04	9/9	40
RW-2	Nov-09	2 4	2.6	0.470	49	23	9.49 150	410
RW-2	Apr-10	1.5	1.0	0.49	200	1.5	66	98
RW-2	Nov-10	0.36	8.1	0.6	34.9	1.0	7.7	23.3
RW-2	Apr-11	1.0	1.5	0.39	146	1.3	27.8	51.7
RW-2	Nov-11	0.96	0.69	0.39	363	4.7	36.5	63.8
RW-2	Apr-12	0.57	13.9	0.74	139	1.0	13.7	17.4
RW-2	Nov-12	0.71	1.0	0.91	196	1.2	11.2	8.3
RW-2	Apr-13	0.47	3.0	0.49	230	2.0	20	6.6 E E
RW-2 RW-2	$\Delta nr_{-}1/$	2 20	4.0	0.25	290	2.9	0.2 84	79
RW-2	Nov-14	2.30	3.2	0.33	460	100	140	140
RW-2	Apr-15	2.20	2.7	0.3	340	28	77	55
RW-2	Nov-15	1.6	2.4	0.15	330	1.9	20	19
RW-2	Apr-16	4.1	50	2.3	250	16	40	31
RW-2	Nov-16	3.6	170	7.2	330	0.98	5.20	1.4
RW-2	Apr-17	1.7	7.4	0.28	150	130	29	15
RW-2	Nov-17	0.89	4.2	0.25	390	2.8	22	9.2
RW-2	Apr-18	1.1	52	2.2	130	6.6	4.9	2.2
RW-2	NOV-18 Apr 10	2.4	10	0.76	180	30	13	59
RW-2	Api-19 Nov-19	0.00	20	3.8	20 16	20	2.0	4.0
RW-2	Average	2.5	15.2	1.1	492	79.7	80.5	222.4
CM 119	Nov 03	2.20			614	20.2	67.0	1/1
GM-115	Aug-04	2.20	57	3 93	506	2 17	49.3	84.1
GM-11S	Feb-05	2.42	25.1	<15	55.6	0.848	25.5	17.3
GM-11S	Nov-05	2.15	37.4	<7.14	124	3.66	13.7	5.34
GM-11S	Mar-06	1.41	17.8	7.5	218	2.5	24.5	5
GM-11S	Nov-06	0.131	10.8	1.05	13.5	0.5	2.86	1.59
GM-11S	May-07	1.68	1.1	0.556	175	2.5	81.2	35.1
GM-11S	Nov-07	2.20	2.34	0.505	56.2	4.16	48.4	34.3
GM-115	Apr-08	1.93	1.23	0.532	00.7 95 3	1.70	185	132
GM-115 GM-115	Apr-09	1.00	0.942	0.472	5 34	0.898	19.1	14.0
GM-11S	Aug-09	1.90	1.2	0.48	71	2.4	37	6.3
GM-11S	Nov-09	1.50	3.6	0.48	36	1.1	48	24
GM-11S	Apr-10	3.00	5	0.5	46	1.6	93	156
GM-11S	Nov-10	1.39	1.8	0.48	42	1.9	64.9	37.1
GM-11S	Apr-11	1.42	0.52	0.4	18.4	1	26.5	20.1
GM-11S	Nov-11	2.28	0.47	0.38	30.9	1.7	22.9	10.3
GM-11S	Apr-12 Nov 12	2.24 0.671	1.1	0.38	33 11 /	1.7	59.2	40.4 27.0
GM-115 GM-11S	Apr-13	0.5	0.05	0.02	20	0.50	23	91
GM-11S pun	nping discontinue	ed in May 201	3 due to line	e blockage a	and concentra	ations mainly	below cleanup le	vels.
GM-11S	Nov-13	0.33	0.47	0.58	4.1	0.6	10	1
GM-11S	Apr-14	1.2	3.9	1.4	10	0.82	23	2.7
GM-11S	Nov-14	0.72	0.83	0.4	6.5	8.7	1.0	3.0
GM-11S	Apr-15	0.2	0.51	0.35	2.0	2.0	3.0	3.0
GM-11S	Nov-15	0.5	0.77	0.41	1.6	0.54	0.52	0.70
GN-115 GM-119	Api-10 Nov-16	0.52	1.1	ι.Ծ 0.21	14.U 0 2	2.U 0.2	3.U 0.2	5.U 0.5
GM_119	Anr-17	0.070	0.04	0.21	20	20	3.0	3.0
GM_119	Nov-17	0.0 0.82	1/	0.20	2.0	2.0	0.0	17
GM-11S	Apr-18	0.03	14	0.98	0.2	0.2	0.2	0.5
GM-11S	Nov-18	0.48	4.8	4.0	0.2	0.2	0.2	0.5
GM-11S	Apr-19	0.3	2.0	0.57	2.0	1.2	0.27	0.5
GM-11S	Nov-19	0.66	2.1	0.75	0.5	2.0	2.0	4.0
GM-11S	Average	1.2	6.1	1.0	69.2	2.9	31.1	25.4
Groundwate	r Cleanup Level	1.0	10.0	10.0	71			
Reporting L	.imits/Units	0.05 mg/l	0.25 mg/l	.750 mg/l	0.5 ug/of 6	3 Varies	Varies	Varies

		Gasoline	Diesel	Oil	Benzene		Ethylbenzene	Xylenes
Well ID	Date	mg/l	mg/l	mg/l	ug/l	Toluene ug/l	ug/l	(total) ug/l

Detection limits for many of the Oil analyses were raised due to sample dilution for diesel analyses.

These samples are listed with a "<" notation.

Values in italics were not detected at the listed reporting limit. Values in bold exceed the cleanup level for confirmational wells.

Note that cleanup levels are included for reference only. Cleanup levels are to be met at points of compliance,

which are downgradient from the recovery wells listed above.



Table 3: Waterfront Probing Field and Laboratory Groundwater Sampling Results

Project: 2019 Waterfront Probing Investigation

Site: Former BP Harbor Island Terminal

			Depth To Groundwater	Groundwater Elevation (NAVD 88	Sheen Strength	Measurable Free LNAPL	Gasoline NTWPH-Gx Result	Diesel NTWPH-Dx Result	Oil NTWPH-Dx Result	Benzene EPA-8260 Result
Boring ID #	Well ID #	Sample Date and Time	(Feet BGS)	Datum)	Observed	Present	(µg/L)	(µg/L)	(µg/L)	(µg/L)
B-01	MW-01-PI	9/27/19 8:17 AM	5.41	6.55	Slight	None	110	460	<750	<0.5
B-03	MW-02-PI	9/26/19 11:05 AM	8.87	5.97	Slight	None	530	13,000	<1,500	<0.5
B-05	MW-03-PI	9/26/19 10:17 AM	8.76	6.32	Slight	None	<50	13,000	2,200	<0.5
B-07	MW-04-PI	9/26/19 9:47 AM	8.73	6.22	None	None	150	2,500	<750	<0.5
B-09	MW-05-PI	9/26/19 9:14 AM	9.09	6.1	None	None	<50	3,400	2,000	<0.5
B-11	MW-06-PI	9/26/19 8:28 AM	8.83	6.28	None	None	52	380	<750	<0.5
B-25	MW-07-PI	9/26/19 12:25 PM	8.98	6.14	Slight	None	240	5,600	860	<0.5
B-33	MW-08-PI	9/27/19 8:50 AM	5.96	6.06	None	None	180	<250	<750	<0.5
B-35	MW-09-PI	9/27/19 9:12 AM	5.67	5.88	Slight	None	530	1,800	<750	<0.5
B-41	MW-10-PI	9/27/19 9:50 AM	5.53	5.36	Slight	None	2,200	6,000	<750	68
B-42	MW-11-PI	9/27/19 10:45 AM	5.51	5.5	Moderate	None	6,300	3,400	<750	63
			Groundwater C	Cleanup Criteria ^(a) :	None ^(a)	<0.01 feet	1,000 ^(a)	10,000 ^(a)	10,000 ^(a)	71 ^(a)
						Reporting Limit:	50	250	750	0.5

Definitions:

µg/L - micrograms per liter

BGS - Below Ground Surface

LNAPL - Light Non-Aqueous Phase Liquid

Notes:

Values in bold exceed the associated cleanup criterion.

^(a) - Associated cleanup criteria are to be met at the point of compliance, which is downgradient from the wells in this study. Data provided for comparison purposes only.



					Field Screening Results					Laboratory Analytical Results						
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
Device ID #		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, U - no)	Result (PPIVI)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	ТРН
B-01 B-01	MW-01	7/30/2019	2	10.26	SP SP	Yes	0	0	None							
B-01	MW-01	7/30/2019	3	9.26	SP	Yes	0	0	None							
B-01	MW-01	7/30/2019	4	8.26	SP	Yes	0	0	None							
B-01	MW-01	7/30/2019	5	7.26	SP	Yes	0	0	None							
B-01	MW-01	7/30/2019	6	6.26		No										
B-01	MW-01	7/30/2019	7	5.26		No										
B-01	MW-01	7/30/2019	8	4.26	SP	Yes	0	0	Moderate							
B-01	WW-01	7/30/2019	9	3.20	52	Yes	1	270	Moderate	2000	12000	1000	16000	1.09/	759/	69/
B-01 B-01	MW-01	7/30/2019	10	1.26	Jr	No	1	270	пеачу	3000	12000	1000	10000	19%	73%	0%
B-01	MW-01	7/30/2019	12	0.26	SP	Yes	1	232	Heavy							
B-01	MW-01	7/30/2019	13	-0.74	SP	Yes	1	199	Heavy							
B-01	MW-01	7/30/2019	14	-1.74	SP	Yes	1	5	Slight							
B-01	MW-01	7/30/2019	15	-2.74	SP	Yes	1	1	Slight							
B-24		7/22/2019	1	14.54	SP	Yes	0	0	None							
B-24		7/22/2019	2	13.54	SP	Yes	0	0	None							
B-24 B-24		7/22/2019	3	12.54	5P 5D	Ves	0	0	None							
B-24		7/22/2019	5	10.54	SP	Yes	0 0	0	None							
B-24		7/22/2019	6	9.54	SP	Yes	0	0	None							
B-24		7/22/2019	7	8.54	SP	Yes	0	0	None							
B-24		7/22/2019	8	7.54	SP	Yes	0	0	None							
B-24		7/22/2019	9	6.54	SP	Yes	1	0	Slight							
B-24		7/22/2019	10	5.54	SP	Yes	1	63	Moderate							
B-24		7/22/2019	11	4.54	SP	Yes	1	54	Moderate							
B-24 B-24		7/22/2019	12	2.54	SP	Ves	1	1029	Hope	1300	9800	940	12040	11%	91%	9%
B-24		7/22/2019	14	1.54	SP	Yes	1	249	Moderate	1500	5866	540	12040	11/6	81/6	878
B-24		7/22/2019	15	0.54	SP	Yes	1	5	Slight							
B-24		7/22/2019	16	-0.46	SP	Yes	1	154	Slight							
B-24		7/22/2019	17	-1.46	SP	Yes	1	100	Slight							
B-24		7/22/2019	18	-2.46	SP	Yes	1	20	None							
B-24		7/22/2019	19	-3.46	SP	Yes	1	0	None							
B-24		7/22/2019	20	-4.46	OH	Yes	1	0	None							
B-14 B-14		7/22/2019	2	13.44	SP	Yes	0	0	None							
B-14		7/22/2019	3	12.44	SP	Yes	0	0	None							
B-14		7/22/2019	4	11.44	SP	Yes	0	0	None							
B-14		7/22/2019	5	10.44	SP	Yes	0	0	None							
B-14		7/22/2019	6	9.44	SP	Yes	0	0	None							
B-14		7/22/2019	7	8.44	SP	Yes	0	0	None							
B-14		7/22/2019	8	7.44	SP	Yes	0	0	None							
B-14 P-14		7/22/2019	9	5.44	58	Yes	0	270	None							
B-14 B-14		7/22/2019	10	4.44	SP	Yes	1	70	Moderate							
B-14		7/22/2019	12	3.44	SP	Yes	1	418	Moderate							
B-14		7/22/2019	13	2.44	SP	Yes	1	694	Moderate	850	15000	1400	17250	5%	87%	8%
B-14		7/22/2019	14	1.44	SP	Yes	1	50	Slight							
B-14		7/22/2019	15	0.44	SP	Yes	1	37	Slight							
B-14		7/22/2019	16	-0.56	SP	Yes	1	130	Moderate							
B-14		7/22/2019	17	-1.56	SP	Yes	1	15	Slight							
B-14 B-14		7/22/2019	10	-2.56	5P 5D	res Ves	1	36	Slight							
B-14		7/22/2019	20	-4.56	OH	Yes	1	0	None							
B-02		7/22/2019	1	14.33	0	No	-									
B-02		7/22/2019	2	13.33		No										



					Field Screening Results					Laboratory Analytical Results						
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
Destine ID #	M-11 1D #	Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	well ID #	Sample Date	BGS)	Datum)	Туре	or NO)	(1 - yes, 0 - no)	Result (PPIVI)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total IPH	IPH
B-02		7/22/2019	3	12.33	SP	Yes	0	0	None							
B-02		7/22/2019	4 5	10.22	SP SD	Yes	0	0	None							
B-02 B-02		7/22/2019	6	9.33	SP	No	0	0	None							
B-02		7/22/2019	7	8.33	SP	Yes	0	0	None							
B-02		7/22/2019	8	7.33	SP	Yes	0	0	None							
B-02		7/22/2019	9	6.33	SP	No										
B-02		7/22/2019	10	5.33	SP	Yes	0	0	None							
B-02		7/22/2019	11	4.33	SP	Yes	0	0	None							
B-02		7/22/2019	12	3.33	SP	Yes	1	330	Heavy	480	29000	4600	34080	1%	85%	13%
B-02		7/22/2019	13	2.33	SP	Yes	1	319	Heavy							
B-02		7/22/2019	14	1.33	SP SD	Yes	1	40	Slight							
B-02 B-02		7/22/2019	15	-0.67	SP SP	Ves	1	170	Hoppy							
B-02		7/22/2019	17	-1.67	SP	Yes	1	68	Slight							
B-02		7/22/2019	18	-2.67	SP	Yes	1	10	Slight							
B-02		7/22/2019	19	-3.67	SP	Yes	1	0	None							
B-02		7/22/2019	20	-4.67	ОН	Yes	1	0	None							
B-25	MW-07	7/23/2019	1	14.56		No										
B-25	MW-07	7/23/2019	2	13.56	SP	Yes	0	0	None							
B-25	MW-07	7/23/2019	3	12.56	SP	Yes	0	0	None							
B-25	MW-07	7/23/2019	4	11.56	SP	Yes	0	0	None							
B-25	NIW-07	7/23/2019	5	10.56	5P	Yes	0	0	None							
B-25	MW-07	7/23/2019	7	9.56	SP SP	Vec	0	0	None							
B-25	MW-07	7/23/2019	8	7.56	SP	Yes	0	0	None							
B-25	MW-07	7/23/2019	9	6.56	SP	Yes	o	ő	Slight							
B-25	MW-07	7/23/2019	10	5.56	SP	Yes	1	17	Slight							
B-25	MW-07	7/23/2019	11	4.56	SP	Yes	1	116	Moderate							
B-25	MW-07	7/23/2019	12	3.56	SP	Yes	1	3058	Heavy	1200	2100	240	3540	34%	59%	7%
B-25	MW-07	7/23/2019	13	2.56	SP	Yes	1	1403	Heavy							
B-25	MW-07	7/23/2019	14	1.56	SP	Yes	1	0	Moderate							
B-25	MW-07	7/23/2019	15	0.56	SP	Yes	1	0	Slight							
B-25	MW-07	7/23/2019	16	-0.44	SP SD	Yes	1	380	Slight							
B-25	MW-07	7/23/2019	18	-1.44	SP	Ves	1	0	None							
B-25	MW-07	7/23/2019	19	-3.44	SP	Yes	1	ő	None							
B-25	MW-07	7/23/2019	20	-4.44	SP	Yes	1	0	None							
B-03	MW-02	7/23/2019	1	14.31		No										
B-03	MW-02	7/23/2019	2	13.31		No										
B-03	MW-02	7/23/2019	3	12.31	SP	Yes	0	0	None							
B-03	MW-02	7/23/2019	4	11.31	SP	Yes	0	0	None							
B-03	MW-02	7/23/2019	5	10.31	SP	Yes	0	0	None							
B-03	NIW-02	7/23/2019	7	9.31	SP SD	Yes	0	0	None							
B-03	MW-02	7/23/2019	8	7 31	SP SP	Vec	0	0	None							
B-03	MW-02	7/23/2019	9	6.31	SP	Yes	1	390	Heavy							
B-03	MW-02	7/23/2019	10	5.31	SP	Yes	1	850	Moderate							
B-03	MW-02	7/23/2019	11	4.31	SP	Yes	1	780	Moderate							
B-03	MW-02	7/23/2019	12	3.31	SP	Yes	1	1543	Moderate							
B-03	MW-02	7/23/2019	13	2.31	SP	Yes	1	2400	Moderate							
B-03	MW-02	7/23/2019	14	1.31	SP	Yes	1	3680	Free LNAPL	53	3100	510	3663	1%	85%	14%
B-03	MW-02	7/23/2019	15	0.31	SP	Yes	1	0	Free LNAPL							
B-03	MW-02	//23/2019	16	-0.69	SP	Yes	1	200	Nioderate							
B-03 B-03	MW-02	7/23/2019	1/	-1.69	52	Yes	1	10/1	None							
B-03	MW-02	7/23/2019	19	-3.69	SP	Yes	1	0	None							
1 2 2 2 1		1/25/2015	1 1	5.65	1		-	ı v	1	1 1	1	1	1			I I

.



							Field Screening Result	S				Laborator	y Analytical Results			
			Commite	Sample		Coll Decement	C		Characterization			0"	Table Toll Caralina			
		Completion 8	Sample	Elevation		Soll Recovered	Groundwater	PID Field	Sneen Test	Gasoline NIWPH-	Diesei NTWPH-		Discal L Oil	% Casalina of	% Discal of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	(NAVD 88	Type	or No)	(1 - ves. 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	TPH
B-03	MW-02	7/23/2019	20	-4.69	., ,, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Ves	1	0	None	(6/6/	(8/8/	(8/8/	1100urt (1116/116)			
B-15	11111 02	7/23/2019	1	14.42		No	-		None							
B-15		7/23/2019	2	13.42		No										
B-15		7/23/2019	3	12.42	SP	Yes	0	0	None							
B-15		7/23/2019	4	11.42	SP	Yes	0	0	None							
B-15		7/23/2019	5	10.42	SP	Yes	0	0	None							
B-15		7/23/2019	7	8.42	SP	Yes	ő	0	None							
B-15		7/23/2019	8	7.42	SP	Yes	0	0	None							
B-15		7/23/2019	9	6.42	SP	Yes	1	0	Heavy							
B-15		7/23/2019	10	5.42	SP	Yes	1	178	Heavy							
B-15		7/23/2019	11	4.42	SP	Yes	1	831	Heavy	1200	9800	1100	12100	10%	81%	9%
B-15 B-15		7/23/2019	12	3.42	SP SP	Yes	1	2262	Noderate							
B-15		7/23/2019	14	1.42	SP	Yes	1	5	None							
B-15		7/23/2019	15	0.42	SP	Yes	1	0	None							
B-15		7/23/2019	16	-0.58	SP	Yes	1	0	None							
B-15		7/23/2019	17	-1.58	SP	Yes	1	0	None							
B-15		7/23/2019	18	-2.58	SP	Yes	1	0	None							
B-15		7/23/2019	19	-3.58	OH	Yes	1	0	None							
B-15 B-26		7/23/2019	20	-4.58	UH	No	1	0	None							
B-26		7/22/2019	2	13.45	SP	Yes	0	0	None							
B-26		7/23/2019	3	12.45	SP	Yes	0	0	None							
B-26		7/23/2019	4	11.45	SP	Yes	0	0	None							
B-26		7/23/2019	5	10.45	SP	Yes	0	0	None							
B-26		7/23/2019	6	9.45	SP	Yes	0	0	None							
B-20 B-26		7/23/2019	8	8.45	SP SP	Yes	0	0	None							
B-26		7/23/2019	9	6.45	SP	Yes	0	0	Slight							
B-26		7/23/2019	10	5.45	SP	Yes	1	350	Free LNAPL							
B-26		7/23/2019	11	4.45	SP	Yes	1	1190	Heavy	580	4200	410	5190	11%	81%	8%
B-26		7/23/2019	12	3.45	SP	Yes	1	760	Moderate							
B-26		7/23/2019	13	2.45	SP	Yes	1	40	Slight							
B-26		7/23/2019	14	0.45	SP	Yes	1	10	None							
B-26		7/23/2019	16	-0.55	SP	Yes	1	970	Moderate							
B-26		7/23/2019	17	-1.55	SP	Yes	1	30	None							
B-26		7/23/2019	18	-2.55	SP	Yes	1	3	None							
B-26		7/23/2019	19	-3.55	ОН	Yes	1	0	None							
B-26 B-04		7/23/2019	20	-4.55	OH	Yes	1	0	None							
B-04		7/22/2019	2	13.44	SP	Yes	0	0	None							
B-04		7/23/2019	3	12.44	SP	Yes	0	0	None							
B-04		7/23/2019	4	11.44	SP	Yes	0	0	None							
B-04		7/23/2019	5	10.44	SP	Yes	0	0	None							
B-04		7/23/2019	6	9.44	SP	Yes	0	0	None							
B-04 B-04		7/23/2019	8	0.44 7 44	SP SP	res Yes	0		None							
B-04		7/23/2019	9	6.44	SP	Yes	1	151	Slight							
B-04		7/23/2019	10	5.44	SP	Yes	1	1023	Free LNAPL							
B-04		7/23/2019	11	4.44	SP	Yes	1	850	Heavy							
B-04		7/23/2019	12	3.44	SP	Yes	1	1691	Moderate	910	3100	350	4360	21%	71%	8%
B-04		7/23/2019	13	2.44	SP / OH	Yes	1	600	Heavy							
B-04 B-04		7/23/2019	14	1.44	SP SP	Yes	1	0	None							
B-04		7/23/2019	16	-0.56	SP	Yes	1	32	None							



							Field Screening Result	5	1			Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	ТРН
B-04		7/23/2019	17	-1.56	SP	Yes	1	0	None							
B-04		7/23/2019	18	-2.56	SP	Yes	1	0	None							
B-04 B-04		7/23/2019	20	-3.50	OH	Vec	1	0	None							
B-05	MW-03	7/23/2019	1	14.47	011	No	-	0	None							
B-05	MW-03	7/22/2019	2	13.47	SP	Yes	0	0	None							
B-05	MW-03	7/23/2019	3	12.47	SP	Yes	0	0	None							
B-05	MW-03	7/23/2019	4	11.47	SP	Yes	0	0	None							
B-05	MW-03	7/23/2019	5	10.47	SP	Yes	0	0	None							
B-05	MW-03	7/23/2019	6	9.47	SP	Yes	0	0	None							
B-05	MW-03	7/23/2019	/	8.47	SP SD	Yes	0	0	None							
B-05	MW-03	7/23/2019	°	6.47	SP SP	Vec	1	0	None							
B-05	MW-03	7/23/2019	10	5.47	SP	Yes	1	100	Slight	8.9	1300	250	1558.9	1%	83%	16%
B-05	MW-03	7/23/2019	11	4.47	SP	Yes	1	85	Slight							
B-05	MW-03	7/23/2019	12	3.47	SP	Yes	1	66	Slight							
B-05	MW-03	7/23/2019	13	2.47	SP	Yes	1	27	None							
B-05	MW-03	7/23/2019	14	1.47	SP	Yes	1	29	None							
B-05	MW-03	7/23/2019	15	0.47	SP	Yes	1	16	None							
B-05	MW-03	7/23/2019	16	-0.53		No										
B-05	MW-03	7/23/2019	17	-1.53		NO										
B-05	MW-03	7/23/2019	19	-3.53		No										
B-05	MW-03	7/23/2019	20	-4.53		No										
B-27		7/23/2019	1	14.53		No										
B-27		7/22/2019	2	13.53	SP	Yes	0	0	None							
B-27		7/23/2019	3	12.53	SP	Yes	0	0	None							
B-27		7/23/2019	4	11.53	SP	Yes	0	0	None							
B-27		7/23/2019	5	10.53	SP	Yes	0	0	None							
B-27 B-27		7/23/2019	7	9.53	SP SP	Yes	0	0	None							
B-27		7/23/2019	8	7.53	SP	Yes	0 0	0	None							
B-27		7/23/2019	9	6.53	SP	Yes	0	0	None							
B-27		7/23/2019	10	5.53	SP	Yes	0	0	None							
B-27		7/23/2019	11	4.53	SP	Yes	1	35	Slight							
B-27		7/23/2019	12	3.53	SP	Yes	1	155	Slight	870	3900	400	5170	17%	75%	8%
B-27		7/23/2019	13	2.53	SP	Yes	1	82	None							
B-2/		7/23/2019	14	1.53	SP SD	Yes	1	33	None							
B-27		7/23/2019	15	-0.47	SP SP	Yes	1	110	Slight							
B-27		7/23/2019	17	-1.47	SP	Yes	1	66	Slight							
B-27		7/23/2019	18	-2.47	SP	Yes	1	0	None							
B-27		7/23/2019	19	-3.47	SP	Yes	1	0	None							
B-27		7/23/2019	20	-4.47	SP	Yes	1	0	None							
B-17		7/24/2019	1	14.49		No										
B-17		7/24/2019	2	13.49	SP	Yes	0	0	None							
B-1/		7/24/2019	3	12.49	58	Yes	U	U	None							
B-17		7/24/2019	4 5	10.49	58	res Vec	0	0	None							
B-17		7/24/2019	6	9.49	SP	Yes	0	0	None							
B-17		7/24/2019	7	8.49	SP	Yes	0	0	None							
B-17		7/24/2019	8	7.49	SP	Yes	0	0	None							
B-17		7/24/2019	9	6.49	SP	Yes	1	7	Slight							
B-17		7/24/2019	10	5.49	SP	Yes	1	77	Moderate							
B-17		7/24/2019	11	4.49	SP	Yes	1	0	None							
B-17		7/24/2019	12	3.49	SP	Yes	1	0	None							
B-1/		//24/2019	13	2.49	I SP	Yes	1	66	Moderate	I		1				



							Field Screening Result	5	1			Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	TPH
B-17		7/24/2019	14	1.49	SP	Yes	1	11	Slight							
B-17		7/24/2019	15	0.49	SP	Yes	1	0	Slight							
B-17		7/24/2019	16	-0.51	SP	Yes	1	5	None							
B-17		7/24/2019	17	-1.51	SP	Yes	1	232	Heavy	430	5500	490	6420	7%	86%	8%
B-1/		7/24/2019	18	-2.51	SP	Yes	1	10	Slight							
B-17 B-17		7/24/2019	20	-3.51	SM / OH	Ves	1	2	None							
B-06		7/24/2019	1	14.48	31017 011	No	1	2	None							
B-06		7/24/2019	2	13.48		No										
B-06		7/24/2019	3	12.48	SP	Yes	0	0	None							
B-06		7/24/2019	4	11.48	SP	Yes	0	0	None							
B-06		7/24/2019	5	10.48	SP	Yes	0	0	None							
B-06		7/24/2019	6	9.48	SP	Yes	0	0	None							
B-06		7/24/2019	7	8.48	SP	Yes	0	0	None							
B-06		7/24/2019	8	7.48	SP	Yes	0	0	None							
B-06		7/24/2019	9	6.48	SP	Yes	1	0	Slight							
B-06		7/24/2019	10	5.48	58	Yes	1	1//	Heavy							
B-06		7/24/2019	12	4.46	5P 5D	Ves	1	265	Free I NAPI							
B-06		7/24/2019	13	2.48	SP	Yes	1	323	Free I NAPI	950	10000	1100	12050	8%	83%	9%
B-06		7/24/2019	14	1.48	SP	Yes	1	187	Moderate							
B-06		7/24/2019	15	0.48	SP	Yes	1	52	Moderate							
B-06		7/24/2019	16	-0.52	SP	Yes	1	321	Moderate							
B-06		7/24/2019	17	-1.52	SP	Yes	1	16	Moderate							
B-06		7/24/2019	18	-2.52	SM / OH	Yes	1	2	Moderate							
B-06		7/24/2019	19	-3.52	SM / OH	Yes	1	2	Slight							
B-06		7/24/2019	20	-4.52	SM / OH	Yes	1	1	Slight							
B-28		7/24/2019	1	14.51	CD.	NO	0	0	Nono							
B-20 B-28		7/24/2019	2	12.51	5P 5D	Ves	0	0	None							
B-28		7/24/2019	4	11.51	SP	Yes	0	0	None							
B-28		7/24/2019	5	10.51	SP	Yes	0	0	None							
B-28		7/24/2019	6	9.51	SP	Yes	0	0	None							
B-28		7/24/2019	7	8.51	SP	Yes	0	0	None							
B-28		7/24/2019	8	7.51	SP	Yes	0	0	None							
B-28		7/24/2019	9	6.51	SP	Yes	1	0	None							
B-28		7/24/2019	10	5.51	SP	Yes	1	0	None							
B-28		7/24/2019	11	4.51	SP	Yes	1	0	None	160	400	120	790	210/	C20/	170/
B-20		7/24/2019	12	2 51	SP	Yes	1	0	Slight	100	490	150	760	21%	63%	1/%
B-28		7/24/2019	14	1.51	SP	Yes	1	0	None							
B-28		7/24/2019	15	0.51	SP	Yes	1	0	None							
B-28		7/24/2019	16	-0.49	SP	Yes	1	12	Moderate							
B-28		7/24/2019	17	-1.49	SP	Yes	1	0	None							
B-28		7/24/2019	18	-2.49	SP	Yes	1	0	None							
B-28		7/24/2019	19	-3.49	ОН	Yes	1	0	None							
B-28		7/24/2019	20	-4.49	ОН	Yes	1	0	None							
B-18		7/24/2019	1	14.45		No										
B-18 D-19		7/24/2019	2	13.45	CD.	NO	0	0	Nono							
B-18		7/24/2019	4	12.45	SP SP	Yes	0	0	None							
B-18		7/24/2019	5	10.45	SP	Yes	0	0	None							
B-18		7/24/2019	6	9.45	SP.	Yes	õ	0	None							
B-18		7/24/2019	7	8.45	SP	Yes	0	0	None							
B-18		7/24/2019	8	7.45	SP	Yes	0	0	None							
B-18		7/24/2019	9	6.45	SP	Yes	1	0	None							
B-18		7/24/2019	10	5.45	SP	Yes	1	36	Heavy							



							Field Screening Result	S	1			Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
Desing ID #		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
BUTTING ID #	weii iD #	Sample Date	BG3)	Datullij	туре		(1 - yes, 0 - 110)	Result (FFIVI)	Result	(IIIg/Kg)	(IIIg/Kg)	(IIIg/Kg)	Result (IIIg/kg)	TOLATIPH	TOLATIFH	IFA
B-18 B-19		7/24/2019	11	4.45	SP	Yes	1	24	Slight	880	6400	710	7000	11%	20%	0%
B-18		7/24/2019	12	2.45	SP	Yes	1	60	Moderate	880	0400	/10	7350	11/6	80%	576
B-18		7/24/2019	14	1.45	SP	Yes	1	10	Slight							
B-18		7/24/2019	15	0.45	SP	Yes	1	8	None							
B-18		7/24/2019	16	-0.55	SP	Yes	1	2	None							
B-18		7/24/2019	17	-1.55	SP	Yes	1	0	None							
B-18		7/24/2019	18	-2.55	SP	Yes	1	0	None							
B-18		7/24/2019	19	-3.55	ОН	Yes	1	0	None							
B-18 B-07	MM-04	7/24/2019	20	-4.55	OH	Yes	1	0	None							
B-07	MW-04	7/24/2019	2	13.44		No										
B-07	MW-04	7/24/2019	3	12.44	SP	Yes	0	0	None							
B-07	MW-04	7/24/2019	4	11.44	SP	Yes	0	0	None							
B-07	MW-04	7/24/2019	5	10.44	SP	Yes	0	0	None							
B-07	MW-04	7/24/2019	6	9.44	SP	Yes	0	0	None							
B-07	MW-04	7/24/2019	7	8.44	SP	Yes	0	0	None							
B-07	MW-04	7/24/2019	8	7.44	SP	Yes	0	0	None							
B-07	MW-04	7/24/2019	9	6.44	SP	Yes	1	0	None							
B-07	MW-04	7/24/2019	10	5.44	SP SD	Yes	1	/8	Moderate							
B-07	MW-04	7/24/2019	11	3.44	SP SP	Yes	1	260	Heavy	840	3800	470	5110	16%	7/%	9%
B-07	MW-04	7/24/2019	13	2.44	SP	Yes	1	105	Slight	040	5666	470	5110	10/8	7470	576
B-07	MW-04	7/24/2019	14	1.44	SP	Yes	1	13	Slight							
B-07	MW-04	7/24/2019	15	0.44	SP	Yes	1	4	Moderate							
B-07	MW-04	7/24/2019	16	-0.56	SP	Yes	1	13	Slight							
B-07	MW-04	7/24/2019	17	-1.56	SP	Yes	1	3	None							
B-07	MW-04	7/24/2019	18	-2.56	SP	Yes	1	3	None							
B-07	MW-04	7/24/2019	19	-3.56	ОН	Yes	1	0	None							
B-07	MW-04	7/24/2019	20	-4.56	ОН	Yes	1	0	None							
B-29		7/24/2019	1	14.51	CD.	No	0	0	Nees							
B-29		7/24/2019	2	13.51	5P	Yos	0	0	None							
B-29		7/24/2019	4	11 51	SP	Yes	0	0	None							
B-29		7/24/2019	5	10.51	SP	Yes	0	0	None							
B-29		7/24/2019	6	9.51	SP	Yes	0	0	None							
B-29		7/24/2019	7	8.51	SP	Yes	0	0	None							
B-29		7/24/2019	8	7.51	SP	Yes	0	0	None							
B-29		7/24/2019	9	6.51	SP	Yes	1	0	None							
B-29		7/24/2019	10	5.51	SP	Yes	1	0	None							
B-29		7/24/2019	11	4.51	SP	Yes	1	130	Slight	490	2900	360	3750	13%	77%	10%
B-29		7/24/2019	12	3.51	SP SD	Yes	1	10	None							
B-29 B-20		7/24/2019	13	2.51	58	Yes	1	0	None							
B-29		7/24/2019	15	0.51	SP	Yes	1	0	None							
B-29		7/24/2019	16	-0.49	SP	Yes	1	0	None							
B-29		7/24/2019	17	-1.49	SP	Yes	1	0	None							
B-29		7/24/2019	18	-2.49	SP	Yes	1	0	None							
B-29		7/24/2019	19	-3.49	ОН	Yes	1	0	None							
B-29		7/24/2019	20	-4.49	ОН	Yes	1	0	None							
B-30		7/24/2019	1	14.55		No										
B-30		7/24/2019	2	13.55	SP	Yes	0	0	None							
B-30		7/24/2019	3	12.55	58	Yes	U	0	None							
B-30 B_20		7/24/2019	4 c	11.55	57	Yes	0		None							
B-30		7/24/2019	6	9,55	SP	Yes	0	0	None							
B-30		7/24/2019	7	8.55	SP	Yes	0	0	None							



							Field Screening Result	S	1			Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	ТРН
B-30		7/24/2019	8	7.55	SP	Yes	0	0	None							
B-30		7/24/2019	9	6.55	SP	Yes	0	0	None							
B-30		7/24/2019	10	5.55	SP	Yes	1	25	Moderate							
B-30		7/24/2019	11	4.55	SP	Yes	1	40	None							
B-30		7/24/2019	12	3.55	SP	Yes	1	210	Moderate	800	4700	320	5820	14%	81%	5%
B-30		7/24/2019	13	2.55	SP	Yes	1	10	Heavy							
B-30		7/24/2019	14	1.55	SP	Yes	1	5	Slight							
B-30		7/24/2019	15	0.55	SP	Yes	1	3	None							
B-30		7/24/2019	16	-0.45	SP	Yes	1	5	None							
B-30		7/24/2019	17	-1.45	SP	Yes	1	3	None							
B-30		7/24/2019	18	-2.45	SP	Yes	1	0	None							
B-30		7/24/2019	19	-3.45	ОН	Yes	1	0	None							
B-30		7/24/2019	20	-4.45	OH	Yes	1	0	None							
B-19 B-10		7/24/2019	2	14.52	SD	Vor	0	0	Nana							
B-19 B-19		7/24/2019	2	12.52	SP	Ves	0	0	None							
B-19		7/24/2019	4	11 52	SP	Yes	0	0	None							
B-19		7/24/2019	5	10.52	SP	Yes	0	0	None							
B-19		7/24/2019	6	9.52	SP	Yes	0	0	None							
B-19		7/24/2019	7	8.52	SP	Yes	0	0	None							
B-19		7/24/2019	8	7.52	SP	Yes	0	0	None							
B-19		7/24/2019	9	6.52	SP	Yes	0	3	None							
B-19		7/24/2019	10	5.52	SP	Yes	1	85	Moderate							
B-19		7/24/2019	11	4.52	SP	Yes	1	360	Moderate							
B-19		7/24/2019	12	3.52	SP	Yes	1	382	Heavy							
B-19		7/24/2019	13	2.52	SP	Yes	1	415	Heavy	2000	17000	1400	20400	10%	83%	7%
B-19		7/24/2019	14	1.52	SP	Yes	1	5	Moderate							
B-19		7/24/2019	15	0.52	SP	Yes	1	1	Slight							
B-19		7/24/2019	16	-0.48	SP	Yes	1	2	Moderate							
B-19		7/24/2019	17	-1.48	SP	Yes	1	130	Slight							
B-19		7/24/2019	18	-2.48	SP	Yes	1	2	None							
B-19		7/24/2019	19	-3.48	ОН	Yes	1	0	None							
B-19		7/24/2019	20	-4.48	ОН	Yes	1	0	None							
B-08		7/24/2019	1	14.45		No										
B-08		7/24/2019	2	13.45	CD.	NO	0	0	Nana							
B-08		7/24/2019	5	12.45	58	Vec	0	0	None							
B-08		7/24/2019	4 5	10.45	5P	Voc	0	0	None							
B-08		7/24/2019	6	9.45	SP	Yes	0	0	None							
B-08		7/24/2019	7	8.45	SP	Yes	0	0	None							
B-08		7/24/2019	8	7.45	SP	Yes	0	o	None							
B-08		7/24/2019	9	6.45	SP	Yes	0	0	None							
B-08		7/24/2019	10	5.45	SP	Yes	1	140	Moderate	540	8200	890	9630	6%	85%	9%
B-08		7/24/2019	11	4.45	SP	Yes	1	100	Moderate							
B-08		7/24/2019	12	3.45	SP	Yes	1	125	Slight							
B-08		7/24/2019	13	2.45	SP	Yes	1	23	Slight							
B-08		7/24/2019	14	1.45	SP	Yes	1	12	Slight							
B-08		7/24/2019	15	0.45	SP	Yes	1	8	Slight							
B-08		7/24/2019	16	-0.55	SP	Yes	1	28	Moderate							
B-08		7/24/2019	17	-1.55	SP	Yes	1	0	Slight							
B-08		7/24/2019	18	-2.55	SP	Yes	1	0	None							
B-08		7/24/2019	19	-3.55	ОН	Yes	1	0	None							
B-08		7/24/2019	20	-4.55	ОН	Yes	1	0	None							
B-20		7/25/2019	1	14.4/	6.0	No	0	0	News							
B-20		7/25/2019	2	13.4/	52	Yes	U		None							
B-20		7/25/2019	3	11.47	58	Voc	0		None							
D-20	ı – – – – – – – – – – – – – – – – – – –	1/25/2019	- 4	1 11.47	1 JP	162	U		INOTIC			1	1			



							Field Screening Result	5				Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
Device ID #		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	well ID #	Sample Date	BG2)	Datum)	Туре	or NO)	(1 - yes, 0 - no)	Result (PPIVI)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	IPH
B-20		7/25/2019	5	10.47	SP	Yes	0	0	None							
B-20 B-20		7/25/2019	7	9.47	SP SD	Yes	0	0	None							
B-20 B-20		7/25/2019	8	7.47	SP	Yes	0	0	None							
B-20		7/25/2019	9	6.47	SP	Yes	1	0	None							
B-20		7/25/2019	10	5.47	SP	Yes	1	130	Free LNAPL							
B-20		7/25/2019	11	4.47	SP	Yes	1	70	None							
B-20		7/25/2019	12	3.47	SP	Yes	1	65	None							
B-20		7/25/2019	13	2.47	SP	Yes	1	48	Free LNAPL							
B-20		7/25/2019	14	1.47	SP	Yes	1	220	Free LNAPL	120	22000	4000		20/	040/	70/
B-20		7/25/2019	15	0.47	SP	Yes	1	243	Heavy	420	23000	1800	25220	2%	91%	1%
B-20 B-20		7/25/2019	10	-0.55	SP SP	Ves	1	3	None							
B-20		7/25/2019	18	-2.53	SP	Yes	1	0	None							
B-20		7/25/2019	19	-3.53	ОН	Yes	1	0	None							
B-20		7/25/2019	20	-4.53	ОН	Yes	1	0	None							
B-31		7/25/2019	1	14.6		No										
B-31		7/25/2019	2	13.6	SP	Yes	0	0	None							
B-31		7/25/2019	3	12.6	SP	Yes	0	0	None							
B-31		7/25/2019	4	11.6	SP	Yes	0	0	None							
B-31		7/25/2019	5	10.6	SP	Yes	0	0	None							
B-31 B-21		7/25/2019	7	9.6	SP SD	Yes	0	0	None							
B-31 B-31		7/25/2019	8	7.6	SP	Yes	0	0	None							
B-31		7/25/2019	9	6.6	SP	Yes	0	0	None							
B-31		7/25/2019	10	5.6	SP	Yes	1	98	Moderate							
B-31		7/25/2019	11	4.6		No	1									
B-31		7/25/2019	12	3.6	SP	Yes	1	274	Heavy	920	7200	360	8480	11%	85%	4%
B-31		7/25/2019	13	2.6	SP	Yes	1	46	Slight							
B-31		7/25/2019	14	1.6	SP	Yes	1	5	Slight							
B-31		7/25/2019	15	0.6	SP	Yes	1	1	Slight							
B-31		7/25/2019	16	-0.4	SP	Yes	1	23	None							
B-31		7/25/2019	1/	-1.4	SP	Yes	1	5	Slight							
B-31 B-21		7/25/2019	18	-2.4	5P	Yes	1	4	Nono							
B-31 B-31		7/25/2019	20	-4.4	ОН	Yes	1	0	None							
B-32		7/25/2019	1	14.63	011	No	-		Hone							
B-32		7/25/2019	2	13.63	SP	Yes	0	0	None							
B-32		7/25/2019	3	12.63	SP	Yes	0	0	None							
B-32		7/25/2019	4	11.63	SP	Yes	0	0	None							
B-32		7/25/2019	5	10.63	SP	Yes	0	0	None							
B-32		7/25/2019	6	9.63	SP	Yes	0	0	None							
B-32		7/25/2019	7	8.63	SP	Yes	0	0	None							
B-32		7/25/2019	8	7.03	5P	Yes	0	0	None							
B-32 B-37		7/25/2019	10	5.63	SP SP	Ves	1	2	Moderate							
B-32		7/25/2019	10	4.63	51	No	1	2	Woderate							
B-32		7/25/2019	12	3.63	SP	Yes	1	683	Free LNAPL	1100	11000	820	12920	9%	85%	6%
B-32		7/25/2019	13	2.63	SP	Yes	1	122	Free LNAPL							
B-32		7/25/2019	14	1.63	SP	Yes	1	5	Slight							
B-32		7/25/2019	15	0.63	SP	Yes	1	3	Slight							
B-32		7/25/2019	16	-0.37		No	1									
B-32		7/25/2019	17	-1.37	SP	Yes	1	231	Moderate							
B-32		7/25/2019	18	-2.37	SP	Yes	1	2	Slight							
B-32		7/25/2019	19	-3.3/		Yes	1	0	Siight							
B-21		7/25/2019	1	14.55	UII	No	1	U	None							
		.,,														



							Field Screening Result	S				Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	ТРН
B-21		7/25/2019	2	13.55	SP	Yes	0	0	None							
B-21		7/25/2019	3	12.55	SP	Yes	0	0	None							
B-21 B-21		7/25/2019	5	10.55	SP SP	Yes	0	0	None							
B-21		7/25/2019	6	9.55	SP	Yes	0	0	None							
B-21		7/25/2019	7	8.55	SP	Yes	0	0	None							
B-21		7/25/2019	8	7.55	SP	Yes	0	0	None							
B-21		7/25/2019	9	6.55	SP	Yes	0	0	None							
B-21		7/25/2019	10	5.55	SP	Yes	1	15	Moderate							
B-21		7/25/2019	11	4.55	6.0	No	1	76		570	10000	969	11 1 20	50/	070/	
B-21 B-21		7/25/2019	12	3.55	SP	Yes	1	76	Slight	570	10000	860	11430	5%	87%	8%
B-21 B-21		7/25/2019	13	1.55	SP	Yes	1	57	Slight							
B-21		7/25/2019	15	0.55		No	1									
B-21		7/25/2019	16	-0.45		No	1									
B-21		7/25/2019	17	-1.45	SP	Yes	1	119	Moderate							
B-21		7/25/2019	18	-2.45	SP	Yes	1	6	Moderate							
B-21		7/25/2019	19	-3.45	ОН	Yes	1	4	None							
B-21		7/25/2019	20	-4.45	ОН	Yes	1	1	None							
B-10 R-10		7/25/2019	1	14.41	SD	NO	0	0	Nono							
B-10 B-10		7/25/2019	3	12.41	SP	Yes	0	0	None							
B-10		7/25/2019	4	11.41	SP	Yes	0	0	None							
B-10		7/25/2019	5	10.41	SP	Yes	0	0	None							
B-10		7/25/2019	6	9.41	SP	Yes	0	0	None							
B-10		7/25/2019	7	8.41	SP	Yes	0	0	None							
B-10		7/25/2019	8	7.41	SP	Yes	0	0	None							
B-10		7/25/2019	9	6.41	SP	Yes	0	0	None							
B-10		7/25/2019	10	5.41	SP	Yes	1	/8	неаvy							
B-10 B-10		7/25/2019	12	3.41	SP	Yes	1	61	Moderate							
B-10		7/25/2019	13	2.41	SP	Yes	1	121	Moderate	150	6000	650	6800	2%	88%	10%
B-10		7/25/2019	14	1.41	SP	Yes	1	5	None							
B-10		7/25/2019	15	0.41	SP	Yes	1	2	None							
B-10		7/25/2019	16	-0.59	SP	Yes	1	0	None							
B-10		7/25/2019	17	-1.59	SP	Yes	1	0	None							
B-10		7/25/2019	18	-2.59	SP	Yes	1	0	None							
B-10 B-10		7/25/2019	20	-3.59	ОН	Yes	1	0	None							
B-22		7/25/2019	1	14.54	0	No	-									
B-22		7/26/2019	2	13.54		No										
B-22		7/25/2019	3	12.54	SP	Yes	0	0	None							
B-22		7/25/2019	4	11.54	SP	Yes	0	0	None							
B-22		7/25/2019	5	10.54	SP	Yes	0	0	None							
B-22		7/25/2019	6	9.54	SP	Yes	0	0	None							
B-22		7/25/2019	/	8.54	SP	Yes	0	0	None							
B-22 B-22		7/25/2019	9	6 54	SP	Yes	0	0	None							
B-22		7/25/2019	10	5.54	SP	Yes	1	51	Heavy							
B-22		7/25/2019	11	4.54		No	1									
B-22		7/26/2019	12	3.54		No	1	58								
B-22		7/25/2019	13	2.54	SP	Yes	1	203	Heavy							
B-22		7/25/2019	14	1.54	SP	Yes	1	378	Moderate	1300	4700	610	6610	20%	71%	9%
B-22		7/25/2019	15	0.54	SP	Yes	1	113	Slight							
B-22 B-22		7/25/2019	16	-0.46		No	1									
B-22		7/25/2019	18	-2.46	SP	Yes	1	35	Moderate							
		,, 23, 2023	1						1	I I	1	1	1			I I



							Field Screening Result	5				Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	ТРН
B-22		7/25/2019	19	-3.46	OL	Yes	1	5	Slight							
B-22		7/25/2019	20	-4.46	ОН	Yes	1	0	None							
Б-44 R-44		7/25/2019	2	14.49		No										
B-44		7/25/2019	3	12.49	SP	Yes	0	0	None							
B-44		7/25/2019	4	11.49	SP	Yes	0	0	None							
B-44		7/25/2019	5	10.49	SP	Yes	0	0	None							
B-44		7/25/2019	6	9.49	SP	Yes	0	0	None							
B-44		7/25/2019	7	8.49	SP	Yes	0	0	None							
B-44 B-44		7/25/2019	8	6.49	52	Yes	0	0	None							
B-44 B-44		7/25/2019	10	5.49	SP	Yes	1	45	Heavy							
B-44		7/25/2019	11	4.49	51	No	1	15	licary							
B-44		7/26/2019	12	3.49	SP	Yes	1	141	Moderate							
B-44		7/25/2019	13	2.49	SP	Yes	1	145	Heavy	380	9700	680	10760	4%	90%	6%
B-44		7/25/2019	14	1.49	SP	Yes	1	20	Slight							
B-44		7/25/2019	15	0.49	SP	Yes	1	17	Slight							
B-44		7/25/2019	16	-0.51	60	No	1	47	CI: L I							
B-44		7/25/2019	1/	-1.51	SP OH (OL	Yes	1	1/	Slight							
Б-44 В-44		7/25/2019	10	-2.51		Yes	1	0	None							
B-44		7/25/2019	20	-4.51	OH/OL	Yes	1	0	None							
B-11	MW-06	7/25/2019	1	14.56		No										
B-11	MW-06	7/25/2019	2	13.56		No										
B-11	MW-06	7/25/2019	3	12.56		No										
B-11	MW-06	7/25/2019	4	11.56	SP	Yes	0	0	None							
B-11	MW-06	7/25/2019	5	10.56	SP	Yes	0	0	None							
B-11 P-11	MW-06	7/25/2019	5	9.56	SP	Yes	0	0	None							
B-11 B-11	MW-06	7/25/2019	8	8.50 7.56	SP SP	Yes	0	0	None							
B-11	MW-06	7/25/2019	9	6.56	SP	Yes	0	0	None							
B-11	MW-06	7/25/2019	10	5.56	SP	Yes	1	72	Moderate							
B-11	MW-06	7/25/2019	11	4.56		No	1									
B-11	MW-06	7/26/2019	12	3.56	SP	Yes	1	97	Moderate							
B-11	MW-06	7/25/2019	13	2.56	SP	Yes	1	164	Heavy	400	7000	650	8050	5%	87%	8%
B-11	MW-06	7/25/2019	14	1.56	SP	Yes	1	15	Slight							
B-11 B-11	MW-06	7/25/2019	15	-0.44	٦r	No	1	27	None							
B-11	MW-06	7/25/2019	17	-1.44	SP	Yes	1	1	Slight							
B-11	MW-06	7/25/2019	18	-2.44	OH / OL	Yes	1	0	None							
B-11	MW-06	7/25/2019	19	-3.44	OH / OL	Yes	1	0	None							
B-11	MW-06	7/25/2019	20	-4.44	OH / OL	Yes	1	0	None							
B-09	MW-05	7/25/2019	1	14.47		No										
B-09	MW-05	7/25/2019	2	13.47		NO										
B-09 B-09	MW-05	7/25/2019	4	11.47		No										
B-09	MW-05	7/25/2019	5	10.47		No										
B-09	MW-05	7/25/2019	6	9.47		No										
B-09	MW-05	7/25/2019	7	8.47		No										
B-09	MW-05	7/25/2019	8	7.47	SP	Yes	0	0	None							
B-09	MW-05	7/25/2019	9	6.47	SP	Yes	0	0	None							
B-09	MW-05	7/25/2019	10	5.47	SP	Yes	0	0	None	<3	450	240	690	0%	65%	35%
B-09	MW-05	7/25/2019	11	4.47		NO										
B-09 B-09	MW-05	7/25/2019	12	2.47		No										
B-09	MW-05	7/25/2019	14	1.47		No										
B-09	MW-05	7/25/2019	15	0.47		No										



							Field Screening Result	S	1			Laborator	y Analytical Results			.
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	ТРН
B-09	MW-05	7/25/2019	16	-0.53		No										
B-09	MW-05	7/25/2019	17	-1.53		No										
B-09	MW-05	7/25/2019	18	-2.53		No										
B-09	MW-05	7/25/2019	19	-3.53		No										
B-09	MW-05	7/25/2019	20	-4.53		No										
B-33	MW-08	7/29/2019	1	11.3		No										
B-33	MW-08	7/29/2019	2	10.3		No										
B-33	MW-08	7/29/2019	3	9.3	SP	Yes	0	0	None							
B-33	MW-08	7/29/2019	4	8.3	SP	Yes	0	0	None							
B-33	MW-08	7/29/2019	5	7.3	SP	Yes	0	0	None							
B-33	MW-08	7/29/2019	6	6.3	SP	Yes	0	0	None							
B-33	MW-08	7/29/2019	7	5.3	SP	Yes	1	0	None							
B-33	MW-08	7/29/2019	8	4.3	SP	Yes	1	188	Moderate	4500	2000	262	2760			
B-33	MW-08	7/29/2019	9	3.3	SP	Yes	1	2042	Slight	1500	2000	260	3760	40%	53%	/%
B-33	IVIV-08	7/29/2019	10	2.3	58	Yes	1	5	Slight							
B-33	IVIV-08	7/29/2019	11	1.3	58	Yes	1	/88	Slight							
B-33	IVIV-08	7/29/2019	12	0.3	58	Yes	1	20	None							
B-33	NIW-08	7/29/2019	13	-0.7	SP OH (OI	Yes	1	2	None							
B-33	NIW-08	7/29/2019	14	-1./		Yes	1	2	None							
D-33	N/W-08	7/29/2019	15	-2.7	UH / SP	No	1	0	None							
B-35	MW-09	7/29/2019	2	9.78		No										
B-35	MW-09	7/29/2019	3	8 78	SP	Yes	0	0	None							
B-35	MW-09	7/29/2019	4	7 78	SP SP	Yes	0	0	None							
B-35	MW-09	7/29/2019	5	6.78	SP	Yes	0	0	None							
B-35	MW-09	7/29/2019	6	5.78	SP	Yes	0	0	None							
B-35	MW-09	7/29/2019	7	4.78	SP	Yes	0	0	Slight							
B-35	MW-09	7/29/2019	8	3.78	SP	Yes	1	28	Moderate							
B-35	MW-09	7/29/2019	9	2.78	SP	Yes	1	203	None	4.7	520	<54	524.7	1%	99%	0%
B-35	MW-09	7/29/2019	10	1.78	SP	Yes	1	82	None							
B-35	MW-09	7/29/2019	11	0.78	SP	Yes	1	0	None							
B-35	MW-09	7/29/2019	12	-0.22	SP / OH	Yes	1	3	None							
B-35	MW-09	7/29/2019	13	-1.22	SP	Yes	1	0	None							
B-35	MW-09	7/29/2019	14	-2.22	ОН	Yes	1	0	None							
B-35	MW-09	7/29/2019	15	-3.22	OH / SP	Yes	1	0	None							
B-38		7/29/2019	1	11.31		No										
B-38		7/29/2019	2	10.31		No										
B-38		7/29/2019	3	9.31	SP	Yes	0	0	None							
B-38		7/29/2019	4	8.31	SP	Yes	0	0	None							
B-38		7/29/2019	5	/.31	SP	Yes	U	U	None							
B-38		7/29/2019	5	6.31 E 21	52	Yes	U		None							
B-38		7/29/2019	/	5.31	52	Yes	0	259	Modorato	1200	5100	1400	7700	1.50/	650/	100/
B-38		7/29/2019	0	4.51	SP	Ves	1	1408	Slight	1200	3100	1400	//00	10%	00%	18%
B-38		7/29/2019	10	2 31	SP SP	Voc	1	1400	None							
B-38		7/29/2019	10	1 31	SP SP	Yes	1	480	Slight							
B-38		7/29/2019	12	0.31	SP/OH	Yes	1	13	None							
B-38		7/29/2019	13	-0.69	SP	Yes	1	4	None							
B-38		7/29/2019	14	-1.69	OH / SP	Yes	1	0	None							
B-38		7/29/2019	15	-2.69	OH/SP	Yes	1	0	None							
B-34		7/29/2019	1	10.97		No										
B-34		7/29/2019	2	9.97	SP	Yes	0	0	None							
B-34		7/29/2019	3	8.97	SP	Yes	0	0	None							
B-34		7/29/2019	4	7.97	SP	Yes	0	0	None							
B-34		7/29/2019	5	6.97	SP	Yes	0	0	None							
B-34		7/29/2019	6	5.97	SP	Yes	0	0	None							
B-34		7/29/2019	7	4.97	SP	Yes	0	0	None							



							Field Screening Result	5				Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	ТРН
B-34		7/29/2019	8	3.97	SP	Yes	0	0	Slight							
B-34		7/29/2019	9	2.97	SP	Yes	1	1600	Moderate	740	1400	590	2730	27%	51%	22%
B-34		7/29/2019	10	1.97	SP	Yes	1	2	None							
B-34		7/29/2019	11	0.97	SP	Yes	1	327	Moderate							
B-34		7/29/2019	12	-0.03	SP	Yes	1	29	None							
B-34		7/29/2019	13	-1.03	SP	Yes	1	2	None							
B-34		7/29/2019	14	-2.03	SP OUL/CD	Yes	1	1	None							
B-34		7/29/2019	15	-3.03	UH / SP	res	1	U	None							
B-39		7/29/2019	2	10.09	SD	NO	0	0	Nono							
B-39		7/29/2019	2	0.09	5P	Voc	0	0	None							
B-39		7/29/2019	3	8.09	SP	Vec	0	0	None							
B-39		7/29/2019	5	7.09	SP	Yes	0	0	None							
B-39		7/29/2019	6	6.09	SP	Yes	0	0	None							
B-39		7/29/2019	7	5.09	SP	Yes	0	0	None							
B-39		7/29/2019	8	4.09	SP	Yes	1	91	Slight							
B-39		7/29/2019	9	3.09	SP	Yes	1	941	Moderate	1000	3400	310	4710	21%	72%	7%
B-39		7/29/2019	10	2.09	SP	Yes	1	6	None							
B-39		7/29/2019	11	1.09	SP	Yes	1	303	Moderate							
B-39		7/29/2019	12	0.09	SP	Yes	1	18	Moderate							
B-39		7/29/2019	13	-0.91	SP	Yes	1	3	None							
B-39		7/29/2019	14	-1.91	OL/OH	Yes	1	1	None							
B-39		7/29/2019	15	-2.91	OL/OH	Yes	1	0	None							
B-40		7/29/2019	1	10.75		No										
B-40		7/29/2019	2	9.75	SP	Yes	0	0	None							
B-40		7/29/2019	3	8.75	SP	Yes	0	0	None							
B-40		7/29/2019	4	7.75	SP	Yes	0	0	None							
B-40		7/29/2019	5	6.75	SP	Yes	0	0	None							
B-40		7/29/2019	6	5.75		No										
B-40		7/29/2019	/	4.75	SP	Yes	1	214	Heavy	/90	4500	380	5670	14%	/9%	7%
B-40		7/29/2019	×	3.75	SP	Yes	1	153	Moderate							
B-40		7/29/2019	9	2.75	57	Yes	1	97	Niderate							
B-40 B-40		7/29/2019	10	1.75	5P	Voc	1	5	Slight							
B-40 B-40		7/29/2019	12	0.75	5P	Voc	1	12	Nono							
B-40		7/29/2019	12	-0.25	SP	Vec	1	2	None							
B-40		7/29/2019	14	-2.25		Yes	1	1	None							
B-40		7/29/2019	15	-3.25	OL / OH	Yes	1	0	None							
B-36		7/29/2019	1	10.66		No										
B-36		7/29/2019	2	9.66	SP	Yes	0	0	None							
B-36		7/29/2019	3	8.66	SP	Yes	0	0	None							
B-36		7/29/2019	4	7.66	SP	Yes	0	0	None							
B-36		7/29/2019	5	6.66	SP	Yes	0	0	None							
B-36		7/29/2019	6	5.66		No										
B-36		7/29/2019	7	4.66	SP	Yes	0	0	None							
B-36		7/29/2019	8	3.66	SP	Yes	1	132	Moderate	1200	4800	160	6160	19%	78%	3%
B-36		7/29/2019	9	2.66	SP	Yes	1	110	Slight							
B-36		//29/2019	10	1.66	SP	Yes	1	7	None							
B-36		7/29/2019	11	0.66	SP	Yes	1	4/	Moderate							
B-36		7/29/2019	12	-0.34	52	Yes	1	3	None							
B-30		7/29/2019	13	-1.34	58	res	1	1	None							
D-30 D-26		7/29/2019	14	-2.34		Voc	1	1	None							
B-30		7/30/2019	1	9.98	017.58	No	1	U	NUTE							
B-37		7/30/2019	2	8.98	SP	Yes	0	0	None							
B-37		7/30/2019	3	7.98	SP	Yes	ő	ő	None							
B-37		7/30/2019	4	6.98	SP	Yes	0	1	None							
-					-		-		1 1 1 2	1		1				



							Field Screening Result	5				Laborator	y Analytical Results			
				Sample												
			Sample	Elevation		Soil Recovered	Groundwater	PID Field	Sheen Test	Gasoline NTWPH-	Diesel NTWPH-	Oil	Total TPH Gasoline			
		Completion &	Depth (Feet	(NAVD 88	USCS Soil	from depth (Yes	Present	Screening	Field Screening	Gx Result	Dx Result	NTWPH-Dx Result	+ Diesel + Oil	% Gasoline of	% Diesel of	% Oil of Total
Boring ID #	Well ID #	Sample Date	BGS)	Datum)	Туре	or No)	(1 - yes, 0 - no)	Result (PPM)	Result	(mg/kg)	(mg/kg)	(mg/kg)	Result (mg/kg)	Total TPH	Total TPH	ТРН
B-37		7/30/2019	5	5.98	SP	Yes	0	185	Moderate							
B-37		7/30/2019	6	4.98	SP	Yes	1	9	Moderate	1000	2000	420	1020	270(con/	20/
B-37		7/30/2019	/	3.98	SP	Yes	1	/80	Heavy	1800	2900	120	4820	37%	60%	2%
B-37		7/30/2019	9	1 98	SP	Yes	1	19	Slight							
B-37		7/30/2019	10	0.98	SP	Yes	1	14	None							
B-37		7/30/2019	11	-0.02	SP	Yes	1	48	None							
B-37		7/30/2019	12	-1.02	SP	Yes	1	11	None							
B-37		7/30/2019	13	-2.02	SP	Yes	1	4	None							
B-37		7/30/2019	14	-3.02	SP	Yes	1	1	None							
B-37		7/30/2019	15	-4.02	OH / SP	Yes	1	0	None							
B-13		7/30/2019	1	11.34		NO										
B-13		7/30/2019	2	9 34	SP	Yes	0	0	None							
B-13		7/30/2019	4	8.34	SP	Yes	0	0	None							
B-13		7/30/2019	5	7.34	SP	Yes	0	0	None							
B-13		7/30/2019	6	6.34		No										
B-13		7/30/2019	7	5.34	SP	Yes	0	0	None							
B-13		7/30/2019	8	4.34	SP	Yes	1	0	Slight							
B-13		7/30/2019	9	3.34	SP	Yes	1	68	Moderate	160	10000	970	11130	1%	90%	9%
B-13		7/30/2019	10	2.34	SP	Yes	1	14	Moderate							
B-13		7/30/2019	12	0.34	SP SP	Ves	1	25	Moderate							
B-13		7/30/2019	13	-0.66	SP	Yes	1	15	Moderate							
B-13		7/30/2019	14	-1.66	OH /OL	Yes	1	0	Slight							
B-13		7/30/2019	15	-2.66	SP	Yes	1	0	None							
B-41	MW-10	7/31/2019	1	10.35		No										
B-41	MW-10	7/31/2019	2	9.35		No										
B-41	MW-10	7/31/2019	3	8.35	SP	Yes	0	0	None							
B-41	MW-10	7/31/2019	4	7.35	SP	Yes	0	0	None							
B-41 B-41	MW-10	7/31/2019	5	5 35	SP SP	Yes	0	87	None							
B-41	MW-10	7/31/2019	7	4.35	SP	Yes	1	195	Moderate							
B-41	MW-10	7/31/2019	8	3.35	SP	Yes	1	415	Moderate	2200	2000	<98	4200	52%	48%	0%
B-41	MW-10	7/31/2019	9	2.35	SP	Yes	1	5	Slight							
B-41	MW-10	7/31/2019	10	1.35	SP	Yes	1	3	None							
B-41	MW-10	7/31/2019	11	0.35	SP	Yes	1	126	Moderate							
B-41	MW-10	7/31/2019	12	-0.65	SP	Yes	1	68	Slight							
B-41	MW-10	7/31/2019	13	-1.65	OH / SP	Yes	1	1	None							
B-41 B-/11	MW-10	7/31/2019	14	-2.05		res Voc	1	0	Slight							
B-41 B-42	MW-11	7/31/2019	1	10.47	011/35	No	1	0	None							
B-42	MW-11	7/31/2019	2	9.47	SP	Yes	0	0	None							
B-42	MW-11	7/31/2019	3	8.47	SP	Yes	0	0	None							
B-42	MW-11	7/31/2019	4	7.47	SP	Yes	0	1	None							
B-42	MW-11	7/31/2019	5	6.47	SP	Yes	0	150	Moderate							
B-42	MW-11	7/31/2019	6	5.47	SP	Yes	1	26	Moderate							
B-42	MW-11	7/31/2019	7	4.47	SP	Yes	1	286	Heavy	790	4100	140	5030	16%	82%	3%
B-42	MW-11	7/31/2019	8	3.47	SP	Yes	1	0	Moderate							
B-42	IVIVV-11	7/31/2019	9	2.4/	SP (OL	Yes	1	2	Slight							
D-42 B-12	MW-11	7/31/2019	10	0.47	SP/UL	Vec	1	0	Slight							
B-42	MW-11	7/31/2019	12	-0.53	SP	Yes	1	0	Slight							
B-42	MW-11	7/31/2019	13	-1.53	SP	Yes	1	0	None							
B-42	MW-11	7/31/2019	14	-2.53	OH / SP	Yes	1	0	None							
B-42	MW-11	7/31/2019	15	-3.53	OH / SP	Yes	1	0	None							
B-43		7/31/2019	1	10.03		No										



							Field Screening Result	S				Laborator	y Analytical Results			
Boring ID #	Well ID #	Completion & Sample Date	Sample Depth (Feet BGS)	Sample Elevation (NAVD 88 Datum)	USCS Soil Type	Soil Recovered from depth (Yes or No)	Groundwater Present (1 - yes, 0 - no)	PID Field Screening Result (PPM)	Sheen Test Field Screening Result	Gasoline NTWPH- Gx Result (mg/kg)	Diesel NTWPH- Dx Result (mg/kg)	Oil NTWPH-Dx Result (mg/kg)	Total TPH Gasoline + Diesel + Oil Result (mg/kg)	% Gasoline of Total TPH	% Diesel of Total TPH	% Oil of Total TPH
B-43		7/31/2019	2	9.03		No										
B-43		7/31/2019	3	8.03	SP	Yes	0	1272	Moderate							
B-43		7/31/2019	4	7.03	SP	Yes	0	1159	Free LNAPL							
B-43		7/31/2019	5	6.03	SP	Yes	0	1408	Free LNAPL							
B-43		7/31/2019	6	5.03	SP	Yes	0	940	Free LNAPL							
B-43		7/31/2019	7	4.03	SP	Yes	0	1472	Free LNAPL	2000	14000	640	16640	12%	84%	4%
B-43		7/31/2019	8	3.03	SP	Yes	0	1508	Moderate							
B-43		7/31/2019	9	2.03	SP	Yes	1	453	Slight							
B-43		7/31/2019	10	1.03	SP	Yes	1	15	Slight							
B-43		7/31/2019	11	0.03	SP	Yes	1	751	Slight							
B-43		7/31/2019	12	-0.97	SP	Yes	1	16	None							
B-43		7/31/2019	13	-1.97	SP	Yes	1	18	Slight							
B-43		7/31/2019	14	-2.97	SP	Yes	1	11	None							
B-43		7/31/2019	15	-3.97	SP	Yes	1	5	None							
													Averages:	13%	78%	9%

Definitions:

BGS - Below ground surface

NAVD 88 - The North American Vertical Datum of 1988

PID - Photoionization detector

PPM - Parts per million

TPH - Total Petroleum Hydrocarbons

USCS - Unified Soil Classification System

Notes:

Cells highlighted in blue represent the sample interval with the highest soil impacts from each boring, and the soils submitted for laboratory analysis.

Cells highlighted in grey indicate no soil was recovered from that interval during probing.

Cells highlighted in orange indicate field screening result that were affected by soil sluffing in boring and likely do not represent actual conditions at listed depth.

Total TPH results in bold exceed the Plant 1 Total TPH Soil Cleanup Level of 10,000 mg/kg

USCS Soil Types:

OH - Organic clays of medium to high plasticity, organic silts.

OL - Organic silts and organic silty clays of low plasticity.

SM - Silty - sands, sand - silt mixtures.

SP - Poorly graded sands, gravelly sands, little or no fines

Date Tidal Stage Boorr Sheen Observations Boorr Sheen Observations Boorr Sheen Observations Boorr Sheen Observations 4/29/1996 high High Ocs Note Tide Rate (New/Not) Sheen (New/Not) S				Loading	Rack Area	Warehou	se Area North	Warehou	se Area South
Date Observations Observations Observations Observations Low, Medium (ebb & flood), (see Notes) Tide Rating (see Notes) Sheen Rating (see Notes)		Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Boom Sheen	
Date Took Medium (ebb & Riood), High Tode Rating (See Notes) Sheen Rating (See Notes) 5/21/1396 1 Ves 1.0 No 0.0 No 0.0 6/13/1396 1 Ves No 0.0 <td< th=""><th>_ .</th><th></th><th></th><th>Ohse</th><th>rvations</th><th>Obs</th><th>ervations</th><th>Ohse</th><th>ervations</th></td<>	_ .			Ohse	rvations	Obs	ervations	Ohse	ervations
Low Tide Rating (See Notes) Sheen (See Notes) Sheen (See Notes) Sheen (Yes/No) Sheen (See Notes) Sheen (Yes/No)	Date	Low Medium		0030		0.03		0.50	
High (See Notes) (Yes/No) (See Notes) (Yes/No) (See Notes) (Yes/No) (See Notes) 4/20/1396 low 0 Yes 2.0 (Yes/No) (See Notes)		(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
4/29/1996 high 2 (100, 10) </td <td></td> <td>High</td> <td>(See Notes)</td> <td>(Yes/No)</td> <td>(See Notes)</td> <td>(Ves/No)</td> <td>(See Notes)</td> <td>(Ves/No)</td> <td>(See Notes)</td>		High	(See Notes)	(Yes/No)	(See Notes)	(Ves/No)	(See Notes)	(Ves/No)	(See Notes)
H30/1996 Iow O Yes 1.0 4/30/1996 Iow 0 No 0.0 5/15/1996 No 0.0 No 0.0 5/22/1996 No No 0.0 0.0 5/22/1996 Ves 1.0 Yes 1.0 6/7/1996 ebb 1 Yes 0.0 6/13/1996 No 0.0 0.0 6/13/1996 No 0.0 0.0 6/13/1996 No 0.0 0.0 6/13/1996 No 0.0 0.0 8/16/1996 No No 0.0 8/16/1996 ebb 1 Yes 1.0 8/16/1996 ebb Yes 1.0 10/1/1996 low No 0.0 No 10/1/1/1996 low No 0.0 No 10/1/1/1996 low No 0.0 No 0.0 10/1/1/1996 low No 0	4/29/1996	high	2	(100,110)	(500 110103)	Yes	2.0	(100/110/	(500 110103)
Transmission Tes 2.0 $y13/1996$ flood 1 No 0.0 $5/20/1996$ No 0.0 1 $5/22/1996$ Ves 1.0 $5/22/1996$ Ves 1.0 $6/13/1996$ Ves 1.0 $6/13/1996$ No 0.0 $6/13/1996$ No 0.0 $6/13/1996$ medium No 0.0 $6/13/1996$ medium No 0.0 $6/13/1996$ medium No 0.0 $6/13/1996$ medium No 0.0 $8/16/1996$ ebb 1 No 0.0 $8/16/1996$ ebb 1 Yes 1.0 $10/3/1996$ low 0 Yes 1.0 $10/3/1996$ low 0 No 0.0 No $10/1/1996$ low No 0.0 No 0.0 $10/3/1996$ low No 0.0 No	4/30/1996	low	0			Yes	1.0		
7/15/190 Inv 0 No 0.0 5/15/1906 No 0.0 No 0.0 5/22/1906 Ves 1.0 Yes 1.0 5/24/1906 Ves 1.0 Yes 1.0 6/1/1906 Ves 0.0 6/16/19/1906 Yes 0.0 6/13/1906 No No 0.0 6/16/19/1906 Yes 0.0 6/13/1906 No No 0.0 0.0 7/30/1906 No 0.0 8/16/1906 ebb 1 No 0.0 1.0 Yes 1.0 8/16/1906 ebb 1 Yes 1.0 1.0 1.0 1.0 1.0 10/3/1906 ebb 1 Yes 1.0 1.	4/30/1996	flood	1			Yes	2.0		
Transmit	5/15/1996	low	0			No	0.0		
5/22/1996 ebb 1 Yes 1.0 5/22/1996 - Yes 1.0 6/17/1996 - Yes 1.0 6/17/1996 - Yes 0.0 6/13/1996 - No 0.0 6/13/1996 high 2 No 0.0 6/24/1996 medium 1 No 0.0 6/24/1996 medium 1 No 0.0 8/14/1996 ebb 1 No 0.0 8/14/1996 ebb 1 Yes 1.0 10/4/1996 ebb 1 Yes 1.0 10/4/1996 ebb 1 Yes 1.0 10/4/1996 low 0 No 0.0 No 10/1/1996 low 0 No 0.0 No 10/1/1/1996 low 0 No 0.0 No 10/1/1/1996 low 0 No 0.0 No <td>5/20/1996</td> <td>1011</td> <td>U U</td> <td></td> <td></td> <td>No</td> <td>0.0</td> <td></td> <td></td>	5/20/1996	1011	U U			No	0.0		
Transmit Tes 1.0 6/17/1996 ebb 1 Yes 1.0 6/13/1996 No 0.0 No 0.0 6/13/1996 high 2 No 0.0 6/13/1996 medium 1 No 0.0 6/24/1996 medium 1 No 0.0 7/30/1996 ebb 1 No 0.0 8/14/1996 medium 1 No 0.0 8/13/1996 ebb 1 Yes 1.0 8/13/1996 ebb 1 Yes 1.0 10/3/1996 low 0 Yes 1.0 10/1/1996 low 0 No 0.0 No 10/1/1996 low 0 No 0.0 No 0.0 10/1/1/1996 low 0 No 0.0 No 0.0 10/1/1/1996 low 0 No 0.0 No 0.0	5/22/1996	ebb	1			Yes	1.0		
$i^{1}_{1}_{1}_{1}_{2}_{2}_{3}_{6}$ $i^{1}_{1}_{2}_{2}_{3}_{6}$ $i^{1}_{2}_{2}_{3}_{6}$ $i^{1}_{1}_{1}_{1}_{2}_{3}_{3}_{6}$ $i^{1}_{2}_{2}_{3}_{6}$ $i^{1}_{2}_{2}_{3}_{6}$ $i^{1}_{1}_{1}_{2}_{3}_{3}_{6}$ $i^{1}_{1}_{1}_{1}_{2}_{1}_{3}_{3}_{6}$ $i^{1}_{1}_{1}_{1}_{2}_{1}_{1}_{2}_{1}_{1}_{1}_{1}_{1}_{1}_{1}_{1}_{1}_{1$	5/24/1996		-			Yes	1.0		
b) c) c) c) ves 0.5 6/13/1996 - No 0.0 6/13/1996 medium 1 No 0.0 6/14/1996 medium 1 No 0.0 7/30/1996 ebb 1 No 0.0 8/14/1996 medium 1 Yes 1.0 8/14/1996 ebb 1 Yes 1.0 8/14/1996 ebb 1 Yes 1.0 8/14/1996 low 0 Yes 1.0 8/14/1996 ebb 1 Yes 1.0 10/3/1996 low 0 No 0.0 No 10/1/1996 filood 1 No 0.0 No 0.0 10/3/1996 low 0 No 0.0 No 0.0 10/3/3/1996 low 0 No 0.0 No 0.0 10/3/3/1996 medium 1 No <	6/7/1996	ebb	1			Yes	1.0		
0/13/1996 Image of the second	6/10/1996		-			Yes	0.5		
013/13/13/13/13/13/13/13/13/13/13/13/13/1	6/13/1996					No	0.0		
0/2/1205 max No No $7/30/1996$ ebb 1 No 0.0 $8/16/1996$ ebb 1 No 0.0 $8/16/1996$ ebb 1 Yes 1.0 $8/19/1996$ ebb 1 Yes 1.0 $8/19/1996$ ebb 1 Yes 1.0 $10/3/1996$ low 0 Yes 1.0 $10/4/1996$ ebb 1 Yes 1.0 $10/7/1996$ flood 1 No 0.0 No $10/7/1996$ flood No 0.0 No 0.0 $10/2/3/1996$ low 0 No 0.0 No 0.0 $10/2/3/1996$ high 2 No 0.0 Yes 2.0 $11/4/1996$ medium 1 No 0.0 No 0.0 $11/2/1996$ low 0 No 0.0 No 0.0 $11/4/1996$ </td <td>6/19/1996</td> <td>high</td> <td>2</td> <td></td> <td></td> <td>No</td> <td>0.0</td> <td></td> <td></td>	6/19/1996	high	2			No	0.0		
0/29/1295 medium 1 No 0.0 8/14/1996 medium 1 No 0.0 8/16/1996 ebb 1 Yes 1.0 8/19/1996 ebb 1 Yes 1.0 8/19/1996 ebb 1 Yes 1.0 10/3/1996 ebb 1 Yes 1.0 10/4/1996 ebb 1 Yes 0.5 10/7/1996 flood 1 No 0.0 No 10/10/1996 low 0 No 0.0 No 0.0 10/2/3/1996 high 2 No 0.0 No 0.0 10/2/3/1996 high 2 No 0.0 No 0.0 10/2/3/1996 high 2 No 0.0 No 0.0 11/4/1996 medium 1 No 0.0 No 0.0 11/2/1996 iow 0 No 0.0 Yes	6/24/1996	medium	2			No	0.0		
N/3/12/36 $Coorder Mathematical Mathematindustripol Mathematical Mathematical Mathemat$	7/30/1996	ehh	1			No	0.0		
3/16/1996 ebb 1 Yes 1.0 $8/19/1996$ ebb 1 Yes 1.0 $8/29/1996$ ebb 1 Yes 1.0 $10/3/1996$ low 0 Yes 1.0 $10/3/1996$ low 0 Yes 1.0 $10/3/1996$ ebb 1 Yes 2.0 $10/10/1996$ low 0 No 0.0 No 0.0 $10/11/1996$ low 0 No 0.0 No 0.0 $10/25/1996$ high 2 No 0.0 No 0.0 $10/32/1996$ high 2 No 0.0 Yes 2.0 $11/1/1996$ medium 1 No 0.0 No 0.0 $11/5/1996$ high 2 No 0.0 Yes 2.0 $11/7/1996$ low 0 No 0.0 Yes 2.0 $11/12/1996$ low 0 No 0.0 Yes 2.0 $11/7/1996$ low<	8/14/1006	medium	1			No	0.0		
3/19/1996 ebb 1 Yes 1.0 $8/12/1996$ ebb 1 Yes 1.0 $10/3/1996$ ebb 1 Yes 1.0 $10/3/1996$ low 0 Yes 1.0 $10/3/1996$ flood 1 No 0.0 Yes 2.0 $10/1/1996$ flood 1 No 0.0 No 0.0 $10/1/1/1996$ low 0 No 0.0 No 0.0 $10/23/1996$ low 0 No 0.0 No 0.0 $10/23/1996$ low 0 No 0.0 No 0.0 $10/23/1996$ high 2 No 0.0 No 0.0 $10/30/1996$ high 2 No 0.0 No 0.0 $11/4/1996$ medium 1 No 0.0 No 0.0 $11/5/1996$ low 0 No 0.0 Yes 2.0 $11/1/1/1996$ low 0 No 0.0 Yes	8/16/1006	ehh	1			Ves	1.0		
0/29/1996 ebb 1 Yes 1.0 $10/3/1996$ low 0 Yes 1.0 $10/3/1996$ low 0 Yes 0.5 $10/7/1996$ flood 1 No 0.0 Yes 2.0 $10/10/1996$ low 0 No 0.0 No 0.0 $10/23/1996$ low 0 No 0.0 No 0.0 $10/23/1996$ low 0 No 0.0 No 0.0 $10/23/1996$ high 2 No 0.0 Yes 2.0 $11/1/1996$ medium 1 No 0.0 Yes 2.0 $11/1/1996$ medium 1 No 0.0 No 0.0 $11/5/1996$ now 0 No 0.0 Yes 2.0 $11/1/1996$ now 0 No 0.0 Yes 2.0 $11/1/1996$ now 0 No 0.0 Yes 1.0 $11/2/1996$ low 0 No	8/10/1990	ebb	1			Voc	1.0		
0/3/1996 low 0 Yes 1.0 $10/4/1996$ ebb 1 Yes 0.5 $10/7/1996$ flood 1 No 0.0 Yes 0.5 $10/10/10/1996$ flow 0 No 0.0 No 0.0 $10/11/1996$ low 0 No 0.0 No 0.0 $10/23/1996$ high 2 No 0.0 No 0.0 $10/23/1996$ high 2 No 0.0 No 0.0 $10/30/1996$ high 2 No 0.0 Yes 2.0 $11/4/1996$ medium 1 No 0.0 Yes 2.0 $11/5/1996$ medium 1 No 0.0 No 0.0 $11/5/1996$ low 0 No 0.0 Yes 0.5 $11/12/1996$ low 0 No 0.0 Yes 1.0 $11/12/1996$ high 2 No 0.0 Yes 1.0 $11/12/1996$ lo	8/19/1990	ebb	1			Voc	1.0		
10/3/1290 How 0 Hes 1.0 10/7/1996 flood 1 No 0.0 Yes 2.0 10/10/1996 low 0 No 0.0 No 0.0 10/11/1996 low 0 No 0.0 No 0.0 10/23/1996 low 0 No 0.0 No 0.0 10/25/1996 high 2 No 0.0 No 0.0 10/30/1996 high 2 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 Yes 2.0 11/1/1996 low 0 No 0.0 Yes 2.0 11/1/1996 low 0 No 0.0 Yes 1.0 11/1/1996 low 0 No 0.0 Yes 1.0 11/1/21/1996	0/29/1990 10/2/1006	low	1			Voc	1.0		
10//1/1996 fead 1 No No 0.0 Yes 2.0 10/10/1996 low 0 No 0.0 No 0.0 10/11/1996 low 0 No 0.0 No 0.0 10/23/1996 low 0 No 0.0 No 0.0 10/30/1996 high 2 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 No 0.0 11/4/1996 medium 1 No 0.0 No 0.0 11/5/1996 low 0 No 0.0 Yes 2.0 11/1/1996 low 0 No 0.0 Yes 2.0 11/1/1996 low 0 No 0.0 Yes 2.0 11/1/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 No 0.0	10/3/1990	iow	0			Vec	1.0		
10/10/1996 10w 0 No 0.0 No 0.0 10/10/1996 10w 0 No 0.0 No 0.0 10/23/1996 10w 0 No 0.0 No 0.0 10/25/1996 high 2 No 0.0 No 0.0 10/30/1996 high 2 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 No 0.0 11/5/1996 medium 1 No 0.0 No 0.0 11/5/1996 iow 0 No 0.0 Yes 2.0 11/1/1/1996 iow 0 No 0.0 Yes 1.0 11/1/1/1996 iow 0 No 0.0 Yes 1.0 11/20/1996 high 2 No 0.0 N	10/4/1996	flood	1	Ne	0.0	Yes	0.5		
10/10/1996 10W 0 No 0.0 No 0.0 10/11/1996 low 0 No 0.0 No 0.0 10/23/1996 high 2 No 0.0 No 0.0 10/25/1996 high 2 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 Yes 2.0 11/4/1996 medium 1 No 0.0 No 0.0 11/5/1996 needium 1 No 0.0 Yes 2.0 11/1/1996 low 0 No 0.0 Yes 2.0 11/7/1996 low 0 No 0.0 Yes 2.0 11/1/1996 low 0 No 0.0 Yes 1.0 11/1/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 Yes 1.0	10/7/1996	1000	1	NO	0.0	res	2.0		
11/1/1996 Iow 0 No 0.0 No 0.0 10/23/1996 high 2 No 0.0 No 0.0 10/30/1996 high 2 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 Yes 2.0 11/4/1996 medium 1 No 0.0 No 0.0 11/4/1996 medium 1 No 0.0 No 0.0 11/4/1996 medium 1 No 0.0 No 0.0 11/1/1996 Iow 0 No 0.0 Yes 2.0 11/1/1/1996 Iow 0 No 0.0 Yes 2.0 11/1/1/1996 Iow 0 No 0.0 Yes 1.0 11/1/1/1996 Iow 0 No 0.0 No 0.0 11/1/1/1/1996 Iow 0 No 0.0 No 0.0 11/1/1/1/1996 Iow 0 No 0.0 No 0.0 <td>10/10/1996</td> <td>IOW</td> <td>0</td> <td>NO</td> <td>0.0</td> <td>NO</td> <td>0.0</td> <td></td> <td></td>	10/10/1996	IOW	0	NO	0.0	NO	0.0		
LU/23/1996 How O No O.0 No O.0 10/25/1996 high 2 No O.0 No O.0 11/2/1996 medium 1 No O.0 Yes 2.0 11/4/1996 medium 1 No O.0 No O.0 11/5/1996 medium 1 No O.0 No O.0 11/5/1996 low 0 No O.0 Yes 2.0 11/7/1996 low 0 No O.0 Yes 0.5 11/1/1/1996 low 0 No O.0 Yes 0.5 11/1/1/1996 low 0 No O.0 Yes 1.0 11/1/1/1996 low 0 No O.0 Yes 1.0 11/21/1996 low 0 No O.0 Yes 1.0 11/21/1996 low 0 No O.0 No 0.0	10/11/1996	low	0	NO	0.0	NO	0.0		
10/25/12956 high 2 No 0.0 No 0.0 10/30/1996 high 2 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 Yes 2.0 11/4/1996 medium 1 No 0.0 No 0.0 11/5/1996 medium 0 No 0.0 Yes 2.0 11/7/1996 low 0 No 0.0 Yes 2.0 11/12/1996 low 0 No 0.0 Yes 2.0 11/12/1996 low 0 No 0.0 Yes 0.5 11/12/1996 low 0 No 0.0 No 0.0 11/12/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 Yes 1.0 11/21/1996 flood 1 No 0.0 No 0.0 <td>10/23/1996</td> <td>IOW</td> <td>0</td> <td>NO</td> <td>0.0</td> <td>NO No</td> <td>0.0</td> <td></td> <td></td>	10/23/1996	IOW	0	NO	0.0	NO No	0.0		
11/10/1996 nigh 2 No 0.0 Yes 2.0 11/1/1996 medium 1 No 0.0 No 0.0 11/4/1996 medium 1 No 0.0 No 0.0 11/4/1996 medium 1 No 0.0 No 0.0 11/4/1996 low 0 No 0.0 Yes 2.0 11/1/1/1996 low 0 No 0.0 Yes 2.0 11/1/1/1/1996 low 0 No 0.0 Yes 2.0 11/1/1/1/1/1/1/1/1/1/1/1/1/1/1/1/1/1/1	10/25/1996	nign	2	NO	0.0	NO	0.0		
11/1/1996 medium 1 No 0.0 Yes 2.0 11/4/1996 medium 1 No 0.0 No 0.0 11/5/1996 No 0.0 No 0.0 No 0.0 11/5/1996 Iow 0 No 0.0 Yes 2.0 11/7/1996 Iow 0 No 0.0 Yes 2.0 11/7/1996 Iow 0 No 0.0 Yes 2.0 11/7/1996 Iow 0 No 0.0 Yes 2.0 11/1/1/1996 Iow 0 No 0.0 Yes 1.0 11/1/1/1996 Iow 0 No 0.0 No 0.0 11/1/1/1996 Iow 0 No 0.0 Yes 1.0 11/20/1996 Iow 0 No 0.0 Yes 1.0 11/21/1996 Iood 1 No 0.0 No 0.0 12/6/1996 flood 1 No 0.0 No 0.0 <t< td=""><td>10/30/1996</td><td>nign</td><td>2</td><td>NO</td><td>0.0</td><td>Yes</td><td>2.0</td><td></td><td></td></t<>	10/30/1996	nign	2	NO	0.0	Yes	2.0		
11/4/1996 medium 1 No 0.0 No 0.0 11/5/1996 Iow 0 No 0.0 No 0.0 11/5/1996 Iow 0 No 0.0 Yes 2.0 11/7/1996 Iow 0 No 0.0 Yes 2.0 11/12/1996 No 0.0 Yes 2.0 11/14/1996 11/14/1996 No 0.0 Yes 0.5 11/14/1996 11/14/1996 No 0.0 No 0.0 10 11/14/1996 11/14/1996 No 0.0 No 0.0 No 0.0 11/14/1996 Iow 0 No 0.0 No 0.0 11/12/1996 Iow 0 No 0.0 Yes 1.0 11/20/1996 Iow 0 No 0.0 No 0.0 12/21/996 flood 1 No 0.0 No 0.0 12/12	11/1/1996	medium	1	NO	0.0	Yes	2.0		
11/5/1996 Iow 0 No 0.0 No 0.0 11/6/1996 Iow 0 No 0.0 Yes 2.0 11/7/1996 Iow 0 No 0.0 Yes 2.0 11/12/1996 No 0.0 Yes 2.0 11/12/1996 11/12/1996 No 0.0 Yes 0.0 11/14/1996 No 0.0 No 0.0 11/11/19/1996 high 2 No 0.0 No 0.0 11/12/1996 Iow 0 No 0.0 Yes 1.0 11/20/1996 Iow 0 No 0.0 Yes 1.0 11/2/1996 Iow 0 No 0.0 Yes 1.0 11/2/1996 Iood 1 No 0.0 No 0.0 12/6/1996 flood 1 Yes 0.5 No 0.0 12/11/1996 flood 1 No 0.0 No 0.0 12/12/1996 flood 1 No	11/4/1996	medium	1	No	0.0	NO	0.0		
11/6/1996 Iow 0 No 0.0 Yes 2.0 11/17/1996 Iow 0 No 0.0 Yes 2.0 11/12/1996 Iow 0 No 0.0 Yes 0.5 11/13/1996 Iow No 0.0 No 0.0 11/14/1996 No No 0.0 Yes 1.0 11/18/1996 high 2 No 0.0 No 0.0 11/14/1996 low 0 No 0.0 Yes 1.0 11/12/1996 low 0 No 0.0 Yes 1.0 11/12/1996 low 0 No 0.0 Yes 1.0 11/20/1996 low 0 No 0.0 No 0.0 12/16/1996 flood 1 No 0.0 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/12/1996	11/5/1996			No	0.0	NO	0.0		
11///1996 Iow 0 No 0.0 Yes 2.0 11/12/1996 I No 0.0 Yes 0.5 11/13/1996 No 0.0 No 0.0 11/14/1996 Nigh 2 No 0.0 Yes 1.0 11/18/1996 high 2 No 0.0 Yes 1.0 11/18/1996 low 0 No 0.0 Yes 1.0 11/19/1996 low 0 No 0.0 Yes 1.0 11/20/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 Yes 1.0 12/2/1996 flood 1 No 0.0 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 Yes 1.0 12/16/1996 flood	11/6/1996	low	0	No	0.0	Yes	2.0		
11/12/1996No 0.0 Yes 0.5 $11/13/1996$ No 0.0 No 0.0 $11/14/1996$ high2No 0.0 Yes 1.0 $11/19/1996$ low 0 No 0.0 Yes 1.0 $11/20/1996$ low 0 No 0.0 Yes 1.0 $11/21/1996$ flood 1 No 0.0 No 0.0 $12/10/1996$ flood 1 Yes 0.5 No 0.0 $12/11/1996$ flood 1 No 0.0 No 0.0 $12/12/1996$ flood 1 No 0.0 No 0.0 $12/16/1996$ flood 1 Yes 2.0 Yes 1.0 $12/17/1996$ flood 1 Yes 3.0 Yes 1.0 $12/18/1996$ flood 1 Yes 3.0 Yes 1.0 $1/2/1997$ high 2 Yes 1.0 No 0.0 $1/2/1997$ high 2 Yes 3.0 Yes 1.0 $1/9/1997$ Yes 3.0 Yes 1.0 No 0.0 $1/9/1997$ ebb 1 Yes 3.0 Yes 1.0	11/7/1996	low	0	No	0.0	Yes	2.0		
11/13/1996 Image: Second S	11/12/1996			No	0.0	Yes	0.5		
11/14/1996	11/13/1996			No	0.0	No	0.0		
11/18/1996 high 2 No 0.0 No 0.0 11/19/1996 low 0 No 0.0 Yes 1.0 11/20/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 Yes 1.0 12/6/1996 ebb 1 No 0.0 No 0.0 12/9/1996 medium 1 No 0.0 No 0.0 12/10/1996 flood 1 Yes 0.5 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/16/1996 flood 1 No 0.0 Yes 1.0 12/17/1996 flood 1 Yes 2.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 1.0 No 0.0 </td <td>11/14/1996</td> <td></td> <td>_</td> <td>No</td> <td>0.0</td> <td>Yes</td> <td>1.0</td> <td></td> <td></td>	11/14/1996		_	No	0.0	Yes	1.0		
11/19/1996 low 0 No 0.0 Yes 1.0 11/20/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 Yes 1.0 12/6/1996 ebb 1 No 0.0 No 0.0 12/9/1996 medium 1 No 0.0 No 0.0 12/10/1996 flood 1 Yes 0.5 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 3.0 No 0.0 <	11/18/1996	high	2	No	0.0	No	0.0		
11/20/1996 low 0 No 0.0 Yes 1.0 11/21/1996 low 0 No 0.0 Yes 1.0 12/6/1996 ebb 1 No 0.0 No 0.0 12/9/1996 medium 1 No 0.0 No 0.0 12/9/1996 medium 1 No 0.0 No 0.0 12/10/1996 flood 1 Yes 0.5 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 No 0.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 3.0 No 0.0	11/19/1996	low	0	No	0.0	Yes	1.0		
11/21/1996 low 0 No 0.0 Yes 1.0 12/6/1996 ebb 1 No 0.0 No 0.0 12/9/1996 medium 1 No 0.0 No 0.0 12/9/1996 medium 1 No 0.0 No 0.0 12/10/1996 flood 1 Yes 0.5 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 Yes 3.0 Yes 1.0 12/17/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 3.0 No 0.0 1/8/1997 high 2 Yes 3.0 Yes 1.0	11/20/1996	low	0	No	0.0	Yes	1.0		
12/6/1996 ebb 1 No 0.0 No 0.0 12/9/1996 medium 1 No 0.0 No 0.0 12/10/1996 flood 1 Yes 0.5 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 No 0.0 Yes 1.0 12/17/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 1.0 No 0.0 1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	11/21/1996	low	0	No	0.0	Yes	1.0		
12/9/1996 medium 1 No 0.0 No 0.0 12/10/1996 flood 1 Yes 0.5 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 No 0.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 1.0 No 0.0 1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 - Yes 3.0 Yes 1.0 Yes 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	12/6/1996	ebb	1	No	0.0	No	0.0		
12/10/1996 flood 1 Yes 0.5 No 0.0 12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 1.0 No 0.0 1/2/1997 high 2 Yes 3.0 No 0.0 1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 - Yes 3.0 Yes 1.0 Yes 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	12/9/1996	medium	1	No	0.0	No	0.0		
12/12/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 No 0.0 No 0.0 12/13/1996 flood 1 Yes 2.0 Yes 1.0 12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 No 0.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 1.0 No 0.0 1/2/1997 high 2 Yes 1.0 No 0.0 1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 - Yes 3.0 Yes 1.0 Yes 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	12/10/1996	flood	1	Yes	0.5	No	0.0		
12/13/1996 flood 1 No 0.0 No 0.0 12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 No 0.0 Yes 1.0 12/17/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 1.0 No 0.0 1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 - Yes 3.0 Yes 1.0 1/9/1997 - Yes 3.0 Yes 1.0	12/12/1996	flood	1	No	0.0	No	0.0		
12/16/1996 flood 1 Yes 2.0 Yes 1.0 12/17/1996 flood 1 No 0.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 1.0 No 0.0 1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 - Yes 3.0 Yes 1.0 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	12/13/1996	flood	1	No	0.0	No	0.0		
12/17/1996 flood 1 No 0.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 1.0 No 0.0 1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 res 3.0 Yes 1.0 No 0.0 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	12/16/1996	flood	1	Yes	2.0	Yes	1.0		
12/18/1996 flood 1 Yes 3.0 Yes 1.0 1/2/1997 high 2 Yes 1.0 No 0.0 1/8/1997 high 2 Yes 3.0 Yes 1.0 1/9/1997 high 2 Yes 3.0 No 0.0 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	12/17/1996	flood	1	No	0.0	Yes	1.0		
1/2/1997 high 2 Yes 1.0 No 0.0 1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 Yes 3.0 Yes 1.0 1/9/1997 Yes 3.0 Yes 1.0	12/18/1996	flood	1	Yes	3.0	Yes	1.0		
1/8/1997 high 2 Yes 3.0 No 0.0 1/9/1997 Yes 3.0 Yes 1.0 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	1/2/1997	high	2	Yes	1.0	No	0.0		
1/9/1997 Yes 3.0 Yes 1.0 1/9/1997 ebb 1 Yes 3.0 Yes 1.0	1/8/1997	high	2	Yes	3.0	No	0.0		
1/9/1997 ebb 1 Yes 3.0 Yes 1.0	1/9/1997		-	Yes	3.0	Yes	1.0		
	1/9/1997	ebb	1	Yes	3.0	Yes	1.0		

			Loading	Rack Area	Warehou	se Area North	Warehou	se Area South
	Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Boom Sheen	
- .			Ohse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium		0,50		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
1/9/1997	high	2	Ves	3.0	Ves	3.0	(103/110)	(See Notes)
1/1//1997	low	0	Ves	1.0	Voc	1.0		
1/15/1997	low	0	Ves	2.0	No	1.0		
1/16/1997	low	0	Ves	3.0	Ves	1.0		
1/17/1997	1000	0	Ves	1.5	No	1.0		
1/20/1997	low	0	Ves	3.0	No	0.0		
1/20/1997	high	2	Ves	2.0	Ves	1.0		
1/21/1997	high	2	Ves	2.0	Voc	1.0		
1/22/1997	flood	1	Ves	1.0	No	0.0		
1/22/1997	flood	1	Ves	1.0	No	0.0		
1/24/1997	flood	1	Ves	2.0	Ves	0.5		
1/27/1997	low	0	Ves	1.0	Ves	1.0		
1/27/1997	low	0	Ves	3.0	No	0.0		
1/28/1997	low	0	No	0.0	Ves	1.0		
1/28/1997	high	2	Ves	2.0	No	1.0		
1/30/1997	low	0	Ves	0.5	Ves	1.0		
1/31/1997	low	0	Ves	0.5	Voc	1.0		
2/3/1997	flood	1	Ves	1.0	Voc	0.5		
2/3/1997	flood	1	Ves	3.0	Voc	3.0		
2/4/1997	high	1	Vos	0.5	Voc	0.5		
2/6/1997	flood	2	Ves	2.0	Voc	0.5		
2/0/1007	flood	1	Ves	2.0	Voc	1.0		
2/10/1997	low	0	No	2.0	No	1.0		
2/10/1007	low	0	No	0.0	No	0.0		
2/12/1997	low	0	No	0.0	No	0.0		
2/12/1997		0	Ves	0.0	Ves	0.0		
2/14/1997	flood	1	Ves	0.5	No	0.0		
2/20/1997	ehh	1	Ves	2.0	Ves	2.0		
12/3/1997	high	2	No	2.0	No	2.0		
12/3/1337	ehh	1	No	0.0	No	0		
12/4/1337	600	1	NO	0.0		0		
1/11/2000	medium	1	Yes	1.0	No	0.0		
1/21/2000	high	2	No	0.0	No	0.0		
2/16/2000	medium	1	No	0.0	No	0.0		
2/22/2000	high	2	No	0.0	No	0.0		
2/23/2000	medium	1	No	0.0	No	0.0		
2/24/2000	low	0	No	0.0	No	0.0		
3/15/2000	medium	1	No	0.0	No	0.0		
3/16/2000	medium	1	No	0.0	No	0.0		
3/21/2000	low	0	Yes	1.0	No	0.0		
4/14/2000	medium	1	Yes	1.0	No	0.0		
6/15/2000	low	0	No	0.0	No	0.0		
6/28/2000	low	0	No	0.0	Yes	1.0		
6/29/2000	low	0	No	0.0	No	0.0		
7/11/2000	high	2	No	0.0	No	0.0		
7/19/2000	low	0	No	0.0	No	0.0		
8/15/2000	low	0	No	0.0	No	0.0		
10/12/2000	low	0	No	0.0	No	0.0		
11/14/2000	medium	1	No	0.0	No	0.0		
12/14/2000	high	2	No	0.0	No	0.0		

			Loading	Loading Rack Area		se Area North	Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Booi	m Sheen
Data		U	Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium		0.000		0.00		0.000	
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
		(0000000)		(000.0000)	(100)1107	(000110100)	(100)110)	(,
1/11/2001	medium	1	No	0.0	No	0.0		
2/15/2001	medium	1	No	0.0	No	0.0		
4/12/2001	medium	1	No	0.0	Yes	1.0		
4/13/2001	medium	1	No	0.0	No	0.0		
5/16/2001	low	0	No	0.0	No	0.0		
5/17/2001	low	0	No	0.0	No	0.0		
5/18/2001	low	0	No	0.0	No	0.0		
5/21/2001	low	0	No	0.0	No	0.0		
5/23/2001	low	0	No	0.0	No	0.0		
5/29/2001	low	0	No	0.0	No	0.0		
6/11/2001	medium	1	No	0.0	No	0.0		
7/23/2001	low	0	No	0.0	No	0.0		
8/21/2001	medium	1	No	0.0	No	0.0		
9/6/2001	high	2	No	0.0	No	0.0		
10/16/2001	low	0	No	0.0	No	0.0		
11/15/2001	medium	1	No	0.0	No	0.0		
12/10/2001	medium	1	No	0.0	No	0.0		
		-						
1/4/2002	high	2	No	0.0	No	0.0		
1/9/2002	medium	1	Yes	1.0	NO	0.0		
1/11/2002	medium	1	Yes	1.0	NO	0.0		
1/10/2002	nign	2	Yes	1.0	NO	0.0		
1/22/2002	low	1	Yes	1.0	No	0.0		
2/4/2002	high	2	No	1.0	No	0.0		
2/4/2002	medium	2	No	0.0	Ves	0.0		
2/21/2002	medium	1	Ves	2.0	No	0.0		
3/21/2002	medium	1	Ves	1.0	No	0.0		
3/25/2002	medium	1	No	0.0	No	0.0		
3/26/2002	medium	- 1	No	0.0	No	0.0		
3/27/2002	medium	- 1	Yes	2.0	No	0.0		
4/4/2002	high	2	No	0.0	No	0.0		
5/3/2002	low	0	No	0.0	No	0.0		
5/7/2002	medium	1	No	0.0	Yes	1.0		
5/21/2002	medium	1	Yes	1.0	Yes	1.0		
6/6/2002	medium	1	No	0.0	Yes	1.0		
6/18/2002	low	0	No	0.0	No	0.0		
6/27/2002	high	2	Yes	1.0	Yes	1.0		
7/10/2002	medium	1	Yes	1.0	Yes	1.0		
7/29/2002	medium	1	No	0.0	Yes	1.0		
8/21/2002	low	0	No	0.0	No	0.0		
9/9/2002	high	2	Yes	1.0	Yes	1.0		
9/20/2002	medium	1	No	0.0	Yes	1.0		
10/9/2002	high	2	No	0.0	No	0.0		
11/25/2002	high	2	No	0.0	No	0.0		
11/27/2002	high	2	No	0.0	No	0.0		
12/19/2002	medium	1	No	0.0	No	0.0		
12/20/2002	high	2	No	0.0	No	0.0		

			Loading	Rack Area	Warehou	se Area North	Warehous	se Area South
	Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Boom Sheen	
			Ohse	rvations	Ohs	ervations	Observations	
Date	Low Medium		0030		0.03		0030	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High		(Yes/No)		(Ves/No)		(Ves/No)	(See Notes)
1/16/2003	medium	1	No	0.0	No		(103/110/	(See Notes)
2/2/2003	medium	1	No	0.0	No	0.0		
2/3/2003	medium	1	No	0.0	No	0.0		
2/10/2003	low	1	No	0.0	No	0.0		
2/10/2003	modium	1	No	0.0	No	0.0		
2/11/2003	high	1	No	0.0	No	0.0		
2/11/2003	low	2	No	0.0	No	0.0		
2/11/2003	modium	1	No	0.0	No	0.0		
2/12/2003	high	1	NO	0.0	No	0.0		
2/13/2003	madium	2	NO	0.0	No	0.0		
2/15/2005	high	1	NO	0.0	No	0.0		
2/14/2003	high	2	No	0.0	No	0.0		
2/20/2003	modium	2	No	0.0	No	0.0		
2/20/2003	Ineurum	1	NO	0.0	No	0.0		
2/20/2003	IUW	0	No	0.0	No	0.0		
2/21/2003	nign	2	NO	0.0	NO	0.0		
2/21/2003	medium	1	NO	0.0	NO No	0.0		
3/3/2003	medium	1	NO	0.0	NO No	0.0		
3/10/2003	meaium	1	NO	0.0	NO	0.0		
3/11/2003	nign	2	NO	0.0	NO	0.0		
3/18/2003	meaium	1	NO	0.0	NO	0.0		
4/1/2003	low	0	No	0.0	NO	0.0		
4/8/2003	high	2	Yes	2.0	No	0.0		
4/15/2003	low	0	Yes	2.0	No	0.0		
4/21/2003	high	2	No	0.0	No	0.0		
5/15/2003	low	0	No	0.0	No	0.0		
5/20/2003	medium	1	No	0.0	No	0.0		
5/21/2003	medium	1	No	0.0	No	0.0		
5/27/2003	low	0	No	0.0	No	0.0		
6/3/2003	medium	1	No	0.0	No	0.0		
6/17/2003	medium	1	No	0.0	No	0.0		
7/15/2003	medium	1	No	0.0	No	0.0		
7/21/2003	low	0	No	0.0	No	0.0		
8/7/2003	low	0	No	0.0	No	0.0		
8/13/2003	medium	1	No	0.0	No	0.0		
9/15/2003	high	2	No	0.0	No	0.0		
9/16/2003	high	2	No	0.0	No	0.0		
9/17/2003	medium	1	No	0.0	No	0.0		
9/19/2003	medium	1	No	0.0	No	0.0		
10/9/2003	medium	1	No	0.0	Yes	1.0		
10/14/2003	high	2	No	0.0	No	0.0		
11/12/2003	high	2	No	0.0	No	0.0		
11/19/2003	high	2	No	0.0	No	0.0		
12/17/2003	medium	1	No	0.0	No	0.0		
12/23/2003	medium	1	No	0.0	No	0.0		
1/13/2004	medium	1	Yes	1.0	No	0.0		
1/24/2004	high	2	No	0.0	No	0.0		
2/10/2004	medium	1	Yes	1.0	No	0.0		
2/23/2004	medium	1	No	0.0	Yes	1.0		
3/17/2004	medium	1	No	0.0	No	0.0		

Date Tidal Stage Bosw Sheen Obswwains Sheen Sheen Rating 3/15/2004 medium 1 No 0.0 No No No				Loading Rack Area V		Warehouse Area North		Warehouse Area South	
DateObservationsObservationsObservationsObservationsObservationsIcov, Medium (seb & flood)Tide Rating (see Notes)Sheen (see Notes) <t< td=""><td></td><td>Tidal</td><td>Stage</td><td>Boor</td><td>n Sheen</td><td>Воо</td><td>m Sheen</td><td colspan="2">Boom Sheen</td></t<>		Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Boom Sheen	
Date Conv. Medium (ebb. 8 flood), High Tide Rating (see Notes) Sheen Sheen Rating (Yes/No) Sheen (See Notes) Sheen (Yes/No) Sheen (See Notes) 3/15/2004 medium 1 No 0.0 No 0.0 4/15/2004 medium 1 No 0.0 No 0.0 4/15/2004 medium 1 No 0.0 No 0.0 5/25/2004 medium 1 No 0.0 No 0.0 5/25/2004 medium 1 No 0.0 No 0.0 6/14/2004 medium 1 No 0.0 No 0.0 6/23/2004 medium 1 No 0.0 No 0.0 6/33/2004 medium 1 No 0.0 No 0.0 6/32/2004 medium 1 No 0.0 No 0.0 7/13/2004 hugh 2 No 0.0 No 0.0 8/12/2004 hugh <td>- .</td> <td></td> <td></td> <td>Ohse</td> <td>rvations</td> <td>Ohs</td> <td>ervations</td> <td>Ohse</td> <td>ervations</td>	- .			Ohse	rvations	Ohs	ervations	Ohse	ervations
Network Tide Rating (see Notes) Sheen (see Notes) 5/23/2004 medium 1 No 0.0 No 0.0 No 0.0	Date	Low Medium		0,50		0.03		0.50	
High (See Notes) (Yes/No) (See Notes) (Yes/No) (See Notes) (Yes/No) (See Notes) 3/19/2004 medium 1 No 0.0 No 0.0 4/15/2004 medium 1 No 0.0 No 0.0 4/15/2004 medium 1 No 0.0 No 0.0 5/24/2004 medium 1 No 0.0 No 0.0 6/13/2004 medium 1 No 0.0 No 0.0 6/13/2004 medium 1 No 0.0 No 0.0 6/23/2004 high 2 No 0.0 No 0.0 6/23/2004 medium 1 No 0.0 No 0.0 6/3/2004 medium 1 No 0.0 No 0.0 7/11/2004 low 0 No 0.0 No 0.0 8/11/2004 high 2 No		(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
3/19/2004 medium 1 No 0.0 No 0.0 4/15/2004 medium 1 No 0.0 Yes 1.0 4/15/2004 medium 1 No 0.0 No 0.0 4/15/2004 medium 1 No 0.0 No 0.0 5/24/2004 medium 1 No 0.0 No 0.0 5/24/2004 medium 1 No 0.0 No 0.0 6/15/2004 medium 1 No 0.0 No 0.0 6/23/2004 medium 1 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 8/12/2004 medium 1 No 0.0 No 0.0 8/12/2004 medium 1 No 0.0 No 0.0		High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
312/200- 4/15/2004 medium 1 No 0.0 Yes 0.0 4/15/2004 medium 1 No 0.0 No 0.0 5/24/2004 medium 1 No 0.0 No 0.0 5/25/2004 medium 1 No 0.0 No 0.0 6/15/2004 medium 1 No 0.0 No 0.0 6/22/2004 low 0 No 0.0 No 0.0 6/22/2004 medium 1 No 0.0 No 0.0 6/22/2004 medium 1 No 0.0 No 0.0 6/22/2004 medium 1 No 0.0 No 0.0 7/12/2004 high 2 No 0.0 No 0.0 8/12/2004 high 2 No 0.0 No 0.0 9/2/2004 high 2 No 0.0 No 0.0	3/19/2004	medium	1	No	0.0	No		(103/110)	(See Notes)
H H NO D.0 NO D.0 4/19/2004 medium 1 NO 0.0 NO 0.0 5/24/2004 medium 1 NO 0.0 NO 0.0 5/24/2004 medium 1 NO 0.0 NO 0.0 6/14/2004 medium 1 NO 0.0 NO 0.0 6/15/2004 low 0 NO 0.0 NO 0.0 6/28/2004 low 0 NO 0.0 NO 0.0 6/28/2004 medium 1 NO 0.0 NO 0.0 6/30/2004 medium 1 NO 0.0 NO 0.0 7/13/2004 low 0 NO 0.0 NO 0.0 8/12/2004 medium 1 NO 0.0 NO 0.0 9/17/2004 medium 1 NO 0.0 NO 0.0 9/17/2004 <td>3/15/2004 1/15/2004</td> <td>medium</td> <td>1</td> <td>No</td> <td>0.0</td> <td>Ves</td> <td>1.0</td> <td></td> <td></td>	3/15/2004 1/15/2004	medium	1	No	0.0	Ves	1.0		
High Ho Do Do Do Do 4722/2004 medium 1 No 0.0 No 0.0 5/24/2004 medium 1 No 0.0 No 0.0 6/14/2004 medium 1 No 0.0 No 0.0 6/14/2004 medium 1 No 0.0 No 0.0 6/13/2004 ligh 2 No 0.0 No 0.0 6/28/2004 low 0 No 0.0 No 0.0 6/28/2004 medium 1 No 0.0 No 0.0 6/28/2004 medium 1 No 0.0 No 0.0 8/11/2004 low 0 No 0.0 No 0.0 8/12/2004 medium 1 No 0.0 No 0.0 9/12/2004 medium 1 No 0.0 No 0.0 9/12/2004	4/19/2004	medium	1	No	0.0	No	0.0		
Tight Tight Tight Tight Tight Tight 5/24/2004 medium 1 No 0.0 No 0.0 5/25/2004 medium 1 No 0.0 No 0.0 6/14/2004 low 0 No 0.0 No 0.0 6/28/2004 low 0 No 0.0 No 0.0 6/28/2004 low 0 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 8/11/2004 low 0 No 0.0 No 0.0 8/11/2004 high 2 No 0.0 No 0.0 8/2/2/2004 high 2 No 0.0 No 0.0 9/16/2004 high 2 No 0.0 No 0.0 9/	4/22/2004	medium	1	No	0.0	No	0.0		
b) b) b) b) b) b) b) b) b) b) b) b) b) b) b)	5/24/2004	medium	1	No	0.0	No	0.0		
3/2/2004 Inclum 1 No 0.0 No 0.0 6/14/2004 Iew 0 No 0.0 No 0.0 6/23/2004 Iew 0 No 0.0 No 0.0 6/23/2004 Iew 0 No 0.0 No 0.0 6/23/2004 redium 1 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 7/13/2004 low 0 No 0.0 No 0.0 8/11/2004 high 2 No 0.0 No 0.0 9/3/2004 high 2 No 0.0 No 0.0 9/16/2004 high 2 No 0.0 No 0.0 9/16/2004 medium 1 No 0.0 No 0.0	5/25/2004	medium	1	No	0.0	No	0.0		
displace Inc. Los No D.0 No D.0 6/13/2004 high 2 No 0.0 No 0.0 6/23/2004 medium 1 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 7/12/2004 low 0 No 0.0 No 0.0 7/12/2004 low 0 No 0.0 No 0.0 8/12/2004 low 0 No 0.0 No 0.0 8/24/2004 medium 1 No 0.0 No 0.0 9/10/2004 high 2 No 0.0 No 0.0 9/11/2004 medium 1 No 0.0 No 0.0 9/11/2004 medium 1 No 0.0 No 0.0 9/	6/14/2004	medium	1	No	0.0	No	0.0		
6/23/2004 high 2 No 0.0 No 0.0 6/28/2004 low 0 No 0.0 No 0.0 6/28/2004 medium 1 No 0.0 No 0.0 6/28/2004 medium 1 No 0.0 No 0.0 6/28/2004 low 0 No 0.0 No 0.0 7/13/2004 low 0 No 0.0 No 0.0 8/12/2004 medium 1 No 0.0 No 0.0 8/24/2004 medium 1 No 0.0 No 0.0 9/2/2004 high 2 No 0.0 No 0.0 9/3/2/204 medium 1 No 0.0 No 0.0 9/3/2/204 medium 1 No 0.0 No 0.0 9/3/2/204 medium 1 No 0.0 No 0.0 10/3/3/204 medium 1 No 0.0 No 0.0 <t< td=""><td>6/15/2004</td><td>low</td><td>0</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	6/15/2004	low	0	No	0.0	No	0.0		
digst n.g. z No 0.0 No 0.0 6/28/2004 medium 1 No 0.0 No 0.0 6/30/2004 medium 1 No 0.0 No 0.0 7/13/2004 low 0 No 0.0 No 0.0 7/13/2004 low 0 No 0.0 No 0.0 8/11/2004 high 2 No 0.0 No 0.0 8/2/2004 medium 1 No 0.0 No 0.0 9/2/2004 migh 2 No 0.0 No 0.0 9/1/2004 medium 1 No 0.0 No 0.0	6/23/2004	high	2	No	0.0	No	0.0		
bit bit bit bit bit bit 6/39/2004 medium 1 No 0.0 No 0.0 6/39/2004 iow 0 No 0.0 No 0.0 7/12/2004 iow 0 No 0.0 No 0.0 8/11/2004 high 2 No 0.0 No 0.0 8/12/2004 ingh 2 No 0.0 No 0.0 8/12/2004 high 2 No 0.0 No 0.0 9/12/2004 high 2 No 0.0 No 0.0 9/12/2004 high 2 No 0.0 No 0.0 9/12/2004 medium 1 No 0.0 No 0.0 9/12/2004 medium 1 No 0.0 No 0.0 10/15/2004 medium 1 No 0.0 No 0.0 10/15/2004 </td <td>6/28/2004</td> <td>low</td> <td>0</td> <td>No</td> <td>0.0</td> <td>No</td> <td>0.0</td> <td></td> <td></td>	6/28/2004	low	0	No	0.0	No	0.0		
0/10/12/2004 medium 1 No 0.0 No 0.0 7/12/2004 low 0 No 0.0 No 0.0 7/13/2004 low 0 No 0.0 No 0.0 8/11/2004 high 2 No 0.0 No 0.0 8/12/2004 low 0 No 0.0 No 0.0 8/12/2004 high 2 No 0.0 No 0.0 9/1/2004 high 2 No 0.0 No 0.0 9/1/2004 high 2 No 0.0 No 0.0 9/1/2004 medium 1 No 0.0 No 0.0 9/1/2004 medium 1 No 0.0 No 0.0 9/1/2004 medium 1 No 0.0 No 0.0 10/15/2004 medium 1 No 0.0 No 0.0	6/29/2004	medium	1	No	0.0	No	0.0		
0 j 0 120-7 medium 1 No 0.0 No 0.0 7/12/2004 low 0 No 0.0 No 0.0 8/11/2004 high 2 No 0.0 No 0.0 8/12/2004 high 2 No 0.0 No 0.0 8/24/2004 medium 1 No 0.0 No 0.0 9/2/2004 high 2 No 0.0 No 0.0 9/1/2004 medium 1 No 0.0 No 0.0 9/1/2004 medium 1 No 0.0 No 0.0 9/1/2/2044 medium 1 No 0.0 No 0.0 9/2/2/2044 medium 1 No 0.0 No 0.0 9/2/2/2044 medium 1 No 0.0 No 0.0 10/15/2004 medium 1 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0	6/30/2004	medium	1	No	0.0	No	0.0		
7/13/2004 Iow 0 No 0.0 No 0.0 8/11/2004 high 2 No 0.0 No 0.0 8/24/2004 medium 1 No 0.0 No 0.0 8/24/2004 high 2 No 0.0 No 0.0 9/2/2004 high 2 No 0.0 No 0.0 9/3/2004 high 2 No 0.0 No 0.0 9/10/2004 low 0 No 0.0 No 0.0 9/16/2004 high 2 No 0.0 No 0.0 9/16/2004 medium 1 No 0.0 No 0.0 9/21/2004 medium 1 No 0.0 No 0.0 10/15/2004 medium 1 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0	7/12/2004	low	0	No	0.0	No	0.0		
A/12/2004 high 2 No 0.0 No 0.0 8/11/2004 low 0 No 0.0 No 0.0 8/24/2004 low 0 No 0.0 No 0.0 8/24/2004 high 2 No 0.0 No 0.0 9/3/2004 high 2 No 0.0 No 0.0 9/3/2004 high 2 No 0.0 No 0.0 9/10/2004 low 0 No 0.0 No 0.0 9/12/2004 medium 1 No 0.0 No 0.0 9/21/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 10/13/2004 medium 1 No 0.0 No 0.0 10/13/2004 high 2 No 0.0 No 0.0 11/42/2004 medium 1 No 0.0 No 0.0	7/13/2004	low	0	No	0.0	No	0.0		
b) 1,12004 Ing. 1 No 0.0 No 0.0 8/24/2004 medium 1 No 0.0 No 0.0 9/2/2004 high 2 No 0.0 No 0.0 9/3/2004 high 2 No 0.0 No 0.0 9/16/2004 medium 1 No 0.0 No 0.0 9/16/2004 high 2 No 0.0 No 0.0 9/16/2004 medium 1 No 0.0 No 0.0 9/21/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 10/15/2004 medium 1 No 0.0 No 0.0 10/15/2004 medium 1 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 11/12/2004 high 2 No 0.0 No 0.0	8/11/2004	high	2	No	0.0	No	0.0		
b) 21/2004 Noi 0.0 No 0.0 g/2/2004 high 2 No 0.0 No 0.0 g/3/2004 high 2 No 0.0 No 0.0 g/3/2004 high 2 No 0.0 No 0.0 g/10/2004 high 2 No 0.0 No 0.0 g/11/2004 medium 1 No 0.0 No 0.0 g/11/2004 medium 1 No 0.0 No 0.0 g/21/2004 medium 1 No 0.0 No 0.0 g/21/2004 medium 1 No 0.0 No 0.0 g/21/2004 medium 1 No 0.0 No 0.0 10/15/2004 medium 1 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 11/14/2004 <	8/12/2004	low	0	No	0.0	No	0.0		
b) 4/2004 high 2 No 0.0 No 0.0 9/3/2004 high 2 No 0.0 No 0.0 9/3/2004 high 2 No 0.0 No 0.0 9/10/2004 low 0 No 0.0 No 0.0 9/16/2004 high 2 No 0.0 No 0.0 9/12/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 10/15/2004 medium 1 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 11/4/2004 high 2 No 0.0 No 0.0 11/12/2004 high 2 No 0.0 No 0.0 11/12/2004 medium 1 No 0.0 No 0.0 11/12/2004 medium 1 No 0.0 No 0.0 12/12/2004	8/24/2004	medium	1	No	0.0	No	0.0		
J.J.2004 high 2 No 0.0 No 0.0 9/3/2004 high 2 No 0.0 No 0.0 9/10/2004 low 0 No 0.0 No 0.0 9/11/2004 high 2 No 0.0 No 0.0 9/21/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 10/13/2004 medium 1 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 11/8/2004 medium 1 No 0.0 No 0.0 11/23/2004 high 2 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 <	9/2/2004	high	2	No	0.0	No	0.0		
J. J. 2004 Ing.n 1 No 0.0 No 0.0 9/1/2/2004 low 0 No 0.0 No 0.0 9/10/2004 high 2 No 0.0 No 0.0 9/11/2/2004 high 2 No 0.0 No 0.0 9/21/2/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 10/5/2004 medium 1 No 0.0 No 0.0 10/13/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 11/12/2004 high 2 No 0.0 No 0.0 11/12/2004 high 2 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 <td>9/3/2004</td> <td>high</td> <td>2</td> <td>No</td> <td>0.0</td> <td>No</td> <td>0.0</td> <td></td> <td></td>	9/3/2004	high	2	No	0.0	No	0.0		
J/1/2004 Inextan 1 No 0.0 No 0.0 9/16/2004 high 2 No 0.0 No 0.0 9/16/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 10/5/2004 medium 1 No 0.0 No 0.0 10/13/2004 high 2 No 0.0 No 0.0 10/18/2004 high 2 No 0.0 No 0.0 11/14/2004 medium 1 No 0.0 No 0.0 12/23/2004 low 0 No 0.0 No 0.0 1/1/2/205 high 2 No 0.0 No 0.0	9/3/2004	medium	2	No	0.0	No	0.0		
3/10/204 high 2 No 0.0 No 0.0 9/11/2004 medium 1 No 0.0 No 0.0 9/21/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 9/23/2004 medium 1 No 0.0 No 0.0 10/5/2004 medium 1 No 0.0 No 0.0 10/13/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 11/12/2004 medium 1 No 0.0 No 0.0 11/23/204 hedium 1 No 0.0 No 0.0 12/21/2004 high 2 No 0.0 No 0.0	9/10/2004	low	0	No	0.0	No	0.0		
9/21/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 9/22/2004 medium 1 No 0.0 No 0.0 10/5/2004 medium 1 No 0.0 No 0.0 10/13/2004 medium 1 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 11/4/2004 medium 1 No 0.0 No 0.0 11/2/2004 medium 1 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 1/14/2005 high 2 No 0.0 No 0.0 <td>9/16/2004</td> <td>high</td> <td>2</td> <td>No</td> <td>0.0</td> <td>No</td> <td>0.0</td> <td></td> <td></td>	9/16/2004	high	2	No	0.0	No	0.0		
3/22/2004 medium 1 No 0.0 No 0.0 9/23/2004 medium 1 No 0.0 No 0.0 10/5/2004 medium 1 No 0.0 No 0.0 10/13/2004 medium 1 No 0.0 Yes 1.0 10/13/2004 medium 1 No 0.0 Yes 1.0 10/15/2004 high 2 No 0.0 No 0.0 10/18/2004 high 2 No 0.0 No 0.0 10/25/2004 low 0 No 0.0 No 0.0 11/12/2004 medium 1 No 0.0 No 0.0 11/23/2004 medium 1 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 1/1/3/2005 high 2 No 0.0 No 0.0	9/21/2004	medium	1	No	0.0	No	0.0		
J22/2004 medium 1 No 0.0 No 0.0 J0/5/2004 medium 1 No 0.0 No 0.0 10/13/2004 medium 1 No 0.0 No 0.0 10/13/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 11/12/2004 medium 1 No 0.0 No 0.0 11/12/2004 medium 1 No 0.0 No 0.0 11/12/2004 medium 1 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 1/14/2005 high 2 No 0.0 No 0.0 <td>9/22/2004</td> <td>medium</td> <td>1</td> <td>No</td> <td>0.0</td> <td>No</td> <td>0.0</td> <td></td> <td></td>	9/22/2004	medium	1	No	0.0	No	0.0		
J2/2004 Intertaint 1 No 0.0 No 0.0 10/5/2004 medium 1 No 0.0 No 0.0 10/13/2004 medium 1 No 0.0 No 0.0 10/13/2004 high 2 No 0.0 No 0.0 10/13/2004 high 2 No 0.0 No 0.0 10/125/2004 low 0 No 0.0 No 0.0 11/4/2004 medium 1 No 0.0 No 0.0 11/13/2004 high 2 No 0.0 No 0.0 11/13/2004 medium 1 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 11/13/2005 high 2 No 0.0 No 0.0 1/13/2005 high 2 Yes 1.0 No 0.0 1/13/2005 high 2 Yes 1.0 No 0.0 <	9/22/2004	medium	1	No	0.0	No	0.0		
10/13/2004 medium 1 No 0.0 Yes 1.0 10/13/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/25/2004 low 0 No 0.0 No 0.0 11/2/2004 medium 1 No 0.0 No 0.0 11/12/2004 high 2 No 0.0 No 0.0 11/23/2004 low 0 No 0.0 No 0.0 12/3/2004 high 2 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 1/4/2005 high 2 No 0.0 No 0.0 1/1/3/2005 high 2 Yes 1.0 No 0.0 1/1/2/2005 high 2 Yes 1.0 No 0.0 <	10/5/2004	medium	1	No	0.0	No	0.0		
10/15/2004 high 1 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/15/2004 high 2 No 0.0 No 0.0 10/25/2004 low 0 No 0.0 No 0.0 11/4/2004 medium 1 No 0.0 No 0.0 11/23/2004 high 2 No 0.0 No 0.0 11/23/2004 high 2 No 0.0 No 0.0 12/3/2004 high 2 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 12/3/2004 medium 1 No 0.0 No 0.0 1/12/3/2004 high 2 No 0.0 No 0.0 1/13/2005 high 2 No 0.0 No 0.0 1/12/2005 high 2 Yes 1.0 No 0.0 <t< td=""><td>10/13/2004</td><td>medium</td><td>1</td><td>No</td><td>0.0</td><td>Ves</td><td>1.0</td><td></td><td></td></t<>	10/13/2004	medium	1	No	0.0	Ves	1.0		
10/13/2004high2No0.0No0.010/18/2004high2No0.0No0.011/25/2004low0No0.0No0.011/4/2004medium1No0.0No0.011/18/2004high2No0.0No0.011/23/2004medium1No0.0No0.012/3/2004low0No0.0No0.012/15/2004high2No0.0No0.012/15/2004high2No0.0No0.012/23/2004medium1No0.0No0.012/15/2004high2No0.0No0.01/4/2005high2No0.0No0.01/4/2005high2No0.0No0.01/21/2005low0No0.0No0.02/1/2005high2Yes1.0No0.02/3/2005medium1Yes1.0No0.02/4/2005low0Yes1.0No0.02/15/2005high2No0.0No0.02/15/2005high2No0.0No0.02/15/2005high2No0.0No0.02/15/2005high2No0.0No0.02/	10/15/2004	high	2	No	0.0	No	1.0		
10/10/100Ing2No0.0No0.010/25/2004low0No0.0No0.011/4/2004medium1No0.0No0.011/18/2004high2No0.0No0.011/23/2004medium1No0.0No0.012/3/2004low0No0.0No0.012/15/2004high2No0.0No0.012/15/2004medium1No0.0No0.012/15/2004medium1No0.0No0.012/15/2004medium1No0.0No0.012/15/2004high2No0.0No0.01/4/2005high2No0.0No0.01/4/2005high2No0.0No0.01/21/2005high2Yes1.0No0.02/3/2005medium1Yes1.0No0.02/3/2005medium1Yes1.0No0.02/4/2005low0No0.0No0.02/3/2005low0No0.0No0.02/15/2005high2No0.0No0.02/15/2005high2No0.0No0.02/25/2005high2No0.0No0.0 <td< td=""><td>10/13/2004</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></td<>	10/13/2004	high	2	No	0.0	No	0.0		
11/2/2004INW0NONO0.0NO0.0 $11/18/2004$ high2No0.0No0.0 $11/18/2004$ high2No0.0No0.0 $11/23/2004$ medium1No0.0No0.0 $12/3/2004$ low0No0.0No0.0 $12/3/2004$ high2No0.0No0.0 $12/15/2004$ high2No0.0No0.0 $12/23/2004$ medium1No0.0No0.0 $12/23/2004$ medium1No0.0No0.0 $12/23/2004$ medium1No0.0No0.0 $12/23/2004$ medium1No0.0No0.0 $1/13/2005$ high2No0.0No0.0 $1/21/2005$ high2Yes1.0No0.0 $2/1/2005$ high2Yes1.0No0.0 $2/3/2005$ medium1Yes1.0No0.0 $2/4/2005$ low0No0.0No0.0 $2/8/2005$ low0No0.0No0.0 $2/25/2005$ high2No0.0No0.0 $2/25/2005$ high2No0.0No0.0 $2/25/2005$ high2No0.0No0.0 $2/25/2005$ high2	10/18/2004	low	2	No	0.0	No	0.0		
11/1/2/2004 high 2 No 0.0 No 0.0 11/18/2004 medium 1 No 0.0 No 0.0 11/23/2004 low 0 No 0.0 No 0.0 12/3/2004 low 0 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 11/4/2005 high 2 No 0.0 No 0.0 1/13/2005 high 2 No 0.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/2/2005 high 2 Yes 1.0 No 0.0	11/4/2004	medium	1	No	0.0	No	0.0		
11/10/2004 Ingin 1 No 0.0 No 0.0 11/23/2004 low 0 No 0.0 No 0.0 12/3/2004 low 0 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 12/23/2005 high 2 No 0.0 No 0.0 1/4/2005 high 2 No 0.0 No 0.0 1/21/2005 high 2 Yes 1.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/3/2005 low 0 Yes 1.0 No 0.0 <	11/18/2004	high	2	No	0.0	No	0.0		
12/2/2004 low 0 No 0.0 No 0.0 12/3/2004 low 0 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 12/4/2005 high 2 No 0.0 No 0.0 1/4/2005 high 2 No 0.0 No 0.0 1/4/2005 high 2 No 0.0 No 0.0 1/21/2005 high 2 No 0.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 low 0 No 0.0 No 0.0	11/23/2004	medium	1	No	0.0	No	0.0		
12/3/2004 high 2 No 0.0 No 0.0 12/15/2004 high 2 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 1/4/2005 high 2 No 0.0 No 0.0 1/13/2005 high 2 No 0.0 No 0.0 1/21/2005 high 2 No 0.0 No 0.0 1/21/2005 high 2 Yes 1.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/2/2005 high 2 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0	12/3/2004	low	0	No	0.0	No	0.0		
12/13/2004 Ingin 2 No 0.0 No 0.0 12/23/2004 medium 1 No 0.0 No 0.0 1/4/2005 high 2 No 0.0 No 0.0 1/4/2005 high 2 No 0.0 No 0.0 1/13/2005 high 2 No 0.0 No 0.0 1/21/2005 low 0 No 0.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/2/2005 high 2 Yes 1.0 No 0.0 2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0	12/15/2004	high	2	No	0.0	No	0.0		
11/12/2004 Inclum 1 No 0.0 No 0.0 1/4/2005 high 2 No 0.0 No 0.0 1/13/2005 high 2 No 0.0 No 0.0 1/21/2005 high 2 No 0.0 No 0.0 1/21/2005 low 0 No 0.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/2/2005 high 2 Yes 1.0 No 0.0 2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0 <t< td=""><td>12/23/2004</td><td>medium</td><td>1</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	12/23/2004	medium	1	No	0.0	No	0.0		
1/4/2005 high 2 No 0.0 No 0.0 1/13/2005 high 2 No 0.0 No 0.0 1/21/2005 low 0 No 0.0 No 0.0 2/1/2005 low 0 No 0.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/2/2005 high 2 Yes 2.0 No 0.0 2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/7/2005 low 0 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 <t< td=""><td>12/23/2004</td><td>meanan</td><td>-</td><td>110</td><td>0.0</td><td></td><td>0.0</td><td></td><td></td></t<>	12/23/2004	meanan	-	110	0.0		0.0		
1/12005 high 2 No 0.0 No 0.0 1/13/2005 high 2 No 0.0 No 0.0 1/21/2005 low 0 No 0.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/2/2005 high 2 Yes 1.0 No 0.0 2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/7/2005 low 0 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0 0/0	1/4/2005	high	2	No	0.0	No	0.0		
1/21/2005 low 0 No 0.0 No 0.0 1/21/2005 logh 2 Yes 1.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/2/2005 high 2 Yes 2.0 No 0.0 2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/7/2005 low 0 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	1/13/2005	high	2	No	0.0	No	0.0		
2/1/2005 high 2 Yes 1.0 No 0.0 2/1/2005 high 2 Yes 1.0 No 0.0 2/2/2005 high 2 Yes 2.0 No 0.0 2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/7/2005 low 0 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/2/2005 high 2 No 0.0 No 0.0 2/2/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	1/21/2005	low	0	No	0.0	No	0.0		
2/2/2005 high 2 Yes 2.0 No 0.0 2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/7/2005 low 0 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	2/1/2005	high	2	Ves	1.0	No	0.0		
2/3/2005 medium 1 Yes 1.0 No 0.0 2/4/2005 medium 1 Yes 1.0 No 0.0 2/7/2005 medium 1 Yes 1.0 No 0.0 2/7/2005 low 0 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	2/2/2005	high	2	Ves	2.0	No	0.0		
2/4/2005 medium 1 Yes 1.0 No 0.0 2/7/2005 low 0 Yes 1.0 No 0.0 2/8/2005 low 0 Yes 1.0 No 0.0 2/15/2005 low 0 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	2/3/2005	medium	- 1	Yes	1.0	No	0.0		
2/7/2005 low 0 Yes 1.0 No 0.0 2/8/2005 low 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	2/4/2005	medium	1	Yes	1.0	No	0.0		
2/8/2005 Iow 0 No 0.0 No 0.0 2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	2/7/2005		0	Vec	1.0	No	0.0		
2/15/2005 high 2 No 0.0 No 0.0 2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	2/8/2005		0	No	0.0	No	0.0		
2/25/2005 high 2 No 0.0 No 0.0 3/2/2005 high 2 No 0.0 No 0.0	2/0/2005	high	2	No	0.0	No	0.0		
3/2/2005 high 2 No 0.0 No 0.0	2/25/2005	high	2	No	0.0	No	0.0		
	3/2/2005	high	2	No	0.0	No	0.0		
1 3/8/2005 I IOW () I NO OO I NO OO I	3/8/2005		0	No	0.0	No	0.0		

			Loading	g Rack Area	Warehou	ise Area North	Warehou	se Area South
	Tidal	Stage	Boor	n Sheen	Вос	om Sheen	Boom Sheen	
.			Obse	rvations	Obs	ervations	Observations	
Date	Low Medium		0,50		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
3/15/2005	high	2	No	0.0	No	0.0	(100/110/	(000 110 100)
4/4/2005	low	0	No	0.0	No	0.0		
4/11/2015	high	2	Ves	1.0	No	0.0		
4/13/2015	medium	1	Ves	2.0	No	0.0		
4/13/2005	high	1	Vas	2.0	No	0.0		
4/14/2005	medium	1	Ves	2.0	No	0.0		
4/13/2005	low	1	No	2.0	No	0.0		
4/18/2005	medium	1	No	0.0	No	0.0		
5/2/2005	low	1	No	0.0	No	0.0		
5/2/2005	medium	1	No	0.0	No	0.0		
5/9/2005	low	1	No	0.0	No	0.0		
5/10/2005	low	0	No	0.0	No	0.0		
5/20/2005	modium	1	No	0.0	No	0.0		
5/25/2005	medium	1	No	0.0	No	0.0		
5/50/2005	medium	1	No	0.0		0.0		
6/6/2005 C/10/2005	medium	1	NO	0.0	NO	0.0		
6/10/2005	mealum	1	NO	0.0	NO	0.0		
6/13/2005	nign	2	NO	0.0	NO No	0.0		
6/20/2005	IOW	0	NO	0.0	NO No	0.0		
6/2//2005	nign	2	NO	0.0	NO	0.0		
7/4/2005	medium	1	NO	0.0	NO	0.0		
7/11/2005	nign	2	Yes	1.0	NO	0.0		
7/15/2005	medium	1	NO	0.0	NO	0.0		
//18/2005	low	0	No	0.0	NO	0.0		
7/25/2005	high	2	No	0.0	No	0.0		
8/1/2005	low	0	No	0.0	No	0.0		
8/8/2005	high	2	No	0.0	No	0.0		
8/12/2005	medium	1	No	0.0	No	0.0		
8/15/2005	low	0	No	0.0	No	0.0		
8/22/2005	medium	1	No	0.0	No	0.0		
8/29/2005	low	0	No	0.0	No	0.0		
9/5/2005	medium	1	No	0.0	No	0.0		
9/12/2005	medium	1	No	0.0	No	0.0		
9/14/2005	low	0	No	0.0	No	0.0		
9/19/2005	medium	1	No	0.0	No	0.0		
9/26/2005	low	0	No	0.0	No	0.0		
10/3/2005	medium	1	No	0.0	No	0.0		
10/10/2005	medium	1	No	0.0	No	0.0		
10/14/2005	low	0	No	0.0	No	0.0		
10/17/2005	medium	1	No	0.0	No	0.0		
10/24/2005	medium	1	No	0.0	No	0.0		
10/31/2005	low	0	No	0.0	No	0.0		
11/7/2005	high	2	No	0.0	No	0.0		
11/14/2005	low	0	No	0.0	No	0.0		
11/21/2005	high	2	No	0.0	No	0.0		
11/23/2005	medium	1	No	0.0	No	0.0		
11/28/2005	low	0	No	0.0	No	0.0		
11/29/2005	medium	1	No	0.0	No	0.0		
11/30/2005	medium	1	No	0.0	No	0.0		
12/1/2005	high	2	No	0.0	No	0.0		
12/2/2005	high	2	No	0.0	No	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
			Ohse	rvations	Ohs	ervations	Observations	
Date	Low Modium		0030		0.03		0030	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Bating	Sheen	Sheen Rating
	High		(Yes/No)	(See Notes)		(See Notes)	(Ves/No)	
12/5/2005	high	2	No	0.0	No.	0.0	(103/110)	(500 1000)
12/6/2005	medium	1	No	0.0	No	0.0		
12/7/2005	high	2	No	0.0	No	0.0		
12/9/2005	high	2	No	0.0	No	0.0		
12/15/2005	high	2	Ves	10	No	0.0		
12/19/2005	high	2	Ves	1.0	No	0.0		
12/13/2003	111611	-	105	1.0		0.0		
1/25/2006	low	0	Yes	2.0	Yes	2.0		
2/8/2006		Ū	Yes	1.0	No	0.0		
2/9/2006			Yes	1.0	No	0.0		
2/10/2006			Yes	1.0	No	0.0		
2/13/2006	medium	1	Yes	1.0	No	0.0		
2/14/2006	medium	1	Yes	1.0	No	0.0		
3/15/2006	low	0	No	0.0	No	0.0		
3/17/2006	low	0	No	0.0	No	0.0		
3/21/2006	high	2	No	0.0	No	0.0		
3/27/2006	low	0	No	0.0	No	0.0		
4/3/2006	high	2	No	0.0	No	0.0		
4/11/2006	medium	- 1	No	0.0	No	0.0		
4/14/2006	medium	-	No	0.0	No	0.0		
4/17/2006	high	2	No	0.0	No	0.0		
4/24/2006	low	0	No	0.0	No	0.0		
4/25/2006	medium	1	No	0.0	No	0.0		
4/26/2006	medium	-	No	0.0	No	0.0		
4/27/2006	medium	-	No	0.0	No	0.0		
4/28/2006	medium	-	No	0.0	No	0.0		
5/1/2006	medium	1	No	0.0	No	0.0		
5/9/2006	low	0	No	0.0	No	0.0		
5/17/2006	high	2	No	0.0	No	0.0		
5/18/2006	high	2	No	0.0	No	0.0		
5/22/2006	low	0	No	0.0	No	0.0		
5/30/2006	medium	1	No	0.0	No	0.0		
5/31/2006	high	2	No	0.0	No	0.0		
6/1/2006	high	2	No	0.0	No	0.0		
6/5/2006	medium	1	No	0.0	Yes	0.5		
6/12/2006	low	0	No	0.0	No	0.0		
6/14/2006	medium	1	No	0.0	No	0.0		
7/12/2006	low	0	No	0.0	No	0.0		
7/19/2006	medium	1	No	0.0	Yes	1.0		
7/24/2006	high	2	No	0.0	No	0.0		
7/25/2006	low	0	No	0.0	Yes	1.0		
7/31/2006	high	2	Yes	1.0	No	0.0		
8/2/2006	high	2	No	0.0	No	0.0		
8/8/2006	high	2	No	0.0	No	0.0		
8/14/2006	high	2	Yes	1.0	Yes	1.0		
8/16/2006	medium	1	Yes	1.0	Yes	2.0		
8/21/2006	low	0	No	0.0	No	0.0		
8/25/2006	high	2	Yes	0.5	Yes	0.5		
8/28/2006	high	2	No	0.0	Yes	0.5		
8/29/2006	high	2	No	0.0	No	0.0		

			Loading	Rack Area	Warehou	se Area North	Warehous	se Area South
	Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Boom Sheen	
_ .			Ohse	rvations	Ohs	ervations	Observations	
Date	Low Medium		0030		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
9/1/2006	medium	1	No	0.0	No	0.0	(100/110/	(500 110103)
9/5/2006	low	0	No	0.0	No	0.0		
9/6/2006	low	0	No	0.0	No	0.0		
9/11/2006	high	2	No	0.0	No	0.0		
9/12/2006	high	2	Ves	10	Ves	1.0		
9/18/2006	low	2	Ves	1.0	No	1.0		
9/19/2006	low	0	No	1.0	Ves	2.0		
9/22/2006	high	2	No	0.0	No	2.0		
9/25/2000	high	2	Ves	10	No	0.0		
9/27/2006	high	2	No	1.0	No	0.0		
10/2/2006	medium	2 1	No	0.0	No	0.0		
10/5/2000	low	0	No	0.0	No	0.0		
10/5/2000	high	2	No	0.0	No	0.0		
10/0/2000	high	2	No	0.0	No	0.0		
10/3/2000	high	2	No	0.0	No	0.0		
10/12/2000	modium	2	No	0.0	No	0.0		
10/10/2000	high	1	No	0.0	Voc	0.0		
10/17/2000	high	2	No	0.0	No	1.0		
10/25/2000	high	2	No	0.0	No	0.0		
10/23/2000	high	2	No	0.0	No	0.0		
10/30/2000	high	2	No	0.0	Voc	0.0		
11/1/2006	modium	2	No	0.0	No	1.0		
11/5/2006	high	1	No	0.0	No	0.0		
11/0/2000	high	2	No	0.0	No	0.0		
11/8/2006	high	2	No	0.0	No	0.0		
11/8/2000	high	2	No	0.0	No	0.0		
11/13/2006	high	2	Ves	10	No	0.0		
11/13/2000	modium	2 1	No	1.0	No	0.0		
11/17/2000	high	1	No	0.0	No	0.0		
11/20/2000	high	2	No	0.0	No	0.0		
11/27/2000	high	2	No	0.0	No	0.0		
12/4/2006	medium	2	Ves	10	No	0.0		
12/4/2000	high	1	No	1.0	Ves	1.0		
12/11/2006	high	2	No	0.0	No	0.0		
12/11/2000	medium	2	No	0.0	No	0.0		
12/12/2000	high	2	No	0.0	No	0.0		
12/13/2000	high	2	No	0.0	No	0.0		
12/15/2006	medium	2	No	0.0	Ves	1.0		
12/15/2000	medium	1	No	0.0	No	0.0		
12/18/2006	medium	1	No	0.0	No	0.0		
12/18/2000	high	1	Ves	10	No	0.0		
12/21/2000	high	2	No	1.0	No	0.0		
12/22/2000	high	2	No	0.0	No	0.0		
12/22/2000	iiigii	۷	NU	0.0		0.0		
1/2/2007	high	2	No	0.0	No	0.0		
1/5/2007	high	2	No	0.0	No	0.0		
1/8/2007	high	2	No	0.0	No	0.0		
1/9/2007	high	2	No	0.0	No	0.0		
1/10/2007	high	2	No	0.0	No	0.0		
1/15/2007	high	2	No	0.0	No	0.0		
1/15/2007	l	2	NU	0.0		0.0		

			Loading	g Rack Area	Warehou	ise Area North	Warehou	se Area South
	Tidal	Stage	Boor	n Sheen	Вос	m Sheen	Воо	m Sheen
. .			Ohse	rvations	Obs	ervations	Ohse	ervations
Date	Low Medium		0,50		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
1/19/2007	high	2	Yes	1.0	No	0.0	(100/110/	(000110100)
1/22/2007	high	2	Yes	0.5	No	0.0		
1/29/2007	high	2	Yes	1.0	No	0.0		
1/31/2007	high	2	No	0.0	Yes	1.0		
2/2/2007	high	2	No	0.0	No	0.0		
2/5/2007	high	2	No	0.0	No	0.0		
2/6/2007	high	2	No	0.0	No	0.0		
2/7/2007	high	2	No	0.0	No	0.0		
2/12/2007	high	2	No	0.0	No	0.0		
2/14/2007	high	2	No	0.0	No	0.0		
2/16/2007	high	2	No	0.0	No	0.0		
2/20/2007	high	2	No	0.0	No	0.0		
2/26/2007	high	2	No	0.0	No	0.0		
3/5/2007	medium	1	No	0.0	No	0.0		
3/7/2007	medium	- 1	No	0.0	No	0.0		
3/13/2007	high	2	No	0.0	No	0.0		
3/16/2007	medium	1	No	0.0	No	0.0		
3/19/2007	low	0	No	0.0	No	0.0		
3/20/2007	medium	1	No	0.0	No	0.0		
3/21/2007	high	2	No	0.0	No	0.0		
3/22/2007	high	2	No	0.0	No	0.0		
3/26/2007	high	2	No	0.0	No	0.0		
3/30/2007	medium	1	No	0.0	No	0.0		
4/2/2007	high	2	No	0.0	No	0.0		
4/6/2007	high	2	No	0.0	Ves	1.0		
4/9/2007	high	2	No	0.0	No	0.0		
4/12/2007	high	2	No	0.0	No	0.0		
4/13/2007	medium	1	No	0.0	No	0.0		
4/16/2007	low	0	No	0.0	No	0.0		
4/19/2007	medium	1	No	0.0	No	0.0		
4/23/2007	high	2	No	0.0	No	0.0		
4/24/2007	high	2	No	0.0	Yes	1.0		
4/26/2007	medium	1	No	0.0	No	0.0		
4/27/2007	high	2	No	0.0	No	0.0		
4/30/2007	low	0	No	0.0	No	0.0		
5/3/2007	medium	1	No	0.0	No	0.0		
5/8/2007	high	2	No	0.0	No	0.0		
5/9/2007	high	2	No	0.0	No	0.0		
5/14/2007	low	0	No	0.0	No	0.0		
5/17/2007	medium	1	No	0.0	No	0.0		
5/21/2007	high	2	No	0.0	No	0.0		
5/23/2007	medium	- 1	No	0.0	No	0.0		
6/1/2007	medium	-	No	0.0	No	0.0		
6/4/2007	high	2	Yes	1.0	Yes	1.0		
6/6/2007	high	- 2	No	0.0	No	0.0		
6/7/2007	medium	-	No	0.0	Yes	1.0		
6/11/2007	low	0	No	0.0	No	0.0		
6/13/2007	low	0	No	0.0	No	0.0		
6/14/2007	low	0	No	0.0	No	0.0		
6/18/2007	medium	1	No	0.0	No	0.0		

			Loading	g Rack Area	Warehou	ise Area North	Warehou	se Area South
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
Data		U	Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium							
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
6/19/2007	high	2	No	0.0	No	0.0	(/ - /	(/
6/25/2007	low	-	No	0.0	No	0.0		
7/2/2007	high	2	No	0.0	Yes	2.0		
7/9/2007	low	0	No	0.0	No	0.0		
7/13/2007	low	0	No	0.0	No	0.0		
7/16/2007	low	0	No	0.0	No	0.0		
7/23/2007	low	0	No	0.0	No	0.0		
7/30/2007	medium	1	No	0.0	No	0.0		
7/31/2007	high	2	No	0.0	Yes	1.0		
8/6/2007	medium	- 1	No	0.0	No	0.0		
8/8/2007	low	0	No	0.0	No	0.0		
8/13/2007	medium	1	No	0.0	No	0.0		
8/16/2007	high	2	No	0.0	No	0.0		
8/20/2007	high	2	No	0.0	No	0.0		
8/22/2007	medium	-	No	0.0	No	0.0		
8/23/2007	medium	1	No	0.0	No	0.0		
8/24/2007	low	0	No	0.0	No	0.0		
8/27/2007	low	0	No	0.0	No	0.0		
8/30/2007	low	0	No	0.0	No	0.0		
9/4/2007	medium	1	No	0.0	No	0.0		
9/10/2007	medium	1	No	0.0	No	0.0		
9/13/2007	medium	1	No	0.0	No	0.0		
9/14/2007	high	2	No	0.0	No	0.0		
9/17/2007	high	2	No	0.0	No	0.0		
9/18/2007	high	2	No	0.0	No	0.0		
9/19/2007	high	2	No	0.0	No	0.0		
9/20/2007	medium	- 1	No	0.0	No	0.0		
9/24/2007	low	-	No	0.0	No	0.0		
10/1/2007	high	2	No	0.0	No	0.0		
10/2/2007	high	2	No	0.0	No	0.0		
10/3/2007	medium	- 1	No	0.0	No	0.0		
10/5/2007	low	-	No	0.0	No	0.0		
10/8/2007	medium	1	No	0.0	No	0.0		
10/9/2007	high	2	No	0.0	No	0.0		
10/11/2007	high	2	No	0.0	No	0.0		
10/15/2007	high	2	No	0.0	No	0.0		
10/17/2007	medium	1	No	0.0	No	0.0		
10/22/2007	low	0	No	0.0	No	0.0		
10/24/2007	medium	1	No	0.0	No	0.0		
10/25/2007	high	2	No	0.0	No	0.0		
10/29/2007	high	2	No	0.0	No	0.0		
10/31/2007	low	0	No	0.0	No	0.0		
11/1/2007	low	0	No	0.0	No	0.0		
11/2/2007	low	0	No	0.0	No	0.0		
11/5/2007	low	0	No	0.0	No	0.0		
11/6/2007	low	0	No	0.0	No	0.0		
11/12/2007	high	2	No	0.0	No	0.0		
11/13/2007	high	2	No	0.0	No	0.0		
11/15/2007	high	2	No	0.0	No	0.0		
11/16/2007	high	2	No	0.0	No	0.0		

Parte Horswers Boorssees Boorssees Boorssees 1000000000000000000000000000000000000				Loading Rack Area		Warehouse Area North		Warehouse Area South	
Date Observations Observations Observations Observations Low, Medium (ebb 8, flood), 11/15/2007 Tide Rating (See Notes) Sheen (See Notes) Sheen Rating (See Notes) Sheen Sh		Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Воо	m Sheen
Date Conv. Medium (ebb & Riool), High Tide Rating (See Notes) Sheen (Yee/No) Sheen (See Notes) Sheen (Yee/No) Sheen (See Notes) Sheen (Yee/No) Sheen (Yee/No) </th <th>_.</th> <th></th> <th></th> <th>Ohse</th> <th>rvations</th> <th>Obs</th> <th>ervations</th> <th colspan="2">Observations</th>	_ .			Ohse	rvations	Obs	ervations	Observations	
Letter Control Tide Rating (See Notes) Sheen (See Notes) 111/15/2007 migh high 2 No 0.0 No 0.0 11/12/2007 high 2 No 0.0 No 0.0 11/12/2007 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/11/2008 high 2 No 0.0 No 0.0 1/11/2008 high 2 No 0.0	Date	Low Medium		0030		0.03		0.50	
High (See Notes) (Yes/No) (See Notes) (Yes/No) (See Notes) (Yes/No) (See Notes) 11/1/5/2007 high 2 No 0.0 No 0.0 11/2/2/2007 high 2 No 0.0 No 0.0 11/2/2/2007 high 2 No 0.0 No 0.0 12/1/2/2007 high 2 No 0.0 No 0.0 12/1/1/2007 high 2 No 0.0 No 0.0 12/1/1/2007 high 2 No 0.0 No 0.0 12/1/2007 high 2 No 0.0 No 0.0 12/2/2007 high 2 No 0.0 No 0.0 11/2008 high 2 No 0.0 No 0.0 1//2008 high 2 No 0.0 No 0.0 1/2/2008 high 2 No 0.0		(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
11/15/2007 Ingh 2 No 0.0 No 0.0 11/16/2007 high 2 No 0.0 No 0.0 11/27/2007 high 2 No 0.0 No 0.0 11/27/2007 high 2 No 0.0 No 0.0 12/1/2007 high 2 No 0.0 No 0.0 12/1/2007 high 2 No 0.0 No 0.0 12/1/2007 high 2 No 0.0 No 0.0 12/2/007 high 2 No 0.0 No 0.0 12/2/2007 high 2 No 0.0 No 0.0 12/2/2008 high 2 No 0.0 No 0.0 1/1/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0		High	(See Notes)	(Yes/No)	(See Notes)	(Ves/No)	(See Notes)	(Ves/No)	(See Notes)
11/25/200 Initial 1 No 0.0 No 0.0 11/27/2007 high 2 No 0.0 Ves 0.5 12/17/2007 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/12/2007 high 2 No 0.0 No 0.0 12/2/2007 medium 1 No 0.0 No 0.0 11/2/2008 high 2 No 0.0 No 0.0 11/1/2008 high 2 No 0.0 No 0.0 11/12/2008 high 2 No 0.0 No 0.0 12/21/208 high 2 No 0.0 No 0.0	11/19/2007	medium	1	No	0.0	No	0.0	(100/110/	(500 110103)
11/27/2007 high 2 No 0.0 Yes 0.5 12/3/2007 high 2 No 0.0 No 0.0 12/10/2007 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/19/2007 high 2 No 0.0 No 0.0 12/20/2007 high 2 No 0.0 No 0.0 12/2/2007 high 2 No 0.0 No 0.0 12/2/2008 high 2 No 0.0 No 0.0 1/1/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0	11/26/2007	high	2	No	0.0	No	0.0		
111/1/2007 high 1 NO 0.0 NO 0.0 12/1/1/2007 high 2 NO 0.0 NO 0.0 12/24/2007 medium 1 NO 0.0 NO 0.0 12/24/2007 medium 1 NO 0.0 NO 0.0 11/1/2008 high 2 NO 0.0 NO 0.0 1/1/1/2008 high 2 NO 0.0 NO 0.0 1/22/2008 high 2 NO 0.0 NO 0.0 1/22/2008 high 2 NO 0.0 NO 0.0 1/24/2008 high 2 NO 0.0 NO 0.0	11/27/2007	high	2	No	0.0	Vec	0.5		
12/10/200 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/11/2007 high 2 No 0.0 No 0.0 12/12/0707 high 2 No 0.0 No 0.0 12/202007 high 2 No 0.0 No 0.0 12/202007 high 2 No 0.0 No 0.0 11/2/2008 high 2 No 0.0 No 0.0 1/1/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0	12/3/2007	high	2	No	0.0	No	0.0		
12/11/200 high 2 No 0.0 No 0.0 12/14/2007 high 2 No 0.0 No 0.0 12/14/2007 high 2 No 0.0 No 0.0 12/14/2007 high 2 No 0.0 No 0.0 12/20/2007 high 2 No 0.0 No 0.0 12/24/2007 medium 1 No 0.0 No 0.0 1/2/2008 high 2 No 0.0 No 0.0 1/1/2008 high 2 No 0.0 No 0.0 1/1/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 <tr< td=""><td>12/10/2007</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></tr<>	12/10/2007	high	2	No	0.0	No	0.0		
12/14/2007 high 2 No 0.0 No 0.0 12/1/7/2007 high 2 No 0.0 No 0.0 12/19/2007 high 2 No 0.0 No 0.0 12/19/2007 high 2 No 0.0 No 0.0 12/24/2007 medium 1 No 0.0 No 0.0 1/2/2008 high 2 No 0.0 No 0.0 1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 <t< td=""><td>12/11/2007</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	12/11/2007	high	2	No	0.0	No	0.0		
12/17/2007 high 2 No 0.0 No 0.0 12/19/2007 high 2 No 0.0 No 0.0 12/19/2007 high 2 No 0.0 No 0.0 12/20/2007 medium 1 No 0.0 No 0.0 1/2/2008 high 2 Yes 1.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/2/2008 high 2 No 0.0 No 0.0 1/2/2008 high 2 No 0.0 No 0.0 1/2/2008 high 2 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0	12/14/2007	high	2	No	0.0	No	0.0		
12/19/2007 high 2 No 0.0 No 0.0 12/20/2007 high 2 No 0.0 No 0.0 12/24/2007 medium 1 No 0.0 No 0.0 1/2/2008 high 2 Yes 1.0 No 0.0 1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 2/11/2008 medium 1 No 0.0 No 0.0 2/11/2008 high 2 No 0.0 No 0.0 <	12/17/2007	high	2	No	0.0	No	0.0		
12/20/2007 high 2 No 0.0 No 0.0 12/24/2007 medium 1 No 0.0 No 0.0 1/1/2008 high 2 Yes 1.0 No 0.0 1 1/1/2008 high 2 No 0.0 No 0.0 1 1/1/1/2008 high 2 No 0.0 No 0.0 1 1/21/2008 high 2 No 0.0 No 0.0 1 1/21/2008 high 2 No 0.0 No 0.0 1 2/14/2008 high 2 No 0.0 No 0.0 1 2/14/2008 high 2 No 0.0 No 0.0 1 <t< td=""><td>12/19/2007</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	12/19/2007	high	2	No	0.0	No	0.0		
12/24/2007 medium 1 No 0.0 No 0.0 1/2/2008 high 2 Yes 1.0 No 0.0 1/1/2008 high 2 No 0.0 No 0.0 1/11/2008 high 2 No 0.0 No 0.0 1/11/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 1/22/2008 high 2 No 0.0 No 0.0 1/24/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/2/2/2008 high 2 No 0.0 No 0.0	12/20/2007	high	2	No	0.0	No	0.0		
1/2/2008 high 2 Yes 1.0 No 0.0 1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/1/1/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 1/2/2/2008 high 2 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/2/2/2008 high 2 No 0.0 No 0.0	12/24/2007	medium	1	No	0.0	No	0.0		
1/2/2008 high 2 Yes 1.0 No 0.0 1/7/2008 high 2 No 0.0 No 0.0 1/11/2008 high 2 No 0.0 No 0.0 1/14/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 1/22/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 2/4/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/2/5/2008 high 2 No 0.0 No 0.0									
1/7/2008 high 2 No 0.0 No 0.0 1/11/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 1/22/2008 high 2 No 0.0 No 0.0 1/28/2008 high 2 No 0.0 No 0.0 2/4/2008 high 2 No 0.0 No 0.0 2/11/2008 medium 1 No 0.0 No 0.0 2/11/2008 high 2 No 0.0 No 0.0 2/11/2008 high 2 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/26/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 <td< td=""><td>1/2/2008</td><td>high</td><td>2</td><td>Yes</td><td>1.0</td><td>No</td><td>0.0</td><td></td><td></td></td<>	1/2/2008	high	2	Yes	1.0	No	0.0		
1/11/2008 high 2 No 0.0 No 0.0 1/14/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 1/22/2008 high 2 No 0.0 No 0.0 1/28/2008 high 2 No 0.0 No 0.0 1/29/2008 high 2 No 0.0 No 0.0 2/4/2008 high 2 No 0.0 No 0.0 2/11/2008 medium 1 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/12/20208 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 <	1/7/2008	high	2	No	0.0	No	0.0		
1/14/2008 high 2 No 0.0 No 0.0 1/21/2008 high 2 No 0.0 No 0.0 1/22/2008 high 2 No 0.0 No 0.0 1/28/2008 high 2 No 0.0 No 0.0 1/29/2008 high 2 No 0.0 Yes 0.5 2/11/2008 medium 1 No 0.0 No 0.0 2/11/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/19/2008 high 2 No 0.0 No 0.0 2/2/2/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/1/2008 high 2 No 0.0 No 0.0 3/1/2/2008 high 2 No 0.0 No 0.0	1/11/2008	high	2	No	0.0	No	0.0		
1/21/2008 high 2 No 0.0 No 0.0 1/22/2008 high 2 No 0.0 No 0.0 1/28/2008 high 2 No 0.0 No 0.0 1/29/2008 high 2 No 0.0 No 0.0 2/4/2008 high 2 No 0.0 No 0.0 2/11/2008 medium 1 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/20/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/4/2008 medium 1 No 0.0 No 0.0 3/1/2008 high 2 No 0.0 No 0.0 <t< td=""><td>1/14/2008</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	1/14/2008	high	2	No	0.0	No	0.0		
1/22/2008 high 2 No 0.0 No 0.0 1/28/2008 high 2 No 0.0 No 0.0 1/29/2008 high 2 No 0.0 No 0.0 2/4/2008 high 2 No 0.0 Yes 0.5 2/11/2008 medium 1 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/19/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 <	1/21/2008	high	2	No	0.0	No	0.0		
1/28/2008 high 2 No 0.0 No 0.0 1/29/2008 high 2 No 0.0 No 0.0 2/4/2008 migh 2 No 0.0 Yes 0.5 2/11/2008 medium 1 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/19/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 <	1/22/2008	high	2	No	0.0	No	0.0		
1/29/2008 high 2 No 0.0 No 0.0 2/4/2008 high 2 No 0.0 No 0.0 2/11/2008 medium 1 No 0.0 No 0.0 2/11/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 <t< td=""><td>1/28/2008</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	1/28/2008	high	2	No	0.0	No	0.0		
2/4/2008 high 2 No 0.0 Yes 0.5 2/11/2008 medium 1 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/19/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 medium 1 No 0.0 No 0.0	1/29/2008	high	2	No	0.0	No	0.0		
2/11/2008 medium 1 No 0.0 No 0.0 2/12/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/19/2008 high 2 No 0.0 No 0.0 2/20/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 <	2/4/2008	high	2	No	0.0	Yes	0.5		
2/12/2008 high 2 No 0.0 No 0.0 2/14/2008 high 2 No 0.0 No 0.0 2/19/2008 high 2 No 0.0 No 0.0 2/20/2008 high 2 No 0.0 No 0.0 2/28/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/1/2008 medium 1 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0	2/11/2008	medium	1	No	0.0	No	0.0		
2/14/2008 high 2 No 0.0 No 0.0 2/19/2008 high 2 No 0.0 No 0.0 2/20/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/1/2008 medium 1 No 0.0 No 0.0 3/1/2008 high 2 No 0.0 No 0.0 3/1/2008 high 2 No 0.0 No 0.0 3/1/2008 high 2 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 <td< td=""><td>2/12/2008</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></td<>	2/12/2008	high	2	No	0.0	No	0.0		
2/19/2008 high 2 No 0.0 No 0.0 2/20/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 2/28/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/4/2008 medium 1 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0	2/14/2008	high	2	No	0.0	No	0.0		
2/20/2008 high 2 No 0.0 No 0.0 2/25/2008 high 2 No 0.0 No 0.0 2/28/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/4/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 medium 1 No 0.0 No 0.0 3/17/2008 medium 1 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0	2/19/2008	high	2	No	0.0	No	0.0		
2/25/2008 high 2 No 0.0 No 0.0 2/28/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/4/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/17/2008 medium 1 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0	2/20/2008	high	2	No	0.0	No	0.0		
2/28/2008 high 2 No 0.0 No 0.0 3/3/2008 medium 1 No 0.0 No 0.0 3/4/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4	2/25/2008	high	2	No	0.0	No	0.0		
3/3/2008 medium 1 No 0.0 No 0.0 3/4/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/17/2008 medium 1 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 <tr< td=""><td>2/28/2008</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></tr<>	2/28/2008	high	2	No	0.0	No	0.0		
3/4/2008 medium 1 No 0.0 No 0.0 3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/17/2008 medium 1 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 <t< td=""><td>3/3/2008</td><td>medium</td><td>1</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	3/3/2008	medium	1	No	0.0	No	0.0		
3/10/2008 high 2 No 0.0 No 0.0 3/11/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/14/2008 medium 1 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/21/2008 low 0 No 0.0 No 0.0 4	3/4/2008	medium	1	No	0.0	No	0.0		
3/11/2008 high 2 No 0.0 No 0.0 3/12/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/17/2008 medium 1 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 3/1/2008 medium 1 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0	3/10/2008	high	2	No	0.0	No	0.0		
3/12/2008 high 2 No 0.0 No 0.0 3/14/2008 high 2 No 0.0 No 0.0 3/17/2008 medium 1 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 <t< td=""><td>3/11/2008</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	3/11/2008	high	2	No	0.0	No	0.0		
3/14/2008 high 2 No 0.0 No 0.0 3/17/2008 medium 1 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 <	3/12/2008	high	2	No	0.0	No	0.0		
3/17/2008 medium 1 No 0.0 No 0.0 3/24/2008 high 2 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/7/2008 high 2 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0	3/14/2008	high	2	No	0.0	No	0.0		
3/24/2008 high 2 No 0.0 No 0.0 3/26/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 No 0.0 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0	3/17/2008	medium	1	No	0.0	No	0.0		
3/26/2008 high 2 No 0.0 No 0.0 3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/7/2008 high 2 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 Yes 0.5 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0 <td>3/24/2008</td> <td>high</td> <td>2</td> <td>No</td> <td>0.0</td> <td>No</td> <td>0.0</td> <td></td> <td></td>	3/24/2008	high	2	No	0.0	No	0.0		
3/31/2008 medium 1 No 0.0 No 0.0 4/1/2008 medium 1 No 0.0 No 0.0 4/7/2008 high 2 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 Yes 0.5 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0 5/2/2008 medium 1 No 0.0 No 0.0 <td>3/26/2008</td> <td>high</td> <td>2</td> <td>No</td> <td>0.0</td> <td>No</td> <td>0.0</td> <td></td> <td></td>	3/26/2008	high	2	No	0.0	No	0.0		
4/1/2008 medium 1 No 0.0 No 0.0 4/7/2008 high 2 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 Yes 0.5 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0 5/2/2008 medium 1 No 0.0 No 0.0	3/31/2008	medium	1	No	0.0	No	0.0		
4/7/2008 high 2 No 0.0 No 0.0 4/10/2008 medium 1 No 0.0 Yes 0.5 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0 5/5/2008 medium 1 No 0.0 No 0.0	4/1/2008	medium	1	No	0.0	NO	0.0		
4/10/2008 medium 1 No 0.0 Yes 0.5 4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0	4/7/2008	high	2	No	0.0	No	0.0		
4/11/2008 medium 1 No 0.0 No 0.0 4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0 5/2/2008 medium 1 No 0.0 No 0.0	4/10/2008	medium	1	No	0.0	Yes	0.5		
4/15/2008 medium 1 No 0.0 No 0.0 4/16/2008 low 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0	4/11/2008	medium	1	NO	0.0	NO	0.0		
4/16/2008 10W 0 No 0.0 No 0.0 4/18/2008 low 0 No 0.0 No 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0 5/5/2008 medium 1 No 0.0 No 0.0	4/15/2008	medium	1	NO	0.0	NO	0.0		
4/18/2008 10W 0 NO 0.0 NO 0.0 4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 Iow 0 No 0.0 No 0.0 5/5/2008 medium 1 No 0.0 No 0.0	4/16/2008	low	U	NO N-	0.0		0.0		
4/21/2008 medium 1 No 0.0 No 0.0 4/22/2008 medium 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0 5/5/2008 medium 1 No 0.0 No 0.0	4/18/2008	IOW	U	NO No	0.0		0.0		
4/22/2008 Intertain 1 No 0.0 No 0.0 4/28/2008 medium 1 No 0.0 No 0.0 5/2/2008 low 0 No 0.0 No 0.0 5/5/2008 medium 1 No 0.0 No 0.0	4/21/2008	medium	1	NO	0.0		0.0		
4/20/2008 Inedium 1 NO 0.0 NO 0.0 5/2/2008 Iow 0 No 0.0 No 0.0 5/5/2008 medium 1 No 0.0 No 0.0	4/22/2008	modium	1	NO	0.0		0.0		
5/5/2008 medium 1 No 0.0 No 0.0	4/20/2008 5/2/2000		I O	No	0.0	No	0.0		
	5/5/2008	medium	1	No	0.0	No	0.0		

			Loading	Rack Area	Warehou	ise Area North	Warehou	se Area South
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
Data		0	Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium							
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
5/12/2008	medium	1	No	0.0	No	0.0	(100)110)	(,
5/16/2008	medium	- 1	No	0.0	No	0.0		
5/19/2008	low	0	No	0.0	No	0.0		
5/21/2008	low	0	No	0.0	No	0.0		
5/23/2008	high	2	No	0.0	No	0.0		
5/27/2008	medium	1	No	0.0	Yes	0.5		
5/29/2008	medium	1	No	0.0	No	0.0		
6/2/2008	low	0	No	0.0	No	0.0		
6/9/2008	medium	1	No	0.0	No	0.0		
6/12/2008	medium	1	No	0.0	No	0.0		
6/17/2008	low	0	No	0.0	No	0.0		
6/18/2008	low	0	No	0.0	No	0.0		
6/19/2008	medium	1	No	0.0	No	0.0		
6/23/2008	high	2	No	0.0	Yes	1.0		
6/25/2008	medium	1	No	0.0	No	0.0		
6/26/2008	medium	1	No	0.0	No	0.0		
6/27/2008	low	0	No	0.0	No	0.0		
6/30/2008	low	0	No	0.0	No	0.0		
7/7/2008	high	2	No	0.0	No	0.0		
7/8/2008	high	2	No	0.0	No	0.0		
7/14/2008	low	0	No	0.0	No	0.0		
7/16/2008	medium	1	Yes	1.0	Yes	1.0		
7/21/2008	high	2	No	0.0	No	0.0		
7/22/2008	high	2	No	0.0	No	0.0		
7/23/2008	high	2	No	0.0	No	0.0		
7/28/2008	low	0	No	0.0	No	0.0		
7/30/2008	low	0	No	0.0	No	0.0		
7/31/2008	low	0	No	0.0	No	0.0		
8/4/2008	high	2	No	0.0	No	0.0		
8/5/2008	high	2	No	0.0	No	0.0		
8/6/2008	high	2	No	0.0	No	0.0		
8/7/2008	high	2	No	0.0	No	0.0		
8/8/2008	medium	1	No	0.0	No	0.0		
8/11/2008	low	0	No	0.0	No	0.0		
8/12/2008	low	0	No	0.0	No	0.0		
8/13/2008	low	0	No	0.0	No	0.0		
8/18/2008	medium	1	No	0.0	No	0.0		
8/19/2008	high	2	No	0.0	Yes	1.0		
8/20/2008	high	2	No	0.0	No	0.0		
8/21/2008	high	2	No	0.0	No	0.0		
8/25/2008	medium	1	No	0.0	No	0.0		
8/2//2008	low	0	No	0.0	NO	0.0		
9/2/2008	medium	1	NO	0.0	NO	0.0		
9/8/2008	medium	1	No	0.0	No	0.0		
9/16/2008	medium	1	NO	0.0	NO	0.0		
9/1//2008	nigh	2	NO	0.0	NO	0.0		
9/18/2008	nign	2	NO No	0.0		0.0		
9/19/2008	nign	2	NO No	0.0		0.0		
9/22/2008	nign	۲ ۱	NO	0.0		0.0		
9/23/2008	medium	1	INO	0.0	I INO	0.0		

	Tidal Stage		Loading Rack Area Boom Sheen		Warehouse Area North Boom Sheen		Warehouse Area South Boom Sheen	
Date								
			Observations		Observations		Observations	
	Low Medium							
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
9/24/2008	low	0	No	0.0	No	0.0	(/ - /	(/
9/29/2008	high	2	No	0.0	No	0.0		
9/30/2008	high	2	No	0.0	No	0.0		
10/1/2008	high	2	No	0.0	No	0.0		
10/2/2008	high	2	No	0.0	No	0.0		
10/6/2008	high	2	No	0.0	No	0.0		
10/13/2008	medium	1	No	0.0	No	0.0		
10/15/2008	medium	1	No	0.0	No	0.0		
10/17/2008	high	2	No	0.0	No	0.0		
10/20/2008	high	2	No	0.0	No	0.0		
10/21/2008	high	2	No	0.0	No	0.0		
10/24/2008	low	0	No	0.0	No	0.0		
10/25/2008	medium	1	No	0.0	No	0.0		
10/27/2008	high	2	No	0.0	No	0.0		
11/3/2008	high	2	No	0.0	No	0.0		
11/6/2008	high	2	No	0.0	No	0.0		
11/10/2008	medium	1	No	0.0	No	0.0		
11/14/2008	high	2	No	0.0	No	0.0		
11/17/2008	high	2	No	0.0	No	0.0		
11/18/2008	high	2	No	0.0	No	0.0		
11/21/2008	medium	1	No	0.0	No	0.0		
11/24/2008	medium	1	No	0.0	No	0.0		
11/25/2008	high	2	No	0.0	No	0.0		
12/1/2008	high	2	No	0.0	No	0.0		
12/2/2008	high	2	No	0.0	No	0.0		
12/3/2008	high	2	No	0.0	No	0.0		
12/8/2008	high	2	No	0.0	No	0.0		
12/11/2008	high	2	No	0.0	No	0.0		
12/12/2008	high	2	No	0.0	No	0.0		
12/15/2008	high	2	No	0.0	No	0.0		
12/16/2008	high	2	No	0.0	No	0.0		
12/17/2008	high	2	No	0.0	No	0.0		
12/23/2008	high	2	No	0.0	No	0.0		
12/29/2008	high	2	No	0.0	No	0.0		
. /= /= ===		-						
1/5/2009	high	2	No	0.0	No	0.0		
1/12/2009	high	2	NO	0.0	NO	0.0		
1/14/2009	nign	2	NO	0.0	NO	0.0		
1/15/2009	nign	2	NO	0.0	NO	0.0		
1/16/2009	nign	2	NO	0.0	NO	0.0		
1/20/2009	nign	2	NO	0.0	NO No	0.0		
1/22/2009	modium	۲ ۱	NO	0.0		0.0		
1/20/2009	high	1 2	NO	0.0		0.0		
1/2//2009	modium	∠ 1	NO	0.0		0.0		
1/20/2009	modium	1	NO	0.0		0.0		
1/20/2009	medium	1 1	No	0.0		0.0		
2/2/2009	high	1 2	No	0.0	No	0.0		
2/2/2009	high	2	Vec	0.0	No	0.0		
2/9/2009	high	2	No	0.0	No	0.0		
2, 3, 2005	1	<u>~</u>	110	0.0	1 10	0.0		
			Loading Rack Area		Warehouse Area North		Warehouse Area South	
-----------	----------------	-------------	-------------------	--------------	----------------------	--------------	----------------------	--------------
	Tidal Stage		Boor	n Sheen	Воо	m Sheen	Boo	m Sheen
Data		C	Obse	rvations	Obs	ervations	Obse	ervations
Date	Low, Medium							
	(ebb & flood),	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes /No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
2/11/2009	medium	1	No	0.0	No	0.0		
2/17/2009	high	2	No	0.0	Yes	0.5		
2/18/2009	high	2	No	0.0	No	0.0		
2/23/2009	high	2	No	0.0	No	0.0		
2/26/2009	medium	1	No	0.0	No	0.0		
3/3/2009	high	2	No	0.0	No	0.0		
3/9/2009	medium	1	No	0.0	No	0.0		
3/11/2009	medium	1	No	0.0	No	0.0		
3/16/2009	medium	1	No	0.0	No	0.0		
3/17/2009	high	2	No	0.0	No	0.0		
3/18/2009	high	2	No	0.0	No	0.0		
3/23/2009	medium	1	No	0.0	No	0.0		
3/30/2009	high	2	No	0.0	No	0.0		
3/31/2009	high	2	No	0.0	No	0.0		
4/6/2009	medium	1	No	0.0	No	0.0		
4/7/2009	medium	1	No	0.0	No	0.0		
4/13/2009	high	2	No	0.0	No	0.0		
4/15/2009	high	2	No	0.0	No	0.0		
4/16/2009	low	0	No	0.0	No	0.0		
4/21/2009	low	0	No	0.0	No	0.0		
4/27/2009	medium	1	No	0.0	No	0.0		
4/28/2009	high	2	No	0.0	No	0.0		
4/29/2009	high	2	No	0.0	No	0.0		
5/4/2009	low	0	No	0.0	No	0.0		
5/11/2009	medium	1	No	0.0	No	0.0		
5/14/0009	high	2	No	0.0	No	0.0		
5/15/2009	high	2	No	0.0	No	0.0		
5/18/2009	medium	1	No	0.0	No	0.0		
5/26/2009	medium	1	No	0.0	No	0.0		
5/27/2009	medium	1	No	0.0	No	0.0		
6/1/2009	medium	1	No	0.0	No	0.0		
6/2/2009	medium	1	No	0.0	No	0.0		
6/4/2009	low	0	No	0.0	No	0.0		
6/8/2009	medium	1	No	0.0	No	0.0		
6/10/2009	high	2	No	0.0	No	0.0		
6/11/2009	medium	1	No	0.0	No	0.0		
6/15/2009	high	2	No	0.0	No	0.0		
6/16/2009	medium	1	No	0.0	No	0.0		
6/19/2009	high	2	No	0.0	No	0.0		
6/22/2009	low	0	No	0.0	No	0.0		
6/25/2009	high	2	No	0.0	No	0.0		
6/29/2009	high	2	No	0.0	No	0.0		
7/6/2009	low	0	No	0.0	No	0.0		
7/13/2009	high	2	No	0.0	No	0.0		
7/15/2009	high	2	No	0.0	No	0.0		
7/16/2009	low	0	No	0.0	No	0.0		
7/20/2009	low	0	No	0.0	No	0.0		
7/22/2009	low	0	No	0.0	No	0.0		
//27/2009	high	2	No	0.0	No	0.0		
8/3/2009	low	0	No	0.0	No	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal Stage		Boor	n Sheen	Воо	m Sheen	Воо	m Sheen
_ .			Ohse	rvations	Obs	ervations	Ohse	ervations
Date	Low Medium		0030		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High		(Yes/No)		(Vec/No)		(Ves/No)	(See Notes)
8/10/2000	high	(JCC NO(C3)	No.		Voc		(163/110)	(500 10003)
8/10/2009	low	2	No	0.0	No	0.5		
8/14/2009	low	0	No	0.0	No	0.0		
8/17/2009	low	0	No	0.0	No	0.0		
8/18/2009	high	2	No	0.0	No	0.0		
8/24/2009	low	2	No	0.0	No	0.0		
0/1/2009	modium	1	No	0.0	No	0.0		
9/1/2009	high	1	No	0.0	No	0.0		
9/8/2009	high	2	No	0.0	No	0.0		
9/11/2009	modium	2	No	0.0	No	0.0		
9/14/2009	modium	1	No	0.0	No	0.0		
9/10/2009	modium	1	No	0.0	No	0.0		
0/18/2009	high	1	No	0.0	No	0.0		
9/18/2009	high	2	No	0.0	No	0.0		
9/21/2009	low	2	No	0.0	No	0.0		
9/28/2009	nodium	0	No	0.0	NO	0.0		
10/1/2009	high	1	No	0.0	NO	0.0		
10/12/2009	modium	2	No	0.0	NO	0.0		
10/12/2009	high	1	No	0.0	NO Voc	0.0		
10/20/2009	high	2	No	0.0	No	0.5		
10/21/2009	modium	2	No	0.0	NO	0.0		
10/20/2009	modium	1	No	0.0	No	0.0		
11/2/2009	modium	1	No	0.0	No	0.0		
11/2/2009	high	1	No	0.0	No	0.0		
11/3/2009	modium	2	No	0.0	No	0.0		
11/10/2009	high	1	No	0.0	No	0.0		
11/10/2009	high	2	No	0.0	No	0.0		
11/17/2009	high	2	No	0.0	No	0.0		
11/18/2009	high	2	No	0.0	NO	0.0		
11/23/2009	high	2	No	0.0	NO	0.0		
11/24/2009	high	2	No	0.0	NO	0.0		
12/2/2009	high	2	No	0.0	NO	0.0		
12/3/2009	high	2	No	0.0	NO	0.0		
12/4/2009	high	2	No	0.0	NO	0.0		
12/7/2009	high	2	No	0.0	NO	0.0		
12/8/2009	high	2	NO	0.0	NO	0.0		
12/9/2009	modium	2	NO	0.0	NO	0.0		
12/10/2009	high	1	No	0.0	No	0.0		
12/11/2009	high	2	No	0.0	Voc	0.0		
12/14/2009	high	2	No	0.0	No	1.0		
12/15/2009	high	2	No	0.0	NO	0.0		
12/16/2009	nign	2	NO	0.0	NO	0.0		
12/1//2009	high	2	NO	0.0		0.0		
12/21/2009	nign biab	2	NO	0.0		0.0		
12/28/2009	nign	۷	NO	0.0	NO	0.0		
1/4/2010	high	2	No	0.0	No	0.0		
1/5/2010	high	2	No	0.0		0.0		
1/5/2010	high	2	No	0.0		0.0		
1/7/2010	high	2	NO	0.0		0.0		
1/11/2010	high	2	NO	0.0		0.0		
1/11/2010	l iigu	Z	NO	0.0		0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal Stage		Boor	n Sheen	Воо	m Sheen	Воо	m Sheen
.			Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium		0,50		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
1/14/2010	high	2	No	0.0	No	0.0	(100/110/	(000110100)
1/19/2010	high	2	No	0.0	No	0.0		
1/20/2010	high	2	No	0.0	No	0.0		
1/21/2010	high	2	No	0.0	No	0.0		
1/25/2010	high	2	No	0.0	No	0.0		
1/27/2010	high	2	No	0.0	No	0.0		
2/1/2010	high	2	No	0.0	No	0.0		
2/2/2010	high	2	No	0.0	No	0.0		
2/8/2010	high	2	No	0.0	No	0.0		
2/9/2010	high	2	No	0.0	No	0.0		
2/16/2010	high	2	No	0.0	No	0.0		
2/17/2010	high	2	No	0.0	No	0.0		
2/18/2010	high	2	No	0.0	No	0.0		
2/19/2010	high	2	No	0.0	No	0.0		
2/22/2010	high	2	No	0.0	No	0.0		
3/1/2010	high	2	No	0.0	Yes	1.0		
3/8/2010	high	2	No	0.0	No	0.0		
3/12/2010	high	2	No	0.0	No	0.0		
3/16/2010	high	2	No	0.0	No	0.0		
3/17/2010	medium	1	No	0.0	No	0.0		
3/19/2010	high	2	No	0.0	No	0.0		
3/22/2010	high	2	No	0.0	No	0.0		
3/25/2010	high	2	No	0.0	No	0.0		
3/30/2010	high	2	No	0.0	No	0.0		
3/31/2010	high	2	No	0.0	No	0.0		
4/1/2010	high	2	No	0.0	No	0.0		
4/2/2010	high	2	No	0.0	No	0.0		
4/5/2010	high	2	No	0.0	No	0.0		
4/6/2010	high	2	No	0.0	No	0.0		
4/9/2010	medium	1	No	0.0	No	0.0		
4/12/2010	medium	1	No	0.0	No	0.0		
4/14/2010	medium	1	No	0.0	No	0.0		
4/15/2010	medium	1	No	0.0	No	0.0		
4/16/2010	medium	1	No	0.0	No	0.0		
4/19/2010	high	2	No	0.0	No	0.0		
4/20/2010	high	2	No	0.0	No	0.0		
4/27/2010	high	2	No	0.0	No	0.0		
4/28/2010	high	2	No	0.0	No	0.0		
4/29/2010	high	2	No	0.0	No	0.0		
5/3/2010	high	2	No	0.0	No	0.0		
5/5/2010	medium	1	No	0.0	No	0.0		
5/6/2010	medium	1	No	0.0	No	0.0		
5/7/2010	medium	1	No	0.0	No	0.0		
5/10/2010	medium	1	No	0.0	No	0.0		
5/1//2010	high	2	No	0.0	NO	0.0		
5/18/2010	high	2	No	0.0	NO	0.0		
5/24/2010	low	U	NO	0.0	NO	0.0		
6/1/2010	meaium	1	NO N-	0.0		0.0		
6///2010	IOW	U	NO No	0.0	NO NE	0.0		
6/9/2010	IOW	U	NO	0.0	I NO	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal Stage		Boor	n Sheen	Воо	m Sheen	Воо	m Sheen
_ .			Ohse	rvations	Obs	ervations	Ohse	ervations
Date	Low Medium		0030		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
6/10/2010	low	0	No	0.0	No	0.0	(100/110/	(500 110103)
6/14/2010	high	2	No	0.0	No	0.0		
6/16/2010	high	2	No	0.0	No	0.0		
6/17/2010	medium	1	No	0.0	No	0.0		
6/21/2010	low	0	No	0.0	No	0.0		
6/24/2010	low	0	No	0.0	No	0.0		
6/28/2010	high	2	No	0.0	No	0.0		
7/6/2010	low	0	No	0.0	No	0.0		
7/8/2010	low	0	No	0.0	No	0.0		
7/12/2010	medium	1	No	0.0	No	0.0		
7/13/2010	medium	1	No	0.0	No	0.0		
7/14/2010	medium	1	No	0.0	No	0.0		
7/15/2010	high	2	No	0.0	No	0.0		
7/16/2010	high	2	No	0.0	No	0.0		
7/19/2010	low	0	No	0.0	Ves	1.0		
7/20/2010	medium	1	No	0.0	Ves	1.0		
7/21/2010	low	0	No	0.0	No	0.0		
7/22/2010	low	0	No	0.0	No	0.0		
7/26/2010	high	1	No	0.0	No	0.0		
7/28/2010	medium	1	No	0.0	No	0.0		
7/29/2010	medium	1	No	0.0	No	0.0		
8/2/2010	medium	1	No	0.0	No	0.0		
8/3/2010	low	0	No	0.0	No	0.0		
8/9/2010	medium	1	No	0.0	No	0.0		
8/11/2010	high	2	No	0.0	No	0.0		
8/16/2010	medium	1	No	0.0	No	0.0		
8/18/2010	low	0	No	0.0	No	0.0		
8/19/2010	low	0	No	0.0	No	0.0		
8/23/2010	medium	1	No	0.0	No	0.0		
8/24/2010	high	2	No	0.0	No	0.0		
8/30/2010	high	2	No	0.0	No	0.0		
8/31/2010	high	2	No	0.0	No	0.0		
9/1/2010	high	2	No	0.0	No	0.0		
9/2/2010	low	0	No	0.0	No	0.0		
9/3/2010	low	0	No	0.0	No	0.0		
9/7/2010	low	0	No	0.0	No	0.0		
9/14/2010	medium	1	No	0.0	No	0.0		
9/15/2010	low	0	No	0.0	No	0.0		
9/16/2010	low	0	No	0.0	No	0.0		
9/20/2010	medium	1	No	0.0	No	0.0		
9/21/2010	medium	1	No	0.0	No	0.0		
9/22/2010	medium	1	No	0.0	No	0.0		
9/27/2010	hiøh	2	No	0.0	No	0.0		
9/30/2010	high	2	No	0.0	No	0.0		
10/4/2010		0	No	0.0	No	0.0		
10/7/2010	medium	1	No	0.0	No	0.0		
10/11/2010	hiøh	2	No	0.0	No	0.0		
10/14/2010	medium	2 1	No	0.0	No	0.0		
10/18/2010	medium	1	No	0.0	No	0.0		
10/19/2010	medium	1	No	0.0	No	0.0		

			Loading	g Rack Area	Warehou	ise Area North	Warehou	se Area South
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boom Sheen	
Data		U	Obse	rvations	Observations		Obse	ervations
Date	Low Medium							
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
10/20/2010	medium	1	No	0.0	No	0.0	(100)110)	(,
10/21/2010	medium	-	No	0.0	No	0.0		
10/25/2010	high	2	No	0.0	No	0.0		
10/29/2010	high	2	No	0.0	No	0.0		
11/1/2010	low	0	No	0.0	No	0.0		
11/2/2010	medium	1	No	0.0	No	0.0		
11/8/2010	high	2	No	0.0	No	0.0		
11/11/2010	high	2	No	0.0	No	0.0		
11/15/2010	medium	1	No	0.0	No	0.0		
11/16/2010	medium	1	No	0.0	No	0.0		
11/17/2010	medium	1	No	0.0	No	0.0		
11/18/2010	medium	1	No	0.0	No	0.0		
11/22/2010	high	2	No	0.0	No	0.0		
11/29/2010	high	2	No	0.0	No	0.0		
11/30/2010	medium	1	No	0.0	No	0.0		
12/1/2010	medium	1	No	0.0	No	0.0		
12/2/2010	medium	1	No	0.0	No	0.0		
12/3/2010	medium	1	No	0.0	No	0.0		
12/6/2010	high	2	No	0.0	No	0.0		
12/7/2010	high	2	No	0.0	No	0.0		
12/8/2010	high	2	No	0.0	No	0.0		
12/13/2010	high	2	No	0.0	No	0.0		
12/14/2010	high	2	No	0.0	No	0.0		
12/15/2010	high	2	No	0.0	No	0.0		
12/16/2010	high	2	No	0.0	No	0.0		
12/20/2010	high	2	No	0.0	No	0.0		
12/22/2010	high	2	No	0.0	No	0.0		
12/23/2010	high	2	No	0.0	No	0.0		
12/24/2010	high	2	No	0.0	No	0.0		
12/27/2010	high	2	No	0.0	No	0.0		
	<u> </u>							
1/3/2011	high	2	No	0.0	No	0.0		
1/10/2011	high	2	No	0.0	No	0.0		
1/17/2011	high	2	No	0.0	No	0.0		
1/18/2011	high	2	No	0.0	No	0.0		
1/19/2011	high	2	No	0.0	No	0.0		
1/24/2011	high	2	No	0.0	No	0.0		
1/27/2011	high	2	No	0.0	No	0.0		
1/31/2011	high	2	No	0.0	No	0.0		
2/4/2011	high	2	No	0.0	No	0.0		
2/7/2011	high	2	No	0.0	No	0.0		
2/8/2011	high	2	No	0.0	No	0.0		
2/14/2011	high	2	No	0.0	No	0.0		
2/15/2011	high	2	No	0.0	No	0.0		
2/16/2011	high	2	No	0.0	No	0.0		
2/22/2011	high	2	No	0.0	No	0.0		
2/25/2011	high	2	No	0.0	No	0.0		
2/28/2011	high	2	No	0.0	No	0.0		
3/2/2011	high	2	No	0.0	No	0.0		
3/9/2011	high	2	No	0.0	No	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Вос	om Sheen	Воо	m Sheen
.			Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium		0,50		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
3/10/2011	high	2	No	0.0	No	0.0	(100/110/	(000 110 100)
3/11/2011	high	2	No	0.0	No	0.0		
3/1//2011	high	2	No	0.0	No	0.0		
3/21/2011	high	2	No	0.0	No	0.0		
3/22/2011	high	2	No	0.0	No	0.0		
3/22/2011	high	2	No	0.0	No	0.0		
3/23/2011	high	2	No	0.0	No	0.0		
3/24/2011	high	2	No	0.0	No	0.0		
3/20/2011	high	2	No	0.0	No	0.0		
3/23/2011 A/A/2011	high	2	No	0.0	No	0.0		
4/4/2011	high	2	No	0.0	No	0.0		
4/3/2011	high	2	No	0.0	No	0.0		
4/11/2011	high	2	No	0.0	No	0.0		
4/12/2011	high	2	No	0.0	No	0.0		
4/13/2011	high	2	No	0.0	No	0.0		
4/19/2011	high	2	No	0.0	No	0.0		
4/20/2011	high	2	No	0.0	No	0.0		
4/21/2011	high	2	No	0.0	No	0.0		
4/22/2011	modium	2	No	0.0	No	0.0		
4/23/2011	medium	1	No	0.0	Vec	0.0		
4/2//2011	high	1	No	0.0	No	1.0		
5/2/2011	high	2	No	0.0	No	0.0		
5/9/2011	modium	2	No	0.0	No	0.0		
5/10/2011	high	1	No	0.0	No	0.0		
5/18/2011	high	2	No	0.0	No	0.0		
5/19/2011	high	2	No	0.0	No	0.0		
6/1/2011	medium	1	No	0.0	No	0.0		
6/6/2011	high	1	No	0.0	No	0.0		
6/10/2011	modium	2	No	0.0	Voc	0.0		
6/12/2011	low	1	No	0.0	No	1.0		
6/14/2011	low	0	No	0.0	No	0.0		
6/15/2011	low	0	No	0.0	No	0.0		
6/20/2011	high	2	No	0.0	No	0.0		
6/20/2011	modium	2	No	0.0	Voc	0.0		
6/22/2011	modium	1	No	0.0	No	0.3		
6/23/2011	low	1	No	0.0	No	0.0		
6/20/2011	modium	1	No	0.0	No	0.0		
7/6/2011	high	1	No	0.0	No	0.0		
7/0/2011	low	2	No	0.0	No	0.0		
7/11/2011	high	2	No	0.0	No	0.0		
7/18/2011	nign	2	NO	0.0	NO	0.0		
7/19/2011	high	2	No	0.0	No	0.0		
7/20/2011	low	2	No	0.0	No	0.0		
7/20/2011	modium	1	No	0.0		0.0		
2/1/2011 2/1/2011	high	1 2	NO	0.0		0.0		
0/1/2011	nign	2	NO	0.0		0.0		
0/0/2011 0/15/2011	low	0	NO	0.0		0.0		
0/15/2011	nign biab	2	NO	0.0		0.0		
0/10/2011	high	2	NO	0.0		0.0		
0/1//2011		2	No	0.0		0.0		
8/22/2011	IOW	U	INO	0.0		0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal Stage		Boor	n Sheen	Воо	m Sheen	Boo	m Sheen
		etuge	Ohse	rvations	Ohs	ervations	Ohse	arvations
Date	Low Modium		0036		003		0030	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Ves/No)	(See Notes)	(Ves/No)	(See Notes)
8/24/2011	high	2	No	0.0	No	0.0	(103/110)	(See Notes)
8/24/2011	medium	2	No	0.0	No	0.0		
0/23/2011	modium	1	No	0.0	No	0.0		
0/6/2011	modium	1	No	0.0	No	0.0		
9/0/2011	high	1	No	0.0	No	0.0		
9/12/2011	high	2	No	0.0	No	0.0		
9/13/2011	high	2	No	0.0	No	0.0		
9/14/2011	modium	2	No	0.0	No	0.0		
9/20/2011	medium	1	NO	0.0	NO	0.0		
9/20/2011	high	1	NO	0.0	NO	0.0		
9/2//2011	high	2	NO	0.0	NO	0.0		
9/20/2011	high	2	NO	0.0	NO	0.0		
9/29/2011	high	2	NO	0.0	NO	0.0		
10/5/2011	high	2	NO	0.0	NO	0.0		
10/10/2011	high	2	No	0.0	No	0.0		
10/11/2011	nign	2	NO	0.0	NO	0.0		
10/12/2011	nign	2	NO	0.0	NO	0.0		
10/17/2011	nign	2	NO	0.0	NO	0.0		
10/18/2011	nign	2	NO	0.0	NO	0.0		
10/19/2011	nign	2	NO	0.0	NO	0.0		
10/20/2011	nign	2	NO	0.0	NO	0.0		
10/24/2011	mealum	1	NO	0.0	NO	0.0		
10/31/2011	nign	2	NO	0.0	NO	0.0		
11/8/2011	mealum	1	NO	0.0	NO	0.0		
11/14/2011	nign	2	NO	0.0	NO	0.0		
11/21/2011	mealum	1	NO	0.0	NO	0.0		
11/22/2011	nign	2	NO	0.0	NO	0.0		
11/23/2011	nign	2	NO	0.0	NO No	0.0		
11/28/2011	nign	2	NO	0.0	NO No	0.0		
11/29/2011	nign	2	NO	0.0	NO No	0.0		
12/5/2011	mealum	1	NO	0.0	NO No	0.0		
12/12/2011	nign	2	NO	0.0	NO No	0.0		
12/13/2011	nign	2	NO	0.0	NO No	0.0		
12/14/2011	nign	2	NO	0.0	NO No	0.0		
12/19/2011	nign	2	NO	0.0	NO No	0.0		
12/20/2011	nign	2	NO	0.0	NO No	0.0		
12/21/2011	nign	2	NO	0.0	NO No	0.0		
12/2//2011	nign	Ζ	NO	0.0	NO	0.0		
1/2/2012	hiah	2	No	0.0	No	0.0		
1/3/2012	nign	2	NO	0.0	NO	0.0		
1/9/2012	nign	2	NO	0.0	NO	0.0		
1/1//2012	high	2	No	0.0	NO	0.0		
1/23/2012	high	2	No	0.0	NO	0.0		
1/24/2012	high	2	No	0.0	NO	0.0		
1/25/2012	high	2	No	0.0	No	0.0		
1/27/2012	high	2	No	0.0	No	0.0		
1/30/2012	high	2	No	0.0	No	0.0		
2/6/2012	high	2	No	0.0	No	0.0		
2/13/2012	high	2	No	0.0	No	0.0		
2/21/2012	medium	1	No	0.0	No	0.0		
2/27/2012	high	2	No	0.0	No	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Boo	om Sheen	Boom Sheen	
.			Observations		Obs	ervations	Obse	ervations
Date	Low Medium		0,50		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
2/24/2012	high	2	No	0.0	No	0.0	(100/110/	(000110100)
3/1/2012	medium	1	No	0.0	No	0.0		
3/2/2012	high	2	No	0.0	No	0.0		
3/5/2012	high	2	No	0.0	No	0.0		
3/12/2012	high	2	No	0.0	No	0.0		
3/13/2012	high	2	No	0.0	No	0.0		
3/14/2012	medium	1	No	0.0	No	0.0		
3/15/2012	high	2	No	0.0	No	0.0		
3/19/2012	high	2	No	0.0	No	0.0		
3/20/2012	high	2	No	0.0	No	0.0		
.3/21/2012	high	2	No	0.0	No	0.0		
3/22/2012	high	2	No	0.0	No	0.0		
3/26/2012	high	2	No	0.0	No	0.0		
3/28/2012	high	2	No	0.0	No	0.0		
4/2/2012	medium	1	No	0.0	No	0.0		
4/5/2012	medium	1	No	0.0	No	0.0		
4/9/2012	high	2	No	0.0	No	0.0		
4/16/2012	medium	1	No	0.0	No	0.0		
4/17/2012	medium	1	No	0.0	No	0.0		
4/18/2012	high	2	No	0.0	No	0.0		
4/19/2012	medium	1	No	0.0	No	0.0		
4/23/2012	medium	1	No	0.0	No	0.0		
4/30/2012	medium	1	No	0.0	No	0.0		
5/2/2012	medium	1	No	0.0	No	0.0		
5/7/2012	high	2	No	0.0	No	0.0		
5/8/2012	high	2	No	0.0	No	0.0		
5/14/2012	medium	1	No	0.0	No	0.0		
5/15/2012	low	0	No	0.0	No	0.0		
5/16/2012	medium	1	No	0.0	No	0.0		
5/21/2012	high	2	No	0.0	No	0.0		
5/22/2012	high	2	No	0.0	No	0.0		
5/23/2012	high	2	No	0.0	No	0.0		
5/24/2012	high	2	No	0.0	No	0.0		
5/29/2012	high	2	No	0.0	No	0.0		
5/31/2012	low	0	No	0.0	No	0.0		
6/4/2012	medium	1	No	0.0	No	0.0		
6/11/2012	medium	1	No	0.0	No	0.0		
6/12/2012	medium	1	No	0.0	No	0.0		
6/13/2012	medium	1	No	0.0	No	0.0		
6/20/2012	high	2	No	0.0	No	0.0		
6/25/2012	medium	1	No	0.0	No	0.0		
7/2/2012	low	0	No	0.0	No	0.0		
7/9/2012	medium	1	No	0.0	Yes	0.5		
7/10/2012	high	2	No	0.0	No	0.0		
7/11/2012	high	2	No	0.0	No	0.0		
7/12/2012	high	2	No	0.0	Yes	0.5		
7/16/2012	low	0	No	0.0	No	0.0		
7/17/2012	low	0	No	0.0	No	0.0		
7/19/2012	low	0	No	0.0	No	0.0		
7/20/2012	low	0	No	0.0	No	0.0		

Date Tidal Stage Boom Sheen Observations Sheen Sheen Rating Sheen Rating Sheen Rating 7/23/2012 Inkin 2 No 0.0 No 0.0 No 0.0 8/13/2012 Inkin 2 No 0.0 No 0.0 No 0.0 8/13/2012 Inkin 2 No 0.0 No 0.0 No 0.0 8/13/2012 Inkin 2 No 0.0 No 0.0 No 0.0 <th></th> <th></th> <th></th> <th colspan="2">Loading Rack Area</th> <th colspan="2">Warehouse Area North</th> <th colspan="2">Warehouse Area South</th>				Loading Rack Area		Warehouse Area North		Warehouse Area South	
Date Observations Observations Observations Observations 1000, Medium (high Tide Rating (see Notes) Sheen (see Notes) Sheen Rating (see Notes) <th></th> <th>Tidal</th> <th>Stage</th> <th>Boor</th> <th>n Sheen</th> <th>Boo</th> <th>m Sheen</th> <th>Boo</th> <th>m Sheen</th>		Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
Date Took, Medium (ebb & flood), High Tode Rating (See Notes) Sheen Rating (See Notes) 8/12/2012 Ing 0 No 0.0 No 0.0 No 0.0 8/12/2012 Ing 0 No 0.0 No 0.0 No 0.0 8/23/2012 Ing 2 No 0.0 No 0.0 No 0.0 9/12/2012 Ing 2 No 0	Data		U	Obse	rvations	Obs	ervations	Obse	ervations
ebb & Rootl, High Tide Rating (see Notes) Sheen Rating (see Notes) 7/23/2012 high 2 No 0.0 No 0.0 8/6/2012 high 2 No 0.0 No 0.0 8/7/2012 medium 1 No 0.0 No 0.0 8/10/2012 medium 1 No 0.0 No 0.0 8/11/2012 low 0 No 0.0 No 0.0 8/14/2012 high 2 No 0.0 No 0.0 8/23/2012 high 2 No 0.0 No 0.0 9/2/2011 high 2 No 0.0 No 0.0 9/3/2012 high 2 No 0.0 No 0.0 9/3/2012 high 2 No 0.0 No 0.0 9/3/2012 high	Date	Low. Medium							
High (See Notes) (Yes/No) (See Notes) (Yes/No) (See Notes) 7/23/2012 high 2 No 0.0 No 0.0 8/3/2012 high 2 No 0.0 No 0.0 8/3/2012 medium 1 No 0.0 No 0.0 8/12/2012 medium 1 No 0.0 No 0.0 8/13/2012 new 0 No 0.0 No 0.0 8/13/2012 low 0 No 0.0 No 0.0 8/20/2012 high 2 No 0.0 No 0.0 8/23/2012 low 0 No 0.0 No 0.0 8/23/2012 low 0 No 0.0 No 0.0 9/4/2012 low 0 No 0.0 No 0.0 9/17/2012 high 2 No 0.0 No 0.0 9		(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
7/23/2012 high 2 No 0.0 No 0.0 7/33/2012 low 0 No 0.0 No 0.0 8/6/2012 ingh 2 No 0.0 No 0.0 8/72/2012 medium 1 No 0.0 No 0.0 8/10/2012 low 0 No 0.0 No 0.0 8/11/2012 low 0 No 0.0 No 0.0 8/14/2012 low 0 No 0.0 No 0.0 8/22/2012 logh 0 No 0.0 No 0.0 8/22/2012 logh 2 No 0.0 No 0.0 9/12/2012 high 2 No 0.0 No 0.0 9/12/2012 high 2 No 0.0 No 0.0 9/12/2012 high 2 No 0.0 No 0.0 <t< td=""><td></td><td>High</td><td>(See Notes)</td><td>(Yes/No)</td><td>(See Notes)</td><td>(Yes/No)</td><td>(See Notes)</td><td>(Yes/No)</td><td>(See Notes)</td></t<>		High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
7/30/2012 how 0 No 0.0 No 0.0 8/6/2012 medium 1 No 0.0 No 0.0 8/12/2012 medium 1 No 0.0 No 0.0 8/13/2012 low 0 No 0.0 No 0.0 8/13/2012 low 0 No 0.0 No 0.0 8/20/2012 high 2 No 0.0 No 0.0 8/22/2012 low 0 No 0.0 No 0.0 8/28/2012 low 0 No 0.0 No 0.0 8/28/2012 low 0 No 0.0 No 0.0 9/4/2012 high 2 No 0.0 No 0.0 9/10/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 <td< td=""><td>7/23/2012</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td>, ,</td></td<>	7/23/2012	high	2	No	0.0	No	0.0		, ,
Šk/s/2012 high 2 No 0.0 No 0.0 8/12/2012 medium 1 No 0.0 No 0.0 8/13/2012 low 0 No 0.0 No 0.0 8/14/2012 low 0 No 0.0 No 0.0 8/14/2012 low 0 No 0.0 No 0.0 8/14/2012 low 0 No 0.0 No 0.0 8/28/2012 low 0 No 0.0 No 0.0 8/28/2012 low 0 No 0.0 No 0.0 9/1/2012 high 2 No 0.0 No 0.0 9/1/2012 ligh 2 No 0.0 No 0.0 9/1/2012 ligh 2 No 0.0 No 0.0 9/1/2012 ligh 2 No 0.0 No 0.0 9/1	7/30/2012	low	0	No	0.0	No	0.0		
\$/7/2012 medium 1 No 0.0 No 0.0 8/10/2012 iow 0 No 0.0 No 0.0 8/13/2012 iow 0 No 0.0 No 0.0 8/13/2012 iow 0 No 0.0 No 0.0 8/13/2012 high 2 No 0.0 No 0.0 8/28/2012 iow 0 No 0.0 No 0.0 8/28/2012 iow 0 No 0.0 No 0.0 9/4/2012 high 2 No 0.0 No 0.0 9/1/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 <	8/6/2012	high	2	No	0.0	No	0.0		
8/10/2012 medium 1 No 0.0 No 0.0 8/13/2012 low 0 No 0.0 No 0.0 8/14/2012 low 0 No 0.0 No 0.0 8/15/2012 low 0 No 0.0 No 0.0 8/20/2012 high 2 No 0.0 No 0.0 8/28/2012 low 0 No 0.0 No 0.0 9/4/2012 high 2 No 0.0 No 0.0 9/4/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 <td< td=""><td>8/7/2012</td><td>medium</td><td>1</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></td<>	8/7/2012	medium	1	No	0.0	No	0.0		
§/13/2012 low 0 No 0.0 No 0.0 8/14/2012 low 0 No 0.0 No 0.0 8/15/2012 low 0 No 0.0 No 0.0 8/23/2012 high 2 No 0.0 No 0.0 8/23/2012 low 0 No 0.0 No 0.0 8/29/2012 low 0 No 0.0 No 0.0 9/4/2012 high 2 No 0.0 No 0.0 9/1/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/	8/10/2012	medium	1	No	0.0	No	0.0		
8/14/2012 low 0 No 0.0 No 0.0 8/15/2012 high 2 No 0.0 No 0.0 8/20/2012 high 2 No 0.0 No 0.0 8/23/2012 low 0 No 0.0 No 0.0 8/23/2012 low 0 No 0.0 No 0.0 9/1/2012 high 2 No 0.0 No 0.0 9/1/2012 high 2 No 0.0 No 0.0 9/1/2012 high 2 No 0.0 No 0.0 9/11/2012 low 0 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/	8/13/2012	low	0	No	0.0	No	0.0		
\$/15/2012 low 0 No 0.0 No 0.0 \$/23/2012 high 2 No 0.0 No 0.0 \$/23/2012 low 0 No 0.0 No 0.0 \$/24/2012 low 0 No 0.0 No 0.0 \$/25/2012 low 0 No 0.0 No 0.0 \$/25/2012 high 2 No 0.0 No 0.0 \$/7/2012 high 2 No 0.0 No 0.0 \$/11/2012 low 0 No 0.0 No 0.0 \$/11/2012 logh 2 No 0.0 No 0.0 \$/11/2012 high 2 No 0.0 No 0.0 \$/21/2012 high 2 No 0.0 No 0.0 \$/21/2012 high 2 No 0.0 No 0.0 \$/21/2012 high 2 No 0.0 No 0.0 \$/21/2	8/14/2012	low	0	No	0.0	No	0.0		
\$/20/2012 high 2 No 0.0 No 0.0 \$/23/2012 low 0 No 0.0 No 0.0 \$/24/2012 low 0 No 0.0 No 0.0 \$/24/2012 low 0 No 0.0 No 0.0 \$/26/2012 high 2 No 0.0 No 0.0 \$/1/2012 high 2 No 0.0 No 0.0 \$/2/2/2012 high 2 No 0.0 No 0.0 \$/2/2/2012 high 2 No 0.0 No 0.0 \$/2/2/2012 high 2 No 0.0 No 0.0 \$/2/2	8/15/2012	low	0	No	0.0	No	0.0		
\$/23/2012 high 2 No 0.0 No 0.0 \$/28/2012 low 0 No 0.0 No 0.0 \$/29/2012 low 0 No 0.0 No 0.0 \$/4/2012 high 2 No 0.0 No 0.0 \$/4/2012 <td>8/20/2012</td> <td>high</td> <td>2</td> <td>No</td> <td>0.0</td> <td>No</td> <td>0.0</td> <td></td> <td></td>	8/20/2012	high	2	No	0.0	No	0.0		
B/28/2012 Iow 0 No 0.0 No 0.0 8/29/2012 Inigh 2 No 0.0 No 0.0 9/4/2012 Inigh 2 No 0.0 No 0.0 9/5/2012 Inigh 2 No 0.0 No 0.0 9/10/2012 Iow 0 No 0.0 No 0.0 9/11/2012 Iow 0 No 0.0 No 0.0 9/11/2012 Inigh 2 No 0.0 No 0.0 9/12/2012 Inigh 2 No 0.0 No 0.0 9/20/2012 Inigh 2 No 0.0 No 0.0 9/21/2012 Inigh 2 No 0.0 No 0.0 9/21/2012 Inigh 2 No 0.0 No 0.0 9/21/2012 Inigh 2 No 0.0 No 0.0	8/23/2012	high	2	No	0.0	No	0.0		
b iow 0 No 0.0 No 0.0 9/4/2012 high 2 No 0.0 No 0.0 9/5/2012 high 2 No 0.0 No 0.0 9/1/2012 high 2 No 0.0 No 0.0 9/11/2012 low 0 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/12/2012 high 2 No 0.0 No 0.0 9/2/2/2012 high 2 No 0.0 No 0.0 9/2/2/2012 low 0 No 0.0 No 0.0 9/2/2/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/1/2/2012<	8/28/2012	low	0	No	0.0	No	0.0		
9/4/2012 high 2 No 0.0 No 0.0 9/5/2012 high 2 No 0.0 No 0.0 9/10/2012 low 0 No 0.0 No 0.0 9/11/2012 low 0 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/14/2012 high 2 No 0.0 No 0.0 9/14/2012 high 2 No 0.0 No 0.0 9/19/2012 high 2 No 0.0 No 0.0 9/20/2012 high 2 No 0.0 No 0.0 9/25/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/1/	8/29/2012	low	0	No	0.0	No	0.0		
9/5/2012 high 2 No 0.0 No 0.0 9/7/2012 high 2 No 0.0 No 0.0 9/10/2012 low 0 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/17/2012 high 2 No 0.0 No 0.0 9/18/2012 high 2 No 0.0 No 0.0 9/19/2012 high 2 No 0.0 No 0.0 9/21/2012 high 2 No 0.0 No 0.0 9/25/2102 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 <td< td=""><td>9/4/2012</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></td<>	9/4/2012	high	2	No	0.0	No	0.0		
9/7/2012 high 2 No 0.0 No 0.0 9/10/2012 low 0 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/13/2012 high 2 No 0.0 No 0.0 9/18/2012 high 2 No 0.0 No 0.0 9/20/2012 high 2 No 0.0 No 0.0 9/21/2012 high 2 No 0.0 No 0.0 9/25/2012 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10	9/5/2012	high	2	No	0.0	No	0.0		
9/10/2012 Iow 0 No 0.0 No 0.0 9/11/2012 Iow 0 No 0.0 No 0.0 9/11/2012 high 2 No 0.0 No 0.0 9/18/2012 high 2 No 0.0 No 0.0 9/12/2012 high 2 No 0.0 No 0.0 9/21/2012 high 2 No 0.0 No 0.0 9/22/2012 low 0 No 0.0 No 0.0 9/25/212 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 <	9/7/2012	high	2	No	0.0	No	0.0		
9/11/2012 low 0 No 0.0 No 0.0 9/17/2012 high 2 No 0.0 No 0.0 9/18/2012 high 2 No 0.0 No 0.0 9/19/2012 high 2 No 0.0 No 0.0 9/21/2012 high 2 No 0.0 No 0.0 9/25/2102 low 0 No 0.0 No 0.0 9/25/2102 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/1/2/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0	9/10/2012	low	0	No	0.0	No	0.0		
9/17/2012 high 2 No 0.0 No 0.0 9/18/2012 high 2 No 0.0 No 0.0 9/19/2012 high 2 No 0.0 No 0.0 9/20/2012 high 2 No 0.0 No 0.0 9/21/2012 high 2 No 0.0 No 0.0 9/25/2102 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/9/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0	9/11/2012	low	0	No	0.0	No	0.0		
9/18/2012 high 2 No 0.0 No 0.0 9/19/2012 high 2 No 0.0 No 0.0 9/20/2012 high 2 No 0.0 No 0.0 9/21/2012 high 2 No 0.0 No 0.0 9/25/2102 low 0 No 0.0 No 0.0 9/26/2012 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 <	9/17/2012	high	2	No	0.0	No	0.0		
9/19/2012 high 2 No 0.0 No 0.0 9/20/2012 high 2 No 0.0 No 0.0 9/21/2012 high 2 No 0.0 No 0.0 9/25/2102 low 0 No 0.0 No 0.0 9/26/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 <t< td=""><td>9/18/2012</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	9/18/2012	high	2	No	0.0	No	0.0		
9/20/2012 high 2 No 0.0 No 0.0 9/21/2012 low 0 No 0.0 No 0.0 9/25/2102 low 0 No 0.0 No 0.0 9/26/2012 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 11/2/2012 high 2 No 0.0 No 0.0 <t< td=""><td>9/19/2012</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	9/19/2012	high	2	No	0.0	No	0.0		
9/21/2012 high 2 No 0.0 No 0.0 9/25/2102 low 0 No 0.0 No 0.0 9/26/2012 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/23/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0	9/20/2012	high	2	No	0.0	No	0.0		
9/25/2102 low 0 No 0.0 No 0.0 9/26/2012 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0	9/21/2012	high	2	No	0.0	No	0.0		
9/26/2012 low 0 No 0.0 No 0.0 9/27/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/19/2012 low 0 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/23/2012 high 2 No 0.0 No 0.0 10/23/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 11/17/2012 high 2 No 0.0 No 0.0 11/17/2012 high 2 No 0.0 No 0.0	9/25/2102	low	0	No	0.0	No	0.0		
9/27/2012 low 0 No 0.0 No 0.0 10/1/2012 high 2 No 0.0 No 0.0 10/9/2012 low 0 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 11/7/2012 high 2 No 0.0 No 0.0 11/13/2012 medium 1 No 0.0 No 0.0 11/13/2012 high 2 No 0.0 No 0.0	9/26/2012	low	0	No	0.0	No	0.0		
10/1/2012 high 2 No 0.0 No 0.0 10/9/2012 low 0 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/23/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 11/7/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0	9/27/2012	low	0	No	0.0	No	0.0		
10/9/2012 low 0 No 0.0 No 0.0 10/15/2012 high 2 No 0.0 No 0.0 10/16/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/23/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 10/29/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/13/2012 medium 1 No 0.0 No 0.0 11/27/2012 high 2 No 0.0 No 0.0	10/1/2012	high	2	No	0.0	No	0.0		
10/15/2012 high 2 No 0.0 No 0.0 10/16/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/23/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 10/29/2012 high 2 No 0.0 No 0.0 11/7/2012 high 2 No 0.0 No 0.0 11/13/2012 high 2 No 0.0 No 0.0 11/28/2012 high 2 No 0.0 No 0.0 11/27/2012 high 2 No 0.0 No 0.0 <	10/9/2012	low	0	No	0.0	No	0.0		
10/16/2012 high 2 No 0.0 No 0.0 10/22/2012 high 2 No 0.0 No 0.0 10/23/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 10/29/2012 high 2 No 0.0 No 0.0 11/7/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/13/2012 medium 1 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/28/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 No 0.0 <t< td=""><td>10/15/2012</td><td>high</td><td>2</td><td>No</td><td>0.0</td><td>No</td><td>0.0</td><td></td><td></td></t<>	10/15/2012	high	2	No	0.0	No	0.0		
10/22/2012high2No0.0No0.010/23/2012high2No0.0No0.010/24/2012high2No0.0No0.010/25/2012high2No0.0No0.010/26/2012high2No0.0No0.010/29/2012high2No0.0No0.011/7/2012high2No0.0No0.011/12/2012high2No0.0No0.011/12/2012high2No0.0No0.011/19/2012high2No0.0No0.011/19/2012high2No0.0No0.011/27/2012high2No0.0No0.011/28/2012high2No0.0No0.012/5/2012high2No0.0No0.012/12/2012high2No0.0No0.012/12/2012high2No0.0No0.012/13/2012medium1No0.0No0.012/13/2012medium1No0.0No0.012/14/2012medium1No0.0No0.012/17/2012high2No0.0No0.012/14/2012medium1No0.0No0.0	10/16/2012	high	2	No	0.0	No	0.0		
10/23/2012 high 2 No 0.0 No 0.0 10/24/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 10/29/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/13/2012 medium 1 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/28/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 No 0.0	10/22/2012	high	2	No	0.0	No	0.0		
10/24/2012 high 2 No 0.0 No 0.0 10/25/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 10/29/2012 high 2 No 0.0 No 0.0 11/7/2012 high 2 No 0.0 Yes 0.5 11/12/2012 high 2 No 0.0 No 0.0 11/13/2012 medium 1 No 0.0 No 0.0 11/19/2012 high 2 No 0.0 No 0.0 11/19/2012 high 2 No 0.0 No 0.0 11/19/2012 high 2 No 0.0 No 0.0 11/27/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 No 0.0	10/23/2012	high	2	No	0.0	No	0.0		
10/25/2012 high 2 No 0.0 No 0.0 10/26/2012 high 2 No 0.0 No 0.0 10/29/2012 high 2 No 0.0 No 0.0 11/7/2012 high 2 No 0.0 Yes 0.5 11/12/2012 high 2 No 0.0 No 0.0 11/13/2012 medium 1 No 0.0 No 0.0 11/19/2012 high 2 No 0.0 No 0.0 11/19/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/12/2012 high 2 No 0.0 No 0.0 11/28/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 No 0.0	10/24/2012	high	2	No	0.0	No	0.0		
10/26/2012high2No0.0No0.010/29/2012high2No0.0No0.011/7/2012high2No0.0Yes0.511/12/2012high2No0.0No0.011/13/2012medium1No0.0No0.011/19/2012high2No0.0No0.011/27/2012high2No0.0No0.011/28/2012high2No0.0No0.012/5/2012high2No0.0No0.012/6/2012high2No0.0No0.012/12/2012medium1No0.0No0.012/12/2012medium1No0.0No0.012/12/2012medium1No0.0No0.012/12/2012medium1No0.0No0.012/12/2012medium1No0.0No0.012/14/2012medium1No0.0No0.012/17/2012high2No0.0No0.012/18/2012high2No0.0No0.0	10/25/2012	high	2	No	0.0	No	0.0		
10/29/2012high2No0.0No0.011/7/2012high2No0.0Yes0.511/12/2012high2No0.0No0.011/13/2012medium1No0.0No0.011/19/2012high2No0.0No0.011/27/2012high2No0.0Yes0.511/28/2012high2No0.0No0.012/5/2012high2No0.0No0.012/6/2012high2No0.0Yes0.512/7/2012high2No0.0No0.012/12/2012medium1No0.0Yes0.512/13/2012medium1No0.0No0.012/14/2012medium1No0.0No0.012/17/2012high2No0.0No0.012/14/2012medium1No0.0No0.012/18/2012high2No0.0No0.0	10/26/2012	high	2	No	0.0	No	0.0		
11//2012high2No0.0Yes0.511/12/2012high2No0.0No0.011/13/2012medium1No0.0No0.011/19/2012high2No0.0No0.011/27/2012high2No0.0Yes0.511/28/2012high2No0.0No0.012/5/2012high2No0.0Yes0.512/6/2012high2No0.0Yes0.512/7/2012high2No0.0Yes0.512/12/2012medium1No0.0Yes0.512/13/2012medium1No0.0No0.012/14/2012medium1No0.0No0.012/17/2012high2No0.0No0.012/17/2012high2No0.0No0.012/17/2012high2No0.0No0.012/18/2012high2No0.0No0.0	10/29/2012	high	2	No	0.0	No	0.0		
11/12/2012 nigh 2 No 0.0 No 0.0 11/13/2012 medium 1 No 0.0 No 0.0 11/19/2012 high 2 No 0.0 No 0.0 11/27/2012 high 2 No 0.0 Yes 0.5 11/27/2012 high 2 No 0.0 No 0.0 11/28/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 Yes 0.5 12/6/2012 high 2 No 0.0 Yes 0.5 12/7/2012 high 2 No 0.0 Yes 0.5 12/12/2012 medium 1 No 0.0 No 0.0 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 <td>11///2012</td> <td>high</td> <td>2</td> <td>NO</td> <td>0.0</td> <td>Yes</td> <td>0.5</td> <td></td> <td></td>	11///2012	high	2	NO	0.0	Yes	0.5		
11/13/2012 medium 1 No 0.0 No 0.0 11/19/2012 high 2 No 0.0 No 0.0 11/27/2012 high 2 No 0.0 Yes 0.5 11/28/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 No 0.0 12/6/2012 high 2 No 0.0 Yes 0.5 12/7/2012 high 2 No 0.0 Yes 0.5 12/12/2012 medium 1 No 0.0 No 0.0 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	11/12/2012	nign	2	NO	0.0	NO	0.0		
11/19/2012 nigh 2 No 0.0 No 0.0 11/27/2012 high 2 No 0.0 Yes 0.5 11/28/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 No 0.0 12/6/2012 high 2 No 0.0 Yes 0.5 12/7/2012 high 2 No 0.0 Yes 0.5 12/12/2012 medium 1 No 0.0 Yes 0.5 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	11/13/2012	medium	1	NO	0.0	NO	0.0		
11/2//2012 nigh 2 No 0.0 Yes 0.5 11/28/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 No 0.0 12/6/2012 high 2 No 0.0 Yes 0.5 12/7/2012 high 2 No 0.0 Yes 0.5 12/12/2012 medium 1 No 0.0 Yes 0.5 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	11/19/2012	nign	2	NO	0.0	NO	0.0		
11/28/2012 high 2 No 0.0 No 0.0 12/5/2012 high 2 No 0.0 No 0.0 12/6/2012 high 2 No 0.0 Yes 0.5 12/7/2012 high 2 No 0.0 Yes 0.0 12/12/2012 medium 1 No 0.0 Yes 0.5 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	11/2//2012	nign	2	NO	0.0	Yes	0.5		
12/5/2012 nigh 2 No 0.0 No 0.0 12/6/2012 high 2 No 0.0 Yes 0.5 12/7/2012 high 2 No 0.0 No 0.0 12/12/2012 medium 1 No 0.0 Yes 0.5 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	11/28/2012	nign	2	NO	0.0	NO	0.0		
12/6/2012 nigh 2 No 0.0 Yes 0.5 12/7/2012 high 2 No 0.0 No 0.0 12/12/2012 medium 1 No 0.0 Yes 0.5 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	12/5/2012	nign	2	NO	0.0	NO	0.0		
12/1/2012 nigh 2 No 0.0 No 0.0 12/12/2012 medium 1 No 0.0 Yes 0.5 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	12/0/2012	nign	2	NO	0.0	res	0.5		
12/12/2012 Interdum 1 No 0.0 Yes 0.5 12/13/2012 medium 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	12/7/2012	nign	2	NO	0.0	NO	0.0		
12/15/2012 Intertain 1 No 0.0 No 0.0 12/14/2012 medium 1 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	12/12/2012	medium	1	NO	0.0	res	0.5		
12/14/2012 high 2 No 0.0 No 0.0 12/17/2012 high 2 No 0.0 No 0.0 12/18/2012 high 2 No 0.0 No 0.0	12/13/2012	modium	1	NO	0.0		0.0		
12/17/2012 high 2 No 0.0 No 0.0	12/14/2012	high	1 2	No	0.0		0.0		
	12/12/2012	high	2	No	0.0	No	0.0		
12/19/2012 high 2 No 00 No 00	12/19/2012	high	2	No	0.0	No	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Вос	om Sheen	Boo	m Sheen
			Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium		0,50		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
12/20/2012	high	2	No	0.0	No	0.0	(100,110)	(000 110 100)
12/20/2012	high	2	No	0.0	No	0.0		
		_		0.0		0.0		
1/2/2013	high	2	No	0.0	No	0.0		
1/3/2013	high	2	No	0.0	No	0.0		
1/7/2013	high	2	No	0.0	No	0.0		
1/14/2013	high	2	No	0.0	No	0.0		
1/22/2013	high	2	No	0.0	No	0.0		
1/23/2013	high	2	No	0.0	No	0.0		
1/28/2013	high	2	No	0.0	No	0.0		
1/30/2013	high	2	No	0.0	No	0.0		
1/31/2013	high	2	No	0.0	No	0.0		
2/1/2013	high	2	No	0.0	No	0.0		
2/4/2013	high	2	No	0.0	No	0.0		
2/11/2013	high	2	No	0.0	No	0.0		
2/19/2013	high	2	No	0.0	No	0.0		
2/20/2013	high	2	No	0.0	No	0.0		
2/21/2013	high	2	No	0.0	No	0.0		
2/25/2013	high	2	No	0.0	No	0.0		
3/5/2013	high	2	No	0.0	No	0.0		
3/6/2013	medium	1	No	0.0	No	0.0		
3/11/2013	medium	1	No	0.0	No	0.0		
3/12/2013	high	2	No	0.0	No	0.0		
3/13/2013	high	2	No	0.0	No	0.0		
3/18/2013	high	2	No	0.0	No	0.0		
3/25/2013	high	2	No	0.0	No	0.0		
4/1/2013	high	2	No	0.0	No	0.0		
4/2/2013	high	2	No	0.0	No	0.0		
4/8/2013	medium	1	No	0.0	No	0.0		
4/9/2013	medium	1	No	0.0	No	0.0		
4/10/2013	high	2	No	0.0	No	0.0		
4/15/2013	high	2	No	0.0	No	0.0		
4/16/2013	high	2	No	0.0	No	0.0		
4/18/2013	high	2	No	0.0	No	0.0		
4/22/2013	medium	1	No	0.0	No	0.0		
4/23/2013	medium	1	No	0.0	No	0.0		
4/24/2013	low	0	No	0.0	No	0.0		
4/25/2013	medium	1	No	0.0	No	0.0		
4/29/2013	high	2	No	0.0	No	0.0		
4/30/2013	high	2	No	0.0	No	0.0		
5/6/2013	low	0	No	0.0	No	0.0		
5/7/2013	medium	1	No	0.0	No	0.0		
5/13/2013	high	2	No	0.0	No	0.0		
5/17/2013	medium	1	No	0.0	No	0.0		
5/20/2013	medium	1	No	0.0	No	0.0		
5/21/2013	medium	1	No	0.0	No	0.0		
5/22/2013	medium	1	No	0.0	No	0.0		
5/23/2013	medium	1	No	0.0	No	0.0		
5/28/2013	high	2	No	0.0	No	0.0		
6/3/2013	medium	1	No	0.0	No	0.0	l .	

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Воо	m Sheen
Data		U	Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium							
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
6/5/2013	low	0	No	0.0	No	0.0	(100/110/	(,
6/10/2013	high	2	No	0.0	No	0.0		
6/11/2013	high	2	No	0.0	No	0.0		
6/12/2013	high	2	No	0.0	No	0.0		
6/17/2013	medium	1	No	0.0	No	0.0		
6/18/2013	medium	1	No	0.0	No	0.0		
6/19/2013	medium	1	No	0.0	No	0.0		
6/24/2013	high	2	No	0.0	No	0.0		
6/25/2013	high	2	No	0.0	No	0.0		
6/26/2013	high	2	No	0.0	No	0.0		
7/1/2013	medium	1	No	0.0	No	0.0		
7/8/2013	medium	1	No	0.0	No	0.0		
7/15/2013	medium	1	No	0.0	No	0.0		
7/13/2013	high	1	No	0.0	No	0.0		
7/10/2013	modium	2	No	0.0	No	0.0		
7/22/2013	medium	1	No	0.0	No	0.0		
7/23/2013	mealum	1	NO	0.0	NO	0.0		
7/24/2013	nign	2	NO	0.0	NO No	0.0		
//29/2013	medium	1	NO	0.0	NO	0.0		
8/5/2013	medium	1	NO	0.0	NO	0.0		
8/12/2013	high	2	No	0.0	No	0.0		
8/19/2013	low	0	No	0.0	No	0.0		
8/20/2013	medium	1	No	0.0	No	0.0		
8/21/2013	high	2	No	0.0	No	0.0		
8/26/2013	high	2	No	0.0	No	0.0		
8/27/2013	medium	1	No	0.0	No	0.0		
9/3/2013	medium	1	No	0.0	No	0.0		
9/9/2013	high	2	No	0.0	No	0.0		
9/10/2013	high	2	No	0.0	No	0.0		
9/11/2013	medium	1	No	0.0	No	0.0		
9/12/2013	medium	1	No	0.0	No	0.0		
9/16/2013	low	0	No	0.0	No	0.0		
9/17/2013	medium	1	No	0.0	No	0.0		
9/23/2013	high	2	No	0.0	No	0.0		
9/24/2013	high	2	No	0.0	Yes	0.5		
9/25/2013	high	2	No	0.0	No	0.0		
9/27/2013	high	2	No	0.0	No	0.0		
9/30/2013	medium	1	No	0.0	No	0.0		
10/2/2013	medium	1	No	0.0	No	0.0		
10/7/2013	high	2	No	0.0	No	0.0		
10/9/2013	high	2	No	0.0	No	0.0		
10/14/2013	low	0	No	0.0	No	0.0		
10/15/2013	low	0	No	0.0	No	0.0		
10/21/2013	high	2	No	0.0	No	0.0		
10/28/2013	high	2	No	0.0	No	0.0		
10/29/2013	medium	1	No	0.0	No	0.0		
10/30/2013	medium	1	No	0.0	Yes	0.5		
10/31/2013	medium	1	No	0.0	No	0.0		
11/4/2013	high	2	No	0.0	No	0.0		
11/11/2013	high	2	No	0.0	No	0.0		
11/13/2013	medium	1	No	0.0	Yes	0.5		

			Loading	Rack Area	Warehou	se Area North	Warehou	se Area South
Tidal St		Stage	Boor	n Sheen	Воо	m Sheen	Boom Sheen	
_			Observations		Observations		Observations	
Date	Low Modium		0030		0.03		0.50	
	(obb & flood)	Tido Pating	Sheen	Shoon Pating	Shoon	Shoon Pating	Shoon	Shoon Pating
	(ebb & fiood),	(Soo Notos)	(Ves/No)			(Soo Notos)	(Voc/No)	(Soo Notos)
11/10/2012		(See Notes)	(Tes/NO)		(Tes/NO)		(165/110)	(See Notes)
11/18/2013	hish	1	NO	0.0	NO No	0.0		
11/19/2013	nign	2	NO	0.0	NO	0.0		
11/20/2013	nign	2	NO	0.0	NO	0.0		
11/25/2013	nign	2	NO	0.0	NO No	0.0		
12/2/2013	nign	2	NO	0.0	NO	0.0		
12/3/2013	nign	2	NO	0.0	NO	0.0		
12/9/2013	nign	2	NO	0.0	NO	0.0		
12/16/2013	high	2	No	0.0	NO	0.0		
12/1//2013	high	2	No	0.0	NO	0.0		
12/18/2013	high	2	No	0.0	No	0.0		
12/23/2013	high	2	No	0.0	No	0.0		
12/30/2013	medium	1	No	0.0	No	0.0		
1/2/2014	ht t	2	N1.	0.0	N	0.0		
1/3/2014	nigh	2	NO	0.0		0.0		
1/6/2014	high	2	No	0.0	NO	0.0		
1/13/2014	high	2	No	0.0	No	0.0		
1/14/2014	high	2	No	0.0	No	0.0		
1/15/2014	high	2	No	0.0	No	0.0		
1/21/2014	high	2	No	0.0	No	0.0		
1/27/2014	high	2	No	0.0	No	0.0		
1/28/2014	high	2	No	0.0	No	0.0		
2/4/2014	high	2	No	0.0	No	0.0		
2/10/2014	high	2	No	0.0	No	0.0		
2/11/2014	high	2	No	0.0	No	0.0		
2/12/2014	high	2	No	0.0	No	0.0		
2/18/2014	high	2	No	0.0	No	0.0		
2/21/2014	high	2	No	0.0	No	0.0		
2/24/2014	high	2	No	0.0	No	0.0		
3/3/2014	high	2	No	0.0	No	0.0		
3/10/2014	high	2	No	0.0	No	0.0		
3/11/2014	high	2	No	0.0	No	0.0		
3/12/2014	high	2	No	0.0	No	0.0		
3/17/2014	high	2	No	0.0	No	0.0		
3/19/2014	high	2	No	0.0	No	0.0		
3/20/2014	high	2	No	0.0	No	0.0		
3/24/2014	high	2	No	0.0	No	0.0		
3/26/2014	high	2	No	0.0	No	0.0		
3/27/2014	high	2	No	0.0	No	0.0		
3/31/2014	high	2	No	0.0	No	0.0		
4/2/2014	high	2	No	0.0	No	0.0		
4/7/2014	high	2	No	0.0	No	0.0		
4/14/2014	medium	1	No	0.0	No	0.0		
4/15/2014	medium	1	No	0.0	No	0.0		
4/16/2014	high	2	No	0.0	No	0.0		
4/17/2014	high	2	No	0.0	No	0.0		
4/21/2014	high	2	No	0.0	No	0.0		
4/22/2014	medium	1	No	0.0	No	0.0		
4/23/2014	medium	1	No	0.0	No	0.0		
4/28/2014	medium	1	No	0.0	No	0.0		
4/29/2014	high	2	No	0.0	No	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
Data		U	Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium							
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
5/5/2014	high	2	No	0.0	Yes	0.5	(100)110)	()
5/12/2014	medium	1	No	0.0	No	0.0		
5/13/2014	medium	1	No	0.0	No	0.0		
5/14/2014	medium	1	No	0.0	No	0.0		
5/19/2014	high	2	No	0.0	No	0.0		
5/20/2014	high	2	No	0.0	No	0.0		
5/21/2014	medium	1	No	0.0	No	0.0		
5/27/2014	low	0	No	0.0	No	0.0		
6/2/2014	high	2	No	0.0	No	0.0		
6/9/2014	low	-	No	0.0	No	0.0		
6/10/2014	medium	1	No	0.0	No	0.0		
6/16/2014	high	2	No	0.0	No	0.0		
6/17/2014	high	2	No	0.0	No	0.0		
6/18/2014	high	2	No	0.0	No	0.0		
6/23/2014	low	0	No	0.0	No	0.0		
6/24/2014	low	0	No	0.0	Ves	1.0		
6/30/2014	high	2	No	0.0	No	0.0		
7/72014	medium	1	No	0.0	No	0.0		
7/8/2014	medium	1	No	0.0	No	0.0		
7/1/2014	high	2	No	0.0	No	0.0		
7/15/2014	high	2	No	0.0	No	0.0		
7/21/2014	low	0	No	0.0	No	0.0		
7/25/2014	medium	1	No	0.0	No	0.0		
7/28/2014	high	2	No	0.0	No	0.0		
7/30/2014	low	0	No	0.0	No	0.0		
8/4/2014	medium	1	No	0.0	No	0.0		
8/11/2014	medium	1	No	0.0	No	0.0		
8/12/2014	high	2	No	0.0	No	0.0		
8/13/2014	high	2	No	0.0	No	0.0		
8/18/2014	low	0	No	0.0	No	0.0		
8/20/2014	medium	1	No	0.0	No	0.0		
8/25/2014	high	2	No	0.0	No	0.0		
9/2/2014	low	0	No	0.0	No	0.0		
9/8/2014	medium	1	No	0.0	No	0.0		
9/9/2014	high	2	No	0.0	No	0.0		
9/10/2014	high	2	No	0.0	No	0.0		
9/16/2014	medium	1	No	0.0	No	0.0		
9/17/2014	medium	-	No	0.0	No	0.0		
9/22/2014	medium	-	No	0.0	No	0.0		
9/29/2014	high	2	No	0.0	No	0.0		
10/6/2014	medium	1	No	0.0	No	0.0		
10/9/2014	high	2	No	0.0	No	0.0		
10/14/2014	high	2	No	0.0	No	0.0		
10/15/2014	high	2	No	0.0	No	0.0		
10/20/2014	medium	- 1	No	0.0	No	0.0		
10/21/2014	medium	- 1	No	0.0	No	0.0		
10/27/2014	high	- 2	No	0.0	No	0.0		
10/28/2014	high	2	No	0.0	No	0.0		
11/3/2014	medium	- 1	No	0.0	No	0.0		
11/10/2014	high	2	No	0.0	No	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Boom Sheen	
_ .			Ohse	rvations	Obs	ervations	Ohse	ervations
Date	Low Medium		0030		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Bating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Ves/No)	(See Notes)	(Ves/No)	(See Notes)
11/17/2014	medium	1	No	0.0	No	0.0	(103/110)	(See Notes)
11/17/2014	medium	1	No	0.0	No	0.0		
11/10/2014	modium	1	No	0.0	No	0.0		
11/13/2014	high	1	No	0.0	No	0.0		
12/1/2014	high	2	No	0.0	No	0.0		
12/1/2014	high	2	No	0.0	No	0.0		
12/2/2014	modium	2	No	0.0	No	0.0		
12/3/2014	high	1	No	0.0	No	0.0		
12/4/2014	high	2	No	0.0	No	0.0		
12/0/2014	high	2	No	0.0	No	0.0		
12/9/2014	high	2	No	0.0	No	0.0		
12/10/2014	high	2	No	0.0	No	0.0		
12/15/2014	high	2	No	0.0	No	0.0		
12/16/2014	nign	2	NO	0.0	NO No	0.0		
12/17/2014	nign	2	NO	0.0		0.0		
12/22/2014	nign	2	NO	0.0		0.0		
12/29/2014	nign	Ζ	NO	0.0	NO	0.0		
1/5/2015	high	2	No	0.0	No	0.0		
1/5/2015	nign	2	NO	0.0	NO	0.0		
1/12/2015	high	2	No	0.0		0.0		
1/13/2015	nign	2	NO	0.0	NO NO	0.0		
1/14/2015	nign	2	NO	0.0	NO NO	0.0		
1/20/2015	nign	2	NO	0.0	NO NO	0.0		
1/20/2015	nign	2	NO	0.0	NO NO	0.0		
1/2//2015	nign	2	NO	0.0	NO NO	0.0		
2/3/2015	nign	2	NO	0.0	NO NO	0.0		
2/4/2015	medium	1	NO	0.0	NO NO	0.0		
2/9/2015	nign	2	NO	0.0	NO No	0.0		
2/10/2015	nign	2	NO	0.0	NO No	0.0		
2/11/2015	nign	2	NO	0.0	NO No	0.0		
2/17/2015	medium	1	NO	0.0	NO No	0.0		
2/18/2015	medium	1	NO	0.0	NO No	0.0		
2/23/2015	nign	2	NO	0.0	NO No	0.0		
2/2//2015	nign	2	NO	0.0	NO No	0.0		
3/2/2015	medium	1	NO	0.0	NO No	0.0		
3/9/2015	nign	2	NO	0.0	NO No	0.0		
3/16/2015	medium	1	NO	0.0	NO No	0.0		
3/1//2015	medium	1	NO	0.0	NO No	0.0		
3/18/2015	nign	2	NO	0.0	NO	0.0		
3/19/2015	nign	2	NO	0.0	I NO	0.0		
3/23/2015	high	2	No	0.0	Yes	0.5		
3/24/2015	nign	2	NO	0.0	NO	0.0		
3/25/2015	high	2	No	0.0	NO	0.0		
3/30/2015	medium	1	No	0.0	NO	0.0		
4/1/2015	medium	1	No	0.0	NO	0.0		
4/6/2015	high	2	No	0.0	No	0.0		
4/7/2015	high 	2	No	0.0	No	0.0		
4/13/2015	medium	1	No	0.0	Yes	0.5		
4/14/2015	low	0	No	0.0	No	0.0		
4/15/2015	low	0	No	0.0	No	0.0		
4/20/2015	l high	2	No	0.0	No	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal S	Stage	Boor	n Sheen	Boo	m Sheen	Boom Sheen	
Data			Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium				0.03			
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
4/21/2015	high	2	No	0.0	No	0.0	(100/110/	(000110100)
4/27/2015	medium	1	No	0.0	No	0.0		
4/28/2015	medium	1	No	0.0	No	0.0		
5/4/2015	medium	1	No	0.0	No	0.0		
5/5/2015	high	2	No	0.0	No	0.0		
5/12/2015	high	2	No	0.0	No	0.0		
5/13/2015	medium	2	No	0.0	No	0.0		
5/14/2015	medium	1	No	0.0	No	0.0		
5/18/2015	high	2	No	0.0	No	0.0		
5/26/2015	low	0	No	0.0	No	0.0		
6/1/2015	low	0	No	0.0	No	0.0		
6/8/2015	high	2	No	0.0	No	0.0		
6/0/2015	high	2	No	0.0	No	0.0		
6/10/2015	high	2	No	0.0	No	0.0		
6/15/2015	medium	2	No	0.0	No	0.0		
6/16/2015	low	0	No	0.0	No	0.0		
6/17/2015	medium	1	No	0.0	No	0.0		
6/22/2015	medium	1	No	0.0	No	0.0		
6/20/2015	low	1	No	0.0	No	0.0		
7/6/2015	high	2	No	0.0	No	0.0		
7/0/2013	low	2	No	0.0	No	0.0		
7/13/2015	low	0	No	0.0	No	0.0		
7/14/2015	low	0	No	0.0	No	0.0		
7/13/2013	high	0	No	0.0	No	0.0		
7/20/2015	high	2	No	0.0	No	0.0		
7/21/2015	modium	2	No	0.0	No	0.0		
7/22/2015	low	1	No	0.0	No	0.0		
7/27/2015	low	0	No	0.0	No	0.0		
7/28/2015	low	0	No	0.0	No	0.0		
0/2/2015	high	0	No	0.0	No	0.0		
0/3/2013 0/10/2015	low	2	No	0.0	No	0.0		
8/10/2015	low	0	No	0.0	No	0.0		
8/11/2015 8/17/2015	high	0	No	0.0	No	0.0		
8/2//2015	low	2	No	0.0	No	0.0		
8/24/2015	high	2	No	0.0	No	0.0		
0/1/2015	high	2	No	0.0	No	0.0		
9/1/2015	low	2	No	0.0	No	0.0		
9/8/2013	high	0	No	0.0	No	0.0		
9/14/2015	high	2	No	0.0	No	0.0		
9/15/2015	high	2	No	0.0	No	0.0		
9/10/2015	high	2	No	0.0	No	0.0		
9/1//2015	modium	2	No	0.0	Voc	0.0		
9/21/2015	high	1	No	0.0	No	0.5		
9/28/2015	high	2	No	0.0	No	0.0		
3/23/2015	modium	∠ 1	NO	0.0		0.0		
10/0/2015		1	NO	0.0		0.0		
10/0/2015	10W	U 2	NO	0.0		0.0		
10/12/2015	nign	2	NO	0.0		0.0		
10/13/2015	high	2	No	0.0		0.0		
10/14/2015	high	2	NO	0.0		0.0		
10/13/2012	l nign	2	INO	0.0		0.0		

			Loading	Rack Area	Warehou	se Area North	Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Воог	m Sheen
Date			Obse	rvations	Obs	ervations	Obse	ervations
Date	Low, Medium							
	(ebb & flood),	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
10/20/2015	high	2	No	0.0	No	0.0		
10/26/2015	high	2	No	0.0	No	0.0		
11/2/2015	high	2	No	0.0	No	0.0		
11/10/2015	medium	1	No	0.0	No	0.0		
11/11/2015	medium	1	No	0.0	No	0.0		
11/13/2015	high	2	No	0.0	No	0.0		
11/16/2015	high	2	No	0.0	No	0.0		
11/17/2015	high	2	No	0.0	No	0.0		
11/18/2015	high	2	No	0.0	No	0.0		
11/23/2015	medium	1	No	0.0	No	0.0		
11/30/2015	high	2	No	0.0	No	0.0		
12/2/2015	medium	1	No	0.0	No	0.0		
12/3/2015	medium	1	No	0.0	No	0.0		
12/7/2015	high	2	No	0.0	No	0.0		
12/9/2015	high	2	No	0.0	No	0.0		
12/10/2015	high	2	No	0.0	No	0.0		
12/14/2015	high	2	No	0.0	No	0.0		
12/15/2015	high	2	No	0.0	No	0.0		
12/16/2015	high	2	No	0.0	No	0.0		
12/21/2015	high	2	No	0.0	No	0.0		
12/28/2015	high	2	No	0.0	No	0.0		
1/4/2016	high	2	Ne	0.0	No	0.0		
1/4/2016	nign	2	NO	0.0	NO	0.0		
1/11/2016	nign	2	NO	0.0	NO No	0.0		
1/12/2016	nign	2	NO	0.0	NO	0.0		
1/13/2016	nign	2	NO	0.0	NO	0.0		
1/19/2016	nign	2	NO	0.0	NO	0.0		
1/20/2016	mealum	2	NO	0.0	NO	0.0		
1/25/2016	nign	2	NO	0.0	NO No	0.0		
2/1/2016	nign	2	NO	0.0	NO No	0.0		
2/8/2016	nign	2	NO	0.0	NO No	0.0		
2/9/2016	nign	2	NO	0.0	NO	0.0		
2/10/2016	nign	2	NO	0.0	NO	0.0		
2/10/2010	nign	2	NO	0.0	NO	0.0		
2/22/2016	mealum	1	NO	0.0	NO	0.0		
2/23/2016	nign	2	NO	0.0	NO	0.0		
2/29/2016	nign	2	NO	0.0	res	0.5		
3/7/2016	nign	2	NO	0.0	NO	0.0		
3/8/2016	nign	2	NO	0.0	NO	0.0		
3/9/2016	nign	2	NO	0.0	NO	0.0		
3/10/2016	liigii	2	No	0.0	No	0.0		
3/14/2016	nign	2	INO	0.0	INO No	0.0		
3/15/2016	nign	2	NO	0.0	NO	0.0		
3/10/2016	nign	2	NO	0.0	NO	0.0		
3/21/2016	nign	2	NO No	0.0	NO No	0.0		
3/22/2010 2/20/2016	nign	2	NO	0.0	NO No	0.0		
2/20/2016	high	2	NO	0.0		0.0		
2/20/2010 2/21/2016	high	2	NO	0.0		0.0		
5/31/2010	nign	۲ ۱	NO	0.0		0.0		
4/4/2016	medium	1	INO	0.0		0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
Data		0	Obse	rvations	Obs	ervations	Obse	ervations
Date	Low. Medium							
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
4/5/2016	medium	1	No	0.0	No	0.0		, ,
4/11/2016	high	2	No	0.0	No	0.0		
4/12/2016	high	2	No	0.0	No	0.0		
4/13/2016	medium	1	No	0.0	No	0.0		
4/18/2016	medium	1	No	0.0	No	0.0		
4/19/2016	medium	1	No	0.0	No	0.0		
4/20/2016	medium	1	No	0.0	No	0.0		
4/25/2016	high	2	No	0.0	Yes	0.5		
4/26/2016	high	2	No	0.0	No	0.0		
5/2/2016	medium	1	No	0.0	No	0.0		
5/3/2016	medium	1	No	0.0	No	0.0		
5/4/2016	medium	1	No	0.0	No	0.0		
5/9/2016	high	2	No	0.0	No	0.0		
5/10/2016	high	2	No	0.0	No	0.0		
5/11/2016	high	2	No	0.0	No	0.0		
5/16/2016	medium	1	No	0.0	No	0.0		
5/17/2016	medium	1	No	0.0	No	0.0		
5/18/2016	medium	1	No	0.0	No	0.0		
5/23/2016	high	2	No	0.0	No	0.0		
5/24/2016	high	2	No	0.0	No	0.0		
5/31/2016	low	0	No	0.0	No	0.0		
6/7/2016	high	2	No	0.0	No	0.0		
6/8/2016	high	2	No	0.0	No	0.0		
6/9/2016	high	2	No	0.0	No	0.0		
6/13/2016	low	0	No	0.0	No	0.0		
6/14/2016	low	0	No	0.0	No	0.0		
6/15/2016	medium	1	No	0.0	No	0.0		
6/20/2016	medium	1	No	0.0	No	0.0		
6/26/2016	medium	1	No	0.0	Yes	0.5		
7/6/2016	medium	1	No	0.0	No	0.0		
7/11/2016	medium	1	No	0.0	No	0.0		
7/12/2016	medium	1	No	0.0	No	0.0		
7/21/2016	high	2	No	0.0	No	0.0		
7/25/2016	high	2	No	0.0	No	0.0		
8/2/2016	low	0	No	0.0	No	0.0		
8/8/2016	high	2	No	0.0	No	0.0		
8/15/2016	low	0	No	0.0	No	0.0		
8/1//2016	medium	1	No	0.0	NO	0.0		
8/18/2016	meaium	1	NO	0.0	NO	0.0		
8/22/2016	high	2	No	0.0	NO	0.0		
8/23/2016	high	2	NO	0.0	NO	0.0		
8/24/2016	nign	2	NO	0.0	NO	0.0		
8/25/2016	nign	2	NO	0.0	NO	0.0		
8/29/2016	IOW	0	NO	0.0	NO	0.0		
9/6/2016	nign	2	NO No	0.0	NO N-	0.0		
9/8/2016	nign	2	NO No	0.0	NO N-	0.0		
9/9/2016	nign	2	NO No	0.0		0.0		
9/12/2010	10W	U n	NO	0.0		0.0		
9/19/2010	nign	2	NO No	0.0		0.0		
9/20/2016	nign	2	INO	0.0	I NO	0.0		

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
			Ohse	rvations	Ohs	ervations	Ohse	ervations
Date	Low Modium		0030		0.03		0.50	
	(ebb & flood)	Tide Pating	Sheen	Sheen Pating	Shoon	Sheen Pating	Shoon	Sheen Pating
	High	(See Notes)	(Yes/No)	(See Notes)	(Ves/No)	(See Notes)	(Ves/No)	(See Notes)
9/21/2016	high	2	No	0.0	No	0.0	(103/110)	(See Notes)
9/26/2016	low	0	No	0.0	No	0.0		
9/28/2016	medium	1	No	0.0	No	0.0		
9/20/2010	medium	1	No	0.0	No	0.0		
10/3/2016	high	1	No	0.0	Ves	0.0		
10/5/2010	high	2	No	0.0	No	0.5		
10/10/2016	high	2	No	0.0	No	0.0		
10/17/2016	high	2	No	0.0	No	0.0		
10/17/2010	ingn	2	No	0.0		0.0		
10/18/2016	high	2	No	0.0	NO	0.0		
10/19/2016	nign	2	NO	0.0	Yes	0.5		
10/24/2016	medium	1	No	0.0	NO	0.0		
10/26/2016	medium	1	No	0.0	No	0.0	Yes	1.0
10/27/2016	medium	1	No	0.0	No	0.0	Yes	1.0
10/31/2016	high	2	No	0.0	No	0.0	No	0.0
11/1/2016	high	2	No	0.0	No	0.0	No	0.0
11/2/2016	high	2	No	0.0	No	0.0	Yes	1.0
11/7/2016	high	2	No	0.0	No	0.0	Yes	1.0
11/9/2016	high	2	No	0.0	No	0.0	Yes	1.0
11/14/2016	high	2	No	0.0	No	0.0	Yes	1.0
11/15/2016	high	2	No	0.0	No	0.0	Yes	1.0
11/16/2016	high	2	No	0.0	No	0.0	Yes	1.0
11/21/2016	high	2	No	0.0	No	0.0	Yes	1.0
11/22/2016	high	2	No	0.0	No	0.0	Yes	1.0
11/28/2016	high	2	No	0.0	No	0.0	No	0.0
12/5/2016	high	2	No	0.0	No	0.0	No	0.0
12/6/2016	high	2	No	0.0	No	0.0	No	0.0
12/7/2016	medium	1	No	0.0	No	0.0	No	0.0
12/12/2016	medium	1	No	0.0	No	0.0	No	0.0
12/13/2016	high	2	No	0.0	No	0.0	No	0.0
12/14/2016	medium	1	No	0.0	No	0.0	No	0.0
12/19/2016	high	2	No	0.0	No	0.0	No	0.0
12/27/2016	high	2	No	0.0	No	0.0	No	0.0
1/2/2017	high	2	No	0.0	No	0.0	No	0.0
1/9/2017	medium	1	No	0.0	No	0.0	No	0.0
1/17/2017	high	2	No	0.0	No	0.0	No	0.0
1/18/2017	high	2	No	0.0	No	0.0	No	0.0
1/23/2017	high	2	No	0.0	No	0.0	No	0.0
1/24/2017	medium	1	No	0.0	Yes	0.5	Yes	0.5
1/27/2017	medium	1	No	0.0	No	0.0	No	0.0
1/30/2017	high	2	No	0.0	No	0.0	No	0.0
2/7/2017	medium	1	No	0.0	No	0.0	No	0.0
2/8/2017	medium	1	No	0.0	No	0.0	No	0.0
2/13/2017	high	2	No	0.0	No	0.0	No	0.0
2/14/2017	high	2	No	0.0	No	0.0	No	0.0
2/15/2017	high	2	No	0.0	No	0.0	No	0.0
2/21/2017	medium	1	No	0.0	No	0.0	No	0.0
2/27/2017	high	2	No	0.0	No	0.0	No	0.0
3/6/2017	high	2	No	0.0	No	0.0	No	0.0
3/7/2017	high	2	No	0.0	No	0.0	No	0.0

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Воо	m Sheen	Boo	m Sheen
.			Obse	rvations	Obs	ervations	Obse	ervations
Date	Low Medium		0,50		0.03		0.50	
	(ebb & flood)	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
3/8/2017	high	2	No	0.0	No	0.0	No	0.0
3/13/2017	high	2	No	0.0	Ves	0.5	Ves	0.5
3/14/2017	high	2	No	0.0	No	0.0	No	0.0
3/15/2017	high	2	No	0.0	No	0.0	No	0.0
3/20/2017	medium	1	No	0.0	No	0.0	No	0.0
3/22/2017	medium	1	No	0.0	No	0.0	No	0.0
3/27/2017	high	2	No	0.0	No	0.0	No	0.0
//3/2017	medium	1	No	0.0	No	0.0	No	0.0
4/5/2017	medium	1	No	0.0	No	0.0	No	0.0
4/10/2017	high	2	No	0.0	No	0.0	No	0.0
4/10/2017	high	2	No	0.0	No	0.0	No	0.0
4/12/2017	low	0	No	0.0	No	0.0	No	0.0
4/17/2017	medium	1	No	0.0	No	0.0	No	0.0
4/24/2017	medium	1	No	0.0	No	0.0	No	0.0
5/1/2017	high	2	No	0.0	No	0.0	No	0.0
5/8/2017	low	0	No	0.0	No	0.0	No	0.0
5/15/2017	high	2	No	0.0	No	0.0	No	0.0
5/16/2017	high	2	No	0.0	No	0.0	No	0.0
5/17/2017	high	2	No	0.0	No	0.0	No	0.0
5/22/2017	low	2	No	0.0	No	0.0	No	0.0
5/22/2017	low	0	No	0.0	No	0.0	No	0.0
5/30/2017	high	2	No	0.0	No	0.0	No	0.0
6/5/2017	medium	1	No	0.0	No	0.0	No	0.0
6/6/2017	medium	1	No	0.0	No	0.0	No	0.0
6/7/2017	medium	1	No	0.0	No	0.0	No	0.0
6/12/2017	high	2	No	0.0	Ves	0.0	Ves	0.0
6/13/2017	high	2	No	0.0	Ves	0.5	No	0.0
6/14/2017	high	2	No	0.0	No	0.0	No	0.0
6/19/2017	low	0	No	0.0	No	0.0	No	0.0
6/26/2017	high	2	No	0.0	Ves	0.0	No	0.0
6/27/2017	high	2	No	0.0	No	0.0	No	0.0
7/6/2017	low	0	No	0.0	No	0.0	No	0.0
7/10/2017	medium	1	No	0.0	No	0.0	No	0.0
7/11/2017	medium	1	No	0.0	No	0.0	No	0.0
7/17/2017	low	0	No	0.0	No	0.0	No	0.0
7/18/2017	low	0	No	0.0	No	0.0	No	0.0
7/19/2017	low	0	No	0.0	No	0.0	No	0.0
7/20/2017	low	0	No	0.0	No	0.0	No	0.0
7/24/2017	high	2	No	0.0	No	0.0	No	0.0
7/31/2017	low	-	No	0.0	No	0.0	No	0.0
8/2/2017	medium	1	No	0.0	No	0.0	No	0.0
8/3/2017	low	0		0.0	No	0.0	No	0.0
8/4/2017	low	0			No	0.0	No	0.0
8/5/2017	low	0			No	0.0	No	0.0
8/7/2017	medium	1			No	0.0	No	0.0
8/8/2017	high	2			No	0.0	No	0.0
8//9/2017	high	2			No	0.0	No	0.0
8/10/2017	high	2			No	0.0	No	0.0
8/14/2017	medium	-			No	0.0	No	0.0
8/18/2017	medium	1			No	0.0	No	0.0

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boom Sheen		Boom Sheen		Boom Sheen	
Data		U	Obse	rvations	Obs	ervations	Obse	ervations
Date	Low, Medium							
	(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
	High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
8/23/2017	high	2		(/	No	0.0	No	0.0
8/24/2017	high	2			No	0.0	No	0.0
8/28/2017	high	- 2			No	0.0	No	0.0
8/29/2017	high	2			No	0.0	No	0.0
9/5/2017	medium	1			No	0.0	No	0.0
9/11/2017	medium	1			No	0.0	No	0.0
9/12/2017	medium	1			No	0.0	No	0.0
9/13/2017	medium	1			No	0.0	No	0.0
0/19/2017	low	1			No	0.0	No	0.0
9/10/2017	medium	1			No	0.0	No	0.0
9/19/2017	high	1			No	0.0	No	0.0
9/20/2017	high	2			No	0.0	No	0.0
9/25/2017	high	2			No	0.0	NO	0.0
9/2//2017	nign	2				0.0	NO	0.0
10/2/2017	low	0				0.0	NO	0.0
10/3/2017	low	0			NO	0.0	NO	0.0
10/5/2017	IOW	0			NO	0.0	NO	0.0
10/9/2017	high	2			No	0.0	No	0.0
10/10/2017	high	2			No	0.0	No	0.0
10/11/2017	high	2			No	0.0	No	0.0
10/12/2017	high	2			No	0.0	No	0.0
10/16/2017	low	0			No	0.0	No	0.0
10/17/2017	low	0			No	0.0	No	0.0
10/18/2017	low	0			No	0.0	No	0.0
10/19/2017	high	2			No	0.0	No	0.0
10/24/2017	high	2			No	0.0	No	0.0
10/25/2017	high	2			No	0.0	No	0.0
10/26/2017	high	2			No	0.0	No	0.0
10/30/2017	medium	1			No	0.0	No	0.0
11/2/2017	medium	1			No	0.0	No	0.0
11/6/2017	high	2			No	0.0	No	0.0
11/13/2017	medium	1			No	0.0	No	0.0
11/14/2017	low	0			No	0.0	No	0.0
11/15/2017	medium	1			No	0.0	Yes	1.0
11/20/2017	high	2			Yes	0.5	No	0.0
11/21/2017	high	2			No	0.0	No	0.0
11/22/2017	medium	1			No	0.0	No	0.0
11/27/2017	medium	1			No	0.0	Yes	0.5
11/30/2017	medium	1			No	0.0	Yes	0.5
12/4/2017	high	2			No	0.0	No	0.0
12/5/2017	high	2			No	0.0	No	0.0
12/11/2017	medium	1			No	0.0	No	0.0
12/12/2017	medium	1			No	0.0	No	0.0
12/13/2017	medium	1			No	0.0	No	0.0
12/15/2017	medium	1			No	0.0	No	0.0
12/18/2017	high	2			No	0.0	No	0.0
12/19/2017	high	2			No	0.0	No	0.0
12/20/2017	high	2			No	0.0	No	0.0
12/26/2017	high	2			No	0.0	No	0.0
	-							
1/2/2018	high	2	No	0.0	No	0.0	No	0.0

		Loading Rack Area		Warehouse Area North		Warehouse Area South		
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
. .			Ohse	rvations	Obs	ervations	Ohse	ervations
Date	Low Medium		0030		0.03		0.50	
	(ebb & flood)	Tide Pating	Sheen	Sheen Pating	Sheen	Sheen Pating	Shoon	Sheen Pating
	(EDD & 11000), High		(Ves/No)		(Vec/No)	(See Notes)	(Vec/No)	(See Notes)
1/0/2010	high	(See Notes)	(TCS/NO)		(Tes/NO)	(See Notes)	(Tes/NO)	
1/0/2010	high	2	No	0.0	No	0.0	No	0.0
1/3/2018	high	2	No	0.0	No	0.0	No	0.0
1/11/2018	high	2	No	0.0	No	0.0	No	0.0
1/10/2018	modium	2	No	0.0	No	0.0	No	0.0
1/1//2018	high	1	No	0.0	No	0.0	No	0.0
1/10/2010	high	2	No	0.0	No	0.0	No	0.0
1/22/2018	high	2	No	0.0	No	0.0	No	0.0
2/5/2018	high	2	No	0.0	No	0.0	No	0.0
2/3/2018	modium	2	No	0.0	No	0.0	No	0.0
2/12/2018	medium	1	No	0.0	No	0.0	No	0.0
2/13/2018	medium	1	NO	0.0	NO No	0.0	NO	0.0
2/14/2018	meaium	1	NO	0.0	NO No	0.0	NO	0.0
2/20/2018	nign	2	NO	0.0	NO No	0.0	NO	0.0
2/21/2018	medium	1	NO	0.0	NO No	0.0	NO	0.0
2/22/2018	medium	1	NO	0.0	NO No	0.0	NO	0.0
2/26/2018	medium	1	NO	0.0	NO No	0.0	NO	0.0
3/1/2018	medium	1	NO	0.0	NO No	0.0	NO	0.0
3/5/2018	nign	2	NO	0.0	NO	0.0	NO	0.0
3/12/2018	nign	2	NO	0.0	Yes	0.5	NO	0.0
3/13/2018	nign	2	NO	0.0	NO No	0.0	NO	0.0
3/14/2018	nign	2	NO	0.0	NO No	0.0	NO	0.0
3/19/2018	nign	2	NO	0.0	NO No	0.0	NO	0.0
3/22/2018	nign	2	NO	0.0	NO No	0.0	NO	0.0
3/23/2018	nign	2	NO	0.0	NO No	0.0	NO	0.0
3/26/2018	medium	1	NO	0.0	NO No	0.0	NO	0.0
3/28/2018	medium	1	NO	0.0	NO No	0.0	NO	0.0
4/2/2018	nign	2	NO	0.0	NO	0.0	NO	0.0
4/9/2018	nign	2	NO	0.0	NO	0.0	NO	0.0
4/10/2018	medium	1	NO	0.0	NO	0.0	NO	0.0
4/16/2018	nign	2	NO	0.0	NO	0.0	NO	0.0
4/1//2018	nign	2	NO	0.0	NO	0.0	NO	0.0
4/18/2018	nign	2	NO	0.0	NO	0.0	NO	0.0
4/23/2018	medium	1	NO	0.0	NO	0.0	NO	0.0
4/25/2018	medium	1	NO	0.0	NO	0.0	NO	0.0
4/26/2018	medium	1	NO	0.0	NO	0.0	NO	0.0
4/2//2018	medium	1	NO	0.0	NO	0.0	NO	0.0
4/30/2018	medium	1	NO	0.0	NO	0.0	NO	0.0
5/2/2018	medium	1	NO	0.0	Yes	0.5	NO	0.0
5/7/2018	nign	2	NO	0.0	NO	0.0	NO	0.0
5/10/2018	high	2	No	0.0	NO	0.0	No	0.0
5/14/2018	medium	1	No	0.0	NO	0.0	No	0.0
5/15/2018	high	2	No	0.0	NO	0.0	NO	0.0
5/16/2018	high	2	No	0.0	NO	0.0	NO	0.0
5/17/2018	high 	2	No	0.0	No	0.0	No	0.0
5/21/2018	medium	1	No	0.0	Yes	0.5	No	0.0
5/31/2018	high 	2	No	0.0	No	0.0	No	0.0
6/4/2018	medium	1	No	0.0	No	0.0	No	0.0
6/5/2018	medium	1	No	0.0	NO	0.0	NO	0.0
6/11/2018	low	U	NO	0.0	NO	0.0	NO	0.0
6/12/2018	low	0	No	0.0	No	0.0	No	0.0

				Loading Rack Area W Boom Sheen		Warehouse Area North		Warehouse Area South	
		Tidal	Stage			Boo	m Sheen	Воо	m Sheen
	Det		- 0 -	Obse	rvations	Obs	ervations	Obse	ervations
	Date	Low Medium		0.000		0.03		0.00	
1		(ebb & flood).	Tide Rating	Sheen	Sheen Rating	Sheen	Sheen Rating	Sheen	Sheen Rating
		High	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)	(Yes/No)	(See Notes)
	6/13/2018	low	0	No	0.0	No	0.0	No	0.0
	6/18/2018	high	2	No	0.0	No	0.0	No	0.0
	6/19/2018	high	2	No	0.0	No	0.0	No	0.0
	6/20/2018	medium	1	No	0.0	No	0.0	No	0.0
	6/25/2018	low	0	No	0.0	No	0.0	No	0.0
	7/6/2018	medium	1	No	0.0	No	0.0	No	0.0
	7/11/2018	low	0	No	0.0	No	0.0	No	0.0
	7/12/2018	medium	1	No	0.0	No	0.0	No	0.0
	7/16/2018	high	2	No	0.0	No	0.0	No	0.0
	7/17/2018	high	2	No	0.0	No	0.0	No	0.0
	7/18/2018	high	2	No	0.0	No	0.0	No	0.0
	7/23/2018	low	0	No	0.0	No	0.0	No	0.0
	7/26/2018	low	0	No	0.0	No	0.0	No	0.0
	7/31/2018	high	2	No	0.0	Yes	0.5	No	0.0
	8/6/2018	low	0	No	0.0	No	0.0	No	0.0
	8/0/2018	high	2	No	0.0	No	0.0	No	0.0
	0/14/2010 0/15/2010	high	2	No	0.0	Voc	0.0	No	0.0
	8/15/2018 8/21/2018	lingii	2	No	0.0	No	0.5	No	0.0
	0/21/2010 0/27/2010	IOW	0	No	0.0		0.0	No	0.0
	8/2//2018	nign	2	NO	0.0	NO No	0.0	NO	0.0
	8/28/2018	nign	2	NO	0.0	NO	0.0	NO	0.0
	8/31/2018	meaium	1	NO	0.0	NO	0.0	NO	0.0
	9/4/2018	IOW	0	NO	0.0	NO	0.0	NO	0.0
	9/10/2018	nign	2	NO	0.0	Yes	0.5	Yes	0.5
	9/11/2018	high	2	No	0.0	NO	0.0	No	0.0
	9/12/2018	high	2	No	0.0	NO	0.0	No	0.0
	9/1//2018	low	0	No	0.0	NO	0.0	No	0.0
	9/18/2018	low	0	No	0.0	No	0.0	No	0.0
	9/19/2018	low	0	No	0.0	No	0.0	No	0.0
	9/24/2018	medium	1	No	0.0	No	0.0	No	0.0
	9/26/2018	high	2	No	0.0	No	0.0	No	0.0
	9/27/2018	high	2	No	0.0	No	0.0	No	0.0
	10/1/2018	medium	1	No	0.0	No	0.0	No	0.0
	10/8/2018	medium	1	No	0.0	No	0.0	No	0.0
	10/15/2018	high	2	No	0.0	No	0.0	No	0.0
ļ	10/16/2018	high	2	No	0.0	No	0.0	No	0.0
	10/17/2018	medium	1	No	0.0	No	0.0	No	0.0
	10/23/2018	medium	1	No	0.0	No	0.0	No	0.0
	10/24/2018	high	2	No	0.0	No	0.0	No	0.0
	10/30/2018	high	2	No	0.0	No	0.0	No	0.0
	11/5/2018	medium	1	No	0.0	No	0.0	No	0.0
	11/8/2018	high	2	No	0.0	No	0.0	No	0.0
	11/12/2018	high	2	No	0.0	No	0.0	No	0.0
	11/13/2018	high	2	No	0.0	No	0.0	No	0.0
	11/14/2018	high	2	No	0.0	No	0.0	No	0.0
	11/19/2018	medium	1	No	0.0	No	0.0	No	0.0
	11/26/2018	high	2	No	0.0	No	0.0	No	0.0
	11/27/2018	high	2	No	0.0	No	0.0	No	0.0
	12/5/2018	high	2	No	0.0	No	0.0	No	0.0

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
			Ohse	rvations	Ohs	ervations	Obse	arvations
Date .	Low Modium		0030		0.03		0030	
	(ebb & flood)	Tide Pating	Sheen	Sheen Pating	Sheen	Sheen Pating	Shoon	Sheen Pating
	(EDD & Hood), High		(Yes/No)		(Ves/No)	(See Notes)	(Ves/No)	(See Notes)
12/6/2018	high	2	No	0.0	No	0.0	No	0.0
12/0/2018	high	2	No	0.0	No	0.0	No	0.0
12/10/2018	high	2	No	0.0	Ves	0.0	Vec	0.0
12/11/2010	high	2	No	0.0	No	0.5	No	0.5
12/12/2010	modium	2	No	0.0	No	0.0	NO	0.0
12/17/2010	medium	1	No	0.0	No	0.0	Voc	1.0
12/10/2010	medium	1	No	0.0	No	0.0	Voc	0.5
12/19/2010	high	1	No	0.0	No	0.0	No	1.0
12/21/2010	high	2	No	0.0	No	0.0	No	0.0
12/20/2018	Iligii	2	NU	0.0	NO	0.0	NU	0.0
1/2/2010	medium	1	No	0.0	No	0.0	No	0.0
1/2/2019	high	1	No	0.0	No	0.0	No	0.0
1/1/2019	high	2	No	0.0	No	0.0	No	0.0
1/14/2019	high	2	No	0.0	No	0.0	No	0.0
1/16/2019	high	2	No	0.0	Ves	0.0	Vec	0.0
1/10/2019	high	2	No	0.0	No	0.5	No	0.5
1/22/2019	high	2	No	0.0	No	0.0	No	0.0
2/6/2019	high	2	No	0.0	No	0.0	No	0.0
2/0/2019	high	2	No	0.0	No	0.0	No	0.0
2/13/2019	high	2	No	0.0	No	0.0	No	0.0
2/14/2019	high	2	No	0.0	No	0.0	No	0.0
2/13/2019	medium	2	No	0.0	No	0.0	Vec	0.0
2/10/2019	low	0	No	0.0	No	0.0	No	0.0
2/15/2015	high	2	No	0.0	Ves	0.0	Ves	0.0
2/20/2015	high	2	No	0.0	No	0.0	No	0.0
3/1/2019	high	2	No	0.0	No	0.0	No	0.0
3/11/2019	high	2	No	0.0	No	0.0	No	0.0
3/12/2019	high	2	No	0.0	No	0.0	No	0.0
3/13/2019	high	2	No	0.0	No	0.0	No	0.0
3/18/20119	medium	1	No	0.0	No	0.0	No	0.0
3/19/20119	medium	1	No	0.0	Ves	1.0	No	0.0
3/20/2019	medium	1	No	0.0	No	0.0	No	0.0
3/25/2019	high	2	No	0.0	No	0.0	No	0.0
4/1/2019	medium	1	No	0.0	No	0.0	No	0.0
4/8/2019	high	2	No	0.0	Yes	0.5	Yes	1.0
4/10/2019	high	2	No	0.0	No	0.0	No	0.0
4/17/2019	medium	1	No	0.0	No	0.0	No	0.0
4/22/2019	high	2	No	0.0	Yes	0.5	No	0.0
4/23/2019	high	2	No	0.0	No	0.0	No	0.0
4/24/2019	medium	- 1	No	0.0	No	0.0	No	0.0
4/29/2019	medium	1	No	0.0	No	0.0	No	0.0
5/3/2019	low	0	No	0.0	No	0.0	No	0.0
5/6/2019	high	2	No	0.0	Yes	0,5	No	0.0
5/8/2019	high	2	No	0.0	No	0.0	No	0.0
5/13/2019	low	0	No	0.0	No	0.0	No	0.0
5/14/2019	low	0	No	0.0	No	0.0	No	0.0
5/15/2019	low	0	No	0.0	No	0.0	No	0.0
5/20/2019	high	2	No	0.0	Yes	1.0	Yes	0.5
5/28/2019	low	0	No	0.0	No	0.0	No	0.0
5/29/2019	low	0	No	0.0	No	0.0	No	0.0

			Loading Rack Area		Warehouse Area North		Warehouse Area South	
	Tidal	Stage	Boor	n Sheen	Boo	m Sheen	Boo	m Sheen
			Ohse	rvations	Ohs	ervations	Obse	ervations
Date	Low Modium		0030		003		0030	
	(ebb & flood)	Tide Pating	Sheen	Sheen Pating	Shoon	Sheen Pating	Shoon	Sheen Pating
	High		(Yes/No)	(See Notes)	(Vec/No)	(See Notes)	(Ves/No)	(See Notes)
5/30/2019	low	0	No		No	0.0	(Tes/NO)	0.0
6/3/2019	medium	1	No	0.0	No	0.0	No	0.0
6/10/2010	modium	1	No	0.0	No	0.0	No	0.0
6/11/2019	low	1	No	0.0	No	0.0	No	0.0
6/17/2019	nodium	0	NO	0.0	Voc	0.0	No	0.0
6/17/2019	medium	1	NO	0.0	No	0.5	No	0.0
6/10/2019	high	1	No	0.0	No	0.0	No	0.0
6/24/2019	modium	2	No	0.0	No	0.0	No	0.0
7/1/2019	low	1	No	0.0	NO	0.0	NO	0.0
7/1/2019	low	0	NO	0.0	NO	0.0	No	0.0
7/8/2019	modium	2	No	0.0	NO Voc	0.0	NO	0.0
7/10/2019	low	1	No	0.0	No	0.5	No	0.0
7/10/2019	low	0	No	0.0	NO	0.0	NO	0.0
7/11/2019	low	0	NO	0.0	NO	0.0	NO	0.0
7/22/2019	nign	2	NO	0.0	NO No	0.0	NO	0.0
7/23/2019	nign	2	NO	0.0	NO No	0.0	NO	0.0
7/24/2019	nign	2	NO	0.0	NO	0.0	NO	0.0
7/25/2019	nign	2	NO	0.0	NO	0.0	NO	0.0
7/26/2019	low	0	NO	0.0	NO	0.0	NO	0.0
7/29/2019	low	0	No	0.0	NO	0.0	No	0.0
//30/2019	low	0	No	0.0	NO	0.0	No	0.0
7/31/2019	low	0	No	0.0	No	0.0	No	0.0
8/5/2019	high	2	No	0.0	No	0.0	No	0.0
8/12/2019	low	0	No	0.0	No	0.0	No	0.0
8/13/2019	low	0	No	0.0	No	0.0	No	0.0
8/14/2019	medium	1	No	0.0	Yes	0.5	No	0.0
8/19/2019	high	2	No	0.0	No	0.0	Yes	0.5
8/26/2019	low	0	No	0.0	No	0.0	No	0.0
9/3/2019	medium	1	No	0.0	No	0.0	No	0.0
9/9/2019	low	0	No	0.0	No	0.0	No	0.0
9/10/2019	low	0	No	0.0	No	0.0	No	0.0
9/26/2019	medium	1	No	0.0	No	0.0	No	0.0
10/1/2019	high	2	No	0.0	No	0.0	No	0.0
10/8/2019	low	0	No	0.0	No	0.0	No	0.0
10/15/2019	high	2	No	0.0	No	0.0	No	0.0
10/16/2019	high	2	No	0.0	No	0.0	No	0.0
10/17/2019	high	2	No	0.0	No	0.0	No	0.0
10/22/2019	low	0	No	0.0	No	0.0	No	0.0
10/29/2019	high	2	No	0.0	No	0.0	No	0.0
11/5/2019	medium	1	No	0.0	No	0.0	No	0.0
11/12/2019	medium	1	No	0.0	No	0.0	No	0.0
11/20/2019	medium	1	No	0.0	No	0.0	No	0.0
11/25/2019	high	2	No	0.0	No	0.0	No	0.0
11/26/2019	medium	1	No	0.0	No	0.0	No	0.0
12/3/2019	medium	1	No	0.0	No	0.0	No	0.0
12/9/2019	high	2	No	0.0	No	0.0	No	0.0
12/10/2019	medium	1	No	0.0	No	0.0	No	0.0
12/11/2019	high	2	No	0.0	No	0.0	No	0.0
12/19/2019	medium	1	No	0.0	No	0.0	No	0.0
12/23/2019	low	0	No	0.0	No	0.0	No	0.0
12/27/2019	high	2	No	0.0	No	0.0	No	0.0

Date	Tidal Stage		Loading Rack Area Boom Sheen Observations		Warehouse Area North Boom Sheen Observations		Warehouse Area South Boom Sheen Observations	
	Low, Medium (ebb & flood), High	Tide Rating (See Notes)	Sheen (Yes /No)	Sheen Rating (See Notes)	Sheen (Yes /No)	Sheen Rating (See Notes)	Sheen (Yes /No)	Sheen Rating (See Notes)

Notes:

Bold entries represent sheen detections.

Loading Rack Area Boom removed in August 2017 with concurrance from Ecology due to persistent lack of sheens.

* Sheen Appearance is rated from 0.0 to 3.0 using criteria below;

0.0 No sheen present

1.0 Light sheen visible in one location

2.0 Sheen visible in several locations and is brightly colored

3.0 Sheen covers large areas of boom, outside boom, and/or LNAPL floating on surface

** Tide Level is rated from 0.0 to 2.0 using the criteria below;

0.0 Low Tide

1.0 Medium Tide (Ebb Tide & Flood Tide)

2.0 High Tide

Plant 1 Waterfront Probing Summary Report Site: Former BP Harbor Island Terminal 2/7/2020

Figures

Figure 1 – Site Location Map
Figure 2 – Areas of Remediation - Plant 1
Figure 3 – Cumulative Waterfront LNAPL Recovery Through September 2019
Figure 4 – Plant 1 Probing Investigation, Boring Locations
Figure 5 – Plant 1 Waterfront Hydrograph, 2012 through 2019
Figure 6 – Plant 1 Waterfront, Existing Well Location Map
Figure 7 – Plant 1 Probing Investigation, A-A' Plant 1 Waterfront Cross Section Map
Figure 9 – Plant 1 Probing Investigation, Groundwater Monitoring Analytical Results
Figure 10 – Plant 1 Probing Investigation, A-A' TPH Contour Cross Section Map
Figure 11 – Plant 1 Probing Investigation, B-B' TPH Contour Cross Section Map
Figure 12 – Plant 1 Probing Investigation, C-C' TPH Contour Cross Section Map













Figure 3. Cumulative Waterfront LNAPL Recovery Through September 2019 Site: Former BP Harbor Island Terminal

Note: Soil vapor extraction recovery occurred January 1996 through May 2008.



Figure 5. Plant 1 Waterfront Hydrograph

2012 through 2019

Site: Former BP Harbor Island Terminal



Note: Groundwater monitoring in well MW-06 is conducted voluntarily by TechSolve and is not part of the required monitoring program.











ct\ARCO 21T\Drawings and CAD files\Autocad files\Quarterly GWM Figures: Plant 1 Probin Inv Figs. dwg October 7, 2019 \\SERVER'




and CAD files/Autocad files/Quarterly GWM Figures: Plant 1 Probin Inv Figs. dwg ______0019 ject\ARCD 21T\Drawings \\SERVER\

Plant 1 Waterfront Probing Summary Report Site: Former BP Harbor Island Terminal 2/7/2020

Appendix A

Groundwater Monitoring and Recovery Well Hydrocarbon Graphs































Plant 1 Waterfront Probing Summary Report Site: Former BP Harbor Island Terminal 2/7/2020

<u>Appendix B</u>

Soil Boring Logs



Project N	lame and Lo	ocation:		Borir	g Number:	B-24		
								Page:1_ of 1
SeaPort	Midstream	Partners,	Seattle	Cont	ractor:	_		Drilling Method:
1652 SW	Lander Str	eet Seattle	e, WA 98134	Casc	ade Drilling	5		Direct Push / Hand Auger
				Drill	Lrew:			Drill Rig:
				Data	/Timo Star	<u>JIC</u>		Date/Time Finished:
				Date			0	7/72 11:50
Surface	Elevation:				Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:
Top of C	asing Elevat	ion:				-		
Well Cor	struction In	formation	: ma					
Screenee	d Interval (ft	: bgs):	NN		Screen:	N	7	Water Level While Drilling (ft bgs):
Filter Pa	ck Interval (ft bgs):	NA		Riser:	N	A	T'
Seal Inte	rval (ft bgs)	:	NA		Seal Type	e: N	A	Water Level at Completion (ft bgs):
Grout In	terval (ft bg	s):	NA		Filter Pac	:k: 🛛 🕅	P.	
Depth	Recovery	Sample	OVM / PID	Sheen Te	St: NS (none),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), I HS (hei	/IS (moderate), avy sheen)	Symbol		
0		0' -	1'-0	1'-	U<	SA	Durk Area	to durk brown sill
1			11-0	1	10 J			
2			L	2 ~	NS	(SP)	Jhads (+	nedium Tr Tibe)
3	<i>۱۱ ا</i>		3-0	1-1-	しく	1Y	NU HO	- Oder
	76		4-4	1.1				
			-1-0	4 -	105			
		51	3-0	51~	いく	SIN	~11 5	
0		-	C-A	61	1		3-51	n-e as above
	-111		0-0	0-0	13		- IV,	HLOJA
8	54		7-0	7- K	15		- N	ctata'Bbs
_ 9		1	8-6	8- 1	JS	₽¥.	- 4	-Slight HCOR
10			9-0	4. 5	4'- SC 516 - 10'- 5			
11		10 -	10-63	14) - HAR JM IC- J				inc as above
12			17-14	11' -	NS		~ 3.	trong HC. adma it
13	23		n si	n'	in 2		11	1 J I C COT LOT
14	59"		12-92	12 -			51	
15	50		170 1000	13', -	45		1+C	macst in phats at 13
16			14-244	14' -	ims		Sm	mple Galtelle Gt 13
17		1	12 . 51	K'	55	K.	-16-5,	in the tables
18		15 -	15 154		14	20		
10			11 17	10 -	< <u></u> < <u>></u>			
20			11-100	17' -	55			
20			18-70-	10'-	NS			
			14-0		Ne			
_ 22			a c	191	500	4	121	
23		10-	60-0	20-	N7	DIM	- 100 Dest	me his court
24						RUX	14.5' -T.	marstern de tit techaten
_ 25						UK,	10.	
							r	D SAND IN CUDITERS

TechSolve Environmental, Inc. 7518 NE 169th Street Kenmore, WA 98028 425-402-8277 Info@techsolveinc.com



Project N	Project Name and Location:			Borin	g Number:	B-14		Page: 1 of 1
SeaPort	Midstream	Partners,	Seattle WA 98134	Contr	actor:	,		Drilling Method: Direct Push / Hand Auger
1052 500	Lander Stre	ser Jearne	, WA 30134	Drill	Crew:			Drill Rig:
					Kyle			Geoprobe 7822
				Date/ 子	Time Start	ed:		Date/Time Finished: 21-2219 12:35
Surface I	Elevation:	0.0.			Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:
Well Con	struction In	formation	NR.					Concrete 27
Screened	Interval (ft	bgs):	1 MA		Screen:	AN		Water Level While Drilling (ft bgs):
Filter Pag	k Interval (f	t bgs):	NK.		Riser:	MA		
Seal Inte	rval (ft bgs):		NA		Seal Type	ALA :		Water Level at Completion (ft bgs):
Grout In	terval (ft bgs	5):	KHA.		Filter Pac	K: NA		
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / PID (ppm)	Sheen Te S5 (slight), M HS (hear	St: NS (none), IS (moderate), wy sheen)	USCS Symbol	-	Sample Description
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	40" 40" 60"	5 10 15 1	4'- Oppm 5'- 3ppm 6'- Oppm 9'- Oppm 9'- Oppm 10'- 270 ppm 10'- 270 ppm 10'- 270 ppm 10'- 270 ppm 10'- 270 ppm 10'- 30 ppm 10'- 50 ppm 10'- 50 ppm 10'- 15 ppm 10'- 15 ppm 10'- 15 ppm 10'- 1 ppm	4'-~ 1 5'-1 10-1-10 10-10 10-		SP JSP JSMOL	D-4" cone 4"-S' Dark Sand Gran O'- Wy 10'- 15' De Silfy Se ODr. 4 Highes 15-20 Soils Decrease at 19'. 20' Trans Silty Silty Same	ete gray to durk brown Silly S. Curedium to fine). No iels. Moist. No hydroarbon meras above. roarbon odor observed. Soils attimed from wass to wet. sit gray medium to fine gravin ands. No graville. wet. Hydroarbon ands. No graville. wet. Hydroarbon to obr, sheen at 13' same as above. Hydrocarbon ode and depthy No noticable odor stion to black to gravy alas at 13'

TechSolve Environmental, Inc. 7518 NE 169th Street Kenmore, WA 98028 425-402-8277 Info@techsolveinc.com

ð

1.00



Project N	Project Name and Location:			Borin	g Number:	B-02		Page: 1_ of 1
SeaPort 1652 SW	Midstream / Lander Stre	Partners, eet Seattle	Seattle e. WA 98134	Contr	actor: de Drilling			Drilling Method: Direct Push / Hand Auger
			.,	Drill	Crew:			Drill Rig:
					Kil	Ú		Geoprobe 7822
		ç		Date/	Time Start	ed:		Date/Time Finished:
Surface	Elevation:			_	togged b	15.50	> M Roberts	Protective Cover:
Top of C	asing Elevati	ion:			LOBBCO D	y. 5. Laise	ny m. noberts	emacuele ~ 4"
Well Con	nstruction In	formation	: NA		7			
Screene	d Interval (ft	bgs):	ALA		Screen:	NY	ł	Water Level While Drilling (ft bgs):
Filter Pa	ck Interval (f	ft bgs):	NA	_	Riser:	N	A	12'
Seal Inte	Seal Interval (ft bgs):					: N	A	Water Level at Completion (ft bgs):
Grout In	terval (ft bg	s):	NA	Char Ta	Filter Pac	k: A	MA	
(ft bgs)	(in.)	Interval	(ppm)	Sneen Te SS (slight), M HS (hear	ST: NS (none), IS (moderate), vy sheen)	Symbol		Sample Description
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	30" 30" 60"	D T 5 1 10 1 15 T	5'- Oppm 6' - Oppm 2'- Oppm 6' - Oppm 6' - Oppm 10' - Oppm 10' - Oppm 12' - 320 ppm 13' - 319 ppm 15' - 13 ppm 16' - 170 ppm 16' - 170 ppm 19' - Oppm 19' - Oppm 20' - Oppm	5'-N 6'-NS 6'-NS 10-NS 11-NS 12'-HS 15'-HS 16'-HS 16'-SS 16'-VS 12'-NS 12'-NS	· · · · · · · · · · · · · · · · · · ·	8	D-4" (on 4"-5' Dar Jofin Gran Odor G-10' Son 10-12' Sam 12'-45' Jark Silty Ugdrz 15-20'-San Deur 15':	sete. & gran to dark brown medim e grain silty sands. No rel. Motst. No hydrocarbon me as above gray medium to fine spain sands. No gravel wet. sands. No gravel wet. sarbon ador. me as above. Hydrocarbon odor case of we deptn. No ador at de collected at 12'



Project N SeaPort 1652 SW Surface Top of C Well Cor Screener Filter Pa Seal Inte	Project Name and Location: SeaPort Midstream Partners, Seattle 1652 SW Lander Street Seattle, WA 98134 Surface Elevation: Top of Casing Elevation: Well Construction Information: 1 ^(*) pre-pre- Screened Interval (ft bgs): 3-22 ^(*) Filter Pack Interval (ft bgs): 5-22 ^(*) Seal Interval (ft bgs): 7-5 ^(*) Grout Interval (ft bgs): 7-5 ^(*)				g Number of tid eg ractor: ade Drilling Crew: Yue Yime Star 23 16 Logged b Il Screen: Riser: Seal Type	$\frac{B-25}{4}$ $\frac{Well}{5}$ $\frac{B'\cdot oO}{5}$ $\frac{15'}{7}$ $\frac{15'}{7}$	n/ M. Roberts	Page: 1 of 1 Drilling Method: Direct Push / Hand Auger Drill Rig: Geoprobe 7822 Date/Time Finished: 42314 9.00 Protective Cover: Concrete Water Level While Drilling (ft bgs): ~10' Water Level at Completion (ft bgs):
Grout In	terval (ft bg	to grade		Filter Pac	: 2/1	2		
Depth	Recovery	Sample	OVM / PID	Sheen Te	St: NS (none),	USCS		Samula Description
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	40" 60"	Ø	5'-Oppm 6'- Oppm 7'- Oppm 8'- Oppm 9- Oppm 10'- 17ppm 10'- 17ppm 10'- 17ppm 12'- 3000 13'- 1403 13'- 1403	HS (head 5'- N 6'- N 9'- N 9'- N 9'- N 9'- SS 10'- SS 10'- SS 11- WS 12- HS 13- HS	vy sheen) ト J J J	SP 1	0-4" cone 4"-5' Dav to fi 	rete to gray to dark brown midler ne grain silty sands. No wel. Moist. No hydrocarbon is ame as above. Glight recarbon odor at 10. ame as above. Soils are to at 11-13. imple centred me soil description Soils are the Hydrocarbon odor at 16.
17 18 19 20 21 22 23 24 25	\$0"		17'-0 18'-0 19'-0 19'-0 20'-0	16-550 15-17-5-175 15-17-175 15-175-175	5		sf samp	k collector at 12'



Droject N	lame and la		Porin	a Numbor	P.02		T			
Project N		cation.		E.	g Number.	TOF	13/10701/	Page: 1 of 1		
SeaPort	Midstroam	Partners (Seattle	Contr	actor:	1) h.	J. DURCEIP	Drilling Method:		
1652 SW	Lander Stre	et Seattle	. WA 98134	Casca	de Drilling			Direct Push / Hand Auger		
1002 011			,	Drill	Crew:			Drill Rig:		
					layle			Geoprobe 7822		
~				Date/	Time Start	ed:		Date/Time Finished:		
				21	23/19	5:20		26319 ,9:50		
Surface E	levation:				Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:		
Top of Ca	asing Elevati	ion:						concrete		
Well Con	struction In	formation	: 1" Pray	04612	·					
Screened	Interval (ft	bgs): 7	-22		Screen:	15'		Water Level While Drilling (ft bgs):		
Filter Pac	ck Interval (f	t bgs): 5	-22'		Riser:	7'		8.5		
Seal Inte	rval (ft bgs):	1 +0	5		Seal Type	: Bent	onte Clips	Water Level at Completion (ft bgs):		
Grout Int	terval (ft bg	s): 2 to	yrudy		Filter Pac	k: 2/17		L		
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / PID (ppm)	Sheen Te SS (slight), N HS (hea	St: NS (none), IS (moderate), vy sheen)	USCS Symbol		Sample Description		
_ 0		-O-T					0-4" con	ente		
_ 1							4-5 Dark	avour to dark brown medin		
_ 2	40"					SP				
3						3	to fi	he silfy sands. No grave.		
_ 4							More	st. No hydrocarbon odor.		
5		2'1	5-7	5- N	5		C 1			
6			JAW ((- N)	<		510° sam	e soil description. Steorig		
7	t an la		6	0 10	-	d	Hyd	rocoston odor at 9. Soils		
8	54"		171	1- 103			- trav	wition form moist to wet at		
9			G - Pro alla	87.4	7 /*		~	Fruflacting product observat		
10		1	9-360 an	9:- 11	0		- Fou			
11		10 -	IA ONDROM	16'-1	15		G			
12	1		11-20000	11 -10	IS IS			13-11-1365		
12	59		121 IFUS	12-10	13		- Sun	ete cullent) it lis!		
- 15			17 189 2ppm	13, -10	15			Contres at 19		
_ 14			14 -2400	15-+	5		101			
15		15-	15-3680	11-1	.(-181	Vo FlL edus and		
16	1.17		16 Uppm	n' N	IS		51	VIL		
_ 17	60		D-200 ppm	LI S.	5		661	nng		
18			18-51 ppm	Nº N	15		- Suils	< 10 che n 31/16'		
19			14 - Ppn	2, 1	5	721		Stat 45 4 Store martin 1-1		
20	20 201 20- 4Ph 60 - NS						-Toursition	to burk grants blank		
21			- mm			SW	Sanlai	Policia		
22			٦				at you and	ewy		
22										
23										
- 24										
_ 25										



Project I	roject Name and Location:			Borin	g Number:	B-15		Page: 1 of 1
SeaPort 1652 SW	Midstream / Lander Str	Partners, seattle	Seattle WA 98134	Contr	actor: de Drilling	1		Drilling Method: Direct Push / Hand Auger
				Drill (Crew: 7/	yle.	1.1.1.4.	Drill Rig: Geoprobe 7822
				Date/	Time Start	ed:	10:25	Date/Time Finished:
Surface	Elevation: asing Elevat	ion:			Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:
Well Cor	nstruction In	formation	ALN :				<u> </u>	(or sice)
Screene	d Interval (ft	bgs):	U.A.		Screen:	N IA		Water Level While Drilling (ft bgs):
Filter Pa	ck Interval (1	ft bgs):	MIA		Riser:	NF	ł	9.5
Seal Inte	Seal Interval (ft bgs):					: NF	ł	Water Level at Completion (ft bgs):
Grout In	terval (ft bg	s):	NA		Filter Pac	k: N	A	
Depth	Recovery	Sample	OVM / PID	Sheen Te	st: NS (none),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	HS (hea	vy sheen)	Symbol		
		4					0-4" Con	ncefe
2	40ª					SP	4-5 Das	barray brown niedhow to fine
- 3							50	hy sands. No grand - MGISI.
5							No	myderiorbon odde.
6		5-	5-Oppm	51-1	NR		5-10' 80	sils description same as above.
8	(0 "		V-Oppur	7-1	15		- Si	et at ~ 7.5. Hydrocarbon ador
9	60		S'-Oppm	8-11 G-1	15 15	2	a	+ 9.5'. Free product at 9.5'
		10 f	W'- IPOppm	10-11	15		10-15 Sc	une as above.
12	100 V		11'- 831 ppm 12'-2262 ppm	12-1	45		14 W	AL Odes ANO Staming
_ 14	a		US-53 10M	14 - K	25		Jumpie	Concret at 11
_ 16		157	15'-0ppm	16-N	JS	11	10-10.5 50	at 15" or helpers.
17			46 -OPPM			alt	19.5-20'	Dack are cill dale blot
19			18 Depm			1000		Son to official stand reach inc.
_ 20		_ 1	19 D Ppm			Ŭ		
21		65	20 Oppm					
23				10				
_ 24				())) .				
_ 25								



Project I	Name and Lo	ocation:		Borin	g Number	: B-26		Page: 1 of 1
SeaPort	Midstream	Partners	Seattle	Contr	actor:			Drilling Method:
1652 SM	/ Lander Str	eet Seattle	e, WA 98134	Casca	de Drilling	g		Direct Push / Hand Auger
				Drill	Crew:			Drill Rig:
					Kyle			Geoprobe 7822
				Date	Time Star	ted:		Date/Time Finished:
				71	25/19	10:53		7731191120
Surface	Elevation:	ion:			Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:
Well Cor	asing Lievan	formation	UF					(oncrete
Screene	d Interval (ft	bgs):	1)A		Screen:	L rA		Water Level While Drilling (ft bgs):
Filter Pa	ck Interval (f	t bgs):	NIA		Riser:	NA		To
Seal Inte	erval (ft bgs):		NA		Seal Type	: NA		Water Level at Completion (ft bgs):
Grout In	terval (ft bg	s):	NA		Filter Pac	k: WA		
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / PID (ppm)	Sheen Te SS (slight), N HS (hea	St: NS (none), IS (moderate), ivy sheen)	USCS Symbol		Sample Description
0 1 2	10"	D			no (neavy sneen)		0'4" Cone 4"-5' dark	rete gray/brow medium to fine grain sands. No gravels, Maist
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	45° 58° 60°	15T W	6'- Oppon 6'- Oppon 2'- Oppon 8'- Oppon 10'- 350 ppon 11-1190 ppon 12'- 760 ppon 13'- 4 Uppon 14'- 2000 14'- 2000 16'- 940 ppon 19'- 30 ppon 19'- Oppon 19'- Oppon 20'- Oppon	5 6 7 8 4 10 11 12 13 14 5 16 17 8 4 0 11 12 13 14 5 16 17 8 4 0	55555555555555555555555555555555555555	SP	19-20' dan Pi-to Dark Citty No M Scilo' Sam Odoc from H4 ode (12 West Contanno Scil Sluf 10-15' Sam 0000 19-20' dan plan odor	aray 16000 medium to the gran sands. No gravels. Moist: indecorrection odor observed. e soil decription. Hyprocolor at -9.5'. Soils transition moist to wet at -16" at 9.5'. Free product hurd) observed at 10' how at 16' may be from fing. ne as above. No hydrocarbox at 10'and below. k gray silty clay. Medium hinty. Wet: No hydrocarbox



Project Name and	d Location:		Borin	g Number:	B-04		Page: 1 of 1_	
SeaPort Midstrea	am Partners, Street Seattle	Seattle e. WA 98134	Contr	actor: de Drilling			Drilling Method: Direct Push / Hand Auger	
		-,	Drill C	rew: Kul			Drill Rig: Geoprobe 7822	
			Date/	Time Start	ed:		Date/Time Finished:	
Surface Elevation	1:			Logged by: S. Larsen/ M. Roberts			Protective Cover:	
Top of Casing Ele	vation:	NA		Concrete				
Well Construction	n Information	1: WA		Screen:	(A)		Water Level While Drilling (ft hgs)	
Filter Pack Interva	al (ft bgs):	MA		Riser:	NM			
Seal Interval (ft b	gs):	NA		Seal Type	: NA		Water Level at Completion (ft bgs):	
Grout Interval (ft	bgs):	NA		Filter Pac	K: NA	ł	NA	
Depth Recove (ft bgs) (in.)	ry Sample Interval	OVM / PID (ppm)	Sheen Te SS (slight), M HS (hear	St: NS (none), IS (moderate), vy sheen)	USCS Symbol		Sample Description	
$ \begin{array}{c} 0\\ 1\\ 2\\ 3\\ 4\\ 5\\ 6\\ 7\\ 8\\ 9\\ 9\\ 0\\ 10\\ 11\\ 12\\ 13\\ 6\\ 14\\ 15\\ 16\\ 17\\ 18\\ 19\\ 20\\ 21\\ 22\\ 23\\ 24\\ 25\\ \end{array} $	D 5 10 15 20	5'= Dppm 6'- Dppm 8'- Oppm 8'- Oppm 10'- 1023 ppm 11'- 800 ppm 12'- 169 ppm 13'- Oppm 14'- Oppm 16'-32 ppm 16'-32 ppm 16'-32 ppm 19'- Oppm 19'- Oppm 20-0 ppm 20-0 ppm	567590112199516759	NNNNS H HANDSANNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNN	SP IS'-H SP OH	0-4" Conc 4"-5' Dash 9ra Son in: 0de 5'-10' Sam che 4"-5' Dash 0de 5'-10' Sam 0de - Free 67 - 01hn 3"5 Sempl 15-729' 16' Ny Pass 19-20' Da	sete k gray/brown medium to fine sined silty sands. No grovel. me fragments of seashells soils. Moist. No hydrocathon a observed. me soil description. 9' soils ange from moist to wet. 9' cre is a strong hydrocathon br. product (hemily weathend) 9.5' Woody debas at 13'W/ sitty Clay lens le Collected at 12' drocaston odor. No odor t 16'. Same soil description. above. itte gray silty glay.	



Project N SeaPort 1652 SW Surface I Top of Ca Well Con Screened Filter Pad Seal Inte	Iame and Lo Midstream Lander Stro Lander Stro Elevation: asing Elevation struction In d Interval (ft ck Interval (ft prval (ft bgs))	Seattle , WA 98134 : Prr (7 - 17 ' b 5 - 17 ' b 2 - 5 ' bq	Borir Dep Cont Casc Drill Date H2 Datk 95	ng Number: t of E.do ractor: ade Drilling Crew: View: View: Crew: Logged b Screen: Riser: Seal Type	B-05 qu (U)e)) g ted: 13'(O y: S. Larse 10' 7' 2: peuts	n/M. Roberts	Page: 1 of 1_ Drilling Method: 0 Direct Push / Hand Auger 0 Drill Rig: Geoprobe 7822 Date/Time Finished: 1340 Protective Cover: 0 Conserve 0 Water Level While Drilling (ft bgs): 9.5 Water Level at Completion (ft bgs): 0	
Grout In: Depth (ft bgs)	terval (ft bg: Recovery (in.)	5): Sample Interval	OVM / PID (ppm)	Sheen To SS (slight), I HS (he	Filter Pac est: NS (none), MS (moderate), avv sheen)	USCS Symbol	2 sand	Sample Description
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	36" 54" 60"	Dr. 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	51-0ppm 6 - 0ppm 71 - 0ppm 8 - 0ppm 10 - 100 ppm 10 - 100 ppm 10 - 100 ppm 12 - 160 ppm 141 - 29 ppm 141 - 29 ppm 15 - 16 ppm	567851121345	NNNNS 5555550 NNNNS 55555555	SP J	0-4" cond 24"-5' Lig Sit wa odd Skils At Wet Hyd 10.15' San 15'-20 N	crete Ud gray medium to fine grain ty sands. Few gravels and 2) particles. Dry. No hydrorado or. Fransition to durk gray 7'B65, soils go trown dry to t. at a.s' B65 he collected at 9.5'B65 he collected at 10.5' m as algore



Project N	Project Name and Location:			Borin	ng Number:	B-27		Page: 1 of 1	
SeaPort 1652 SW	Midstream / Lander Stre	Partners, Seattle	Seattle e. WA 98134	Cont	ractor: ade Drilling			Drilling Method: Direct Push / Hand Auger	
1001 011		or ocurre	.,	Drill	Crew:			Drill Rig:	
					Kylo			Geoprobe 7822	
				Date	/Time Start	ed:		Date/Time Finished:	
				F	123/19	14:20		2/23/19/4:50	
Surface I	Elevation:	00:	٨		Logged by: S. Larsen/ M. Roberts			Protective Cover:	
Well Cor	asing Lievan	formation	ALL				i	1 Um were	
Screener	d Interval (ft	has).	· NH		Screen:	ALA		Water Level While Drilling (ft bgs):	
Filter Pa	ck Interval (f	NA NA		Riser:	N/It_		- Water Level While Drining (in Dgs).		
Seal Inte	Filter Pack Interval (ft bgs): NA				Seal Type	· NA		Water Level at Completion (ft hgs):	
Grout In	ton/al (ft has	.).	NA		Filtor Pac	k. NIN		water zever at completion (it bgs).	
Denth	Recovery	Sample	OVM / PID	Sheen T	est' NS (none)	LISCS			
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), I HS (he	MS (moderate), avy sheen)	Symbol		Sample Description	
_ 0		OT					0-4" Conc	ikete	
							4-5 Light	gray transitioning to dark.	
<u> </u>						SP	C.C.	when the survey of the	
_ 3	48						01100	protocol weblocked to this	
4							gra	in sitty samos. No gravels	
5			5- DOOM	5'-1	S		Me	ist. No hydrocarbon odor.	
6		>1		6 - 1	5	11			
7			mago-0	7-10	15	0	5-10' Sam	e as aborro	
8	40"		7-8 pm	8 - 1	5			·····	
0	1-		3- Doom	0 - 10					
_ 10			GI Brain	9-N	5				
_ 10			-088M	10-1	20		10-15' sau	we cal description. H. Anderbon	
_ 11		101	10- Open	11-5	5		maha	abrand of 11' Sail transition	
_ 12			11-3500M	12-5	5.5		0000	organica as it. soit transminist	
13	11		(2'-155pam)	13'-	Ū<		101	n moist to wet at N.	
14	60		17 - 320-10	14'-	hie	1	S 1	- 11 12 1 12/	
15			14/2000	16'			Sample	Collected of IL	
16		1.th	151 150014	12	ivs				
10		10	10-10 ppm	16-	55		15-20 C	une soil description M. 1	
_ 1/			16 - KOPPM	17-:	55		90	and sold october 100 10 10 10 10 10 10 10 10 10 10 10 10	
18	r d		17-6000M	18-1	NS	1	000	or at 18 or below.	
19	00		12 - Alema	14	Je				
20	20				14-NS				
21		105	the about	20-	20-NS				
22			20 - OROM						
- 22			ort						
- 23	_ 23								
_ 24									
_ 25									



Project N	lame and Lo	cation:		Borin	g Number:	B-17		Page: of 1
SeaPort 1652 SW	Midstream Lander Stre	Partners, Seattle	Seattle , WA 98134	Contr Casca	actor: de Drilling			Drilling Method: Direct Push / Hand Auger
				Drill (Crew:			Drill Rig: Geoprope 7822
				Date/	Time Start	ed:		Date/Time Finished:
Surface E	levation:			-17	Logged b	v: S. Larse	n/ M. Roberts	Protective Cover:
Top of Ca	asing Elevati	on:		-				Concrete
Well Con	struction In	formation	:			. 1.3		
Screened	d Interval (ft	bgs): (V	A		Screen:	JA		Water Level While Drilling (ft bgs):
Seal Inte	rval (ft bgs)	T Dgsj: N	A		Seal Type	NA VA		Water Level at Completion (ft bgs):
Grout Int	terval (ft bgs	s): NA			Filter Pac	K: NA		IVA
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / PID (ppm)	Sheen Te SS (slight), M HS (hea	St: NS (none), IS (moderate), vy sheen)	USCS Symbol		Sample Description
_ 0		ÐT			D-4" Con			crete
	719					50	4-5 Do	rk gray lbrown modium to
2	20					12	fi	we grained silty sands. No
4							91	revels. Moist. No hydrocarbor
_ 5			5' Paul	51	h)r		0	Jor.
_ 6		ST	5 oppmi		NS		5-10' 50	une soil description. Soils
_ 7			0-0ppM	6 -			tr	ausition from moist to Wet
8	60 4		2'-OPPM	7-1	NS	+ ~9 ! Hydrocarbon odor at		
10			9-70pm	4'-	55		~	1.
11		10+	10'-77ppm	16-	MIS		10-15 ~	-2' of sluff from heaving
_ 12			11-0 ppm	11 ~	NS:79-	RC	Sa	nos. Same soil description
_ 13	60'		12- Oppm	12-11	USU		- au	w conditions as previous of
- 14			14-11-010	12) - A	5		0	lescribed. some woold vering
_ 15		5+	15-Oppm	15-5	S		17'-500	mele cullected (0:1, 417)
_ 17	11	1	10 - 5 pp M	12-1	US J-SIM	AF.	1572/0	1' DA. CE P I
18	(00		17-232 ppm	17,-	HS		15-60 - 12	1 21 silver them heaving
_ 19			18 - 10 ppm	18-	55			Shinds. June Skil discription
20			19 - 5ppm	14	1		2	13 45 452VR.
- 21		20	20 -2 ppn	26'-	-NS	71	TNOF	76 Oder 47 19.5
22						SM	- Transta	-to black to Jurk Asan
25						OH	Silty	Sandon Clay. 1
25								2 2



Project Name and Location: SeaPort Midstream Partners, Seattle 1652 SW Lander Street Seattle, WA 98134 Surface Elevation: Top of Casing Elevation: Well Construction Information:					ractor: ade Drilling Crew: Yyl /Time Start 24/19 Logged by	B-06 ed: 8:45 y: S. Larse	n/ M. Roberts	Page: <u>1</u> of 1 Drilling Method: Direct Push / Hand Auger Drill Rig: Geoprobe 7822 Date/Time Finished: July GI(5 Protective Cover: Concrete
Screenec	Interval (ft	bgs):	IA IA		Screen:	A //	4	Water Level While Drilling (ft bgs):
Filter Pag	ck Interval (1	t bgs):	NIA NA		Riser:	R/I	A	
Seal Inte	rval (ft bgs):		NA		Seal Type	: N	A	Water Level at Completion (ft bgs):
Grout Int	terval (ft bg	s):	NA		Filter Pac	k: N	IA	
Depth	Recovery	Sample	OVM / PID	Sheen Te	St: NS (none),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), N HS (hea	no (moderate), avy sheen)	Symbol		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	36° 40° 36°	5 15 15	S- Oppm 6- Oppm B'- Oppm 9'- Oppm 10-177 ppm 10-177 ppm 12'-265 ppm 12'-265 ppm 14'-187 ppm 14'-187 ppm 14'-285 ppm 14'-285 ppm 14'-285 ppm 14'-28 ppm 20'-189 pm	5-1-11-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	NS] SIAFF 15 15 15 15 15 15 15 15 15 15	SP J J OH. SP	4"-5' Light Dark Grain Bragn Moire 6-10' it a Dru 8-10 9' k No 8-10 9' k No 10-15' sa 15-18' Sil 18-19' Sil Plo No 19-20 dar silty sand Oder.	quay toansitioning to a quay toansitioning to a quay boom medium to five yed silly sands. Some brickftein- news at ~3'. Dry to t at 5'. No indrocarbon odor. ppears the top 2' are sluft from heaving sands. " moist to wet sands we inag the tansi tion · 10' reaving sands. " moist to wet sands we inag the tansi tion · 10' reaving the tansi tion · 10' rea



Project Name and Location: SeaPort Midstream Partners, Seattle					ng Number:	B-28		Page: of 1	
					ractor: ade Drilling	,		Drilling Method: Direct Push / Hand Auger	
1032 344		et seattle	, AAW 20134	Drill	Crew:			Drill Rig:	
					legele			Geoprobe 7822	
				Date	/Time Star	ed:		Date/Time Finished:	
					124/19	5:30		7/24/19 10:00	
Surface El	evation:				Logged by: S. Larsen/ M. Roberts			Protective Cover:	
Top of Cas	sing Elevati	on:	NA					Congrete	
Well Cons	struction in	formation	: NA		Scroon.			Water Loval While Drilling (ft has)	
Screened	Interval (ft	bgs):	NA		Biser:	NA		a,	
Soal Inter	val (ft bas)	t bgs):	NA		Seal Type	· 10	Λ	Water Level at Completion (ft bgs):	
Grout Inte	erval (ft bgs).	:):	1 114		Filter Pac	k: N	1		
Depth	Recovery	Sample	OVM / PID	Sheen To	est: NS (none),	USCS			
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), I	MS (moderate),	Symbol		Sample Description	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	42" 44" 493" 54"	57 7	5- 0ppm 6-0ppm 9-0ppm 9-0ppm 9-0ppm 10-0ppm 10-0ppm 10-0ppm 12-0ppm 15-0ppm 16-0ppm 16-0ppm 18-0ppm 19-0ppm 201-0ppm	5 6789111234567540	NNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNN	SP J	44-5 Day 912 tex bai 5-10' Sa tr ad 10-15' Se 00 56mple 15'-19' So vo 16'-20' 8 P. 01	ste gray / bravin medium to fine nin silty sands. Wo gravels. o 'small instances of silty day. inds/ champs. Moist. No Grocubor odor. me soil description. Soils anaition from moit to wet a mi. Wo hyproxaboa odor. ame soil description. Hydrorasho for only observed at ~12'. called at 12' ame soils description. Some with swift to 16'. Some ody debn's at 18' loth given silty day. Hudium "estring wet. No hydrorashon dor.	



Project Name and Location: SeaPort Midstream Partners, Seattle 1652 SW Lander Street Seattle, WA 98134					g Number:	B-18		Page: 1 of 1	
					ractor: ade Drilling			Drilling Method: Direct Push / Hand Auger	
			,	Drill	Crew:			Drill Rig:	
					Kyle			Geoprobe 7822	
				Date	/Time Start	ed:		Date/Time Finished:	
Curface				1 7 24 19 10:20				- + (24)19 10:50	
Top of C	asing Flevati	on:		Logged by: S. Larsen/ M. Roberts			ny IVI. Roberts	Accession of the second	
Well Cor	struction In	formation	:AIA					(underte	
Screened	d Interval (ft	bgs):	Alla		Screen:	NIK	1	Water Level While Drilling (ft bgs):	
Filter Pa	ck Interval (f	t bgs):	MA		Riser:	N	4	~?`	
Seal Inte	rval (ft bgs):		NA		Seal Type	: NI	A	Water Level at Completion (ft bgs):	
Grout In	terval (ft bgs	5):	NA		Filter Pac	k: N	A		
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / PID (ppm)	Sheen Te SS (slight), M HS (hea	est: NS (none), NS (moderate), avy sheen)	USCS Symbol		Sample Description	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	36° 48° 54″	07	5- 0ppm G'- 0ppm J'- 0ppm B'- 0ppm G'- 0ppm G'- 0ppm U'- 24ppm U'- 24ppm 12'- 180ppm 12'- 180ppm 14'- 0ppm 14'- 0ppm 14'- 0ppm 16'- 2ppm 16'- 2ppm 16'- 0ppm 16'- 0ppm 16'- 0ppm 16'- 0ppm 16'- 0ppm 16'- 0ppm	5	NNNNS SSSNNNNNNNNNNNNNNNNNNNNNNNNNNNNN	SP	0-4" 60 4'-5' - Di fin M 5-10' Si tri at 10'-15' S 10 15-18' So W 15-18' So W 15-20' Si W	Accele ark gray Ibrown medium to a grain silty sands. No gravels oist. No hydroarton odor. ance soil description. Soils ansition from moist to wet ~9'. Hydroarbon odon observed ilow 9'. same soil description. Soils from -11' appear to be shott. It appear to description. No It appear to description. No It appear to description. No	



i i F Soil Boring Log

Project Name and Location:					ng Number:	B-07		Page:1_ of 1		
SeaPort	Midstream	Partners, S	Seattle	Cont	ractor:			Drilling Method:		
1652 SW	/ Lander Stro	eet Seattle	e, WA 98134	Casc	ade Drilling	5		Drill Pige		
				Drin	crew:			Geoprobe 7822		
				Date	/Time Start	ed:		Date/Time Finished:		
				71.	24/15	CUELL		21-4/10 10:50		
Surface I	Elevation:		·····		Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:		
Top of Ca	asing Elevati	ion:						Concrete.		
Well Con	nstruction In	formation	: l'u pres	pack						
Screened	d Interval (ft	bgs):	7-171	1	Screen:	10'		Water Level While Drilling (ft bgs):		
Filter Pag	ck Interval (t bgs):	5-171		Riser:	5'		9.5'		
Seal Inte	erval (ft bgs):		3-51		Seal Type	: bento	nite chips	Water Level at Completion (ft bgs):		
Grout In	terval (ft bg:	5):	3- 800	face	Filter Pac	k: 2/17	2 Sand			
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / PID (ppm)	Sheen To SS (slight), I HS (he	est: NS (none), MS (moderate), avy sheen)	USCS Symbol		Sample Description		
0		6T.		-			D-4" Conc	rete		
						SP	4-5' - li	alt army and the stand		
_ 2							6	the grad metric bane grades		
_ 3	36					1.1	5	ity sands, some she oraquients		
_ 4							-	in gravels, Dry, No hydrocation		
_ 5			5- Open	5'-1) (C	obon .		
6		37		6'-A			E'ala' C			
_ 7	1.	6-дррм	7 -NE		1 2-10 Ju.	is some as above. Transition				
8	50"		7- Oppm	81-17			tro	- moist to wat at 4.5 by		
9	10,		8'- Oppm	g'-N) .5		W	Wondy debris at 9.5 bys.		
10			S'- APOM	10-A	10	. 2	LI			
11		1 st f	10'+ 10'-70		112-43		r'ye	crucation ddy and statining		
12			10- Ppn	12'-1	40	4	at	9.5' 1565		
12	UEH		12-710	12'-1	he		5.			
10	95		13' - 105-00	14'-	ŚŚ		Shm	ple callecter Gt IC		
_ 14			141-3pon	15'-	<<		10-15' 4	the comment of		
_ 15)	1611	15-20	111-	MS		10 19 50	15 some as above.		
16			16 Teppin	16)	26					
17			2 - 13ppm	17.2.	35		15-1000	soile sauce as almal. No		
18	427		IT Sppm	18/-	lus		0 10.5			
19	6 C		18-3ppm	14'-	NS		he	procarbon odor below 16.		
_ 20		19'-01	19'-Oppur	201-	NS	1	18.5-20'	Date away silty day. wet		
_ 21		20-	20- Oppm			OHO				
22								No hyporcesson ocor.		
_ 23										
24										
25	1									



Project Name and Location:					ng Number	: B-29		Page: of 1	
SeaPort	Midstream	Partners,	Seattle	Cont	ractor:	a		Drilling Method: Direct Push / Hand Auger	
1052 54	F Lander Str	eet Seattle	, WA 30134	Drill	Crew:	10		Drill Rig:	
				Date	/Time Star	ted:	17:50	Date/Time Finished:	
Surface	Elevation:				Logged b	v: S. Larse	n/ M. Roberts	Protective Cover:	
Top of C	asing Elevat	ion:	NA					Concrete	
Well Cor	nstruction Ir	nformation	: NA		1	.1.			
Screene	d Interval (f	t bgs):	NA		Screen:	NA		Water Level While Drilling (ft bgs):	
Filter Pa	ck interval (T Dgs):	A A		Seal Type	NA - 114		Water Level at Completion (ft bgs):	
Grout In	terval (ft bg	(s): N	A		Filter Pac	k: NA		The second completion (it bgs).	
Depth	Recovery	Sample	OVM / PID	Sheen Te	est: NS (none),	USCS		Samula Description	
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), M HS (hea	WS (moderate), avy sheen)	Symbol		Sample Description	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	40" 54" 48"	0 5 10 15 20	5'- Orpm 1'- Orpm 7'- Orpm 8'- Orpm 10'- Orpm 10'- Orpm 12'- 10 ppm 12'- 10 ppm 13'- Orpm 14'- Orpm 14'- Orpm 14'- Orpm 14'- Orpm 14'- Orpm 14'- Orpm 14'- Orpm 14'- Orpm 14'- Orpm 15'- Orpm 14'- Orpm 15'- Orpm 14'- Orpm 15'- Orpm 15'- Orpm 15'- Orpm 10'- Orpm	5-1-8-10-11-13-11-5	ちちちちななななななななななななな	SP J OH	0-4" Conc 4 ¹² -5' Dow 912 Mo 5-10' So 92 Mo 0 HIC 10-15' Sa 4 50-15 Sa 15-185 Sa 18-5-20 A	rete. the gray borown incolumnts fine ind sitty sands: Few graveles ist. No hydrocalbon ador. me soil description. No avels. Soils transition from oist to wet at ~9. Hydrocabon dor at ~10:5' - Oden at 11' bys me soil conditions. Hydrocalbo. om 11-12'. mpte collected at 11.' Same soil conditions. No hydrocabo. odors dark gray sitty day, medion plasticity. Wet. No hydrocabon odor.	
23 24 25									



Project Name and L		Borin	g Number:	B-30		Page: 1 of 1	
SeaPort Midstream 1652 SW Lander St	Seattle e, WA 98134	Contr Casca	ractor: ade Drilling			Drilling Method: Direct Push / Hand Auger	
			Drill (Crew: Kyle			Drill Rig: Geoprobe 7822
			Date/	line Start	ea:		Traine Inisned:
Surface Elevation:			112	Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:
Top of Casing Eleva	tion:	MA				10 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -	Concrete
Screened Interval (f	t bgs):	NA		Screen:	N/M		Water Level While Drilling (ft bgs)
Filter Pack Interval	(ft bgs):	NA		Riser:	IN I	A	
Seal Interval (ft bgs):	NA		Seal Type	: W	n	Water Level at Completion (ft bgs):
Grout Interval (ft bg	gs):	NA		Filter Pac	k:	A	
Depth Recovery (ft bgs) (in.)	Sample Interval	OVM / PID (ppm)	Sheen Te SS (slight), M HS (hea	ST: NS (none), AS (moderate), avy sheen)	USCS Symbol		Sample Description
$ \begin{array}{c} 0\\ 1\\ 2\\ 3\\ -3\\ -4\\ -5\\ -6\\ -7\\ -8\\ -9\\ -9\\ -10\\ -11\\ -12\\ -13\\ -14\\ -15\\ -16\\ -17\\ -18\\ -19\\ -20\\ -21\\ -22\\ -23\\ -24\\ -25\\ -25\\ -25\\ -25\\ -25\\ -25\\ -25\\ -25$	0'T 5'T 10 10 15 20	5'-0ppm G'-0ppm G'-0ppm G'-0ppm O'-25ppm 10'-25ppm 10'-25ppm 12'-210ppm 12'-210ppm 14'-5ppm 14'-5ppm 16'-5ppm 16'-5ppm 16'-5ppm 16'-9ppm 20'-0ppm	5-NSWNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNN	5 5 5 5 5 15 15 5 5 15 5 15 5 15 5 15	SP J J SP J D H	0-4" (oncon 4"-5' Darl 9va 000 5-10' Sam 0200 10-15' Sa 15-20' 45 00 16 19	ete / terra cotta c gran/brown silty sands. No vel. Moist. No undocarbon s. a soil conditions. Hypocarbon is observed at 10' me soil conditions. -16' appears to be shalf w! neter collegrand brick/terr otta. -20' Dark gran silty clay. Modum plasticity. Wet. No undocarbon odors


Project Name and I	Project Name and Location:			ig Number:	: B-19		Page: of 1
SeaPort Midstrean	n Partners, reet Seattle	Seattle e. WA 98134	Cont	ractor: ade Drilling	7		Drilling Method: Direct Push / Hand Auger
1002 off Lander of	icer search	.,	Drill	Crew:	>		Drill Rig:
				Kule			Geoprobe 7822
			Date	/Time Start	ted:		Date/Time Finished:
			7	24/19	13:40		7/24/15 14:10
Surface Elevation:				Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:
Top of Casing Eleva	tion:	KHA					Concrete
Well Construction	nformation	I: WA		L,	· · · · · ·		
Screened Interval (ft bgs):	IN TRA		Screen:	NIA		Water Level While Drilling (ft bgs):
Filter Pack Interval	(ft bgs):	NA		Riser:	N.M.		10
Seal Interval (ft bgs	:	NIA		Seal Type	: NA	<u></u>	Water Level at Completion (ft bgs):
Grout Interval (ft b	gs):	2 114		Filter Pac	: n/4	4	
Depth Recovery	Sample	OVM / PID	Sheen Te	St: NS (none).	USCS		
(ft bgs) (in.)	Interval	(ppm)	SS (slight), N	AS (moderate),	Symbol		Sample Description
$ \begin{array}{c} 0 \\ 1 \\ 2 \\ 3 \\ 4 \\ 5 \\ 4 \\ 5 \\ 6 \\ 7 \\ 8 \\ 9 \\ 9 \\ 4 \\ 5 \\ 6 \\ 7 \\ 8 \\ 9 \\ 9 \\ 4 \\ 7 \\ 10 \\ 11 \\ 12 \\ 13 \\ 14 \\ 60^{\circ} \\ 15 \\ 16 \\ 17 \\ 18 \\ 19 \\ 20 \\ 21 \\ 22 \\ 23 \\ 24 \\ 25 \\ \end{bmatrix} 60^{\circ} $	0 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	5- 2ppm b' - 2ppm 21- 2ppm 21- 2ppm 2- 3ppm 10- 35ppm 10- 35ppm 11- 360 12- 392 ppm 13- 415ppm 14- 5ppm 15- 1 ppm 16- 2 ppm 16- 2 ppm 18- 2 ppm 19- 2ppm 20- 2ppm	5 6 7 8 9 10 1 13 11 5 1 7 18 4 -	NNN NNN NNN NNN NNN NNN NNNNNNNNNNNNNN	SP J J J J	041" Conc 4"-5' Dar 9m 5-10' So from - Friz p- 12-13' b Shipk 15'-20'1	rete k gran/brown silty sample. No wel: Moist. No hydrocodoon odors. we soil conditions. me soil conditions. Soil transition moist to wet at 10: reduct observed form 35 collected at 13' 15-18.5 same soil conditions. 13.5-20. Dark gran silty clay. No grands. Wet No hydrocation odds.



Project Name and Location: Boil SeaPort Midstream Partners, Seattle Cold 1652 SW Lander Street Seattle, WA 98134 Car Dri Dail Surface Elevation: Dail Top of Casing Elevation: MA Well Construction Information: MA Screened Interval (ft bgs): MA Filter Pack Interval (ft bgs): MA Seal Interval (ft bgs): MA					Riser:	B-08 3 ted: 4:15 y: S. Larsen MA WA e: MA	n/ M. Roberts	Page: 1 of 1 Drilling Method: Direct Push / Hand Auger Drill Rig: Geoprobe 7822 Date/Time Finished: $21241814:50$ Protective Cover: $44:50$ Water Level While Drilling (ft bgs): ~ 9.5 Water Level at Completion (ft bgs):			
Grout Int Depth	Recovery	s): Sample	OVM / PID	Sheen Te	Filter Pacest: NS (none),	USCS	A	L			
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), f HS (hei	WS (moderate), avy sheen)	Symbol		Sample Description			
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	36" 42" 36"	5	5' Oppm 6' Dppm 7' Oppm 8' Oppm 10' - 140ppu 10' - 140ppu 12' - 12 ppm 13' - 23 ppm 14' - 12 ppm 13' - 23 ppm 14' - 12 ppm 13' - 3 ppm 13' - 0 ppm	5678911234567942	NINNIN MASSISS MASKNON	S.R. JAJ	0-4" Conce 4"-5' Lig GVI 00 5-10' Do NI 01 10-15' So 15-20 : 11	nete in gran sitty sands. Fow avels. Dry. No hydroarton or. Terrn cottal brick at 5' ve grant brown sitty sands. o gravel Soils trainsition from wisct to wet at I.S'. Hydrocardon dor at 10'. unesoil conditions. mple collected at 10' bys 5-18.5 same soil conditions. No Hydrocarbon odor bedow W. 10.5-20 dask gray sitty rlay. Medium plushing. Wet. No hydrocarbon odor			



Project I	Name and Lo	cation:		Borin	g Number:	B-20		Page:1_ of 1
SeaPort	Midstream	Partners, S	Seattle WA 98134	Contr	actor: de Drilling			Drilling Method: Direct Push / Hand Auger
1001 000			,	Drill C	Crew:			Drill Rig:
				1	inte			Geoprobe 7822
				Date/	Time Start	ed:		Date/Time Finished:
Curford	Flouration			1 4	125119	4:60	M Pohorte	7/25116 8:15 Protective Cover:
Top of C	asing Flevati	ion:	ALA		rogged by	. S. Laiser	I/ IVI. RODEILS	Comprete
Well Cor	struction In	formation	NA					
Screene	d Interval (ft	bgs):	MA		Screen:	MA		Water Level While Drilling (ft bgs):
Filter Pa	ck Interval (f	t bgs):	NA		Riser:	NA		~9'
Seal Inte	erval (ft bgs):		NA		Seal Type	: Nt	+	Water Level at Completion (ft bgs):
Grout In	terval (ft bgs	s):	NA	Cl	Filter Pac	k: N	A	1
Depth (ft bgs)	Recovery (in.)	Interval	(ppm)	SS (slight), M HS (hear	ST: NS (none), IS (moderate), vy sheen)	Symbol		Sample Description
_ 0		OT					0-4" Con	erate
						20	4"-5 Da	rk grown I brown sitty sends.
- 2	11-1					Dr	No	Branels. Moist. No hydrocarboon
3	92						. de	or.
- 4			5'- gppm	5-1	ک			
- 5		ST	1. Drama	1-1	VC		5'-10' S	ame soil description. Silty clay
- 0			D-OFFI	71-	NC		t	ayer at ~ 7 about 1" tuck. Soils
	4.81"		7 -oppm	01-	Ne	. •	-	transition from moist to wet
- 0			0-oppm	g' -	NS		0	at ~ ?: Undrocation odor 1 staining
- 10			9-Oppm	10'-	214	1		but ~ 9.5'
11		107	10 - 130,00m	11 - 11	. 12		~ F	Free product (ald # Weerbliered)
- 11			11- 20	17' N	" MENNEN		0	bscruez from 10-15' bas
12	0 11		171 10	12/1	D Kurrin		10-15	a la la para intra Mar ula
11	30		131 110	13 -]	-15		-	ance sons description. Very diry,
			10-108	19 -	15			
- 15			19-220	5-	HIS		15-16" 0	in all description the
10		nT	13-245	16	5			and soils best provide No
- 1/			16 - Hoppin	17-	NK		t.	upprocation over below lo.
10	(D"		17 -3ppm	18-	NS	-		dais harding
- 19	QU		181 - OP PM	16-1	NS	64/0	119.20 E	Sette gray silvy ways main
20			191-Oppm				1	plastizity. Wet. No hydrocal bon
- 21		20	20'-Oppm	0-1	C.W		(odors.
- 22								
- 23								
- 24								
L 25								
			/					



Project N	lame and Lo		Borin	ng Number:	B-31		Page: 1 of 1	
SeaPort	Midstream	Partners,	Seattle WA 98134	Cont	ractor: ade Drilling			Drilling Method: Direct Push / Hand Auger
1052 500	Lander Str	eet seatth	, WA 50134	Drill	Crew:			Drill Rig:
					Kyle			Geoprobe 7822
				Date	/Time Start	ed:		Date/Time Finished:
				Ft	25/15	8:20		7/25/15 5:00
Surface E	Elevation:	<u></u>			Logged b	y: S. Larse	n/ M. Roberts	Protective Cover:
Top of Ca	asing Elevati	ion:	NA			······································		Concrete
Well Con	istruction In	formation	NA		Contraction			Mater Lough M/bile Drilling /ft has)
Screened	d Interval (ft	bgs):	NA		Screen:	V	14	water Level while Drilling (ft bgs):
Filter Pac	ck Interval (1	rt bgs):	WA		Riser:	W	A	Water Level at Completion (ft has):
Seal Inte	rval (π bgs):		NA		Sear Type	<u> </u>	/A	water Level at completion (it bgs):
Grout in	Recovery	S):	OVA (PID	Shoon To	Filler Pac		JA	
(ft bgs)	(in.)	Interval	(ppm)	55 (slight), M HS (hea	US (moderate), WS (moderate), avy sheen)	Symbol		Sample Description
0 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 17 18 19	48" (00" 4:2:#	0 5 15	5'-0700 6'-0700 3'-0700 8'-0700 10'-90700 10'-90700 10'-90700 10-90700 12-274 13-46000 14-5000 14-5000 17-60000 18-40000	5-10 	s s s s s s s s s s	SP	0-4" Ce 4"-5' Da 00 5-10' So fr 4 29 10-15' S 1im fed fr Sample 15-15' ve 16	and gray / brown silty sands. a gray / brown silty sands. a growels. Moist. No hydrovation brs une soils description. Soils transition om moist to wat at 95. ydrovation oddr/staining at .5. antie soil description as above is product of scivil at 12' bys Cellected at 12' bys S-16' likely sluff from heaving Sands. 5-14' same soils description as above. No hydrocarbon ador.
21 22 23 24 25		20	20-OppM	20-N	S	04((l pl hu	astrity. No gravels. Wet. No procerbon of Dr.



Project Name and Location:					g Number:	B-32		Page: 1 of 1	
SeaPort	Midstream	Partners, S eet Seattle	eattle . WA 98134	Contr Casca	actor: de Drilling			Drilling Method: Direct Push / Hand Auger	
1002 011	Lunder our		,	Drill	Crew:			Drill Rig:	
					Kyle			Geoprobe 7822	
				Date/	Time Start	ed:		Date/Time Finished:	
Surface E	levation:				Logged by	/: S. Larse	n/ M. Roberts	Protective Cover:	
Top of Ca	asing Elevati	on:	NA					Concrete	
Well Con	struction In	formation	NA						
Screeneo	Interval (ft	bgs):	NA		Screen:	NU		Water Level While Drilling (ft bgs):	
Filter Pac	ck Interval (f	t bgs):	NA		Riser:	MM		4,5	
Seal Inte	rval (ft bgs):		NA		Seal Type	: N¥	f	Water Level at Completion (it bgs):	
Denth		Sample		Sheen Te	st: NS (none)		V4		
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), N HS (hea	IS (moderate), vy sheen)	Symbol		Sample Description	
_ 0		9-			2	,	0-4" Con	nérete.	
						SP	4"-5' De	ack area house silter sands	
						2	N	Decranel Same man and	
_ 3	42."						S	Pashall Consiste Marth No	
- 4							k	uland day	
_ 5		5+	5- Opon	5-1	15			-1010101000n 0000.	
_ 6			6 - Open	6'-1	NY		5-10 8	same soil description. No wood,	
	48"		7' - Denn	フレト	JS		ę	Beschell or gravely Soil touritie	
_ 8	-1.0		g' - Digoin	81-1	15	4	4	rom magnist to writ -1 a's'	
9			9 - Alpan	9-1	US		· 4		
		10 2	10' - Znam	1001	MS			Jore Carbon oder and stamming	
_ 11		10	11 -	11,-10	recorres			17 01.5 Bgs.	
_ 12			17 123004	12,-1	-15		12	-13 bys free product observed	
_ 13	hov		13-1220011	13-1	-15		(1	reavity weathered 17 BKC	
14	48		14-15-001	14-	55		- 5	imple collection of the sign	
15			17-5 ppm	16-5	55		10-15 Sa	nue sous vesa p	
_ 16		15 -	15-3ppm	12 -N	6 rusum				
17			16	17)-1	MS7-STLA	67	15-18 00	and spile description. Soils at	
18	$(1n)^{n}$		17-231000	1 5 1	51		~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	7 and lileely slutt from heaving	
19	40		10 - 2	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	J			and No hand marbon olde	
20			10-2ppm	14-	55	\odot	1	(m) 19	
21		20	20, - Oran	201-	-NS		V2	and the	
22						OHO	13' - AN	ic group silty day layer. Wet	
23						SP	No No	lugorocerton adors.	
24		- 1				OH	15-20- 5	ame silty day.	
25						UN		, –	
25									



Project N	lame and Lo	cation:		Borin	g Number:	B-21		
					•			Page: of 1
SeaPort	Midstream I	Partners, S	eattle	Cont	ractor:			Drilling Method:
1652 SW	Lander Stre	et Seattle	, WA 98134	Casca	ade Drilling			Direct Push / Hand Auger
				Drill	Crew:			Drill Rig:
				Data	/Time Start	<u>d</u>		Date/Time Einished:
				Date	Alachia Sisp			2/25/15 15:1D
Surface F	levation:			TL	Logged by	: S. Larse	n/ M. Roberts	Protective Cover:
Top of Ca	asing Elevati	on:	NA					Concrete
Well Con	struction In	formation:	MA					
Screeneo	d Interval (ft	bgs):	NA		Screen:	NA		Water Level While Drilling (ft bgs):
Filter Pag	ck Interval (f	t bgs):	NR		Riser:	WA		0
Seal Inte	rval (ft bgs):		MΑ		Seal Type	: WA		Water Level at Completion (ft bgs):
Grout In	terval (ft bgs	s):	NA	ch	Filter Pac	K: NA	-	
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / PID (ppm)	Sheen 16 SS (slight), I HS (he	EST: NS (none), MS (moderate), avy sheen)	Symbol		Sample Description
0		θŢ					0-4" Con	crete
_ 1		1 N 1				SP	4"-5 1	
_ 2						1		It gran silty sands. ten gravely.
3	u						Sou	me silly day bands at ~ 3' bas.
_ 4	42						ni	oist. No undrocarbon odor.
5		51	5'-Oppm	5-	NS	-		
6		5	6 - Droand	6'-	105	· LP	5-10' 1.	
_ 7			J' - Propul	7/-	NC	1	in tra	hgran transitioning to one
8	Inu		C' a	d)	15		Sva	"I brown silty sands. No grands.
9	00		0 - oppon	0	N K		501	is go tom moist to wet at ~10.
_ 10			7 - Oppun	9,-	105		K.	Decasbon olor at 10.
_ 11		Tw	10-15 ppm	10-1	NS		10	
_ 12			0-	11 5	Ny PILOVIN		10-10 50	une soils as above.
_ 13	10		12-76 ppm	12,-	- 55		SU	rple callected at 12 bas
14	20		13.84 ppm	121	-50			ч <u> </u>
15			14-57 ppin	16-	-N. MIN			,
16		151	15-No Recov	en 1	h and	2 2	15-20' 16	slutt.
_ 17			16-swff	16	hor alwer	50.2	12	- 15 game Soil, Description as
18			17-11Appm	11	- 55	itt i		above.
19	42		19-Ceppin) ()).s			1 1 alla damanation
20		, ,	19-4ppin	19		64	19.	20 Darlegrung citing conferme
21		2	20-1 ppm	20	-NS			plasticity. Weta No hydrocabo
22								odor.
23								
24	-		·					
25								
<u> </u>								

TechSolve Environmental, Inc. 7518 NE 169th Street Kenmore, WA 98028 425-402-8277 Info@techsolveinc.com

L. F. F. M



Project Name and Lo		Boring	g Number:	B-10		Page: 1 of 1	
SeaPort Midstream I 1652 SW Lander Stre	Partners, S et Seattle,	eattle WA 98134	Contr Casca	actor: de Drilling			Drilling Method: Direct Push / Hand Auger
			Drill C	Crew:			Drill Rig: Geoprobe 7822
			Date/	Time Starte	ed:		Date/Time Finished:
Surface Elevation:			+2	5/19 10:20 Logged by: S. Larsen/ M. Roberts			Protective Cover:
Top of Casing Elevation.	on:	NTN.		LOBBER	. 5. 20150		Poncrete
Well Construction In	formation:	NA					
Screened Interval (ft	bgs):	NA		Screen:	L.	Au	Water Level While Drilling (ft bgs):
Filter Pack Interval (f	t bgs):	NA		Riser:	4	AL	9,5
Seal Interval (ft bgs):		NA		Seal Type	:	AU	Water Level at Completion (ft bgs):
Grout Interval (ft bgs): Comple	NA OVM (DID	Shoon To	Filter Pack		NA	
(ft bgs) (in.)	Interval	(ppm)	SS (slight), M HS (hea	IS (moderate), vy sheen)	Symbol		Sample Description
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0 5 10 10 10 10 10 10 10 10	5 - 0 9 - 0 9 - 0 10 - 73 ppm 13 - 121 ppm 14 - 5 ppm 14 - 5 ppm 14 - 0 18 - 0 18 - 0 19 - 0 10 - 7 10 - 7	HS (head	vy sheen) VS VS VS VS VS VS VS VS VS VS	SP I I I I I I I I I I I I I I I I I I I	0-4" Conce 4"-5' No Unde 5-10' No Dert gra Wet at 10-15 Sau Sampte 15-18' Sa 18-20' Ou W	recovery. Void space observed recovery till 6 Egran/brown silty sands. No Nels. Moist. transitioning to at ~9.5's Hydrocarbon odor staining at ~9.5' me soil description as above. Gliczted at 13' bgs me soil description as above. ark gray silty clay. Some body debn's. medium plasticity. No hydrocarbon odor.

TechSolve Environmental, Inc. 7518 NE 169th Street Kenmore, WA 98028 425-402-8277 Info@techsolveinc.com

.



Project N	ame and Lo		Boring	g Number:	B-22			
								Page:1 of 1
SeaPort N	Midstream F	Partners, S	eattle	Contr	actor:			Drilling Method:
1652 SW	Lander Stre	et Seattle,	, WA 98134	Casca	de Drilling			Direct Push / Hand Auger
				Drill C	Crew:			Drill Rig: Geoprope 7822
				Date	Fyle Time Starte	-de		Date/Time Finished:
				Date/	25/19 11			7/25/19 4125
Surface E	levation:				Logged by	: S. Larser	/ M. Roberts	Protective Cover:
Top of Ca	sing Elevati	on:	NM					Concrete
Well Con	struction In	formation:	NWA					
Screened	Interval (ft	bgs):	AW		Screen:	NA		Water Level While Drilling (ft bgs):
Filter Pac	k Interval (f	t bgs):	NA		Riser:	NA		10
Seal Inter	rval (ft bgs):		NA		Seal Type	NA		Water Level at Completion (it bgs):
Grout Int	erval (ft bgs	s):	WA OVM (DID	Shoop To	Filter Paci			
Depth (ft bgs)	Recovery (in.)	Interval	(ppm)	SS (slight), N HS (hea	ISL: NS (none), NS (moderate), NV sheen)	Symbol		Sample Description
0		0-					GEH" Concra	ete
_ 1						SP	415' Da	- k gray brown silta, crueds
2						[Feu	I could winor siller of a land
3	36						at	"3' Ciletrancelia Cilay band
4							to	which at 275' all in thom thom only
5			51-Oppm	5'- NS	>		60	oc
6		5	6'-Oppm	6-N	5		5-10 50	me as above. Soils frangition
7			7'-Oppon	7'-N	5		. tz	wet at ~10°. Hud marken -).
8	t'		B'- O DOM	8'- N	15		0u	nd staining at Ma'
9	54		5'-1	9-1	US			a contraction of the contraction
10			10'-5 m	10'-1	15			
_ 11		1º T	J. ppm	11, -10,	12200215		10-15 - 10	1-12' appeur to be slutt. (moist
12	3111		121-59	12 - 12	e recovery			Silty sands w/ gravel).
13	50	8 ·····	12-203	13'-H	S);	
14			14-378000	14- M	2/	1	U U	as Daric gray silty sands,
15			15'-11300m	15-55	5		5. 1	No gravel. Wet.
_ 16		12	16	16, -1	JE Milon		> public 1	Chiletted at 14
_ 17			17	[] • \$1 _ N	V2 ACLOVE	5	15-20 1	no recovery. I have about
18	7111		13-35 ppm	1, -1	CC		1 13'50	me soilidescription and transito
19	69		19 5 ppm	14 -	NS	DL	19'	Das & grand Silvey curry
20		1	20 - 4000	20		OH	le la	voin low plasticity to thigh
21		04	co o lan				F	Justicity at 20' Wet. No
22							U	my drocalbon otor.
23								
24								
25								
	×.,							

TechSolve Environmental, Inc. 7518 NE 169th Street Kenmore, WA 98028 425-402-8277 Info@techsolveinc.com ni

()		Ê	(_	sumpress.		S	(\mathcal{C}		V	1		
C	Ε	N	V		R	0	N	M	5	N	Ĩ	A	L	

Project N	ame and Lo		Borin	g Number:	B-44		Dago: 1 of 1	
								Proget
SeaPort I	Midstream I	Partners, S	Seattle	Conti	ractor:			Drilling Method:
1652 SW	Lander Stre	et Seattle	, WA 98134	Casca	ade Drilling			Direct Push / Hand Auger
				Drill	Crew:			Drill Rig:
					Kyle			Geoprobe 7822
				Date,	/Time Start	ed:		Date/Time Finished:
				7	25/19	12:50		7125119 3,20
Surface E	levation:				Logged by	: S. Larsen	/ M. Robert	s Protective Cover:
Top of Ca	asing Elevati	on:	NA					Concrete
Well Con	struction In	formation	NA			1		
Screened	l Interval (ft	bgs):	NR		Screen:	NΩ		Water Level While Drilling (ft bgs):
Filter Pac	k Interval (f	t bgs):	NA		Riser:	NA		9.5
Seal Inte	rval (ft bgs):		NA	(A	Seal Type	: NK	2	Water Level at Completion (ft bgs):
Grout Int	terval (ft bgs	5):	NA		Filter Pac	k: 1/1	6	
Depth	Recovery	Sample	OVM / PID	Sheen Te	est: NS (none),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), f HS (he	vis (moderate), avy sheen)	Symbol		Sample Beschption
0		0 -					0-4"	Concrete
- 1								
- 1						60	4-5	light gray to dark gray/ man
	A ON					01		Silver and I and
_ 3	42	1 1. 1				- 1		or and sonos wigrand. Dry transition
_ 4								to maist at ~3. No unprocarbon
5		1	5- ADOM	5'-	NS			odr.
6		ST	6		110		C	5 -) bur Wet -7
- 7			-Oppm	6	NJ		5-1D	Sime Suils as aspect of ap
	e!		7' Oppu	7'-	NS	2	0	9.5 bys. Hydre unber oder
_ 0	60		8' Orean	8-	NS		0	and standard 95 bas
_ 9			9' Amola	G)_	ale			
_ 10		1	10' UE I		NJ			grands.
11		10	10 - 45ppp		P17			a) and the Price
12			12 - 141 PPU	- 11 -	No rocorn		10-15	10-12 appear to be show noun
13	48"		13 -145 PPW	12-	MS			hearing sands. [light group, dry 1
14	10		14'-20 pour	13-	HS			w(arnivels).
1			-1	14'-	57			
_ 15	1.12	15-	15-17ppm	10)	51	· · ·		13-15 Dark grow brown silver
_ 16	1.2.		16		77			Saude Chonge undoscuston
17	3,11	3/11	17-17ppm	16-N	6 Pebbying			a la com Calacimica
18	26"	- 6	18 1 2014	11-	55		5	allering of 13' bas
19			10 - Pru	15-	NS		Sar	where concertage of 10 -305
20			ppu ppu	19/-	NS	. A		
_ 20		CA I	20-0 pph	70'-	NC		15-18	Dark gray sith sands Wot. No
_ 21			۱۳ ⁻				40	
_ 22							2	Grivels No suprocasion apor below
23								· CP.
24						DIM	1 18220	Durk gray sitty day. Low to.
25						040		medium dation. but 11.
25								THE THE PROPERTY I THAT IN TA
25	<u> </u>							least of a second

chSolve Environmental, Inc. 7518 NE 169th Street Kenmore, WA 98028 425-402-8277 Info@techsolveinc.com

·



Project N	ame and Lo	cation:		Boring	g Number:	B-16		Page:1_ of 1
SopPort I	Midstroom [Partners S	eattle	Contr	actor:			Drilling Method:
1652 SW/	Lander Stre	et Seattle	. WA 98134	Casca	de Drilling			Direct Push / Hand Auger
1032 340	Lanuer Stre	et seattle	,	Drill C	rew:			Drill Rig:
					Kula		5	Geoprobe 7822
				Date/	Time Start	ed:	5. H	Date/Time Finished:
				7/25	IA IZ	25		2/25/19 13:50
Surface F	levation:				Logged by	: S. Larser	n/ M. Roberts	Protective Cover:
Top of Ca	sing Elevati	on:	NA		,			Concrete
Well Con	struction In	formation	MA					
Screened	Interval (ft	bgs):	NA		Screen:	AU		Water Level While Drilling (ft bgs):
Filter Pac	k Interval (f	t bgs):	NA		Riser:	YNA		
Seal Inte	rval (ft bgs):		NA		Seal Type	: NA	f -	Water Level at Completion (ft bgs):
Grout Int	erval (ft bgs	5):	NA		Filter Pac	K: N	1A.	
Depth	Recovery	Sample	OVM / PID	Sheen Te	st: NS (none),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), N HS (hea	15 (moderate), vy sheen)	Symbol		
0		0T					0-4" Cono	rate No recover until 5 bus.
							1 -1	
	4					2.12	4-5-	
	\smile							
3							N 1	
4 5		-					ADU	indoned / Brizles
	7 K	OT					422	Jebris doce not
_ 7	* *			a la			-11,1	Attempted
							GIIOW	in the providence of the provi
_ 10		10'1					7 bo	mgs I doot apart
					7		of ed	all other of Similar
13							resul	Its.
- 15								1
∟ 16	· · · ·							
∟ 17			1 A A					· .
18								
19							,	
20								
21						· ·		
21								
·⊢ 23								
_ 24								
L 25								



		2						
Project N	ame and Lo	cation:		Borin	ig Number:	B-11		Page: 1 of 1
SeaPort I	Midstream F	Partners, S	eattle	Cont	ractor:			Drilling Method:
1652 SW	Lander Stre	et Seattle	, WA 98134	Casca	ade Drilling			Direct Push / Hand Auger
				Drill	Crew:			Drill Rig:
					Kyle			Geoprobe 7822
				Date	/Time Start	ed:		Date/Time Finished:
			FL	5/19 14	00		7/25/19 14:30	
Surface E	levation:	1			Logged by	y: S. Larse	n/ M. Roberts	Protective Cover:
Top of Ca	asing Elevati	on:						Concrete
Well Con	struction In	formation:	1" prespo	at				
Screeneo	Interval (ft	bgs):	7-17'	2GS	Screen:	10'		Water Level While Drilling (ft bgs):
Filter Pac	k Interval (f	t bgs):	5-17'6	255	Riser:	7	1	9.5
Seal Inte	rval (ft bgs):		3-5'65	5	Seal Type	: bent	onite chips	Water Level at Completion (ft bgs):
Grout Int	terval (ft bgs):	3- 600)	C	Filter Pac	k: 21	2 Sare	
Depth	Recovery	Sample	OVM / PID	Sheen To	est: NS (none),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), I HS (he	MS (moderate), avy sheen)	Symbol		Sample Description
0		0					D-4" Con	rete
		T						+ hown sitty sands w/ gravel.
	24"		8			SW	7-5-49	Osha andrea alla i
_ 2	P						Dr	y. No undrocarbor and
_ 3			1					
4			· ·	5'				
5		r 1	5- Oppm	, -1	NS			2
6		21		6 -	NS			
			00	71-	NIC		5-15 - 1	D mol
	42"		7-0	/	100	C.0	0 10 1265	k grayibrown silty sames. NO grave
<u> </u>	-12		0.0	8'-	NS	SP	Wet	- 17 9.5 bys. Hydroccob
_ 9	8		9-B	9'-	NS	1	. 02	or and Staining at 9.5' but
10				10'-	N/S			
11		10T	10 - t2ppm	, ,	n D			1 June 1 al aland
12			1) -	11,~	NO KEENIZ		10-15 San	re soils described above.
- 12			12 - 970000	12,-	145		Sprak	Collected from 13 1365
_ 13	42"	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	13-14	13'-1	F1 5		, P	
_ 14			101 pp/m	141-5	55		-	
15			19-15 ppm	10'- N) <	d	15-18° C.	min cails rescribed above.
16		151	15-5 ppm	15 1	1. ff		10 50	source source descentation of the
17		10	16-27pm	16 ,0	-		(47)	our bon our not observed below
	110		12-10-10	17-	05		L 1	
18	60		10 ppm	18-	NS	-	16.	
_ 19			to Sppm	14-	NS	OH	18-20 1	ask gray silty play. Mehry
20		701	19 Oppm	21/-	11/5	.0.	1	- 1 Sta Lator Lucia
21		41	20 - Oppm	90	100		T	Man plasticity. Wet. No
22			~				1	hydrocas bon adar.
22								,
- 23								
_ 24							×	
∟ 25								



Project N	ame and Loo	cation:		Boring	g Number:	B-09		Page: <u>1</u> of 1
SeaPort I	Vidstream F	Partners, S et Seattle	eattle , WA 98134	Contr Casca	actor: de Drilling		÷	Drilling Method: Direct Push / Hand Auger
1052 544				Drill C	Crew:			Drill Rig:
				Data	Kyle .	ad.		Geoprobe 7822 Date/Time Finished:
				Date/	slic is	1,50		2/25/16 15:20
Surface E	levation:			-116	Logged by	: S. Larse	n/ M. Roberts	Protective Cover:
Top of Ca	asing Elevati	on:						Concrete
Well Con	struction Inf	formation:	1 pre	epsck u	Screen	101		Water Level While Drilling (ft bgs):
Screened	h Interval (ft	bgs):	5-17		Riser:	<u>ں،</u> `د		
Seal Inte	rval (ft bgs):	. 253/.	3-5'		Seal Type	: ben	torite chips	Water Level at Completion (ft bgs):
Grout Int	terval (ft bgs	s):	3' - gvaide	,	Filter Pac	(: 2/1	2 sand	
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / [*] PID (ppm)	Sheen Te SS (slight), N HS (hea	S t: NS (none), AS (moderate), Ivy sheen)	USCS Symbol		Sample Description
0	1	0-T					D-4" Concr	ete 1 DUE
	6						4"-5'- Pea	gravel back fill Trom
2	Ð						62;	acing storm scholar
5			- I	<'- 1.	City	SP	5-10-1	Slack Silts Sands W/
6		5-	S	2-1	1)		pen gi	much heaving. No
7	2111			6			HC ad	W GIZ NO HC
8	56			7-))		52	(42-Haderearbon)
_ 9			B- Oppm	8-)	15		DIGINIO	
_ 10		10'1	9- Oppm	0-11	5		Feg gro	evel heaving into hale
			10- Doom	10-1	NJ	Ì	[0521]	i lot of of
-12	A						14. 7	1' ha recovery from
13							10 +0 (u bgs.
14		13						
16	· ·						Sample	WilleArd at 10 365
17							,	
18	0							
19								
_ 20		2021						
21								
22								
23								
_ 24								
<u> </u>		× .						



Project N	ame and Lo	cation:		Boring	g Number:	B-12		
								Page:of 1
SeaPort I	Vidstream F	Partners, S	eattle	Contr	actor:			Drilling Method: Direct Push / Hand Auger
1652 SW	Lander Stre	et Seattle,	, WA 98134	Casca	row:			Drill Rig:
					1410			Geoprobe 7822
				Date/	Time Start	ed:		Date/Time Finished:
				71-	210/15	8:00		7/26/19 8:20
Surface E	levation:				Logged by	/: S. Larser	n/ M. Roberts	Protective Cover:
Top of Ca	sing Elevati	on:	NA					Concrete
Well Con	struction In	formation:	NA		· · · ·			
Screened	Interval (ft	bgs):	NA		Screen:	NA	t	Water Level While Drilling (ft bgs):
Filter Pac	k Interval (f	t bgs):	NA		Riser:	NV	1	Water Level at Completion (ft hgc):
Seal Inte	rval (ft bgs):	,	NA		Seal Type		A	
Grout Int	erval (ft bgs): Samala	OVM / PID	Sheen To	Filter Pac		M.	1
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), N HS (hea	IS (moderate), Ny sheen)	Symbol		Sample Description
0							O- 6" Con	erete
1							6-5' 1	What ismulte boring,
2	D						e lo	NO NOT COMPETING
3	0						at.	~2' bas driller het an
4			Υ.				065	truction that we could
5							10.05	+ 2 old past. Topiers to a
6		·	ал.	1			VLO	
7							be	a tooting for The
8							wa	schase wall.
9							·	England uses alread doned
10								sound was assented .
11								
12								
13								
15								
16	· ·							
F 19								
-21								
- 22								
23							25°	
L_ 24								
L_ 25		·						



Project N	lame and Loc	ation:		Borin	g Number:	B-33		
i i oject iv				Dit	of Ecology	well)	D: 15LK-301	Page: of 1
SeaPort	Midstream P	artners, S	eattle	Contr	actor:	(Drilling Method:
1652 SW	Lander Stre	et Seattle,	WA 98134	Casca	de Drilling			Direct Push / Hand Auger
				Drill C	Crew:			Drill Rig:
					Lyle	- d -		Date/Time Finished
				Date/	lime Start	ed:		Thetic 9:35
Curfer 1			/	+12	Logged by	: S. Larse	n/ M. Roberts	Protective Cover:
Surface I	asing Flovation:	on:			10000000		,	Asphalt
Well Cor	struction Inf	ormation:	1 prenal	< Well				
Screenee	d Interval (ft	bgs):	3-13'		Screen:	1D'	4	Water Level While Drilling (ft bgs):
Filter Pa	ck Interval (f	t bgs):	2-13'		Riser:	2'3"		.7
Seal Inte	erval (ft bgs):		1:5-2 bas	senternit	Seal Type	: berto	nite chips	Water Level at Completion (ft bgs):
Grout In	terval (ft bgs):	1.5-grade		Filter Pac	k: 2/12	sand	
Depth	Recovery	Sample	OVM 7 PID	Sheen Te SS (slight), M	EST: NS (none), MS (moderate),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	HS (hea	avy sheen)	Symbol	0, "	A Soft deacto 3.5' w/ hand
_ 0		DT	. i.				06 ashp	NUT- SA CLEW (0 SID -CT NAM)
_ 1							Diggi	
2						1	6-2' 0:-	turbed fill material. Sands and
3	36						grav	vel.
4							2' -' 2.1	ligun silty sands. No
5			E'- Prom	5-1	25	SP	D-D DAT	a gran i di la hada calena odor.
6		ST	Broppin	6'	UC .	1	900	wel. Maist. Do unaroco de
			6 -Opport	51-1	5		Le la lat	k argun (brown silty sands, lew
			1-Oppm	8'-1	AF	×)	5-10 000	Do Morst transitioning to we
	60/		B-183 m	G1-	2		grad	~7' Hiderarhon abo/staining et
- 10	0		5'-2042pp	1,1,-	22		~3!.	1 Matel at 5:
			10-50000	10	22		* Sa	mple bullet & to
		NO T	11.200	11 -	21		10-15 Sa	une soils description. 10-11 is
_ 12			17 TOppm	12 -	NS		like	ly shot.
_ 13	0		12-20 ppm	13'-	NS		in -	lick gray silty day. No gravels.
14	a		15-7 ppm	141-	NE	6406	19	ind had prad on allow
_ 15			19-2 ppm					Wet. No hyprocessor open.
16		10.	12-26em	15-1	24	0	19-11	4.5 same soils as previousin
17					r	St	1 1 1	described.
18			8			DH	14.	g-15. Durk Gray String
19								Medium plasticity. Wet, No
20								und concertan allar
20								and a see not all a st
- 23								
_ 24								
L 25						1		



Project N	ame and Lo	cation:		Borin	g Number:	B-35		
FIOJECTIN		cation.		1.1	f. I. Iw	. Well I	N. BLK 302	Page:1_ of 1
Coo Dort I	Aidstroom [Dartnore Se	aattle	Cont	actor:	group		Drilling Method:
Seaport I	Landor Stro	ot Soattlo	WA 98134	Casca	de Drilling			Direct Push / Hand Auger
1052 500	Lander Stre	et Seattle,	WA 30134	Drill	rew.			Drill Rig:
			1					Geoprobe 7822
				Date	/Time Start	ed.		Date/Time Finished:
	ð.,			Fla	alig to	(a)		7/29/15 15:30
Curface F	lovation			110	Logged by	: S. Larse	n/ M. Roberts	Protective Cover:
Surface E	ievation.	0.0.1			Lobbed			Asolalt
Top of Ca	struction Int	formation:	1" la m	KIN				
Sereen Con	Interval (ft	bac).	1 prepar	·	Screen:	10'		Water Level While Drilling (ft bgs):
Screened	Interval (It	bgs/.	5-15 643	>	Riser:	2101	×	8
Filter Pac	R Interval (T	r ngs):	2-16		Seal Type	· lauda	Le duine	Water Level at Completion (ft bgs):
Seal Inte	rval (it bgs):		6-1.56q5		Filter Pac	k. alia	and and	
Grout Int	Percei (IT bgs	Sampla	OVM / PID	Sheen Te	st: NS (none)	USCS	Same	
(ft bgs)	(in)	Interval	(mag)	SS (slight), I	VIS (moderate),	Symbol		Sample Description
(it bgs)	()		(PP)	HS (he	avy sheen)		0,14,0	0+.
_ 0		-O-					O-B aspra	
_ 1							(0'-2' - fill	moterial. Disturbed sand and
2								
3	30					20	1 -1 -1	the and the count low
	00					8	2-5 - Da	I'L Gray Brown 81 0rg sunds, 400
			- · · ·		10	1	gr.	wels. Moist. Walundrocarbon addi
_ 5		-'+	S-Oppm	5 (VY			
_ 6		2	6 DROM	6 -	15		5-10 Sam	ne soils description. Soils
L 7			21 2000	7/-	55		1	to the they at to make made at the
_ 8	15		01 0	-/_	nAS		TVa	
9	(00)		0-20ppn	8,	10.5	1	B.	Hydro car leon odor/Staining out
10			- Sezbow	9-	NS		(R)	
11			10-92 m	14)	1)(L	A Same	le colleter at 3.
12		101	11 - 0 10.	10 0	105		The Samp	call designation bacrowels.
			12 OPPM	11 -	NS		10-15 .50	we solis borrighter i gree
13			-> ppm	121 -	NS	1	- Si	Ity day benerat 12', 14, 14.5-1.
_ 14	Gol		1>- Sppin	13' -	15	OH.		
15	0		14- Appen	w -	101		-110	the debris at ~13.
16		15-	15-Anon	171				
17			Plan	15 -	NY XV		× 1	
12								54 S.
10								
_ 19								
L 20								
_ 21								
22								
22								
23								
L 25								
	1	1		1				



Project N	ame and Lo	cation:		Borin	g Number:	B-38		
								Page: 1 of 1
SeaPort I	Midstream F	Partners, S	eattle	Cont	ractor:			Drilling Method: Direct Push / Hand Auger
1652 SW	Lander Stre	et Seattle,	, WA 98134	Drill	Crew.			Drill Rig:
				V	ala.			Geoprobe 7822
				Date	/Time Start	ed:		Date/Time Finished:
				71:	zalia in	:50		7/29/19 11:20
Surface E	levation:				Logged by	: S. Larser	n/ M. Roberts	Protective Cover:
Top of Ca	asing Elevati	on:	NA					Asphalt.
Well Con	struction In	formation:	NA	dk. U	vell			And the second s
Screened	Interval (ft	bgs):	NA		Screen:	NA	·	Water Level While Drilling (π bgs):
Filter Pac	ck Interval (f	t bgs):	MA		Riser:	NA		Water Level at Completion (ft hgs):
Seal Inte	rval (ft bgs):		NA		Seal Type	N	A	
Grout Int	terval (ft bgs	5):	NA-	Shoon T	Filter Pac		R	
Depth (ft bgs)	Recovery (in.)	Sample Interval	(ppm)	Sneen 10 SS (slight),	MS (moderate),	Symbol		Sample Description
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	42" 52" 60"	5	5'-Orpm 6'-Orpm 7'-Orpm 9'-353 ppm 10'-Sppin 10'-Sppin 11-400 ppm 12-13 ppm 13-4ppm 14-Oppm 15-Orpm	567 850723195	~5 15 15 15 15 15 15 15 15 15 15 15	SP J 6H	6-2' Fill 2-5' Even 5-7: San 7' 2' 8-10' San We Shimple (10-14' San huid 12.5 14'	materia : sand, and gravel. M/brown silty sands. No gravel st. No highrowbon of or ne soil description. layer of concrete ashlate me soil description. Soil ase t at 0'. Hydrocarbon odor. Collected of 8 bys. ne soil description. No observable reasbon odor below 12'. " Durkgran cilty clay layer i'y "twick. dark gran silty clay layer G' thick.
20 21 22 23 24 25								



Project Na	ame and Lo	cation:		Borin	g Number:	B-34		Page: 1 of 1
SeaPort N	Aidstream F	Partners, S	eattle WA 98134	Contr	actor:			Drilling Method: Direct Push / Hand Auger
1022 200	Lanuer Stre	et Seattle,	, WA 50154	Drill	Crew:			Drill Rig:
				k	-12/0			Geoprobe 7822
				Date	Time Start	ed:		Date/Time Finished:
				3h	6/19 12	10		7/25/19 15:30
Surface E	levation:				Logged by	: S. Larse	n/ M. Roberts	Protective Cover:
Top of Ca	sing Flevati	on:	10				-	Asonalt
Well Con	struction In	formation:	N/A					
Screened	Interval (ft	bgs):	n nA		Screen:	UN	1	Water Level While Drilling (ft bgs):
Filter Pac	k Interval (f	t bgs):	NIA		Riser:	61	A.	8.5'
Seal Inter	rval (ft høs):		N/A		Seal Type	: W	A	Water Level at Completion (ft bgs):
Grout Int	erval (ft bgs	;);	0.110		Filter Pac	k: N	A	
Depth	Recovery	Sample	OVM / PID	Sheen Te	est: NS (none),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), f HS (he	MS (moderate), avy sheen)	Symbol		
0		6-					0-6° aspre	lt.
							("E' 1")	5 Fill cauds and granels.
						- ×	0-0:6-1	.S FILL Suit
- 2	·					00	1.5'-5': 6	my bark brown sitty sands
_ 3	46					Sr	0	A way of No.
_ 4							te	w gravels. Mors 1 100
_ 5			SEPON	5-1	N		W	norocarbon olor.
_ 6		ST	o opport		10		G W	Cili Cili
_ 7			6 8 Kpm	6 -	02		5-10 - Se	anne soil descriptions soils
8	p.		F DROW	7'-	NS	\checkmark	fro	msition from moist to wet as
9	48		" Draw	0'-	22		8.	5. Hydrocarbon oder and staining
10			S' 11 Prov	0	Ъ. С		ad	t~8.5
11		a t	1 - 1600ppm		MZ		C C	11-12-1 64 9'
12		0	10-24pm	C0' =	NS		26	mple Concertainty
12			11 - SC+pp m	n'~	MS			The and No observable
	(0°		12 - 29 ppm	12-	NS	4	10-14.5	same sol description. Po
	60		1y - Cppm	13, -	NS			mydro.carbons below 12.
16		152	15-Doom	14,-	NJ	6H	14,5-15	Der gray gity day medium
17	2			15 -1	VS			to i sha Restrict I allo som all.
								to high pusha ignore graves
								Net Wo hydrolas bon odos.
_ 19								
L 20								
_ 21								
_ 22								
23	,							
24								
25								
⊢ ²³								



.

Project Name and Lo	ocation:		Borin	g Number:	B-39		Page: 1 _of 1
SeaPort Midstream 1652 SW Lander Str	Partners, S eet Seattle	eattle , WA 98134	Contr Casca	ractor: ade Drilling			Drilling Method: Direct Push / Hand Auger
			Drill (Crew:			Drill Rig:
			D	/Time Steri	od.		Geoprobe 7822
			Date,	alia start	18:40		ghalla KU:00
Surface Elevation:	,			Logged by	: S. Larser	n/ M. Roberts	Protective Cover:
Top of Casing Elevat	ion:	WA					Asphalt.
Well Construction Ir	formation	MA					Water Lovel While Drilling (ft has):
Screened Interval (ff	bgs):	NA	· · · · ·	Screen:	MA		
Filter Pack Interval (ft bgs):	NA		Riser:	N	<u>7</u>	Water Level at Completion (ft bgs):
Seal Interval (ft bgs)	: c):	NA		Filter Pac	. № k:	11/	
Denth Recovery	Sample	OVM / PID	Sheen Te	est: NS (none),	USCS		
(ft bgs) (in.)	Interval	(ppm)	SS (slight), M HS (he	MS (moderate), avy sheen)	Symbol		Sample Description
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0-10-10-10-10-10-10-10-10-10-10-10-10-10	5- Oppm 6- 71 - Oppm 8-94 ppm 10 - 6 Frm 11 - 303 ppm 13 - 30 ppm 13 - 3 ppm 13 - 3 ppm 13 - 3 ppm 15 - Oppm 15 - Oppm	56789112345	NS NS NS NS NS NS NS	50	D- Asy -2' Fill 2'5' Gr We 00 5-10' So 01 7 5-10' So 01 7 10-14' So 14-15' D U 01 01 01 14-15' D U	phalt. 1: Sands, colles. an i dark brown stilly sands. 9 ravel. Moist. No Indrocarbon loc. me Soil Description. Soils transitor ommoist to wet at ~ g. Undrocarbon doc and staining observed at 1.5! de collected at 9' ame soils description. ort gray silter clay. No gravels. jot. No my orarbon odor. Wady eloris present in soils.



Project Name and Lo	cation:		Borin	g Number:	B-40		Page:1_ of 1
SeaPort Midstream	Partners, S et Seattle,	eattle , WA 98134	Contr Casca	actor: ade Drilling			Drilling Method: Direct Push / Hand Auger
	,		Drill Crew:				Drill Rig:
			Data	Kyll Time Start	ed.		Date/Time Finished:
			Date	Talle Start	415		2/2/19 14:40
Surface Elevation:				Logged by	: S. Larse	n/ M. Roberts	Protective Cover:
Top of Casing Elevati	on:	NA					Asphalt
Well Construction In	formation:	WA					Model And Maile Drilling (ft hgs):
Screened Interval (ft	bgs):	NA		Screen:	NN		Water Level While Drilling (it bgs):
Filter Pack Interval (f	t bgs):	NA		Riser:	Pr	<u>\</u>	Water Level at Completion (ft bgs):
Seal Interval (ft bgs):		NA		Sear Type		F N	
Depth Recovery	Sample	OVM / PID	Sheen Te	est: NS (none).	USCS		
(ft bgs) (in.)	Interval	(ppm)	SS (slight), M HS (her	MS (moderate), avy sheen)	Symbol		Sample Description
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	D	5-1 ppm 6'- 7'-214 ppm 8'-977 ppm 10-3ppm 10-3ppm 11-72ppm 13-2 ppm 14-12ppm 15-2 ppm 15-2 ppm	5 - NU 7 - HS 9 - NJ 13 - NJ 13 - NJ 15 - NJ 15 - NJ	s S S S S S S S	SP 2014 SP	D-6 asphal 6-12" ba Col 12"-5'- d 5-10" Sa fr 4 5-10" Sa fr 4 5ample 10-14" sa 14-14-5:6" MU 94 14.5 15.0	t de fill - light gran sandy bles. ark gran lorown silty sands. vo grovels. Moist. No hydrocerbon odor. me soils description. Soils' transiti om moist to whet at ~ 7'. Grocarbon odor and staining at 1' and below. Collected at 7' bgs. me soil description. band of dark gray silty day. edium to high plasticity. No' avels. Wet. No hydrocarbon odors. Jork gray silty sands. No gravel Wet. No hydrocarbon odor.

Project N	ame and Lo	cation:		Borin	g Number:	B-36		Page: 1 of 1
				-				Drilling Method:
SeaPort I	Midstream F	Partners, S	eattle	Contr	ractor:		5	Direct Push / Hand Auger
1652 SW	Lander Stre	et Seattle,	, WA 98134	Casca	cade Drilling			
				Drill	Drill Crew:			Geoprope 7822
					Ryle			Geoprobe 7822
				Date	/Time Starte	ed:		Date/Time Finished.
				312	6/19 1	4:50	/ M. Dahanta	Protective Cover:
Surface E	levation:				Logged by	: S. Larser	n/ M. Roberts	Protective cover.
Top of Ca	asing Elevati	on:			· · · ·			HSphall
Well Con	struction In	formation:	NA					Water Level While Drilling (ft hgs);
Screeneo	l Interval (ft	bgs):	NA		Screen:	WA		water Level while Drilling (it bgs).
Filter Pac	k Interval (f	t bgs):	WA		Riser:	NA		T,S
Seal Inte	rval (ft bgs):		NA		Seal Type	: NA		Water Level at Completion (π bgs):
Grout Int	erval (ft bgs	s):	NA		Filter Pac	K: NE	t	
Depth	Recovery	Sample	OVM / PID	Sheen Te	est: NS (none),	USCS		Sample Description
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), I HS (he	ws (moderate), avy sheen)	Symbol		cumpic bescription
0		DT					D-6" Aca	A PA
1			· · · · · · · · · · · · · · · · · · ·				1	
_ <u>1</u>							6-2' fill:	sandy cobbles.
_ 2	1-11					00	51 10 5	
3	42					R	1-5 Der	le gray/brown silty sands.
4				,			No	gravel. Moist. No undrocarbon
5			5-17	5-NS			a ch	
6		ST	ppm	6'-5h	FA		000	
			6 ~ ppm	11 N	IC		E. F.	11 a. J. Lastan
\vdash '			71- D ppm		~ <	(SP	5-65 9	revelly saws. light going.
8	(0)		81-132 000	8 - 10	12	9.	hlo	ist. No lup racestoon odor.
_ 9	0		GI = 110 m	9'-5	5	CD	6.5.10 D	art aroun house citize courts
10			, ito ppor	10-1	15	>1		
11		して	10 - F ppm	1) -n	NC		1 W	& gravel. Joils change soom
12			11-47 ppm	17	110			wist to wet at ~7.5. Hyprocol
12			12-3 pm	10-	NS		D	dor and staining at ~7.51 and
13	C N		12 1 0000	13'-1	NS			(upper)
_ 14	120		US-1 PPT	111-	1)c		5. 2.	-11-12-1 47 8 has
15		L	10-1 ppm	1,0	1V-1		-ombio	Collectice of o roga
16		18	15 - Oppm	15 -	NS		10-13 2010	ndtaissel lis .
17						d	10 5 Sam	a soil description
12							13-13.5	Dark aroun silter clause Hash
						64		and the start walle
- 19								plasticity. No growels e Wet.
L 20			- 20-					Ala 14 June day adams
_ 21						SP		100 physica and acres.
22							1-5,5-14	Same SP
23						OH	120-110-	Caller OL
23							rint.s	Jame Utt.
						SP	14.5515	same sp
L 25								

TechSolve Environmental, Inc. 7518 NE 169th Street Kenmore, WA 98028 425-402-8277 Info@techsolveinc.com

CTECHSOLVE ENVIRONMENTAL TECHSOLVE

Soil Boring Log

Project Name and Education: Page: 1 of 1_ SeePort Midstream Partners, Seattle Contractor: 1652 SW Lander Street Seattle, WA 98134 Contractor: Contractor: Drilling Method: Direct Push / Hand Auger Drilling Method: Direct Push / Hand Auger Drill Rig: Geogrobe 7822 Date/Time Started: Date/Time Started: Date/Time Finished: Date/Time Started: Date/Time Finished: Date/Time Started: Date/Time Finished: Screened Interval (ft bgs): NPA Screened Interval (ft bgs): NPA Seal Type: NPA Seal Type: NPA Vertex and the search of		
SeaPort Midstream Partners, Seattle Contractor: Drilling Method: 1652 SW Lander Street Seattle, WA 98134 Contractor: Drilling Direct Push / Hand Auger 1652 SW Lander Street Seattle, WA 98134 Contractor: Drilling Direct Push / Hand Auger 1652 SW Lander Street Seattle, WA 98134 Direct Push / Hand Auger Direct Push / Hand Auger 1652 SW Lander Street Seattle, WA 98134 Direct Push / Hand Auger Direct Push / Hand Auger 1652 SW Lander Street Seattle, WA 98134 Direct Push / Hand Auger Direct Push / Hand Auger 1652 SW Lander Street Seattle, WA 98134 Direct Push / Hand Auger Direct Push / Hand Auger 1652 SW Lander Street Seattle, WA 98134 Direct Push / Hand Auger Direct Push / Hand Auger 1652 SW Lander Street Seattle, WA 98134 Direct Push / Hand Auger Direct Push / Hand Auger 165 Surface Elevation: Logged by: S. Larsen/ M. Roberts Direct Push / Hand Auger 160 Surface Elevation: Logged by: S. Larsen/ M. Roberts Direct Push / Hand Auger 160 Surface Elevation: Logged by: S. Larsen/ M. Roberts Protective Cover: Construction Information: 170 of Casing Elevation: NA Screeen: NA Screeen: NA 16		
Searor ministream Partners, seature Contractor: Contractor: Cascade Drilling Direct Push / Hand Auger Direct Push / Hand Auger <tr< td=""><td></td></tr<>		
1652 SW Lander Street seattle, WA 98154 Cascade Untilling Didle Generation Drill Crew: Geoprobe 7822 Date/Time Started: Date/Time Finished: Top of Casing Elevation: Logged by: S. Larsen/M. Roberts Protective Cover: Well Construction Information: NA Screeen: Mater Level While Drilling Filter Pack Interval (ft bgs): NA Seal Type: NA Seal Interval (ft bgs): NA Seal Type: NA Grout Interval (ft bgs): NA Filter Pack: NA Depth Recovery Sample OVM / PID Sheen Test: NS (none), St (sight), MS (moderate), Symbol Sample Description 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 0 2 3 42 ^{1/4} 4 ¹ -lppm 4 ¹ -N		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$		
Bate/Inite Stated. Date/Inite Stated. Atsplie g:40 Atsplie Surface Elevation: Logged by: S. Larsen/M. Roberts Protective Cover: Consete Well Construction Information: NA Screened Interval (ft bgs): NA Screen: MA Screen: NA Screened Interval (ft bgs): NA Screen: Main Screen: NA Seal Interval (ft bgs): NA Seal Type: NA Seal Type: NA Grout Interval (ft bgs): NA Filter Pack: Depth Recovery Sample OVM / PID Sheen Test: Ns (none), (ft bgs) Interval USCS O O Signed of the stated of the stat		
TACSOLUTION Surface Elevation: Logged by: S. Larsen/ M. Roberts Protective Cover: Conserte: Well Construction Information: NA Screened Interval (ft bgs): NA Screened Interval (ft bgs): NA Seal Interval (ft bgs): NA		
Surface Elevation: Logged by: S. Larsen, M. Roberts Projective cover. Top of Casing Elevation: NA Consete Well Construction Information: NA Screene: MA Screened Interval (ft bgs): NA Screen: NA Filter Pack Interval (ft bgs): NA Screen: NA Seal Interval (ft bgs): NA Seal Type: NA Grout Interval (ft bgs): NA Seal Type: NA Depth Recovery Sample OVM / PID Sheen Test: Ns (none), St (light), MS (moderate), HS (mode		
Top of Casing Elevation: Other Street Well Construction Information: NA Screened Interval (ft bgs): NA Filter Pack Interval (ft bgs): NA Seal Interval (ft bgs): NA Filter Pack: NA Bepth Recovery Sample OVM / PID Sheen Test: Ns (none), Symbol Statisfield Ns (moderate), Symb		
Well Construction Information: NA Screen: MA Water Level While DrillingScreened Interval (ft bgs): NA Riser: NA $\sim 5.5^{'}$ Seal Interval (ft bgs): NA Seal Type: NA $\sim 5.5^{'}$ Seal Interval (ft bgs): NA Seal Type: NA Water Level while DrillingGrout Interval (ft bgs): NA Seal Type: NA Water Level at CompletionGrout Interval (ft bgs): NA Filter Pack: NA SampleDepthRecoverySampleOVM / PIDSheen Test: Ns (none), St slight), MS (moderate), HS (heavy sheen)USCS SymbolSample Description0 0 0 0 0 0 0 0 2 1 1 1 1 1 1 2 3 $42^{''}$ $4'$ -lpp M $4' - NS$ 1 1 4 5 $5'$ -l85 ppm $5'$ -MS 1 1 1 5 $5'$ -l85 ppm $5'$ -MS 1 0 0 0 6 $5'$ $6''$ 3 4 1 1 7 $4''$ 1 2 1 1 1 6 $5'$ $6''$ 1 1 1 1 7 $4''$ 1 1 1 1 1 8 $60''$ 1 1 2 1 1 9 1 1 1 1 1 1 1 9 1 1 1 1 <t< td=""><td></td></t<>		
Screen: MAScreen: MAWater Level while binningFilter Pack Interval (ft bgs):MAReserver in the binningSeal Interval (ft bgs):NASeal Type: NAWater Level at CompletionGrout Interval (ft bgs):NASeal Type: NADepthRecoverySampleOVM / PIDOVM / PIDSheen Test: Ns (none), S (slight), MS (moderate), HS (neav sheen)USCS SymbolOOVM / PIDSheen Test: Ns (none), S (slight), MS (moderate), HS (neav sheen)USCS SymbolOOVM / PIDSheen Test: Ns (none), S (slight), MS (moderate), HS (neav sheen)USCS SymbolSample OUOVM / PID (ppm)Sheen Test: Ns (none), S (slight), MS (moderate), HS (neav sheen)USCS SymbolSample OUOVM / PID (ppm)Sheen Test: Ns (none), S (slight), MS (moderate), HS (neav sheen)USCS SombolSuperificationSuperificationAO'AO'SO'SSSO'S <th colsp<="" td=""><td>(ft hgs):</td></th>	<td>(ft hgs):</td>	(ft hgs):
Filter Pack Interval (ft bgs):NARiser:NASeal Interval (ft bgs):NASeal Type:NAGrout Interval (ft bgs):NAFilter Pack:NADepthRecoverySampleOVM / PIDSheen Test: NS (none), (ft bgs)USCS SymbolSample Description00000000210000000220000000342''4'-lpp4'-NSSP6'-5'00455'-l85pp5'-l85pp5'-MS000555'-l85pp5'-MS000065'-l85pp5'-MS00007-HS860''7'-HS000860''7'-PB05'-lu'805'-lu'	(10 063).	
Seal Interval (ft bgs):NASeal Type:NAWater Level at CompletionGrout Interval (ft bgs):NAFilter Pack:NASampleWater Level at CompletionDepth (ft bgs)Recovery (in.)Sample IntervalOVM / PID (ppm)Sheen Test: NS (none), SS (slight), MS (moderate), HS (heavy sheen)USCS SymbolSample Description0000000021000000342"4'-lpp m4'-NSSP6'-5'00455'-l85 ppm6'-MS1110565'-l85 ppm6'-MS00006766'-MS00007860"7'-HS05'-lw00860"7'-RS05'-lw000	(ft bgs):	
Grout Interval (ft bgs):NAFilter Pack:NADepth (ft bgs)Recovery (in.)Sample IntervalOVM / PID (ppm)Sheen Test: NS (none), SS (slight), MS (moderate), HS (heavy sheen)USCS SymbolSample Description -0 0 0 0 0 0 0 0 0 0 -2 1 0 0 0 0 0 0 0 0 -3 42^{11} 4^{1} 4^{1} 105 5^{1} 0 0 -3 42^{11} 5^{1} 5^{1} 105 0 0 0 -3 42^{11} 5^{1} 5^{1} 105 0 0 0 -3 42^{11} 5^{1} 100 0 0 0 0 -3 42^{11} 5^{1} 100 0 0 0 0 -3 6^{1} 100 0 0 0 0 0 -3 42^{11} 5^{1} 100 0 0 0 0 -3 42^{11} 5^{1} 100 0 0 0 0 -3 42^{11} 5^{1} 100 0 0 0 0 -3 42^{11} 5^{1} 100 0 0 0 0 -3 0 0 0 0 0 0 0 0 -3 0 0 0 0 0 0 0 0 -3	(ir ngs):	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		
- 0 - 1 - 2 - 3 - 4 - 5 - 5 - 6 - 5 - 10 - 4 - 1 - 4 - 1 - 4 - 1 - 4 - 1 - 4 - 1 - 4 - 1 - 1 - 4 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1		
1 2 3 4 4 5 5 6 7 8 60'' 1 1 1 1 1 1 1 1 1 1 1 1 1		
A A A A A A A A A A A A A A	sands.	
3 42" 4'-NS 4 5 5 5'-185 ppm 5'-MS 6 5'-185 ppm 6'-MS 7 6'-9' pm 6'-MS 8 60" 7 7'-780 ppm 7', -HS	T of	
4 4 5 5 6 7 8 60'' 5 4'-lppm 4'-NS 5'-lbt ppm 6'-MS 6'-MS 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 60'' 7'-HS 8 10''' 10''' 10''' 10''' 10''' 10''' 10''' 10'''' 10''''' 10''''''''''	5 9.00	
5 6 7 8 60" 5 1 5'-185 ppm 5'-M5 6'-9' pem 6'-M5 7'-H5 8 60" 7'-H5 8 60" 5'-185 ppm 5'-M5 6'-M5 7'-H5 6'-115 7'-H5 6'-115 7'-H5 6'-115 6'-115 7'-H5 6'-115 7'-115 7'-115 7'-115 6	cond ~2"	
5 1 S'-185 ppm 5'-MS 6 7 7 8 60" 5 1 S'-185 ppm 6'-MS 7 - HS 8 60" 5 1 S'-185 ppm 6'-MS 9 - MS 9 - MS 0000 and striving at au) below. 5'-10' Plus - 10 - 10 - 10 - 10 - 10 - 10 - 10 - 1	alloon	
6 7 8 60" (-9' per 6'-MS) odoc and structure a 2'-780pon 7',-HS au) below. 5'-10" Plus - 10 5 million	A ATI	
- 7 - 8 60" - 7', - HS - and below.	× 43	
- 8 60'' 1-700pm 9' - 100 5'-10' 81.1 - 11 5'-10'		
	Magul	
9 8-1173 pull 5 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -	000000	
10 51 10 m 4 3 3	5.	
10 11 10 10 10 10 10 10		
- 11 I T I with I with the sample Culleched at The	5	
$ -12 $ $ '_{5}48900 _{12}-135$		
13 12 MPPM 13 15 N 12 15		
14 07 13'-4 pp 10' be 10-14.0 guille soit description		
15 10'- I prim 19 - 10 - 15 - Aark over silty a	lay weden	
16 15- Origin 5- NS DH 19.5 1	of wet	
plasticity. No grau		
Up by drocarbon odd		
	· · · ·	
20		
21		
	, a a a	
25		

TechSolve Environmental, Inc. 7518 NE 169th Street Kenmore, WA 98028 425-402-8277 Info@techsolveinc.com

.



						D 42		
Project N	ame and Lo	cation:		Borin	g Number:	B-13		Page: 1 of 1
			ttla	Contr	actor:			Drilling Method:
SeaPort I	Midstream I	Partners, S	eattle	Contr	actor:			Direct Push / Hand Auger
1652 SW	Lander Stre	et Seattle,	, WA 98134	Daill				Drill Rig:
				Drill	Lrew:			Geonrobe 7822
				Data	Joe Time Start	od.		Date/Time Finished:
				Date	lime Start	eu:		altalia loit
				718	015 1	0:25	/M Deherte	Protective Cover:
Surface E	levation:				Logged by	y: S. Larse	h/ IVI. Roberts	Protective cover.
Top of Ca	asing Elevati	on:	10					Asphall
Well Con	struction In	formation:	NR		-	6.1		Water Level While Drilling (ft hgs):
Screeneo	d Interval (ft	bgs):	NA		Screen:	NI	ł	water Level write Drining (it bgs).
Filter Pag	ck Interval (f	t bgs):	NA		Riser:	NI	·	7.5
Seal Inte	rval (ft bgs):		NA		Seal Type	: NK		\downarrow water Level at Completion (π bgs):
Grout In	terval (ft bgs	s):	NA		Filter Pac	k: Ⅳ	9	
Depth	Recovery	Sample	OVM / PID	Sheen Te	st: NS (none),	USCS	× *	Sample Description
(ft bgs)	(in.)	Interval	(ppm)	SS (slight), M HS (hea	avy sheen)	Symbol	·	Sumple Description
0		DT					D-4" ason	ult .
					2.98	1. 1.		
						SP	4-5 Dark	gray brown silty sands
2								
3	214						NOS	revels. Moist. No more carles
4	20						0)20	
5			- 1	1				
6		5+	S - Sppm	5 - 1	15		5-10' Sau	me sal description. Soils transition
7			6 -	6-			fre	AF to tout + true
			21			6	100	had a lower of the
			7 - O ppl	7-N	5		644	procarbon odor and sturing as
_ 9	47"		0.0 ppm	B-CC	1.151		1. ~	8.5.
10	1C		9-68 polo	0- 75			X sample	collected at 9
11		1n-t	102	5- M	÷ .	00	In-14 San	ne sol description.
12			19991	10 - W		N	10-11-11-0-1-1	·
12			11-700M	1.0 100			14-14.5 00	whe grow of they day. Low plasman
	1 de		12-25. 11	1(-M	8	OL	wet.	No hudrocarbon odor.
L 14	(00		13-15- Popul	12 M	5	CO.	14.5-15. 8	arake SP and description.
L 15			ILL SPPW	12-41	G	R		01 01 00 01
16		15-	O ppln	n	25	2		
17			10- Dem	14-	. 12			
18			(Pr	15-	NS			
								•
F 19								
_ 20								· · · · · · · · · · · · · · · · · · ·
_ 21				1.00				
22						1		
22								
23								
L 25							1.5	



.

Project N	ame and Lo	cation:		Boring	g Number:	B-01		
				Dent	ofEcdes	Well Ta	g: 154×304	Page: of 1
SeaPort I	Midstream I	Partners, S	eattle	Contr	actor:		(Drilling Method:
1652 SW	Lander Stre	et Seattle,	WA 98134	Casca	de Drilling			Direct Push / Hand Auger
				Drill C	rew:			Drill Rig: Geoprope 7822
				Data	Joe Time Start	ad:		Date/Time Finished
				Date/	Golia	INDD		2/30/19 10.50
Surface F	levation:				Logged by	: S. Larse	n/ M. Roberts	Protective Cover:
Top of Ca	asing Elevati	on:					2	Asphalt
Well Con	struction In	formation:	1ª prespre	le well				
Screeneo	l Interval (ft	bgs):	3-15'		Screen:	10'		Water Level While Drilling (ft bgs):
Filter Pac	ck Interval (f	t bgs):	2-13		Riser:	2'8"		~ Ø.S
Seal Inte	rval (ft bgs):		1:5-2		Seal Type	: bent	prite chips	water Level at Completion (ft bgs):
Grout Int	terval (ft bgs	s):	1.5 - gray	Share To	Filter Pac	K: 2/17	2 sand	
Depth (ft bgs)	Recovery (in.)	Sample	(ppm)	Sneen Te SS (slight), N	SC: NS (none), IS (moderate), vy sheen)	Symbol		Sample Description
0		9-		no (nea	.,		0-6" ason	alt
1							GUE DI	A sur le matter a star de 199
2						SP	D Vask	gray is round story saws, the
3	— . 1						qra	ver moisi, No un urocasoon odor.
4	36						E IO' -	a I gave a trian Q I.
5			5-Anon	<i>г</i> 1.	x		5-10 Sam	Le soit description - soils
6		51	e' opposit	5-1			transi	tion from woist to wet at NBS.
7			21-	6-			1 Jug roc	arbon o dor and staining at 8.5.
8	26		8-0-00	4' - 11	~ <		Fra P	reduct (Henril webter 1) cherry
9	20		GI UPPM	0 -10	10		From	10-13 bas
_ 10			- 100pph		15		45.1	
_ 11		101	=270 pm	10 -1	101/1		27 Compl	e concerted at 10 bys
_ 12			N ~	11-10	"	2		
_ 13	44"		12-932 DOM	12-	r1s		10-15 Sav	ue soils description.
_ 14			13-199	13-	1)			
15			14=5 19	19-3	35			
16	9°	152	15-1 PPM	15-3	55			
_ 17			POW					
18								
19	2							
20								
21								
22								
23								
24								
25				1997 - A				
- 2J								



Project Name and Location:				Boring	g Number:	B-41				
				Ew	ogy we	(1D:	5114303	Page: <u>1</u> of 1		
SeaPort Midstream Partners, Seattle				Contr	actor: de Drilling			Drining Method: Direct Push / Hand Auger		
1652 SW	Lander Stre	et seattle	, WA 30134	Drill C	Crew:			Drill Rig:		
					Ky	c		Geoprobe 7822		
					Time Start	ed:	7:40	Date/Time Finished: 7/31/19 8:15		
Surface Elevation:					Logged by: S. Larsen/ M. Roberts			Protective Cover:		
Top of Ca	sing Elevati	on:	. 11		1.			Un Lore +C		
Well Con	struction Int	formation	3-18	prepar	Screen:	15		Water Level While Drilling (ft bgs):		
Screened	k Interval (ft	t bgs): 7	2'-18'	9	Riser: 7 all			6		
Seal Inte	rval (ft bgs):	7-1	5		Seal Type	: bento	nitechips	Water Level at Completion (ft bgs):		
Grout Int	terval (ft bgs	5): 1.5-	grade		Filter Pac	k: 2/12	saul			
Depth (ft bgs)	Recovery (in.)	Sample Interval	OVM / PID (ppm)	Sheen Te SS (slight), N HS (hea	S t: NS (none), MS (moderate), Ny sheen)	USCS Symbol		Sample Description		
0		O'T					D-1' Conce	rete.		
_ 1			. *				1-5' Ar-11	Craus / brown silter sauds,		
_ 2	311"				SP 1-5 DEFE			mul. Mast. No und rocarbon		
_ 3	50						no q	ravel polaret. Not well		
_ 4				.1	×C	V	own			
_ 5		51	5-Open	5'-L)>		5-10 same soil description. Soils			
6			6-82 FPM	6'-1	VIS		5 10 500	metion form maist to wet		
⊢ 7	Cal		2-195 ppm	7'-N	15			() Il for a char and		
	60		3-415 ppm	8'-W	2/		of 6. Hydrocarvon door and			
			9-5 001	g' - 5	15		- 5. 5	Stuning at ~6 and bern.		
		Lol	(1)-3 00M		10		JAN	pic collected at 8 695		
		1	11-1210 ppm				10-13' sa	une soil description.		
	60"		12 - 600 ponta	17'-	is S	50	1 - Carl			
		60" 13- 14	13-100m .71-		NK	SP/OF	13-15	Several Small of burnes of		
			14 - ADOW	12,	e i	0.44		Ourle gray still bay medium		
10		15-	15 - Door	14-	20			to high plasticity. wet, No		
17		~V	- T	15-1	NS			Observable undrocarbon odos		
								clay bands are mixed in		
								The same siller sands as		
- 19							· · · · ·	previously described.		
22										
23										
25										
- 25										



·				D	Number	P. 13			
Project Na	ame and Loc	ation:		Boring	g Number:	D-42	11 705	Page: 1 of 1	
				tiole	gy Well	ID: B	16 200	Drilling Method:	
SeaPort Midstream Partners, Seattle				Contr	actor:			Direct Push / Hand Auger	
1652 SW	Lander Stre	et Seattle,	WA 98134	Casca	ae Drilling			Drill Rig	
				Drill C	rew:			Geoprobe 7822	
					yle	- d.		Date/Time Finished:	
				Date/	Time Starte	ed:		Floring G! AF	
	~			73	7/31/19 8:20			Protective Cover:	
Surface E	levation:				Logged by: S. Larsen/ M. Roberts			folective cover.	
Top of Ca	ising Elevation	on:						LONCOLTE	
Well Con	struction Inf	ormation:	1 prepa	ek well	well			Water Level While Drilling (ft bgs):	
Screened	l Interval (ft	bgs):	3-10' 695		Screen:	15			
Filter Pac	k Interval (f	t bgs):	2-18' bass		Riser:	2:8	~	Water level at Completion (ft hgs):	
Seal Inte	rval (ft bgs):		1.5.2'		Seal Type	bent	mite chips		
Grout Int	erval (ft bgs):	1.5 = grade	/	Filter Pac	K: 2/12	sand		
Depth	Recovery	Sample	OVM / PID	Sheen Te	st: NS (none),	USCS		Sample Description	
(ft bgs)	(in.)	Interval	(ppm)	HS (hea	avy sheen)	Symbol		•	
0		B					D-9" Cons	crete	
		-				0	94 5 Das	kgray silty sands into	
	42"					SF	60	uels. Some woods debris at	
	-1- 2		· · ·				Q	2' a si ui mart. He incoles	
_ 3						1	~ ~ ~	so ,	
_ 4			4-1000	4' -h	10		000	or at ~5.	
5			C.IM	11	>		b ist	57	
6		51	o lo ppm	5- N	15	-	6-10' Sar	ue soils description. Graviels	
			6-26ppm	6-1	15		obs.	erved at ~6 (4" band).	
			7, -296 ppm	7-2)	5		9.1	le transition from innet to	
⊢ ŏ	10	· · · · ·	9-Doom	8-N	15		200	at at 1 the day the adar	
L 9	00		9-2 pplaa	6- 4	<			G. G. Unorocasion oddi	
10			10 - ADOM	4 3	2		a	10 strining from 5-9.	
11		107	1' 2.	10-5	5	-	Sum de	with the at 7'has	
12			11 - opom	11', -	55		Jampi	+ 1D	
12		·	12 - EVEN	12-	55	01	- in' . 2"	band of silth clay are	
L 13			13- Ann	13'-	NIC	OL	Cou	Tomedium plasticity. were	
_ 14	. 60		141 0 00		100	0	10-13 Darl	caray silty sands. No growels.	
L 15			1- OPBM	14 -	-NS	36	1,10-	t. No hydracarlos afor.	
16		15-	12-OPEW	16'-	-NS			. 1 64	
17						(01)	15-145 Dar	le gray silly day. High plasticity	
12						OM	(1)	t. No un Droices bon opon	
10						· 21	ure ure	a non a characteristic analis	
- 19							14.5-15 -	siame special and than Wood	
L 20								Valante al sono description, wood	
_ 21								years Observed ~19.5.	
22									
22									
- 25									
_ 24									
_ 25									
	1	1		1			1		



Project Name and Location:				g Number:	B-43		Page: 1 of 1
SeaPort Midstream Par 1652 SW Lander Street	Contra Casca Drill C Date/	actor: de Drilling Trew: Time Starte	ed:		Drilling Method: Direct Push / Hand Auger Drill Rig: Geoprobe 7822 Date/Time Finished:		
Surface Elevation:				13 (19 9:50 Logged by: S. Larsen/ M. Roberts			Protective Cover:
Top of Casing Elevation:	:	NA					Concrete
Well Construction Infor	mation:	NA					
Screened Interval (ft bg	(s):	WA		Screen:		A	Water Level While Drilling (ft bgs):
Filter Pack Interval (ft b	gs):	NVA		Riser:	N	r M	9
Seal Interval (ft bgs):		NA		Seal Type	: .	A	Water Level at Completion (ft bgs):
Grout Interval (ft bgs):	amala	NA OVM (DID	Shoon To	Filter Pac			
DepthRecoverySi(ft bgs)(in.)In	nterval	(ppm)	SS (slight), N HS (hea	IS (moderate), vy sheen)	Symbol		Sample Description
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	5	3'- 1272 ppu 4'- 1159 ppu 5'- 1408 ppu 6' 1472 ppu 3'- 1472 ppu 3'- 1472 ppu 9' 1578 ppu 10'- 15781 10 15781 12-16 ppu 13-18 ppu 14-11 ppu 15-57000	39567860121919		SP	B-1 Coner 11-5' Durk Wood and 2 belo 5-10' Sa tran 29! Samp Herry H-ry 10-15' Sam	ete gran silly sands. Moist. A Debris at 4'. Hydrocalbon obr Staining present at I' and w. me soils description. Soils sition from neoist to wet at Wholy debris at ~0'. The collected at 7' bys. continuation absenced from 1th free preduct observed, e soil description as alloove.

Plant 1 Waterfront Probing Summary Report Site: Former BP Harbor Island Terminal 2/7/2020

Appendix C

Sampling and Analysis Plan

Exhibit F of the Consent Decree Original and unedited copy November 1999

SAMPLING AND ANALYSIS PLAN ARCO HARBOR ISLAND TERMINAL 21T SEATTLE, WASHINGTON

November 1999

Prepared for

ARCO Products Company 4 Centerpointe Lane, Suite 100 La Palma, California 90623

Prepared by

TechSolv Consulting Group, Inc. 12510 128th Lane NE, Kirkland, Washington 98034 (425) 823-4337 FAX (425) 823-2346

TABLE OF CONTENTS

		Page
1. INTROE	DUCTION	1-1
2. OBJECT	IVES	2-1
2.1 Gro	DUNDWATER SAMPLING	2-1
2.2 Son	L SAMPLING	2-1
2.3 SED	IMENT AND BIOTA SAMPLING	2-1
2.4 DAT	TA QUALITY OBJECTIVES	2-2
2.4.1	Level of Quality Control Effort	2-2
2.4.2	Accuracy and Precision	2-3
2.4.3	Completeness, Representativeness, and Comparability	2-3
2.4.3.1	Completeness	2-3
2.4.3.2	2 Representativeness	2-3
2.4.3.3	Comparability	2-4
3. FIELD P	ROCEDURES	3-1
3.1 GEN	VERAL	3-1
3.1.1	Documentation	3-1
3.1.2	Coded Sample Identification System	3-2
3.1.2.1	Primary Samples	3-2
3.1.2.2	2 Field Replicate Samples	3-3
3.1.3	Equipment Calibration	3-3
3.1.4	Equipment Decontamination	3-4
3.1.4.1	Drilling and Soil Sampling Equipment	3-4
3.1.4.2	2 Water Monitoring and Sampling Equipment	3-5
3.1.5	Sample Handling	3-5
3.1.6	Waste Management and Disposal	3-6
3.1.6.1	Drill Cuttings	3-7
3.1.6.2	Excavated Soils	3-7
3.1.6.3	B Purge and Decontamination Rinse Water	3-7
3.1.6.4	Disposable Equipment	3-8
3.2 Gro	DUNDWATER MONITORING WELL INSTALLATION	3-8
3.2.1	Preparation for Drilling, Sampling, and Well Installation	3-8
3.2.2	Drilling Methods	3-9
3.2.2.1	Soil Sampling and Logging	3-10
3.2.2.2	2 Well Installation	3-11
3.2.2.3	Well Development	3-12
3.2.2.4	Documentation	3-13
3.3 Hot	SPOT SOIL EXCAVATIONS	3-13
3.3.1	Sampling Locations	3-14
3.3.2	Sampling Preparation	3-14
3.3.3	Soil Sample Collection	3-14
3.3.4	Sample Documentation	3-15
3.4 GRO	DUNDWATER SAMPLING	3-15
3.4.1	Sampling Locations and Frequency	3-15
3.4.2	Sampling Preparation	3-15
3.4.3	Groundwater Sample Collection	3-16
3.4.4	Sample Documentation	3-17
4. ANALY	FICAL PROCEDURES	4-1

4.1	ANALYTICAL METHODS AND PROCEDURES	
5. QU	ALITY CONTROL PROCEDURES	
5.1	FIELD ACTIVITIES (MEASUREMENTS AND SCREENING)	
5.2	LABORATORY ANALYSIS	
6. DA	TA ASSESSMENT PROCEDURES	
6.1	FIELD DATA VALIDATION	
6.2	LABORATORY DATA VALIDATION	

TABLES

- 1. Instrument Calibration Frequency
- 2. Decontamination Procedures by Activity
- 3. Sample Container Filling Order

1. INTRODUCTION

This Sampling and Analysis Plan (SAP) was prepared by TechSolv Consulting Group, Inc. (TechSolv) in conjunction with ARCADIS Geraghty & Miller (Geraghty & Miller) as part of the remedial design activities and development of the Groundwater Compliance Monitoring and Contingency Program for the ARCO Products Company (ARCO) Harbor Island Terminal 21T in Seattle, Washington. Field and analytical activities as well as site history and conditions are discussed in general terms in the Groundwater Compliance Monitoring and Contingency Program. The SAP establishes specific procedures and protocols for sample collection and field analysis activities to be conducted during implementation of the remediation activities and Groundwater Compliance Monitoring and Contingency Program. This SAP fulfills the requirement for a sampling and analysis plan as defined in the Washington State Model Toxics Control Act (MTCA) (WAC 173-340-820) and meets the provisions of the Agreed Order between ARCO and the Washington State Department of Ecology (Ecology).

The sections that follow describe detailed procedures to be followed for tasks to be completed during the field program, as well as specific laboratory analytical procedures and data quality assurance. All personnel carrying out field tasks as part of the remedial design and Groundwater Compliance Monitoring and Contingency Program will be required to adhere to the health and safety provisions set forth in the Site Health and Safety Plan (SHSP). The Groundwater Compliance Monitoring and Contingency Program, SAP, and SHSP will be available to all field personnel for procedure guidance.

2. OBJECTIVES

The overall objective of the sampling program is to obtain data of known and defensible quality. In addition, the data collected during the remedial design and compliance monitoring program must be comparable with data collected during the RI activities previously completed at the site. To achieve these goals, data collection, sample collection, sample handling, and field documentation conventions must be well-defined prior to the initiation of the remedial design and compliance monitoring program and stringently adhered to during the course of the program. A detailed account of the procedures to be followed during the remedial design and compliance monitoring program is provided below. If any procedures are modified during the course of the procedure.

In addition to the overall objective, a number of the field program objectives are specific to individual tasks. The specific objectives are detailed below:

2.1 Groundwater Sampling

Groundwater samples will be collected from selected monitoring wells installed at the site to verify trends in water quality, confirm cleanup objectives have been attained, and verify cleanup objectives are maintained.

2.2 Soil Sampling

Soil samples may be collected during the installation of performance/confirmational monitoring wells along the waterfront to further define the vertical extent of petroleum hydrocarbons in the soils beneath the warehouse.

Soil samples will be collected from the perimeter of the inland soil excavations in both Plant 1 and Plant 2 to assist in quantifying any residual total petroleum hydrocarbons (TPH) remaining in the inland soils.

2.3 Sediment and Biota Sampling

One round of sediment/biota sampling will be conducted following completion of product removal and compliance monitoring to ensure protection of the aquatic organisms in the Duwamish waterway. Sediment/biota sampling will be implemented in accordance with the procedures outlined in the State of Washington Sediment Management Standards.

2.4 Data Quality Objectives

The overall quality assurance (QA) objective is to develop and implement procedures for field sampling, chain-of-custody, laboratory analysis, and reporting that will provide data of a known quality which are usable, defensible, and can be used for decision making.

Specific procedures for sampling, chain-of-custody, calibration, laboratory analysis, reporting of data, internal quality control (QC), audits, preventive maintenance, and corrective action are described in other sections of this SAP. The purpose of this section is to address the specific objectives for accuracy, precision, completeness, representativeness, and comparability.

2.4.1 Level of Quality Control Effort

Field blank, trip blank, field duplicate, and matrix spike samples will be analyzed to assess the quality of the data. Field and trip blanks provide a means to assess the quality of the data resulting from the field sampling program. Field blank samples are analyzed to check for procedural contamination at the site, which may cause sample contamination. Trip blanks are used to assess the potential for contamination of samples due to contamination migration during sample shipment and storage. Field duplicate samples are analyzed to check for sampling and analytical reproducibility. Matrix spike and matrix duplicate (MD) samples provide information about the effect of the sample matrix on the digestion and measurement methodology. For organic analyses, all matrix spikes are performed in duplicate and are hereinafter referred to as MS/MSD samples. For inorganic analyses, QC samples include MS's and MD's.

The general level of the QC effort will be one field duplicate and one field blank for every ten or fewer investigative samples. One volatile organic compound (VOC) trip blank consisting of distilled deionized ultra-pure water will be included with each shipment of aqueous VOC samples. One set of MS/MSD samples will be collected for every twenty or fewer investigative samples collected for organic analyses. Soil MS/MSD samples require no extra volume for VOCs or extractable organics; however, aqueous MS/MSD samples must be collected at triple the volume for VOCs and triple the volume for extractable organics. The quantities may vary when field conditions or sample screening dictates a change to the SAP.

2.4.2 Accuracy and Precision

Accuracy is the degree of agreement of a measured value with the true or expected value of the quantity of concern. In other words, accuracy answers the question of how close the measurement is to the true value. The farther the measurement is from the true value, the less accurate.

Precision is the degree of mutual agreement, characteristic of independent measurements, that occurs as the result of repeated application of a process under specified conditions. Precision is a measure of the reproducibility of a given measurement system and is concerned with the closeness of results.

The fundamental QA objective with respect to accuracy and precision of laboratory analytical data is to achieve the QC acceptance criteria of the analytical protocols. The accuracy and precision requirements are specified in the method protocols for organics and inorganics.

2.4.3 Completeness, Representativeness, and Comparability

2.4.3.1 Completeness

Completeness is a measure of the amount of valid data obtained from a measurement system compared to the amount that was expected to be obtained under normal conditions. It is expected that the subcontracted laboratory will provide data meeting QC acceptance criteria for 95 percent or more for all samples tested. Following completion of the analytical testing, the percent completeness will be calculated by the following equation:

 $Completeness(\%) = 100 x \frac{Number of measurements judged valid}{Total number of measurements necessary}$ to achieve a specified statistical level
of confidence in decision making

2.4.3.2 Representativeness

Representativeness expresses the degree to which data accurately and precisely represent a characteristic of a population, parameter variations at a sampling point, a process condition, or an environmental condition. Representativeness is a qualitative parameter which is dependent upon the proper design of the sampling program and proper laboratory protocol. The sampling network was

designed to provide data representative of the site conditions. During development of this network, consideration was given to past disposal practices, existing analytical data, and physical setting and processes. Representativeness will be satisfied by ensuring that the proper sampling techniques are used, proper analytical procedures are followed, and holding times of the samples are not exceeded in the laboratory. Representativeness will be assessed by the analysis of field duplicate samples.

2.4.3.3 Comparability

Comparability expresses the confidence with which one data set can be compared with another. The extent to which existing and planned analytical data will be comparable depends on the similarity of sampling and analytical methods. Depending on the analytical procedures and QA objectives, the analytical data collected under this program may or may not be directly comparable to existing data; however, an attempt will be made to collect data in a manner consistent with data collected under the USEPA RI/FS at Harbor Island and previous RI data collected for ARCO.

3. FIELD PROCEDURES

The following procedures will be used by all personnel when conducting field activities at ARCO Terminal 21T. Field activities consist of all site visits, including visits to collect data and samples.

3.1 GENERAL

3.1.1 Documentation

All field activities will be documented in field notes using a pen with indelible ink bound in a three-ring binder and stored at the consultant's office. Information to be recorded in the notes will include the following:

- Date
- Weather conditions
- Names of the field team members
- Times of site arrival and departure
- Documentation of all field activities conducted
- Details regarding equipment calibration and malfunctions
- Odd or unusual occurrences
- Site visitors

The field supervisor will sign the field notes at the end of each day of field work. Attachment sheets will be used to record detailed information for specific activities including, but not necessarily limited to, soil borings, well installations, soil sampling, water monitoring, and water sampling. All attachment sheets will be included with the field notes in the three-ring binder and stored at the consultant's office.

Field equipment calibration procedures will be documented in the field notes during or immediately following calibration of the instrument. The following information will be recorded for each calibration event:

- Date
- Time
- Instrument make and model
- Instrument reading of standard prior to calibration
- Instrument reading of standard following calibration
- Calibration standard used
- Any deviation from standard calibration procedures
- Reason for calibration

3.1.2 Coded Sample Identification System

Every sample collected will be labeled according to the coded identification system outlined below. Each identification code will be unique to each individual sample collected, according to the guidelines detailed below.

3.1.2.1 Primary Samples

Matrix identifiers consisting of a two-letter code will be used to designate the type of sample collected. The complete list of eligible matrix identifiers is as follows:

Code	Matrix Identifier
SD	Sediment
SS	Soil (Surface or subsurface)
GW	Groundwater
FB	Field Blank
EB	Equipment Rinsate Blank
TB	Trip Blank

A one- to three-letter code followed by a number will identify the sample location. A list of representative sample location designators follows:

Designator	Sample Location
B001	Boring B001
EX001	Excavation Sample EX001
AMW01	ARCO Monitoring Well AMW01
SW1	Surface Water Monitoring Station SW1

If more than one sample of the same matrix type is collected on a given day at the same sample location, the samples will be differentiated with a unique sample identification number separated by a hyphen from the end of the sample location designator. Notes regarding the depth of sample collection and requested sample analyses will be recorded in the field notes.

A complete sample identification code will first identify whether the sample was collected in Plant 1 or Plant 2; second, identify a sample matrix type; third, provide a unique

sample location designator; and fourth, identify the quarter and year of sample collection. The following are examples of appropriate sample identification codes:

P1-SSEX004-100-1, where:

P1 indicates the sample was collected in Plant 1,

SS identifies the sample as a soil sample,

EX004 indicates the sample was collected from an excavation at location EX004,

100 indicates the sample was collected in the First Quarter of the year 2000, and

1 indicates a unique sample identifier.

P2-GWAMW06-100, where:

P2 indicates the sample was collected in Plant 2;

GW indicates the sample is a groundwater sample;

- AMW06 indicates the sample was collected from ARCO Monitoring Well 06; and
- 100 indicates the sample was collected in the First Quarter of the year 2000
- *Note*: As only one sample was collected at this location on this date, no unique identifier follows the sample location designator.

3.1.2.2 Field Replicate Samples

Field replicate samples (splits and blind duplicates) will be identified using the matrix codes and adding 200 to the sample location designator. All sample labeling will be recorded on soil or water sampling logs so that the sample can be traced to any field data recorded or analytical data received for that sample.

3.1.3 Equipment Calibration

All field instruments will be operated, calibrated, and maintained according to the manufacture's instructions using industry-accepted calibration standards. Operation and maintenance manuals for all field equipment will be provided to field personnel and kept in a field file in the consultant's office for reference. Field instruments will be calibrated and checked for proper operation at least as often as that listed in Table 1. Instruments will be calibrated on an as-needed basis should they exhibit evidence of power failure, instrument drift, or general failure.

The following instruments will be used to collect field data during the groundwater compliance monitoring program:

- Organic Vapor Meter (OVM) photoionization detector (PID)
- Explosimeter/combustible and toxic gas indicator
- Conductivity Meter
- pH Meter
- Electronic water-level indicator
- Oil/water interface probe

3.1.4 Equipment Decontamination

During remedial and groundwater compliance monitoring program field activities, equipment decontamination will be required to prevent cross-contamination of borings, wells, and samples as well as to prevent the potential transport of contaminants off site. Activity-specific decontamination procedures are summarized in Table 2 and described below.

3.1.4.1 Drilling and Soil Sampling Equipment

All drilling equipment will be cleaned prior to drilling each soil boring. This cleaning will consist of a high-pressure rinse with potable water to remove any grease, oil, or soil on the rig. Following this rinse, the drill rig will be steam cleaned with potable water. Drill rods, auger flights, bits, and sampling equipment will be steam cleaned with potable water. Steam-cleaning activities will take place in a designated area at least 50 feet from any boring location. The designated area will be lined with plastic and constructed such that decontamination rinsate products can be easily collected for storage and disposal, as outlined in Section 3.1.6, Waste Management and Disposal.

All reusable soil sampling equipment will be decontaminated before each use with a tap water rinse, followed by a detergent wash (Liquinox, Alconox, or a comparable laboratory-grade detergent), a 10-percent hydrochloric or nitric acid solution rinse, a 10-percent n-propyl alcohol or methanol solution rinse, and a final distilled water rinse.

Acid and solvent rinses will be stored in spray bottles pending use. All other decontamination washes/rinses will be stored in stainless steel buckets equipped with dedicated cleaning brushes. Rinsate from decontamination washes and rinses will be handled as described in Section 3.1.6, Waste Management and Disposal.

3.1.4.2 Water Monitoring and Sampling Equipment

Prior to measuring water levels and between each individual water level measurement, the instrument used for measurement will be decontaminated by first washing the instrument with a laboratory-grade detergent (Liquinox, Alconox, or other) and then rinsing with distilled water. If liquid hydrocarbons are encountered when collecting water level data, the probe used for measurement will be washed with a laboratory-grade detergent, rinsed with a 10-percent n-propyl alcohol or methanol solution, and then rinsed with distilled water.

Prior to monitoring well development and groundwater sampling, any reusable downhole equipment, such as pumps, will be decontaminated by first washing the equipment with a laboratory-grade detergent and then rinsing it with potable water. During decontamination, approximately 5 gallons of each laboratory-grade detergent solution and distilled water will be circulated through any pump used to purge or sample wells. Reusable bailers, spigots, and beakers will be decontaminated by washing with a laboratory-grade detergent, rinsing with distilled water, rinsing with 10-percent hydrochloric or nitric acid, a second distilled water rinse, a 10-percent n-propyl alcohol or methanol rinse, and a final rinse of distilled water.

Dedicated or disposable equipment will be used for water sampling whenever feasible to reduce the possibility of sample cross-contamination. Any equipment for which decontamination is either impossible or impractical, due to either the natures of the equipment or the nature of the contaminants, will be disposed of following each sampling event.

3.1.5 Sample Handling

All samples collected during the groundwater compliance monitoring program will be processed in a consistent manner. Water and soil samples collected for chemical and physical analysis will be immediately stored in an insulated container with solid cold packs pending delivery to the laboratory. Ice will not be used to cool samples.

Samples will be accompanied by a properly completed chain-of-custody form provided by ARCO or the contracted laboratory The sample numbers and locations will be listed on the chain-of-custody form. When transferring the possession of samples, the individuals relinquishing and receiving will sign, date, and note the time on the record. This record documents transfer of custody of samples from the sampler to another person, to the permanent laboratory, or to/from a secure storage area. Samples will be properly packaged for shipment and dispatched to the appropriate laboratory for analysis with a separate signed custody record enclosed in each sample box or cooler. Shipping containers will be secured with strapping tape for shipment to the laboratory. The cooler will be strapped shut with strapping tape in at least two locations.

The chain-of-custody record identifies the contents of the shipping cooler and the analyses to be performed on the samples contained therein. The original and one copy of the associated chain-of-custody record will accompany each shipment to the laboratory; the original will be attached to the report of analytical results sent to the consultant. The third copy of the chain-of-custody record will be retained by the sampler and returned to the consultant's office where it will be maintained in the permanent project files.

Field personnel will not retain any sample longer than two days after collection before shipment to the laboratory. One trip blank will be included in each sample shipping container possessing samples to be analyzed for volatile organic compounds. Samples will be delivered to the laboratory by the consultant's personnel whenever practicable. When personal delivery is not practicable, a bonded, private courier service that offers same-day or next-day service will be used. The day of delivery will be dependent on the proximity of the selected analytical laboratory to the site.

Bonded couriers handling sample shipments will not be required to sign off on the custody form as long as the custody forms remain sealed inside the sample cooler and the strapping tape on the cooler remains intact.

3.1.6 Waste Management and Disposal

All generated wastes will be dealt with in accordance with the guidance provided below. Wastes include drilling/sampling-generated soil and liquid, decontamination rinsate, personal protective equipment, and disposable sampling equipment. Wastes will fall into one of the following categories:

- Drill cuttings and soil samples not submitted for laboratory analysis.
- Excavated soils.
- Purge water from sampling and development of monitoring wells.
- Decontamination rinse water.
- Disposable personal protection equipment (PPE) (Tyvek® suits, gloves, etc.).

• Disposable sampling equipment (bailers, etc.).

3.1.6.1 Drill Cuttings

Drill cuttings and soil samples not submitted for laboratory analysis will be contained in sealed 55-gallon drums and temporarily stored on-site. The drums will be labeled with the following information:

- Date
- Waste category
- A unique identification number corresponding to the physical location at which the waste was generated.

Disposal of the drums will depend on the results of chemical analyses performed on the contained materials.

If encountered, hazardous waste materials will be segregated from non-hazardous wastes. Hazardous wastes will be stored in appropriate containers and labeled in conformance with Washington State Department of Transportation requirements. Disposal of both non-hazardous and hazardous wastes and will be handled through a licensed transport, storage, and disposal company.

3.1.6.2 Excavated Soils

Prior to soil excavation field activities at both Plants 1 and 2, representative soil samples will be collected from the excavation areas and analyzed for waste disposal parameters. These insitu soil samples will identify the waste classification (hazardous or non-hazardous) of the excavated soils prior to excavation activities. When the excavations occur, the soils will be able to immediately be properly transported off-site by a licensed transport, storage, and disposal company.

3.1.6.3 Purge and Decontamination Rinse Water

Purge water and decontamination rinse water will be temporarily contained in 55-gallon drums and transported to the on-site groundwater treatment system in operation along the waterfront. All purge water and decontamination rinse water will be processed through the treatment system prior to discharge to the sanitary sewer in accordance with the existing King County Department of Natural Resources Industrial Waste Division permit in place for the treatment system.

3.1.6.4 Disposable Equipment

Disposable PPE and sampling equipment will be contained in drums temporarily stored at the terminal. The terminal properly disposes of this type of waste as part of the normal waste management practices of the facility.

3.2 Groundwater Monitoring Well Installation

Five groundwater monitoring wells will be installed prior to beginning performance and confirmational sampling along the waterfront in the warehouse and loading rack area at the site to enable long-term collection of chemical and physical data regarding the aquifer beneath the warehouse and loading rack. Details regarding the locations of the wells are included in the Groundwater Compliance and Contingency Plan. Drilling and installation of groundwater monitoring wells at the site will conform to *Minimum Standards for Construction and Maintenance of Wells* (Chapter 173-160 WAC) and the guidance provided in the *Handbook of Suggested Practices for the Design and Installation of Groundwater Monitoring Wells* (EPA 1991).

During the installation of each well, soil samples may be collected and submitted for chemical characterization.

3.2.1 Preparation for Drilling, Sampling, and Well Installation

Following the selection and marking of monitoring well locations, the area will be checked for subsurface lines and utilities. Utilities and lines will be located and cleared by a professional locating service using active electrical or magnetic anomaly methods. Borings will be considered cleared for drilling if no utilities, lines, or subsurface structures are located within a 5-foot radius of the monitoring well location. In the event that utilities, lines, or subsurface structures are identified within this 5-foot radius, the monitoring well location will be re-evaluated and either relocated or abandoned.

Before beginning drilling, field personnel conducting well installation and associated sampling activities will assemble the appropriate equipment. In addition, all forms identified in this table will be completed to the extent practicable before drilling begins. All field instruments to be used during monitoring well installation will be checked to ensure proper calibration, as described in Section 3.1.3, Equipment Calibration. All equipment which will come in contact with the samples will be decontaminated before drilling commences and prior to the collection of each individual sample. Decontamination of equipment will be performed in accordance with the procedures outlined in Section 3.1.4, Equipment Decontamination.

Well construction materials, such as well screen, filter pack, well casing, and grout, will be assembled and inspected prior to drilling. Should unsuitable materials or equipment be identified during this initial inspection, drilling will be postponed until adequate materials can be assembled.

3.2.2 Drilling Methods

A qualified drilling subcontractor will complete borings for all monitoring wells using the hollow-stem auger drilling method under the supervision of field personnel. Two drill rigs will be used at the site to cope with variable access conditions. In areas where access is restricted by the presence of surface piping, buildings, or overhead obstacles, borings will be drilled using a limited access drill rig equipped with an 10.25-inch-diameter hollow-stem auger. In areas where access is unrestricted, borings will be drilled using a drill rig equipped with an 10.25-inch-diameter hollow-stem auger.

The wells will be drilled in two stages to prevent the shallow groundwater zone where product has been encountered from cross-contaminating the deeper groundwater zone. During the first stage, the boring will be advanced to a depth of approximately 20 feet below ground surface (bgs) using large diameter hollow-stem augers. Temporary conductor casing will then be installed to a depth of approximately 20 feet. Grout will be installed in the bottom of the borehole and allowed to set, such that any floating contaminants occurring at the top of the water table will be sealed out of the temporary conductor casing. Following the placement of this conductor casing, all drilling equipment will be thoroughly steam-cleaned. Drilling will then proceed to the desired depth of the well by drilling through the conductor casing using smaller diameter hollow-stem augers.

Borings will be advanced to the desired depths for soil sampling and well installation as described below. Water will not be added to the borehole to suppress heaving sands encountered during soil sampling. Water may be added during well construction if it is deemed that heaving sands will otherwise prevent successful well construction. Any water added during well construction will be sampled and submitted to a laboratory for analysis for the analytes that will be monitored for the well according the compliance monitoring program.

3.2.2.1 Soil Sampling and Logging

If soil samples are collected during drilling, the soil samples will be collected using a decontaminated stainless-steel sampling spoon. Soil will be placed directly in a laboratory-supplied glass sample jar with a screw-tight lid. Each sample jar will be completely filled with sample prior to sealing. In addition to filling the sample container, one air-tight plastic bag will be filled with a soil sample and retained for description and headspace analysis as described below. Plastic bags used to contain samples for headspace analysis will be labeled with the sample number.

When the sample jar is completely filled with sample, it will be closed by hand tightening the cap. The capped sample container will then be wiped clean and clearly labeled per Section 3.1.2, Coded Sample Identification System. The capped and labeled sample will then be handled in accordance with the procedures outlined in Section 3.1.5, Sampling Handling. For each sample collected, new vinyl gloves will be worn both as a health and safety precaution and to reduce the possibility of sample contamination from external sources. Should the gloves come in contact during sampling with material other than the sampling equipment, the sample material, or the sample container, the soiled gloves will be replaced with clean gloves prior to continuing with sampling.

During the advancement of boreholes, soil samples may be obtained at 2.5-foot intervals for the purpose of describing the materials encountered. The sampler will be driven using a 140-pound hammer designed to free-fall 30 inches before driving the sampler and drill rod.

Soils collected at the surface and retrieved in the sampler will be logged by field personnel using the ASTM Visual-Manual Procedure (D 2488). Soil descriptions, blow counts, and the results of field screening will be recorded on a Soil Boring Log. Sampled collected in the upper half of the sampler will be used for descriptive and field screening purposes. Soils collected in the lower half of the sampler will be retained for possible submission for chemical analysis, as dictated by the sample depth and the field-screening process.

Soil samples will be quickly transferred from the sampler with minimal disturbance to a precleaned glass sample jar. Glass sample jars will be closed by hand tightening the cap. The capped sample container will then be wiped clean and clearly labeled per Section 3.1.2, Coded Sample

3-11

Identification System. The capped and labeled sample will then be handled in accordance with the procedures outlined in Section 3.1.5, Sample Handling. New vinyl gloves will be worn during sample collection and the same precautions against sample cross-contamination will be observed.

Soils retained for field screening will be placed in clearly-labeled, air-tight plastic bags and set aside. These samples will then be screened for the presence of volatile organic compounds using the PID. The PID will be used to sample the headspace in the sealed plastic bag for each portion of sample set aside for field screening. The highest reading noted on the PID for each screening sample will be noted on the Boring Log.

Samples will be selected for submission to the laboratory on the basis of field parameters of headspace analysis, presence of staining or odor, and variation in lithology. The criteria and reasoning used for selection of samples from each borehole will be documented in the field notebook. If, during the course of the investigation, it is necessary to further define the vertical distribution of subsurface contaminants, additional samples will be collected from each boring. Soil samples will be collected below the encountered water table in selected locations and submitted for analysis for physical characteristics as described in the groundwater compliance monitoring program.

3.2.2.2 Well Installation

The monitoring wells will be 2 inches in diameter. Well construction details will be determined using data collected during the RI investigations. These details will be submitted to Ecology in a letter format for approval prior to the installation of any wells at the site.

All wells will be constructed using 2-inch Schedule 40 PVC threaded casing. The well screen will consist of 10-foot-long sections of continuous wrap, v-slot, Schedule 40 PVC screen. The well screen slot size will be determined based on the results of grain-size analyses conducted on soil samples collected during the RI investigation, but is assumed to be 0.01-inch. Casing and screen will be assembled according to the well specifications and lowered down the inside of the auger. All threaded sections of the well will be sealed with Viton® O-rings. All personnel handling well screen will wear clean vinyl or cotton gloves to keep the casing clean during installation. Should the casing or screen become visibly soiled during installation, it will be disassembled and cleaned to the specifications of the on-site field personnel.

The annular space between the well screen and the borehole wall will be backfilled with a filter pack of clean, No. 10-20 sorted silica sand. The sand will be poured inside the auger stem and allowed to fall into the borehole as the auger is slowly withdrawn. The filter pack will be slowly added in this fashion by alternately adding sand and withdrawing the auger until the filter pack extends at least 3 feet above the top of the well screen. A 0.5- to 1-foot-thick layer of clean, No. 35 grit silica sand will be placed on top of the filter pack before installing the well seal. The depth of the filter pack will be constantly monitored during installation by sounding the well with a weighted tape.

Well seals will consist of bentonite grout. A one foot layer of bentonite pellets will be placed atop the filter pack and allowed to hydrate for approximately 30 minutes, then bentonite grout will then be introduced into the boring annulus using a tremie pipe. The auger will be withdrawn in less than 5-foot steps as the annulus is filled with grout. The bentonite grout will seal the well from the top of the filter pack to approximately 1 foot bgs. The bentonite seal will be installed so that it fills any potential gap between the well casing and the conductor casing installed during boring advancement.

Wells will be protected at land surface with flush-mount, traffic-grade well monuments. Each monument will be set in concrete to a depth of at least 1 foot. In paved areas, the concrete will extend from the well monument at least 1.5 feet in all directions and will be poured flush with the existing pavement. A locking cap will be installed at each wellhead.

3.2.2.3 Well Development

All monitoring wells installed will be developed prior to collection of any water level data or water quality samples. The goals of well development are to remove fines left in the borehole and restore the hydraulic connection of the well to the aquifer following well drilling and installation. Monitoring well development will conform to the standards presented in *Minimum Standards for Construction and Maintenance of Well, Resource Protection Wells* (WAC 173-160-500) as well as the guidance presented in *Monitoring Well Installation and Development* (EPA 1991).

Prior to development, all wells will be checked for the presence of free product using a clear bailer. The bailer will be lowered into the well such that water or product surface is intersected. The bailer will then be withdrawn and examined. Should free product be discovered in the well, the well will not be developed. If, during continued monitoring, free-product is found to no longer occur in the well, the well will be developed prior to the initiation of groundwater sampling.

Wells will be developed using a surge-and-pump method. Surging will be conducted by using a loose-fitting block to develop differential pressures across the screened interval of the well. The surging action will settle the sand pack and loosen fines that remain in the borehole following well installation. Following surging, the well will be purged using either a hand bailer or an electric submersible pump. Each well will be surged at least three times during development.

During development, at least three well volumes of water will be removed from the well. Following the removal of each well volume, turbidity, temperature, specific conductivity, and pH will be recorded. Well development will continue either until turbidity, temperature, conductivity, and pH stabilize or five well volumes of water are removed.

All equipment used in well development will be decontaminated between wells using the procedures outlined in Section 3.1.4, Equipment Decontamination. Purge water and sediments removed from the wells will be drummed and managed in accordance with the procedures detailed in Section 3.1.5, Waste Management and Disposal.

3.2.2.4 Documentation

Soil sampling and boring information will be recorded on Soil Sampling and Boring Logs, respectively. The required information will be recorded on these logs at the time of collection or observation. Laboratory tracking and chain-of-custody documentation will be conducted as discussed in Section 3.1.5, Sample Handling.

Well construction logs will be completed for each monitoring well installed at the site. This log will be used to record details of well construction. Details regarding the development of groundwater monitoring wells will be recorded on a Well Development Log.

3.3 Hot Spot Soil Excavations

Excavations of TPH hot spots will occur at Plant 1 and Plant 2 as identified and detailed in the Groundwater Compliance Monitoring and Contingency Plan. Soil samples will be collected from the excavation perimeters following the specific procedures discussed in the sections below.

3.3.1 Sampling Locations

Soil samples will be collected from the perimeter of each of the excavations in Plants 1 and 2. The exact locations of the soil samples will be determined in the field during the excavation activities based on the actual extents of the excavations.

3.3.2 Sampling Preparation

Prior to sampling, field personnel will assemble the appropriate sampling equipment. Equipment that will come in contact with the soil will be decontaminated before each sample is collected. Decontamination will be performed in accordance with the procedures outlined in Section 3.1.4, Equipment Decontamination. Field instruments will be checked to ensure proper calibration according to the manufacturer's specifications and the procedures detailed in Section 3.1.3, Equipment Calibration.

3.3.3 Soil Sample Collection

Perimeter soil samples will be collected by using either a decontaminated hand auger or directly from the bucket of the excavation equipment using a decontaminated stainless-steel sampling spoon. Soil will be placed directly in a laboratory-supplied glass sample jar with a screw-tight lid. Each sample jar will be completely filled with sample prior to sealing. In addition to filling the sample container, one air-tight plastic bag will be filled with a soil sample and retained for description and headspace analysis as described below. Plastic bags used to contain samples for headspace analysis will be labeled with the sample number.

When the sample jar is completely filled with sample, it will be closed by hand tightening the cap. The capped sample container will then be wiped clean and clearly labeled per Section 3.1.2, Coded Sample Identification System. The capped and labeled sample will then be handled in accordance with the procedures outlined in Section 3.1.5, Sampling Handling. For each sample collected, new vinyl gloves will be worn both as a health and safety precaution and to reduce the possibility of sample contamination from external sources. Should the gloves come in contact during sampling with material other than the sampling equipment, the sample material, or the sample container, the soiled gloves will be replaced with clean gloves prior to continuing with sampling.

3.3.4 Sample Documentation

Sampling information including sample location and sample depth collected will be recorded on a Soil Sampling Log. In addition, a record of sample designation, the date, and time of collection will be recorded in the field notes. Laboratory tracking and chain-of-custody documentation is discussed in Section 3.1.5, Sample Handling.

3.4 Groundwater Sampling

Groundwater samples will be collected from monitoring wells installed during the RI and groundwater compliance monitoring program following the specific procedures discussed in the section below.

3.4.1 Sampling Locations and Frequency

Groundwater samples will be collected from the monitoring well network identified in the Groundwater Compliance and Contingency Plan, Exhibit F of the Cleanup Action Plan (CAP). The frequency of groundwater monitoring activities is also detailed in the Groundwater Compliance and Contingency Plan.

3.4.2 Sampling Preparation

Prior to sampling, field personnel will assemble the appropriate sampling equipment. Equipment that will come in contact with the ground water will be decontaminated before each sample is collected. Decontamination will be performed in accordance with the procedures outlined in Section 3.1.4, Equipment Decontamination. Field instruments will be checked to ensure proper calibration according to the manufacturer's specifications and the procedures detailed in Section 3.1.3, Equipment Calibration.

All wells will be sounded to verify identity and to determine the column of water currently present in each well. Static water-level measurements will be taken in all wells using the north side of the well casing as the measuring point (unless otherwise indicated) with an electric water level meter or interface probe. Water level measurements will be collected as described in Section 3.4.2. If product is observed in the well, both the water and product level measurements will be recorded using an interface probe. The electric water level meter and the interface probe will be decontaminated after use in each well in accordance with the procedures outlined in Section 3.1.4, Equipment Decontamination. Notes regarding the condition and integrity of the well will be recorded in the field

notes at the time of water level measurement collection. Should product be encountered in a well, the well will not be purged or sampled.

Prior to sampling, each well will be purged 3 to 5 times the volume of standing water present in the well casing. The groundwater will be purged using either a submersible pump or a disposable polyethylene hand bailer. Purge water will be pumped into 5-gallon buckets before being emptied into 55-gallon drums. The approximate number of gallons purged from each well will be recorded on the Water Sampling Log. Purge water will be handled in accordance with the procedures outlined in Section 3.1.6, Waste Management and Disposal.

Temperature, specific conductivity, and pH will be monitored during well purging. These field parameters will be monitored at least once for each well volume removed and should stabilize with continued purging. Stabilized field parameters will indicate that the well is ready for sampling. Each measurement obtained of field parameters will be recorded on the Water Sampling Log. After purging, the pump or bailer will be removed from the well and decontaminated in accordance with the procedures outlined in Section 3.1.4, Equipment Decontamination. During sampling, plastic sheeting will be placed around the wellhead to provide a relatively clean surface on which to place all sampling-related equipment.

3.4.3 Groundwater Sample Collection

To collect a groundwater sample, a disposable polyethylene bailer will be gently lowered down the well using nylon twine. The bailer will be allowed to fill and then will be quickly but gently raised out of the well to avoid agitation of the sample. Sample containers will be filled in the order listed on Table 3. The time of sampling will be immediately recorded onto the sampling log, chain-of-custody form, and sample label. Samples will be immediately placed in a cooler with ice packs pending transport to the laboratory. Laboratory instructions regarding preservation and handling of the sample will be observed at all times. Samples will be handled in accordance with Section 3.1.5, Sample Handling.

During sample collection, all sampling personnel will wear disposable vinyl gloves while handling any of the sampling equipment. Care will be taken not to allow the bailer or the twine to come in contact with possible contaminants near the well. If the bailer should become soiled during sampling, it will be cleaned using decontamination procedures outlined in Section 3.1.4, Equipment Decontamination. If the bailer cannot be adequately cleaned, it will be replaced before sampling continues.

3.4.4 Sample Documentation

Water level data and sampling information will be recorded on a Water Sampling Log. In addition, a record of sample designation, the date, and time of collection will be recorded in the field notes. Laboratory tracking and chain-of-custody documentation is discussed in Section 3.1.5, Sample Handling.

4. ANALYTICAL PROCEDURES

The analyses to be performed on the soil and water samples collected are described in the Groundwater Compliance Monitoring and Contingency Plan. The methods referenced and their associated quantitation limits will be followed exactly.

4.1 Analytical Methods and Procedures

All samples collected for chemical analysis will be submitted to analytical laboratories accredited by the State of Washington [Chapter 173-50 WAC (Washington Administrative Code)].

Assessment of laboratory performance in analytical parameters will be done through analysis of laboratory method blanks, spiked samples, surrogate spiked samples, duplicate samples, and laboratory control samples.

Groundwater parameters such as pH, specific conductivity, salinity, temperature, dissolved oxygen, and redox potential will be measured and recorded during sample collection events by trained personnel using properly calibrated instruments.

5. QUALITY CONTROL PROCEDURES

Quality control (QC) procedures will be implemented in both the field and laboratory. The kinds of internal checks that will be carried out are described below.

5.1 Field Activities (Measurements and Screening)

QC procedures for field measurements are limited to checking the reproducibility of the measurement by obtaining multiple readings or by calibrating the instruments, when appropriate.

QC of field sampling will involve collecting field duplicates, equipment blanks, and trip blanks.

5.2 Laboratory Analysis

Internal QC procedures are specified in the method protocols. These specifications include the compounds and concentrations to be used, the QC acceptance criteria for these audits, and the types of QC checks required (method blanks, reagent/preparation blanks, MS/MSDs, calibration standards, internal standards, surrogate standards, the frequency of each audit, the specific calibration check standards, laboratory duplicate analysis).

Laboratories have a written QA/QC program which provides rules and guidelines to ensure the reliability and validity of work conducted at the laboratory. Compliance with the QA/QC program is coordinated and monitored by the laboratory's QA department, which is independent of the operating departments.

The stated objectives of a laboratory QA/QC program are to as follows:

- Ensure that all procedures are documented, including any changes in administrative or technical procedures.
- Ensure that all analytical procedures are conducted according to sound scientific principles and have been validated.
- Monitor the performance of the laboratory by a systemic inspection program and provide for corrective action as necessary.
- Collaborate with other laboratories in establishing quality levels, as appropriate.
- Ensure that all data are properly recorded and archived.

All laboratory procedures are documented in writing as either Standard Operating Procedures (SOP's) or method procedures (MP). Internal QC procedures for analytical services

will be conducted by both analytical laboratories chosen to perform work under the RI/FS in accordance with their SOP's and the individual method requirements. The specifications for the SOP's and method requirements must include the type of audits required (sample spikes, surrogate spikes, reference samples, controls, blanks), the frequency of each audit, the compounds to be used for sample spikes and surrogate spikes, and the QC acceptance criteria for these audits.

The subcontracting laboratory will document in each data package provided that both initial and ongoing instrument and analytical QC functions have been met. Any samples analyzed in nonconformance with the QC criteria, excluding matrix effects, will be reanalyzed by the laboratory if sufficient sample volume is available.

6. DATA ASSESSMENT PROCEDURES

Following receipt of all analytical data for each sampling event, the field and laboratory data will be validated to determine whether the data quality objectives have been achieved. Data validation will be performed as described in the following sections.

6.1 Field Data Validation

After completing a sampling program, the field data package (field notes, calibration records, chain-of-custody forms, etc.) will be reviewed for completeness and accuracy. Items to be considered in the field data package validation procedure will include, but are not limited to, the following:

- A review of field data contained in sampling logs for completeness.
- A verification that equipment blanks, trip blanks, and field duplicates were properly prepared, identified, and analyzed.
- A verification of the calibration of field equipment.
- A review of chain-of-custody forms for proper completion, verification of signatures of field personnel and the laboratory sample custodian, and custodial dates.

6.2 Laboratory Data Validation

After validation of the field data package, validation of the analytical data package will be performed upon receipt of the data from the laboratory.

Items to be considered in the analytical data package validation procedure will include, but are not limited to, the following:

- A comparison of sampling dates, sample extraction dates, and analysis dates to check that samples were extracted and analyzed within proper holding times.
- A review of analytical methods and required quantitation limits to verify that they agree with the Groundwater Compliance Monitoring and Contingency Plan, Exhibit F of the CAP.
- A review of field and laboratory blanks to evaluate possible contamination sources (consideration should be given to preparation techniques and frequencies as well as the analytical results).
- A review of field duplicate data for evaluation of sampling and analytical precision.
- A review of selected laboratory QA data (blanks, surrogate spike recoveries, laboratory control sample recoveries, QC check sample recoveries, laboratory duplicate recoveries, etc.) for compliance with required acceptance criteria.

Data will be validated using the USEPA laboratory data validation functional guidelines for organic and inorganic review (USEPA 1993 and 1994). Appropriate data qualifier codes (B, J, R, U, UJ, as described in the USEPA functional guidelines) will be applied to those data for which QC results do not meet acceptance standards.

TABLES

Table 1. Instrument Calibration Frequency Sampling and Analysis Plan ARCO Harbor Island, Seattle, Washington

Instrument	Calibration Frequency
OVM PID	Once per day
Explosimeter	Once per day (1)
PH meter	Once per day
DO meter	Once per day
Salinity/conductivity meter	Once per day
ORP Meter	Once per day
Water-level indicator/interference probe	Once at beginning of field program

(1) Instrument calibration will be checked once per day. If calibration is correct, instrument will be recalibrated.

Table 2. Decontamination Procedures by Activity Sampling and Analysis Plan ARCO Harbor Island Terminal 21T, Seattle, Washington

	Procedure
Drill rug and excavation equipment	Pressure wash
	Steam clean (potable H2O)
Drill rods, auger flights, drill bits, split- spoon sampler	Steam clean (potable H2O)
Hand auger, sample barrels, sampling spoons,	Potable H2O rinse
Hydropunch	Detergent wash
	10% HCL or HNO3 rinse
	Distilled H2O rinse
	10% methanol
	Distilled H2O rinse
	Air Dry
Pumps	5 gallons of circulated detergent
	solution/detergent wash
	5 gallons of circulated potable H2O
	Outside Housing of pumps also
	decontaminated with the following:
	10% HCL of HNO3 rinse
	Distilled H2O rinse
	10% methanol rinse
	Distilled H2O rinse
	Air dry
Downhole equipment (sounding line, water-	Detergent wash
level indicator)	Distilled H2O rinse
Water sampling equipment (field instruments,	Potable H2O rinse
beakers)	Detergent wash
	10% HCL or HNO3 rinse
	Distilled H2O rinse
	10% methanol rinse
	Distilled H2O rinse
	Air dry

Table 3. Sample Container Filling Order Sampling and Analysis Plan ARCO Harbor Island Terminal 21T, Seattle, Washington

- 1. Volatile Organic Compounds
- 2. Total Petroleum Hydrocarbon Analysis
- 3. Carcinogenic Polynuclear Aromaic Hydrocarbons (cPAH's)
- 4. Conventional Water Quality Parameters
- 5. Metals (total and dissolved)

Plant 1 Waterfront Probing Summary Report Site: Former BP Harbor Island Terminal 2/7/2020

Appendix D

Data Validation Summary Sheets for Soil and Groundwater

Project Name: Ha Project Number: TM Site Location: Se	irbor Island IMS 10289 attle, Wash	Termina nington	l	-	Sampling Dates: Matrices: Laboratory:	7/22/19 - 7/31/19 Soil ALS
Parameter: Analytical Method:	Gasoline Washin	e gton Stat	te Metho	od NWTP	Lab Report ID: H-Gx	EV19070155
Sample ID	Sample	ID		Sample	ID	Sample ID
B-24-13-072219	B-17-17	-72419		B-31-12	-72519	B-38-8-72919
B-14-13-072219	B-06-13	-72419		B-32-12	-72519	B-34-9-72919
B-02-12-072219	B-28-12	-72919		B-21-12	-72519	B-39-9-72919
B-25-12-72319	B-18-12	-72419		B-10-13	-72519	B-40-7-72919
B-03-14-72319	B-07-12	-72419		B-22-14	-72519	B-36-8-72919
B-15-11-72319	B-29-11	B-29-11-72419			-72519	B-37-7-73019
B-26-11-72319	B-30-12	-72419		B-11-13-72519		B-13-9-73019
B-04-12-72319	B-10-13	-72419		B-9-10-7	72519	B-1-10-73019
B-05-10.5-72319	B-08-10	B-08-10-72419			72919	B-41-8-73119
B-27-12-72319	B-20-15	B-20-15-72519 B-35-8-			72919	B-42-7-73119
						B-43-7-73119
	Rep	orted	Qualif Req	ication uired		
Criteria Reviewed	Yes	No	Yes	No	Cc	omments
Field Records	X			X		
Chain of Custody	X			X		
Equipment Rinsate		N/A				
Field Duplicates		N/A				
Trip Blanks		N/A				
Preservation	X			X	One cooler temperatu above the recommend qualified on this basis	re was slightly (0.2° C) ded range. No data are
Holding Times	X			X		

Holding Times	Х		Х	
Reporting Limits	Х		Х	
Surrogate Spikes	Х		X	Many surrogate recoveries could not be determined due to high dilution. Recoveries in low or undiluted samples are within control limits. No data are qualified on this basis.
Method Blank	Х		Х	
Laboratory Control Sample	Х		X	The relative percent difference of one LCS duplicate pair is slightly above the control limit. The recoveries are within the control limits. No data are qualified on this basis.
Matrix Spike	Х	Х		The relative percent difference of one matrix
Matrix Spike Duplicate	Х	Х		spike/matrix spike duplicate pair is above the control limit.
Laboratory Duplicate	Х		Х	
Initial Calibration	Х		Х	
Continuing Calibration	Х		Х	

Project Name:	Harbor Island Terminal	Sampling Dates:	7/22/19 - 7/31/19		
Project Number:	TMMS 10289	Matrices:	Soil		
Site Location:	Seattle, Washington	Laboratory:	ALS		
		Lab Report ID:	EV19070155		
Parameter	Gasoline				
Analytical Method:	Washington State Metho	Washington State Method NWTPH-Gx			

Summary of Data Qualifiers

All data are considered quantitative except for the samples and constituents listed below:

Sample ID	Analyte	Qualifier	Reason for Qualification
B-11-13-72519	Gasoline	J	Matrix Spike (Note 1)

Qualifiers:

U The analyte was not detected at or above the quantitation limit.

J The associated value is approximate.

UJ The analyte was not detected at or above the quantitation limit, which is approximate.

The sample results are unusable. The analyte may or may not be present. R

Explanation:

(1) The relative percent difference for the matrix spike/matrix spike duplicate prepared from this sample is greater than the control limit. The sample result is, therefore, qualified as an approximate value (J).

Validator: Dion Valdez

Signed & Dated: Drow Odde_____

August 21, 2019

Data Validation Summary

Project Name: Harb Project Number: TMM Site Location: Seatt Parameter:	or Island S 10289 Ie, Wash Benzene	Termina ington	I	-	Sampling Dates: Matrices: Laboratory: Lab Report ID:	9/26/19 - 9/27/19 Water ALS EV19090220	
Analytical Method:	USEPA N	Aethod 8	3260				
Sample Date(s): Sample IDs:	9/26/2019 P1-GWMW-02-P1-919 P1-GWMW-02-P1-919 P1-GWMW-03-P1-919 P1-GWMW-04-P1-919 P1-GWMW-05-P1-919 P1-GWMW-06-P1-919 P1-GWMW-07-P1-919 RINSATE-1-P1-919 TRIP BLANK-1-P1-919			9/27/2019 P1-GWMW-01-P1-919 P1-GWMW-08-P1-919 P1-GWMW-09-P1-919 P1-GWMW-10-P1-919 P1-GWMW-210-P1-919 P1-GWMW-11-P1-919 RINSATE-2-P1-919 TRIP BLANK-2-P1-919			
	Repo	orted	Qualif Req	ication uired			
Criteria Reviewed	Yes	No	Yes	No	Com	ments	
Field Records	X						
Chain of Custody	X			X			
Equipment Rinsate	X						
	X						
Trip Blanks	X						
	X						
Holding Times							
Reporting Limits							
Method Blank							
Laboratory Control Sample	× ×				Lah also analyzed LCS o	tunlicate	
Matrix Snike	X		×		Percent recoveries in MS	S and MSD prepared from	
Matrix Spike Dunlicate	x		×	<u> </u>	one sample are below co	ontrol limit.	
Laboratory Duplicate		X	~	x	Lab analyzed MS/MSD ir	nstead.	
Initial Calibration	x						
Continuing Calibration	X			X			

Data Validation Summary

Project Name:	Harbor Island Terminal	Sampling Dates:	9/26/19 - 9/27/19
Project Number:	TMMS 10289	Matrices:	Water
Site Location:	Seattle, Washington	Laboratory:	ALS
		Lab Report ID:	EV19090220
Parameter:	Benzene		
Analytical Method:	USEPA Method 8260		

Summary of Data Qualifiers

All data are considered quantitative except for the samples and constituents listed below:

Sample ID	Analyte	Qualifier	Reason for Qualification
P1-GWMW-10-P1-919	Benzene	J	MS and MSD (1)

Qualifiers:

The analyte was not detected at or above the quantitation limit. U

J The associated value is approximate.

UJ The analyte was not detected at or above the quantitation limit, which is approximate.

The sample results are unusable. The analyte may or may not be present. R

Explanation:

(1) The percent recoveries in the matrix spike and matrix spike duplicate prepared from Sample P1-GWMW-10-P1-919 are below the control limit. This sample result is, therefore, qualified as an estimate value (J).

Validator: Dion Valdez

Hrow alde ____ Signed & Dated:

October 11, 2019

Note: Data validation was performed in accordance with EPA National Functional Guidelines for Organic Data Review.